AFWAL-TR-82-3108

AIRPLANE ACTUATION TRADE STUDY

Rockwell International Corporation North American Aircraft Operations P.O. Box 92098 Los Angeles, Cal 90009

February 1983

Final report for period June 1979 to July 1982

A sproved for public release; distribution unlimited

Copy available to DTIC does not permit fully legible reproduction

Prepared for

FILE COPY

E

AIR FORCE FLIGHT DYNAMICS LABORATORY Air Force Systems Command Wright-Patterson Air Force Base, Ohio 45433





· 83 07 26

NOTICE

When Government diawings, specifications, or other data are used for any purpose other than in connection with a definitely related Government procurement operation, the United States Government thereby incurs no responsibility nor any obligation whatsoever; and the fact that the government may have formulated, furnished, or in any way supplied the said drawings, specifications, or other data, is not to be regarded by implication or otherwise as in any manner licensing the holder or any other person or corporation, or conveying any rights or permission to manufacture use, or sell any patented invention that may in any way be related thereto.

This report has been reviewed by the Office of Public Affairs (ASD/PA) and is releasable to the National Technical Information Service (NTIS). At NTIS, it will be available to the general public, including foreign nations.

This technical report has been reviewed and is approved for publication.

Cecine GREGORY/J. DECERE

Program Manager Control Techniques Group

EVARD H. FLINN, Chief Control Systems Development Branch Flight Control Division

nneth & Binns

KENNETH E. BINNS Program Manager Fluid Power Group

Carrie A. Schon

DAVID H. SCHORR, Maj, USAF Chief, Power Systems Branch Aerospace Power Division

FOR THE COMMANDER

Karles K Goss /

CHARLES R. GOSS, JR, Lt Col/ USAF -Acting Chief, Flight Control Division Flight Dynamics Laboratory

FOR THE COMMANDER MES D. REAMS

Chief, Aerospace Power Division Aero Propulsion Laboratory

Copies of this report should not be returned unless return is required by security considerations, contractual obligations, or notice on a specific document.

DISCLAIMER NOTICE

THIS DOCUMENT IS BEST QUALITY PRACTICABLE. THE COPY FURNISHED TO DTIC CONTAINED A SIGNIFICANT NUMBER OF PAGES WHICH DO NOT REPRODUCE LEGIBLY.

REPORT DOCUMENT	TATION PAGE	READ INSTRUCTIONS BEFORE COMPLETING FORM
EPORY NUMBER	2. GOVT ACCESSION N	0. 3. RECIPIENT'S CATALOG NUMBER
AFWAL-TR-82-3108	ADA130709	
TITLE (and Subrine)		5. TYPE OF REPORT & PERIOD COVERED FINAL REPORT
AIRPLANE ACTIVATION ARADE ST	TI IIIY	25 JINE 79 - 1 JULY 82
And LARE ACTORTION TRADI, 5.	1001	6. PERFORMING ORG. REPORT NUMBER
		, NA-79-492-2
AUTHOR(s)		S. CONTRACT OR GRANT NUMBER(a)
C.W. Helsley		F33615-79-C-3615
PERFORMING ORGANIZATION NAME AND Rockwell International Cor	ADDRESS poration	10. PRUGRAM ELEMENT, PROJECT, TASK AREA & WORK UNIT NUMBERS
P.O. Box 92098	-	Project 24030249
Los Angeles CA 90009		
CONTROLLING OFFICE NAME AND ADD	RESS	12. REPORT DATE
Flight Dynamics Laboratory	(AFWAL/FIGL)	January 1983
Air Force Wright Aeronaut.	Cal Laboratories	13 NUMBER OF PAGES
ADNITORING AGENCY NAME & ADDRESS	WFAFB, UNIO 45455 Stil dillerent from Controlling Office)	15. SECURITY CLASS. (of this report)
		Unclassified
		158. JECLASSIFICATION DOWNGRADING SCHEDULE
ISTRIBUTION STATEMENT (of the about	e; distribution unlimi	Led.
DISTRIBUTION STATEMENT (of the about	e; distribution unlimi	Led.
Approved for public release Approved for public release DISTRIBUTION STATEMENT (of the about SUPPLEMENTARY NOTES	e; distribution unlimi	ied.
DISTRIBUTION STATEMENT (of the Repo Approved tor public release DISTRIBUTION STATEMENT (of the about SUPPLEMENTARY NOTES	e; distribution unlimi ect entered in Block 20. II different i ect entered in Block 20. II different i	ted. Trom Report)
DISTRIBUTION STATEMENT (of this Repo Approved for public release DISTRIBUTION STATEMENT (of the observe SUPPLEMENTARY NOTES KEY WORDS (Continue on reverse aids If no Actuation, Power-By-Wire, Aircraft Actuation, Hydrau	e; distribution unlimi ec: entered in Block 20. II different i eccessery and identify by block number All-Electric Airplane, lic Actuation, Seconda	ted. Trom Report) "", Electric Actuation, .ry Power Systems
Approved for public release Approved for public release DISTRIBUTION STATEMENT (of the observed SUPPLEMENTARY NOTES KEY WORDS (Continue on reverse aids If no Actuation, Power-By-Wire, A Aircraft Actuation, Hydrau ABSTRACT (Continue on reverse aids If no Ohis report contains the r Program. The study, condu actuation requirements; de all-electric airplane) and minor variants. The trade to weight, reliability, ma	e; distribution unlimi e; distribution unlimi e; entered in Block 20, 11 different i eccessery and identify by block number All-Electric Airplane, lic Actuation, Seconda cessery and identify by block number esults of the Airplane cted in three phases, sign of two airplaness a trade study of the study includes quanti intainability, and life	"" "" Electric Actuation, "" Actuation Trade Study included establishment of (a baseline airplane and an two airplanes plus several tive comparison data relative The cycle cost.

Ì

فالمماسينين كملك

UNCLASSIFIED

SECURITY CLASSIFICATION OF THIS PAGE(When Deta Entered)

20. ABSTRACT (continued)

The study results indicate that the "All-Electric" approach did not provide a viable alternative to the more conventional "Hydraulic Electric" approach as these approaches would be developed and applied to aircraft of the mid-1990 time period.

SECURITY CLASSIFICATION OF THIS PAGE(When Date En'ered)

FOREWORD

This Final Technical Report summarizes the work accomplished through June 1982 by Rockwell International's North American Aircraft Operations on Air Force contract F33615-79-C-3615, Airplance Actuation Trade Study. The contract was initiated under Project Number 2403 entitled "Flight Control System Development". The USAF Project Engineers for the trade study are Greg Cecere of AFWAL/FIGL and Kenneth Binns of AFWAL/POOS. This report covers work performed between 25 June 1979 to 1 July 1982.

The principal contributors to this activity at Rockwell International were C. W. Helsley Jr., A. Davanzo and W. Frantz.

Accession For	7
NTIS GRA&I	-
DTIC TAB	- J
Unannounced	1
Justification	1
	-
Ву	1
Distribution/	1
Availability Codes	1
Avel und or Dist Special	LII COPW INSPECTED

iii

TABLE OF CONTENTS

Section				Page
1.0	INTRODUCT	ION		
	1.1	Status		1
	1.2	Scope		1
		1.2.1		1
	1.3	Backgr	ound	1
2.0	STUDY AIR	CRAFT D	EFINITION	- 3
	2.1	Baseli	ne Air-to-Ground Tactical Fighter	
		Requir	rements	3
		2.1.1	Design Mission Profile Requirements	1
		2.1.2	Additional Design Criteria	1
		2.1.3	Service Life and Usage Criteria	13
		2.1.4	Structural Mancuvering Design	
			Criteria	13
		2.1.5	Temperature Design Data	13
		2.1.6	Vibration/Acoustics	13
		2.1.7	Reliability	13
	2.2	Basepc	int Tactical Fighter Configuration	18
		2.2.1	General	18
		2.2.2	Power Requirements,	18
		2.2.3	Flight Control System	22
		2.2.4	Environmental Control System	34
		2.2.5	Aircraft Configuration	34
		2.2.6	Armament Subsystem	34
		2.2.7	Engine Starting Loads	34
	2.3	Histor	rical Review	39
	2.3.	1 Data	a Revisions and Additions	39

PROCE ING INCH MARK-INT FILMED

ν

Section

	2.4	Trade Study Ground Rules	49
		 2.4.1 H, draulic System Burst Factors 2.4.1 H, draulic System Design Pressures 2.4.3 varaulic System Design Criteria 2.4.4 Peak Actuation Loads 2.4.5 Actuation Configuration 2.4.6 Trade Study Aircraft 	49 49 55 55 55
3.0	CONCLUSION	S AND RECOMMENDATIONS	
	3.1 3.2	Conclusions Recommendations	57
4.0	TRADE STUD	Y AIRCRAFT DESCRIPTION	61
	4.1	Aircraft	61
	4.2	 4.1.1 Power Generation and Starting 4.1.2 Power Generation and Distribution 4.1.3 Aircraft Inboard Profile 4.1.4 Electrical System Load Analysis 4.1.5 Power Utilization 4.1.6 Starter/Generator System Definition 4.1.7 APU Driven Generator Sizing 4.1.8 APU Sizing 4.1.9 Aircraft I Electrical System Weight Analysis 	61 70 81 87 .13 147 151 151 153 164
		4.2.1 Hydraulic System Description4.2.2 Aircraft II Electrical System General Description	167 226
5.0	TRADE STUI	DY RESULTS	
	5.1 5.2 5.3	Weight Trade Reliability and Mainaainability Trades Life Cycle Costs	2 39 246

Page

vi

Section			Page
APPENDIX A	AIRPLANE ACTUATION TRADE	STUDY	255
APPENDIX B	AIRPLANE ACTUATION TRADE	E STUDY	273
APPENDIX C	AIRPLANE ACTUATION TRADE	E STUDY	285
APPENDIX D	AIRPLANE ACTUATION TRADE	E STUDY	325

.

LIST OF JLLUSTRATIONS

r	<u>+</u>	~
г	1	gure

Title

1	Basepoint High Altitude Design Profile Performance 4
2	Basepoint Alternative Performance Evaluation Mission 5
3	Peacetime Training and Support Missions
4	Composite Lifetime Usage Mission
5	Acoustic Design Criteria
6	Basepoint Advanced Tactical Fighter
7	Power Source Utilization (Housekeeping)
8	Power Source
9	Power Source Utilization (Housekeeping Plus Actuation) 26
10	Aircraft I & Aircraft II ECS System
11	Starting Torque and Speed Requirements
	(Sea Level Static Conditions)
12	Engine Actuation Functions
13	Outboard Trailing Edge (Aileron) Installation Envelope 45
14	Inboard Trailing Edge Surface (Flap) Instal. Envelope 46
15	Modified Goodman Diagram
16	Candidate Configuration Concepts
17	Aircraft I Secondary Power Generation System
18	Aircraft I Electrical Power Generation and Distribution
	System Schematic
19	Aircraft 1 EPGDS Schematic
20	Aircraft I 270 VDC Starting Arrangement
21	Aircraft I Engine Starting Torque Converter Characteristi 79
	Characteristics.
22	ATS Inboard Profile
23	Electric Actuation System Spatial Arrangement
24	Flight Control Actuator Load Characteristics
	During Combat.
25	Inboard Flap Installation
26	Midspan Flap Installation.
27	Aileron Installation
28	Upper Rudder Installation
29	Lower Rudder Installation.
30	L.E. Flan Installation
31	Canard Installation 123
32	Inboard Flap Motor Characteristics
33	Heat Transfer with Change in Phase
34	Evaporative Cooling of Electronic Equipment
35	Wing Trailing Edge Surface Power Control 143
36	Wing Trailing Edge System Actuator Declutching
20	Connections 145
	$Confrections \dots \dots \dots \dots \dots \dots \dots \dots \dots $

Figure

_

Page

37	Signal and Power Hookups	6
38	Motor Weight	5
39	Ballscrew Actuator *Assembly Weight	6
40	Electrical Component Locations	9
41	ATS Weight Trends at Various System Pressure 16	6
42	Hydraulic System Block Diagram	8
43	Schematic-Hydraulic Pressure Generation & Distribution	
	ATS	0
44	Hydraulic System Spatial Agreement	2
45	Aircraft II Secondary Power Generation System 17	3
46	Aileron Actuator Instl	;2
47	Aileron Actuator	3
48	Canard Actuator Design	6
49	Hydraulic Actuator Weight	9
50	Compression Factor (2) 19	4
51	Accumulator Weight	6
52	Motor Sizing	9
53	Reservoir Weight	5
54	Aircraft II Electrical Power Generation	
	System Schematic	:7
55	Final APU/EPU Configuration	56

. . . .

LIST OF TABLES

Table

Title

Page

 Basepoint Flight-By-Flight Composite Load Factor Spectrum (Combined Combat and Training Missions)
(Combined Combat and Training Missions).105.Basepoint Composite MIssion Flight Design Temperature Data Standard Day104.2.2M Mission Reliability Allocation Probability of Success.175.Actuation Loads (KW)276.Canard Positions287.Summary of Control Surfaces/Devices.308.Flight Control Actuation Requirements329.Baseline Utility Actuator Requirements3510.ECS Heat Load.3511.Actuator Output Load Requirements.4412.Baseline Flight Control Actuator Requirements.4813.Burst Pressure Comparison.50
 Basepoint Composite MIssion Flight Design Temperature Data Standard Day 2.2M Mission Reliability Allocation Probability of Success. Actuation Loads (KW) Canard Positions Canard Positions Summary of Control Surfaces/Devices. Summary of Control Actuation Requirements Saseline Utility Actuator Requirements ECS Heat Load. Actuator Output Load Requirements. Actuator Output Load Requirements. Summary Sources Baseline Flight Control Actuator Requirements. Summary Sources Baseline Flight Control Actuator Requirements. Summary Sources Sources Sources<
Standard DayStandard Day4.2.2M Mission Reliability Allocation Probability of Success.5.Actuation Loads (KW)6.Canard Positions7.Summary of Control Surfaces/Devices.8.Flight Control Actuation Requirements.9.Baseline Utility Actuator Requirements.10.ECS Heat Load.11.Actuator Output Load Requirements.12.Baseline Flight Control Actuator Requirements.13.Burst Pressure Comparison.
4.2.2M Mission Reliability Allocation Probability of Success.5.Actuation Loads (KW)6.Canard Positions7.Summary of Control Surfaces/Devices.8.Flight Control Actuation Requirements.9.Baseline Utility Actuator Requirements.10.ECS Heat Load.11.Actuator Output Load Requirements.12.Baseline Flight Control Actuator Requirements.13.Burst Pressure Comparison.
Success.17Success.17Actuation Loads (KW)27Canard Positions28Canard Positions28Summary of Control Surfaces/Devices.308.Flight Control Actuation Requirements.329.Baseline Utility Actuator Requirements3510.ECS Heat Load.3511.Actuator Output Load Requirements.4412.Baseline Flight Control Actuator Requirements.4813.Burst Pressure Comparison.50
5.Actuation Loads (KW)276.Canard Positions287.Summary of Control Surfaces/Devices308.Flight Control Actuation Requirements329.Baseline Utility Actuator Requirements3510.ECS Heat Load3511.Actuator Output Load Requirements4412.Baseline Flight Control Actuator Requirements4813.Burst Pressure Comparison50
6.Canard Positions287.Summary of Control Surfaces/Devices308.Flight Control Actuation Requirements329.Baseline Utility Actuator Requirements3510.ECS Heat Load3511.Actuator Output Load Requirements4412.Baseline Flight Control Actuator Requirements4813.Burst Pressure Comparison50
7.Summary of Control Surfaces/Devices.508.Flight Control Actuation Requirements.529.Baseline Utility Actuator Requirements5310.ECS Heat Load.5511.Actuator Output Load Requirements.4412.Baseline Flight Control Actuator Requirements.4813.Burst Pressure Comparison.50
8.Flight Control Actuation Requirements.529.Baseline Utility Actuator Requirements3310.ECS Heat Load.3511.Actuator Output Load Requirements.4412.Baseline Flight Control Actuator Requirements.4813.Burst Pressure Comparison.50
9.Baseline Utility Actuator Requirements3510.ECS Heat Load.3511.Actuator Output Load Requirements4412.Baseline Flight Control Actuator Requirements4813.Burst Pressure Comparison50
10.ECS Heat Load.5511.Actuator Output Load Requirements.4412.Baseline Flight Control Actuator Requirements.4813.Burst Pressure Comparison.50
11.Actuator Output Load Requirements.4412.Baseline Flight Control Actuator Requirements.4813.Burst Pressure Comparison.50
12.Baseline Flight Control Actuator Requirements.4813.Burst Pressure Comparison.50
13. Burst Pressure Comparison. 50
14 Contain Duo sumo
14. System Pressure
15. Supplemental Electrical Load Analysis Item Number
Breakdown
16. Combat - 5 Sec Loads 4 Channel Operative
17. Aircraft I - Electrical System Load Summation 103
18. Approximate Drive Sizing
19. Approximate Gearhead Sizing
20. Approximate Motor Sizing
21. Inverter Characteristics
22. Inverter Loses
23. Evaporate Cooling Fluid Properties
24. Evaporate Cooling Press/Temp Conditions
25. Electrical Actuation Function Weight
26. Weight - Electrical Power Wiring (Page 1 of 2 Pages) 160
27. Aircraft I Electrical System Weight Summary
28. ATS Hydraulic System Weights At Various Pressures
29. Aircraft II Loads During Combat (5 Sec. Loads)
30. Hydraulic Linear Actuators
31. APU/EPU Power Requirements Definition
32. 3AL-2.5V Titanium Tubing Data
33. Fitting Data
34. Landing Gear P At Rated Flow and -20° F
35. Notor Characteristics and Comparisons

Table

.

36.	Actuator Swept and Exchange Volumes	211
37.	Master Equipment List (Page 1 of 9 Pages	213
38.	Hydraulic Power Generation, Distribution, and	
	Utilization System (HPGDUS)	224
39.	Aircraft Generating System Comparisons	232
40.	Aircraft II Control and Power Supply Interrelationship	235
41.	Aircraft II Electrical System Weight.	237
42.	Aircraft II Total Weight Subject to Trade	238
43.	Weight Summary	240
44.	Reliability-Maintainability Data Aircraft I	
	(Electric) (Sheet 1 of 2)	242
45.	Reliability-Maintainability Data Aircraft II	
	(Hydraulic) (Sheet 1 of 2)	244
46.	Aircraft I Life Cycle Costs - 1982 Dollars	247
47.	Aircraft II Life Cycle Costs - 1982 Dollars	248
48.	Aircraft I Life Cycle Costs - 1995 Dollars	249
49.	Aircraft II LIfe Cycle Costs - 1995 Dollars	250

Page

INTRODUCTION

1.1 Status - Current aircraft are characterized by two main forms of on-board secondary power generation distribution, and utilization, i.e., electrical power and hydraulic power. In general, hydraulic power is generated, distributed, and utilized for the majority of the high power output actuation functions such as primary and secondary flight control surfaces, landing gear extension and retraction, brakes, nose wheel steering, etc. and electrical power is used for everything else. This division of functional responsibility developed over the years. largely as a result of the ever increasing demands of high performance aircraft for higher levels of controllability in the presence of high "G" forces, thus making it necessary to amplify pilot forces with power actuators. The accepted fact that hydraulic actuators have enjoyed many advantages over electromechanical actuators for high torque, high horse power application shifted the pendulum in their direction. In the interim, since hydraulic actuation was accepted approximately twenty five years ago, many changes have occurred. These changes are discussed more fully in paragraph I.3 but they lead up to the objectives of this study which are stated here: "Establish advantages/disadvantages and life cycle cost impact of hydraulic actuation and power-by-wire actuation of aircraft in the 1990 + time frame." A secondary objective of this effort was to identify technology needs and development requirements for future aircraft actuation systems.

1.2 <u>Scope</u> - This program was conducted to satisfy the objectives listed above under the following guide lines.

1.2.1 Conduct a trade-off study between a power-by-wire actuation airplane and one that retains an engine-driven hydraulic syste for actuation. NOTE: A power-by-wire actuation airplane was defined as either (1) removal of all engine driven hydraulic pumps and hydraulic power distribution systems and replacement with electrical power generation and distribution systems to the actuator location where electrical power was then converted to hydraulic power for actuation or (2) same as above except that the electrical power was converted to mechanical power for actuation directly or (3) some combination of (1) and (2). Hybrid systems retaining engine driven pumps for specific functions were considered viable options.

1.2.1.1 Use the ATS concept as the point of reference airplane on which the trade study was to be conducted.

1.2.1.2 Use the 1990+ time frame as a design reference for all system options included above.

1.2.1.3 Include other utility functions such as environmental control systems if they became relevant to the basic trade.

1.2.1.4 Assess the trade on the basis of performance, reliability, maintainability, weight, life cycle costs, growth potential, survivability, and environmental constraints.

1.3 <u>Background</u> - The advent of jet engines in the early fifties greatly increased the performance of military aircraft and made it necessary to supplement pilot control forces with power amplification (actuators) at the control surfaces. At the time these actuators became necessary there was, as there is now, two possible power choices, electrical or hydraulic. As the weight and space penalties were examined to make the choice, i.e., hydraulic or electrical actuation, there was no real contest. Hydraulic actuation was clearly superior, if not indispensable. The development of hydraulic actuation had reached the point where primary control surface actuation had become synonymous with hydraulic actuation. Nor was hydraulic actuation limited to primary control surfaces; landing gear retraction and extension,

1.3 (cont.)

brakes, flaps and slats operation, and auxiliary functions such as nose wheel steering, etc. were all conventionally done with hydraulic actuation on most airplanes. The development of engine driven hydraulic power systems to service these actuation needs had therefore also evolved and matured over the years and had reached a high degree of refinement. However, looking down the road to future design, several significant factors were emerging which suggested that primary flight control actuation, as well as other actuation functions, need not continue to be done in the conventional manner.

Perhaps the single greatest factor that was stimulating the need to examine powerby-wire actuation, as an alternative to conventional hydraulic actuation, was the increasing importance of avionics and in particular fly - by-wire. Fly-by-wire signal transmission dictates absolute electrical power reliability as a paramount design requirement for future airborne electrical power systems. There was, and is, a direct conflict between the independent redundancy required for electrical power support of fly-by-wire and the longstanding independent hydraulic power redundancy requirements to support flight control actuation. For example, on a single engine airplane, with a four channel fly-by-wire system, if carried to the extreme, this approach could have resulted in four engine driven electrical generators plus the normal two hydraulic pumps.

A second factor was the mounting cost to design, develop and maintain the engine driven hydraulic power and distribution systems that were being implemented to utilize the generally admirable qualities of hydraulic actuators. These hydraulic systems were plaged with leaks, conta mination, flammability and generally high life cycle costs, particularly maintenance costs.

A third factor concerned the credibility of power-by-wire actuation as an alternate to conventional hydraulic actuation, i.e., hydraulic actuators supplied by engine driven hydraulic systems. In the twenty-five years since the adoption of hydraulic actuators, several technologies in the electrical and electromechanical area had emerged which warranted a reinvestigation of electromechanical actuation for application to primary flight control and other actuation systems. Some of these advancements were high voltage power supplies, permanent magnet motors using rare earth magnets, electronic commutation and an improvement in solid state power switching devices.

2.0 STUDY AIRCRAFT DEFINITION

2.1 <u>Baseline Air-to-Ground Tactical Fighter Requirements</u> - An integrated baseline set of 1990 tactical air-to-ground fighter mission requirements was selected and is presented in this section. This set of baseline requirements provided a foundation and framework within which the trade studies could be conducted. The selection was based on recent studies (reference 31 through 33) and represent a very likely set of requirements for the time period.

2.1.1 Design Mission Profile Requirements - The baseline "most probable" 1990's advanced tactical mission profile requirements turned out to be a high altitude supersonic design mission with a mach 2.0 class penetration speed supplemented by an alternate low level terrain following profile capability in the high subsonic speed category. The specifics of the design sizing profile and the alternate capability profile are presented in figures 2-1 and 2-2. The mach 2.2 penetration for the high altitude profile was selected because it represented the approximate upper speed boundary for use of the composite materials generally expected to be employed in the time period. The mach 0.9 penetration selected for the alternate capability low level mission was selected as a reasonable compromise between higher speeds providing better survivability and lower speeds providing better target acquisition. The two penetration design points selected also presented moderately high requirements to the aircraft actuation system design in terms of hinge moments, temperature environment and actuation rate requirements. Thus, these baseline design profiles, while representing expected future mission requirements, also presented moderate challenges to actuation system technology without preselecting a particular type of actuation system through selection of extreme combinations of requirements. The remaining requirement particulars indicated on figures 2-1 and 2-2 (distances, payload, combat allowances, takeoff and landing distances, etc.) were selected as nominal representative values from recent industry/government studies (reference 31 through 33). These latter factors influenced size of the aircraft but had only secondary influence on secondary power and actuation system concepts, design and technology requirements.

Other mission profiles considered prior to selection of the mach 2.0 class aircraft were; mach 3.0 high altitude penetrator, 0.7 low level penetrator, and mach 1.2 low level penetrator.

2.1.2 Additional Design Criteria - The following paragraphs present additional design criteria that have significant effects on actuation system designs. They have been derived from the same industry/government studies (reference 31 through 33) that established the basic mission profiles and were selected for their compatibility with the specific basepoint design mission profiles.

2.1.3 Service Life and Usage Criteria - The total service life requirement for this type of aircraft was 8000 total flight hours, based on current DOD policy to maintain a major aircraft in the inventory for 15 years plus the assumption of one major short war. The combat usage consisted of 60 flights of each of the combat profiles illustrated in figures 1 and 2.

Training usage was developed around the philosophy of providing adequate flight time to complement a significant ground simulator training program. Limited actual supersonic flight and low-level terrain following were scheduled to



T.O. & LDG DISTANCE: ≤ 3000 FT GND ROLL

MISSION RADIUS - 550 N MI

(A/G + A/A WEAPONS)

WEAPONS PAYLOAD - 5,680 LBS (A/G WPNS + FAIRINGS) PAYLOAD DROPPED - 5,030 LBS

	MISSION SECHENT	DISTANCE	масн	ALTITUDE (ET) end	TIME (MIN)
				(11) Circ	
A.	TAKEOFF AND ACCEL TO Initial climb velocity	σ	<u>0-0-81</u>	o	15.8
8.	CLIMB AND ACCEL TO CRUISE CONDITION	38.0	0.81-0.9	31,500	4.2
Ç٠	CRUISE-QUT	244.2	0.9	32,800	27.8
D.	CLIMB AND ACCELERATE TO DASH CONDITION	67.8	0.9-2.2	59,000	3.9
Ε.	DASH TO TARGET	200	2.2	60,100	9.5
F.	COMBAT ALLOWANCE*	o	2,2	50,000	2.8
G.	DASH - TARGET TO INITIATION OF RETURN DESCENT/DECEL	200	2.2	62,300	9.5
н.	RETURN DESCENT/DECEL	185.7	2.2-0.9	39,600	13.3
١.	RETURN CRUISE	164.3	0.9	40,300	19.0
J.	LANDING/LOITER	0	0.4-0	0	20.0

* COMBAT ALLOWANCE 360° TURN AT P_s = 0, MAX A/B 2.2M/50,000 FT

Figure 1. Basepoint High Altitude Design Profile Performance



T.O. & LDG DISTANCE: ≤3000 FT GND ROLL

MISSION RADIUS - MAX ACHIEVABLE

(A/G + A/A WEAFONS) (A/G WPNS + FAIRINGS) WEAPONS PAYLOAD - 5,680 L3S PAYLOAD DROPPED - 5,030 LBS

	MISSION SEGMENT	DISTANCE (N MI)	MACH	ALTITUDE (FT)	TIME (MIN)
Α.	TAKEOFF AND ACCEL TO Initial climb velocity	ბ	0-0.81	0	0.8
8.	CLIMB AND ACCEL TO CRUISE CONDITION	38	0.81-0.90	0-31500	4.2
C.	CRUISE-OUT	312	0.90-0.90	31500-33000	35.50
D.	DESCENT TO DASH CONDITION	0	-	33000-0	0
Ε.	DASH TO TARGET	165	0.9	0.0	16.6
F.	COMBAT ALLOWANCE *	O	0.9	0.0	0.45
G.	DASH - TARGET TO IN!TIATION OF RETURN CLIMB	165	0.9	v.o	16.6
н.	RETURN CLIMB	26	0.9-0.9	0.0-39000	2.8
١.	RETURN CRUISE	324	0.9-0.9	39000-40200	37.5
J.	LANDING/LO!TER	0	0.38	0	20.0

* COMBAT ALLOWANCE :

360' TURN AT $P_5 = 0$, MAX A/8 0.9M/0.0 FT BCM - BEST CRUISE MACH BCA - BEST CRUISE ALTITUDE BLM - BEST LOITER MACH

Figure 2. Basepoint Alternative Performance Evaluation Mission

minimize structural temperature and fatigue design requirements, however, adequate airborne experience was accumulated in high- and low-level missions, navigation, inflight refueling weapon delivery tactics, etc. Allowances were included for the extra landings due to touch-and-go practice and routine around-the-field maintenance checkout flights, etc. Six basic training mission profiles were developed based on current tactical fighter training schedules, modified as appropriate for the capabilities and tactical employment envisioned for the advanced fighter. The field-go-around profile was added to the other profiles or conducted independently to represent touch-and-go training activities or short maintenance checkout flights, etc. The basic mission characteristics are illustrated in figure 3.

The average number of flights flown per aircraft each year on each of these profiles were:

Mission Profile	Avg. No. of Flights/Yr.
Ground attack tactics	54.2
Mission support	21.0
Low-level strike with refueling	20.1
High-level strike	16.9
High-level strike with refueling	12.3
Supersonic combat profile	2.4
Extra field-go-around plus landing	153.1
	280.0

The flight operations indicated above impose 4320 ground-air-ground cycles on the average aircraft for the specified usage. The flight-hour usage accumulated in each of the mission legs illustrated by the totality of flight operations indicated in figure 3 plus the combat operations was the basis for projecting the structural fatigue life spectra, thermal environment design criteria and secondary power and actuation system component duty cycle requirements. Estimated vehicle total lifetime hours usage is shown on table 1.

Based on the usage presented in table 1, a single composite mission was developed which if repetitively flown would produce approximately the same cumulative individual leg usage as noted in the table. This composite mission profile is presented in figure 4. Use of a single design mission of this type facilitated development of detailed design criteria for the aircraft usage. Because aircraft actuation system component duty cycles are influenced by the maneuvering and gust upset restoring requirements by mission leg, the load factor spectra was developed based upon the data of MIL-A8866B for the cumulative life of the aircraft as flown over the composite mission. Table 2 presents this design data. The spectrum includes the effects of long-term, peacetime training usage and an allowance for a representive high-intensity, short-duration wartime employment. The combined flight and ground-air-ground cycle design spectra resulting from this usage is presented as a flight-byflight composite mission load spectrum containing an appropriate amount of flight time in each of the mission legs in table 1. The composite mission has a duration of slightly over 2.7 hours. A total of 2,935 such missions with an extra field-go-around and landing on every other flight provides the total design life usage.



TABLE 1. VEHICLE TOTAL LIFETIME HOURS USAGE

MISSION SEGMENTS	SUPERSONIC COMBAT	SUBSONIC COMBAT	PEACETIME MISSIONS (1) - (5)	SUPERSONIC TRAINING	FIELD GO-AROUND	TOTAL HOURS
TAXI, T.O. AND CLIMB	18	30.	1150	22		1220
ENROUTE NAV CRUISE	48	69	2561	38		2707
AERIAL REFUEL			315	16		331
LOW LEVEL TERRAIN FOLLOWING		36	402			438
SUPERSONIC ACCEL/CLIMB	(9)			(9)		(12)
GROUND ATTACK TACTICS			1082			1082
SUPERSONIC CRUISE	(24)			(27)		(12)
WEAPON DELIVERY TURN		12				12
FLIGHT MANEUVER (LOLTER)		Q	234			240
SUPERSONIC WEAPON DELIVERY	(12)					(12)
SUPERSONIC DESCENT	(9)			(11)		(11)
SUBSONIC DESCENT	Q	12	596	2		619
TRAFFIC PATTERN (LONTER)	Q	9	665	Q		683
LAND AND TAX!	9	Ģ	211	2		294
T.O. AND CLIMB					24	24
TRAFFIC PATTERN (LOITER)					223	223
LAND AND TAX:					35	35
	132	168	7282	136	282	0008
	(85)			(44)		(76)

* See figure 3
NOTE: Supersonic hours enclosed in parentheses

_





		LOAD FACTOR			
MISSION SEGMENT	MAX	MIN	CYCLES/MISSION		
			_		
1. Taxi	1.3	0.7	1		
	1.2	0.8	15		
2. Takeoff & Climb	3.3	1.0	0.01		
	2.7	1.0	0.1		
	2.1	1.0			
	1.9	1.0			
	1./	1.0	4		
	1.5	1.0	0		
2 Francis Neutranian Outbound	1.5				
3. Enroute Navigation, Outpound	4.0	1.0	0.01		
		1.0	0.01		
	4.1	1.0	0.1		
		1.0	1		
	2.8	1.0			
	2.0	1.0	2		
	2.7	1.0	7		
	2.0	1.0	4		
	1.7	1.0	4		
	1.4	1.0	8		
	1.1	1.0	8		
4. Aerial Refuel	1.7	; 0.4	0.01		
	1.5	0.6	0.1		
	1.3	0.7	1		
	1.3	0.7	2		
	1.2	0.8	28		
5. Low Level Terrain Following	2.2	0.4	0.01		
	2.1	0.4	0.21		
	2.0	j 0 . 4	1		
	2.0	0.5	1		
	1.9	0.5	4		
	1.8	0.5	4		
	1.7	0.5	4		
	1.6	0.6	8		
	1.5	0.6	8		
	F+4	0.6	16		
	1 1.2	<u>c.8</u>	16		
6. Supersonic Climb/Accelerate	2.3	; 1.0	0.01		
	1.6	1.0	0.1		

TABLE 2. BASEPOINT FLIGHT-BY-FLIGHT COMPOSITE LOAD FACTOR SPECTRUM (COMBINED COMBAT AND TRAINING MISSIONS)

میں باد اور در میں ایک کر اور اور اور اور اور اور اور اور اور او		LOAD	FACTOR
MISSION SEGMENT	MAX	MIN	CYCLES/MISSION
7. Ground Attack Tactics	7.4	-0.3	0.01
	6.3	0.1	0.1
	5.2	0.4	1
	4.8	0.5	Z
	4.3	0.6	4
	3.6	0.7	8
	2.7	0.8	16
	1.7	0.9	27
8. Supersonic Cruise	2.9	1.0	0.01
	2.8	1.0	0.01
	2.6	1.0	0.01
9. Weapon Delivery Turn	7.1	-0.1	0.01
	5.9	0.2	0.1
	4.1	0.5	
	4.1	0.6	2
	3.4	0.7	4
	2.5	0.8	8
	1.7	0.7	11
IC. Flight Maneuver	4.2	1,0	0.01
	4.1	1.0	0.01
	3.1	1.0	0.2
	2.0	1.0	
	1.9	1.0	
	1.5	1.0	2
11 Superconic Weapon Delivery	<u> </u>	1.0	
The Supersonic Weapon Delivery	4 1	0.4	
	1 1 8	1 0	0.1
	2.3	1.0	1.0
	2.1	0.7	1.0
12. Supersonic Descent	1 2.3	1 1.0	0.01
13. Enroute Navigation. Return	4.5	1.0	0.01
	4.0	1.0	0.1
	2.7	1.0	1
	2.1	1.0	2
	1.6	1.0	4
14. Subsonic Descent	4.5	j 1.0	0.01
	3.9	1.0	0.1
	2.9	1.0	1
	2.4	1.0	2
	1.7	1.0	i 4

TABLE 2. BASEPOINT FLIGHT-BY-FLIGHT COMPOSITE LOAD FACTOR SPECTRUM (COMBINED COMBAT AND TRAINING MISSIONS) (CONTINUED)

TABLE 2. BASEPOINT FLIGHT-BY-FILIGHT COMPOSITE LOAD FACTOR SPECTRUM (COMBINED COMBAT AND TRAINING MISSIONS) (CONCLUDED)

			LOAD	FACTOR
	MISSION SEGMENT	МАХ	21 W	CYCLES/MISSION
15.	Traffic Pattern	4.8	1.0	0.01
		3.7	0.1	0.1
		2.7	1.0	_
		2.3	1.0	2
		6.1	1.0	4
		1.4	1.0	60
16.	Landing	4.3	1.0	0.01
-		3.7	1.0	0.1
		2.4	1.0	
		1.8	1.0	-1
17.	Taxi	1.3	C.7	
		1,2	0.8	15

2.1.4 <u>Structural Maneuvering Design Criteria</u> - The structural design load factor requirements were typical standard values for Air Force air-to-ground tactical fighters.

Subsonic	+7.33g	-3.0g
Supersonic	+6.50g	-3.0g

Maximum maneuver roll rate for structural design was 270 degrees per second.

2.1.5 <u>Temperature Design Data</u> - Flight design temperature data were developed to complement the composite design mission leg described above. Table 3 presents the design temperature data. The design standard-day temperatures for each leg of the composite design mission are presented for critical and typical locations on the wing/fuselage structure; e.g., upper and lower surface, 1 foot and 3 feet back from the leading edge. Total stagnation temperature would exist inside the engine inlet duct.

2.1.6 <u>Vibration/Acoustics</u> - Approximate prediction of the vibration/acoustic environment was made using the basepoint aircraft flight envelope and operating characteristics of the propulsion system. Predictions for boundary layer and maximum power engine noise, and weapons bay acoustic environment are shown in figure 5.

2.1.7 <u>Reliability</u> - Air vehicle subsystem reliability, maintainability, and survivability characteristics were considered very important to operating costs and effectiveness of a tactical air-to-ground fighter. Design goals in these areas could significantly affect selection of subsystems concepts arrangements. Overall mission reliability allocation is shown in table 4. The basis for these allocations is reference 4.

			DESTGN TEMP		F	
		REPRESENTATIVE	TOTAL	FUSELAGE	WING	
MIS	SION SEGMENT	MACH/ALT	TEMP	} 		
1.	Taxi	0 / SL	59	59	59	
2.	Takeoff & Climb			ļ		
	Low	0.39M/SL	75	74	74	
	High	0.9M/32K'	12	16	15	
3.	Enroute Navigation			ł		
	Low	0.9M/31K'	16	20	19	
	High	0.9M/36K'	-6	0	-1	
4.	Aerial Refuel	0.7M/25K	12	18	17	
5.	Low Level Terrain Following	C.9M/SL	143	139	138	
6.	Supersonic Climb/Accel				1	
_	Low	0.9M/36K'	2	7	6	
	High	2.2M/51K	242	230	229	
7.	Ground Attack Tactics	0.9M/SL	143	139	138	
8.	Supersonic Cruise					
	Low	2.2M/39KI	242	226	225	
	High	2.2M/51K'	242	230	229	
9.	Weapon Delivery Turn			• •	1 1 1	
{	Low	0.95M/49K'	1	7	6	
	High	0.95M/51KI	1	14 14	13	
10.	Flight Maneuver			i I	L i l 1	
1	Low	0.9M/31K	16	20	19	
	High	0.9M/36K'	-6	0	-1	
111.	Supersonic Weapon Delivery					
1	Low	2.2M/39K	242	226	225	
	High	2.2M/51K	242	230	229	
12.	Supersonic Descent					
	Low	0.9M/36K	-6	0	-1	
	High	2.2M/51K	242	230	229	
ł						

TABLE 3. BASEPOINT COMPOSITE MISSION FLIGHT DESIGN TEMPERATURE DATA STANDARD DAY CRITICAL SOLAR EFFECTS INCLUDED

「おおいた

TABLE 3. BASEPOINT COMPOSITE MISSION FLIGHT DESIGN TEMPERATURE DATA STANDARD DAY

!

:

(CONCLUDED)	
INCLUDED	
EFFECTS	
SOLAR	
CRITICAL	

		DES	JGN TEMP ≥°	u.
MISSION SEGMENT	REPRESENTATIVE MACH/ALT	TOTAL	FUSELAGE	DNIM
13 Enroute Navigation, Return		-6	0	-
Low	0.9M/36K'	-6	0	
			2	Ð.
14. Subsonic Descent				
Low	0.39M/SL	75	79	78
High	0.9M/36K'	-9	0	ī
15. Traffic Pattern	0.39M/\$L	75	79	78
16. Landing	0°2M/SL	67	69	68
17. Taxi	1S/WO	59	59	59



.

TABLE 4. 2.2M MISSION RELIABILITY ALLOCATION PROBABILITY OF SUCCESS (Ps)

Subsystem	Baseline Ps
Avionics	0.9811
Power Plant	0.9934
Structure	0.9989
Armament	0.9983
Flight Controls	0.9964
Fuel	0.9947
Environmental	0.9947
Landing Gear	0.9976
Actuation	0.9976
Electrical	0.9984
Displays/Lighting	0.997
Auxiliary Power	0.9993
Crew Accommodations	.0.9995
Other	0.998
	Subsystem Avionics Power Plant Structure Armament Flight Controls Fuel Environmental Landing Gear Actuation Electrical Displays/Lighting Auxiliary Power Crew Accommodations Other

Total

P

4

í

0.95

.....

...

2.2 Basepoint Tactical Fighter Configuration

2.2.1 General - A representative airplane concept designed to the baseline mission requirements was established. Figure 6 illustrates the general appearance and features of the airplane. The airplane is primarily designed and sized to the high altitude 2.2 Mach number penetration mission but has significant alternate capability on the terrain following 0.9 Mach number mission. The propulsion system elements and blended wing body shaping are optimized for continuous supersonic operation. A retractable canard in combination with a vectoring 2-D nozzle provides good takeoff and landing performance with a modest installed thrust to weight ratio. The wing, canard and empennage are fabricated from advanced integral graphite/epoxy composite materials. The forward fuselage uses advanced super aluminum alloys primarily. High stress concentration areas and the aft portions of the fuselage, designed by the high ambient temperatures of the propulsion system and APU installations, are constructed of advanced superplastic formed/diffusion bonded titanium, including silicon-carbide fiber reinforced filament technol gy in selected areas. The wing is aeroelastically tailored and employs variable geometry features. The engine employs 3000°F turbine inlet temperatures, carbon/carbon nozzle technology and selected other advanced internal component design and material improvements. Thrust reverse capability is provided and facilitated by the 2-D nozzle configuration. The avionics installation includes a full complement of advanced technology offensive, defensive and M&TC equipment to deal with the sophisticated dual mission target and threat systems requirements. Advanced tandem mounted conformal weapons that provide standoff weapon deliveries against heavily defended targets are carried on the lower fuselage centerline. An advanced cockpit, designed around an increased seatback angle and multiple-function integrated displays to conserve space and fuselage depth, provides suitable forebody wave drag characteristics and low radar cross section (RCS) with a gold flashed canopy. The general fuselage shaping, inlet location, use of special antenna design treatments and radar absorbent materials at critical locations provides low RS characteristics.

The flight control system is characterized as a digital 3-channel fly-by-wire system with selected 4-channel portions for critical functions. The flight control will function as part of an integrated flight/fire/propulsion control system. Relaxed static stability control requirements are prescribed because it is believed that, by the 1990's, virtually all new tactical aircraft will incorporate this beneficial feature.





2.2.2 <u>Power Requirements, General</u> - In determining the power requirements the following approach was used. Aircraft power needs were generalized and established in three categories.

I Housekeeping Loads (see paragraph 4.1.4 and Appendix E)

Communications Engine Environmental Control Lighting Fuel System Information Management Navigation Target Acquisition Defensive Avionics

II Actuation Loads (see paragraph 4.1.4 and Appendix E)

Flight Controls Utility

Armament

III Total Power Needs

Sum of Housekeeping and Actuation Loads

With the exception of environmental control, all loads under the Housekeeping Load heading were assumed to remain constant throughout the study. This was done because they represented power supplied to static (avionic black box) type of devices and thus were not considered actuation functions. All the loads under the actuation load heading plus the environmental control system loads were assumed to involve actuation functions. As such they were initially considered proper subjects for this study and ones in which the type and quantity of the power supplied would vary as different actuation methods were used during the trade study activities.

in accordance with the contract an electrical load analysis was made per MIL-E-7016 in which it was assumed that all loads were powered electrically. It was further assumed that the electrical system was 270 VDC since this type of electrical power appeared to be a likely candidate for application to 1990's aircraft. The various aircraft mission segments considered and the since the load analysis are as follows:

K-1 Engine start
K-2 Warm-up/take-off
K-3 Climb
K-4 Cruise
K-5 Penetration
K-6 Combat (including gun operation)

- K-7 Descend
- K-8 Landing
- K-9 Emergency (one of the two generators failed)

Rockwell procedure for encoding this information is available in reference 36. The data was processed by a series of computer programs, and resulted in Power Source Utilization graphs such as figure 7, 8 and 9 of this report. The solid line in these figures represent the load, which is generalized into power expressed as amperes at 270 V. (The load can also be plotted as KVA or kilowatts with equal facility). The other (dotted) lines represent system capacity and interval ratings as indicated in sheet 1 of figure 7.

A separate power source utilization graph was created for each of the following Housekeeping Loads (fig. 7), actuation loads (fig. 8) and combined housekeeping plus actuation loads (fig. 9). These loads represent power supplied at the input terminals of the various output electrical devices, whether they are static black boxes or actuators. However, for the actuators their load was defined by their output (i.e. the load incident to driving compressors, powering control surfaces, etc.) and, therefore, until the type of actuator was defined, an actuator efficiency (or internal power loss) was assumed to determine the power required at the input terminals. For the purpose of this initial electrical load analysis an overall efficiency of 60% was assumed for all actuators.

The three horizontal dotted lines on each of figures 7, 8, and 9 were the interval ratings of the generator and represented;

- 1200 AMPS continuous capacity at 100% generation output (two 600 AMP 270 volt D.C. generators)
- 1800 AMPS available for 2 minutes (150% total generator capacity, derated to .91.5%)
- 2400 AMPS 5 second overload capability (200% generator capacity, derated to 91.5%)

Note the short term overload capability of electrical power supplies compared to mechanically powered hydraulic systems whose overload capability and continuous load capacity are essentially equal.

The initial electrical load analysis was documented in Appendix A of reference 8 is included as Appendix E of this report. A breakdown of the actuation loads, which provided the basis for the "lumped" actuation load shown in reference 8, is shown in table 5.

2.2.3 Flight Control System - The aircraft is a variable stability, Control Configured Vehicle (CCV) that employs variable camber leading and trailing edge devices, a 2-D vectorable nozzle, and a variable area canard. Static longitudinal stability is set with the canard fully extended at $\delta C_{\rm I}/\delta C_{\rm I}$ = +.020 and M = 0.2. Aircraft stability is varied by canard extension or Tetraction. This feature allows a higher trimmed CL_{MAX} with canard extended, and a reduction in wetted area for low C_L penetration and acceleration legs. Extension of the canard for transonic maneuvering allows the variable camber devices to deflect in the proper direction for both low drag and trim. Supersonic flight will require some canard extension to minimize trum drag. Table 6 summarizes

LEGEND

LOAD REQUIREMENTS -----

SISTEM INTERVAL RATINGS

5-SEC		200%	ł			
2-MIN	************	150%	Ł	0F	GENERATOR	CAPACITY
CONTINUOUS		100%)			

TIME INTERVAL DESIGNATIONS






....

Figure 7. Power Source Utilization (Housekeeping) (Sheet 2 of 2)

24



......**`**..

. ----

Figure 8. Power Source Utilization (Actuation)



Figure 9. Power Source Utilization (Housekeeping Plus Actuation)

26

....

Ì

TABLE 5. ACTUATION LOADS (KW)

Ł

	ENGINE	START	WARH-UP	17.0.	4112 C F IV	96	ceu:	5E	PENET	Nollan	87 87 9	F	DESCE	AN	LAND	NG	EMER	SENCY
	5 664	CM1	2 5 E C	CONT	5 SEC	CONT	5 366	CONT	SSEC	CONT	5560	FVO	5550	CONT	2352	CMJ	5365	CONT
	1	1	\$5.50	10.63	31.90	10.45	31.90	10.63	22.7E	10.63	53.7	3.9	5.75	/a.(3	15.42	59-4	24.59	59.Q
			5.32	6/-2	9 2-7	21.3	6.38	5.5	44.1	2./3	10.65	5.2	5.32	2.13	\$32	6.5	5.32	£.5
ર			/ 42	.57	1.70	24%	02.1	.67	86.1	ß	2.63	13.	1.42	-52	24:1	-57	1	I
			1.57	55	164	451	1.64	•55	16.1	25.	2.73	. 55	18.1	SS.	16.1	55	/6/	Ş
		ł	/ 30	ŝ	1.86	130	1.56	.52	181	.52	8.60	.śt	oK./	.52	× 10	i Ş	o£./	Ś,
3	l		3.74	64%	4.48	3.74	8 .49	64.1	5.23	64.1	14.1	64.1	3.74	/ 49	3.74	1.49	1	ļ
ŝ			. 29	1	ف 4	.29	¥.	1	¥.	Ņ.	Ŀ,	".	62.	1	.29	3	1	1
Ê	1	1	١		A 82	541	10.82	5.41	18.04	205	18.04	20.9	\$.6r	14.2	10.04	9.02	- 104	20.6
E		1	١	1	17.83	8.92	17.83	8.92	29.72	14.86	29.72	14.86		ļ	1		1	1
3		!	91.70	45.85	91.70	\$1.85	91.70	41.85	91.70	4585	01.19	45.85		1	1	1		ł
રે			ļ		ļ	I	1	l	191.28	70.64	14/.28	20.64	1		141.20	70.64		ļ
3	ł		1.40		1		1		1				1	1	**		1	l
(H (S) (S)			(16 I) (16 I)	\¥. (1€.1)			111	111		111			111	111	(16.1) (18.1)	4 . ((6)	111	11)
эĘ				1	1	1				1	ł		1	1			10.	۱
(5		ļ		ļ	I	1	!	1		1							01.	1
Ę	<i>)</i> 0.	io.	10.	ò				1					1	1	ò.	ò	1	ł
	11	!!	136.28	62.37	112.22	80.25	1.2.22	72.78	16.965	156.34	360.74	15,37	50.85 3.3.90	15.12	103 90 135.92	96.18	<u>52.73</u> 35./5	22.85
3.	76 60	89 .60	5 00		12.27	1		32 15	14.47	156.34	35.25	15451	33.90	21.41	/35.91	8; 96	35./5	22.85

TABLE 6. CANARD POSITIONS

MISSICN LEG	CANARD POSIT	ICN
Takeoff and Landing	Fully Extended	100%
Subscnic Cruise	Retracted	0%
Transonic Maneuver	Fully Extended	100%
Supersonic Cruise	Half Extended	50%
Supersonic Maneuver	Fully Extended	100%
Supersonic Penetration	Retracted	0%

canard positions versus mission legs, and table 7 summarizes surfaces/ devices.

The wing trailing edges are used for pitch and roll control and, as variable camber devices with the inboard trailing edge device, the primary pitch control for longitudinal trim. The trimming function is intended to be compatible with the variable camber function of each device. A primary aerodynamic advantage of an unstable aircraft is that control deflection needed for proper camber variation is in the positive lift direction. Nose-up moments produced by an unstable aircraft as it increases angle of attack requires downward deflection of a trailing edge flap for trim and to produce the increased camber for reduced induced drag. Camber variations require a different 'neutral' setting for each flap depending on mach number. A C_{mo} change can be provided by scheduling the 2-D nozzle versus M. Therefore, the inboard trailing edge device should have the 'neutral' point scheduled versus M and a deflection versus angle of attack schedule which itself may be a function of M. This variable camber system 'automatically' produces the highest C_I available with the given planforms and control surfaces. Takeoff and landing require different flap deflections to maximize lift at zero angle of attack, C_{Lo}. This second scheduling replaces conventional 'flap' settings and requires deflections of the canard and 2-D nozzle.

The midspan trailing edge flap, in addition to the inboard, is used for pitch trim and variable camber. Also, this flap is used for high speed roll control and may supplement the outboard trailing edge flap for low speed roll control. This midspan surface is used for high speed roll control in order to avoid any control reversal on the outboard surface at high q.

The outboard trailing edge is the primary low speed roll control Its function as a variable camber device is somewhat restricted in low speed flight in order not to use up control authority needed for the low speed time-to-roll requirement. Aileron 'droop' in the takeoff and landing mode can be considered useful.

The leading edge devices are primarily variable camber devices requiring scheduling versus M and angle of attack. A fixed takeoff and landing position has advantages.

The thrust vector vane is primarily a moment producing device. It optimizes flap deflections to obtain minimum drag and maximum lift. This requires scheduling as a function of M and angle of attack.

2.2.3.1 <u>Flight Control Actuator Requirements</u> - In arriving at the flight surface hinge moments, use was made of HiMAT generated data. The HiMAT wing planform is similar to the baseline vehicle. The ATS hinge moment data was estimated by ratioing the relative areas, chords and dynamic pressures, i.e.,

$$(HM)_{ATS} = \frac{(S)_{ATS}}{(S)_{HIMAT}} \frac{(C)_{ATS}}{(C)_{HIMAT}} \frac{(q)_{ATS}}{(q)_{HIMAT}} \times (HM)_{HIMAT}$$

TABLE 7. SUMMARY OF CONTROL SURFACES/DEVICES

<u>Con</u>	trol		Travel	
1.	INBOAF	D TRAILING EDGE	-30 ⁰ , +	+450
	a.	Primary pitch control $f(M, \prec)$		
	ь.	High lift device and decambering $f(M)$		
2.	MIDSPA	N TRAILING EDGE	-30 ⁰ , +	450
	a.	Pitch control f(M , A)		
	ь.	High lift device and decambering $f(M)$		
	c.	High and low speed roll control		
3.	<u>007304</u>	RD TRAILING EDGE	<u>+250</u>	
	a.	Primary low speed roll control		
	ь.	Decambering device f(M)		
4.	CANARE			
	a.	Variable stability device - see table I		
5.	LEADIN	G EDGE DEVICE		
	a.	Variable camber device $f(M, r)$		
6.	THRUST	VECTOR VANE	+200	
	a.	Pitch trim f(M ,)		
7.	RUDDER	<u>s</u>	<u>+</u> 25 ⁰	
	a.	Directional control		
	ь.	Speed brakes		

The ATS rudder hinge moments were computed from the ATS midspan trailing edge flaps (the HiMAT has all movable flip-flcp verticals). The ATS canard loads were estimated from the canard lift force calculated at maximum design load factor (7.3 g's) at maximum dynamic pressure (1750 PSF). The leading edge design loads were based on an assumed 10 psi maximum pressure acting on an arm equal to one half the flap chord. The thrust vector actuator loads were taken from NASA CR 135252 which used an exhaust nozzle and engine similar to the baseline ATS.

Three of the trailing edge control surfaces were examined for flutter stiffness. Estimates of frequency were derived from empirical data. For the inboard flap, the frequency was estimated to be 47.9 hz, 95.8 hz for the upper rudder. The mach number used was 1.25 while for the aileron the Mach number was .95. The stiffness computed from the empirical data was increased by 50 percent to account for surface wind up. The backup structure was thus assumed to have equal stiffness with the actuator. The stiffness requirements for the three surfaces considered are given in table 8 which presents the performance requirements for the actuation systems for the various flight control surfaces/devices identified for the baseline vehicle.

The actuation requirements listed in table 8 need not be satisfied simultaneously. For a worst case analysis, the simultaneous requirements are: 1) .70 percent for inboard and midspan flaps, 2) 50 percent for the rudders, and 3) 100 percent for the allerons. The other flight control actuation devices can be assumed to be stationary during this simultaneous demand.

As indicated earlier, the flight control system is a fly-by-wire design with fail-operate-twice capabilities for the flight critical surfaces. These surfaces are at least triple redundant, electrically and mechanically.

For the other surfaces/devices, triple redundancy is provided electrically but only dual redundancy mechanically. For the flight critical surfaces full actuation performance is required after any two electrical or mechanical failures or any combination of single electrical and mechanical failures. Any further failure will not produce a hardover deflection. For the non-flight critical surfaces/devices, full actuation performance is required after any single failure. After any further failure, the surface/device will be capable of being recentered and locked.

The actuator position loop closures and the monitoring/switching provisions will be included in the actuation system. Electrical digital signal inputs, when used, will be 5 volts.

Table 8 specifies frequency response in two methods. Method one establishes actuator output response requirements for a fixed input signal of increasing frequency, and method two establishes output response requirements for fixed amplitude output at increasing frequency response. Loads are considered to be essentially inertia. For purposes of this study linear load variation with stroke/rotation will be considered.

2.2.3.2 <u>Utility Actuation</u> - Table 9 summarizes utility actuation characteristics.

and a support of a local sector of a

FLIGHT CONFROL ACTUATION REQUIREMENTS TABLE 8.

•

-

LOA	E CH	ARACIERIST	105 (3)	NO. OF	POWER PER	ļ	Θ	TOTAL ERROR			FAILURE	ANPLIT	E CO
NO LOAD DESIGN	DESIGN		MAX. LOAD	SURF -	SURFACE AT	202	RESO -	SUR (E) OF	200	STIFF -	MODE	INPUT IC	UIPU-
(MAXIMUN)LOAD AND RATE RATE	LOAD AND RATE	-	AND/OR STALL	ACES	DESIGN (3)	CYCLE	LUTION (R)	RESOL	REQUENCY	NESS	NERTIA(1)	Θ	0
100" PER 161,664	161,664	1	215,552				R ≤ 0.05 %	E < 0.2 %	MPL 200	182 1106	FOLFS	+1	
SEC. IN-LB AT	IN-LB AT		IN-LB.	2	15.95 KW		FULL		XAM"CE- 0	וא-ום/	1=19330	0 25	<u>.</u>
SO / SEC.	50 / SEC.						STROKE	STROKE	AT 3HZ	/RAD	LB-IN ²		
100 PER 32,344	32,344	~	43,125			z	R ≤ 0.05 Z	E ≤ 0 27.	BOS JUM		FO ² F5	+1	
SEC. IN-LB AT	IN-LB AT	_	IN-LB.	~	3.9 Kv/	.e e	FULL	FULL	6 -30"1141			0.25	ġ
50 PER 21.453	21.453		25.603	ļ		ε	R5 0.27	E 1 0 5 7 0	807. John	4.57×105	FO FS	- -	
SEC. IN-LB AT	IN-LB AT		N-LB	~	0.85KV/	đ٢	FULL	FULL	0 - 45 MAX	IN-18/	1 = 914	.9 0	<u>.</u>
20 */5EC	20*/5EC	-				÷۷	STROKE	STROKE	AT 2 H 2	RAD.	1 NI-81		
40 PER 13.671	12.671		18,495			1.6	R \$ 0.2 %	E 4 0 5 %	AMPL - 208	2 56AI04	F02 F5	* 1	
SEC. IN-LBAT	IN-LB AT		IN-LB	2	0 82 KW	ξ	FULL	FULL	A -30 MAY	N LB/	1 = 6.07	0.5	ö
30°/ 5EC.	30°/ 5EC.				_	٤	STRUKE	STROKE	AT 3H2	d W	2 NI -01		
40 PER 19,816	19,816		26,421			ď	R≤02%	E < 0.5%	AMPL : 208		FO ² FS	+1	
SEC. IN-LBAT	IN-18 AT		1N-15	~	C-78 KW	5	FULL	FULL	¢ - 30°MAX			0	ò
201/540.	20/250					00	5TRONE	51KOKE	AT 3HL				ŀ
5 PER 413,032	413,032	-	520,335			G	RS 0 57	E \$ 1.02	AMPL , JUMA		Fo FS	9.0	0
SEC. IN-LBAT	IN-LBAT	_	IN-LE	~	2.24 KW	×6	FULL	FULL	A - 45 MAK			LE D	Ē
2.4 %/SEC	2.4 %SEC					• =	STROKE	STROKE	AT 2 HZ			-	
3 PER 2000 LB.	2000 LB.					- "	R ≤ 0.57	E 4 10%	AMPL * 200		FO FS		<u>o</u> '
SEC. AT 1.0 /360	AT 1.0 /3EC		2400 LB.	~	0.17 KW	111	FULL	FULL	45°MW			E X 1.	1.1
2.3 IN SEC 077 IN/SEC	077 IN/SEC	_				15	STROKE	STROKE	<u>N 2 H 2</u>			046NG	<u>z</u> 5
3. PER 64,01718	64,01718					EE	RS 0.47	E 1.07	1401 : 308		FO FS	0. 1	0.1
SEC. ATLO /SE	ATLO / 56	J	75,314 LB	~	0.72 KW	d 	FULL	PLL	Ø -45"MV			10101	0.0
0.31 IN/SEC 0 10 W/SEC	0.10 V/SEC						STROKE	STROKE	AT 2 H 2			ž	ž
TUGE UP		Θ	INPUT SU	ENAL	AMPLITUO:	-NOU -	STANT AT	EO.	FAIL OP	- RATF			

TEU = IKALING EDGE UN TED = TRALING EDGE DOWN LED = LEADING EDGE DOWN EXT = EXTEND

. -----

U INPUT SIGNAL AMPLITUDE CONSTANT AT
 IMPLS SESOUTION (METHOD ")
 IMPUT SIGNAL AMPLITUDE WARED TO MAINTAIN OUTPUT AMPLITUDE CONSTANT (METHOD "2)
 AT CONTROL SURFACE

TU - TAIL UPERATE FO² = FAIL OPERATE FS = FAIL SAFE OB = DECIDELS

BASIJUNE UTILITY ACTUATOR REQUIREMENTS TABLE 9.

L				10 OACTED			CALLED AT			ALL CA		
		TRME				OPERATING	NTERFACE	PAILUKE		5074	5@	COMMENTS
			STALL	OPERATING	LOAD/RATE	DAD-RATEO	3	RESPONDE	AC.	HYD	ELEC	
Е	EXTERNAL	10.0 IN	8600 LB	7 310 LB	5.00 1%	413 KW (5.54HP)	1	FAIL SAFE (FS)	4	HLA	MBS	MODULTES
NIS	PLUG THROAT	9.25 IN	57,400 LB	46,790LB	4.6'3 ^{IN} /sec	25.52 KW (34.21 HP)	28.67 KW (38.43 HP)	(FS)	2	MPT	TqM	POWER TRAIN - 36 BALL SCREWY, 95 RT 2 RED. GERBOUK 96 LURKAR - 89 10,000 ROMAT NITERACE. MODULATES
EN	THRUST	12.32 M	16,60018	14,110 LB	12.32 ¹¹ XEC	16.89 KW	19.64 KW (26.33HP)	(FS)	2	1dW	Τqμ	2 POSITION - POVER 16411 1 = 99 WHUT X - 95 GB" 1 × - 95 GB" 2 × 96 DALL SCREW = 166 - 10,000 RDM 46UT
<u> </u>	NOSE GEAR	5,00 IN	1	14,900LB	13.03 Mysec	(1.88HP)		(FS) OPERABLE	-	HLA	MBS	
/39 !	MAIN GEAR	6.00 M		24,00018	1.DO IN/SEC	(3.63HP)	1	AFTER ONE POWER	2	HLA	MBS	
DNIQ	NOSE GEAR	±101°	1	18,000mu	120/5EC (2 RPM)	0.43KW (0.58HP)	0.48KW (0.64HP)	(FS) TRAIL	1	MPT	MPT	GEAR TRAIN N= .90 MODULATES PRODOD RPM AT INTERFACE
רעא	MAIN GEAR BRAKES	0.10 IN	75,282LB	33,800LB	0 50 Kec	1.91 KW (2.56HP)	3.18 KW (4.27 HP)	(E3)	と	đΣ	đ	HYD. CIRCUIT 6 = 87 MOTOR 7 .80 PUMP x .80 VALVING = .60 18.00 RM 1 AT INTERFACE
.2YS.	RIGHT RAM AIR SCOOP	2.52 IN		4100 LB	2.0 IN/SEC	0.93KW (125HP)		(FS)	-	HLA	MBS	
TN.	LEFT RAM AIR SCOOP	1.00 IN	± 1200 LB	±650 LB	out5 Mged	0.01 KW	1	(FS)		HLA	:485	
IR. CO	EXH. DOOR EQUIP BAY	2.66 IN	1300LB	81 006	ASKIEU.O	0.01 KW		(FS)	-	HLA	MBS	
٤ил	GRD COOLING FUEL HT: SINK	°06	ISON-LB	€1-NI 06 ∓	10°//3EC	.003KW (004 HP)	.006 KW (008 HP)	(FS)	-	MBS	MBS	LUMA 5 FLECTRICALLY ACTUMED- 25000 PPM AT INTERFACE - TOO SHALL A POWER RED FOR HY DRAULICS
האכד.	ARMAMENT	G.E. 30MM (< 5.0 36C.B	CANNON É AN NRSTS - NO O	MUNITION FE	ED SYSTEM	25.05 KW (47, 00HP)	1	(FS)	-	Σ	Σ	PROVATIVIED FIRING CYCLE WITH REVERSE ROTATION AT END OF CYCLE TO RECOUP UNFIDED SHELLS
1 YTL	ENGINE START	MAX. STAR	TING TIME P	ER ENGINE -	40 SEC	89.49KW (120 HP)	1	(FS)	2	ATS	GEN	
1110.5	REFUEL RECEPTALE	(3000 PSI	UNERSI) REPU HYDRAULIC)	HELANG RECE	PACLE	0.54 KW (0.72 HP)	0.83KW (1.11HP)	(F3)	-	dΨ	dμ	HYI' CIRCUIT N= .85 PUHPX .80 VALVE X.95 ACTUATOR = .45 20,000 RPH AT WTEURILE, 2 POSITION
รเพ												
Θ	POWER PER	ACTUATOR D	JELIVERED /	NT THE SUR	FACE OR FU	NCTION BEI	NG OPERATE	D B H	P = M(R TURBI	NE STA	RTER GEN=GENERATOR M = MOTOR
9	POWER DELN WHICH EVER WHETHER TI	rered at a 1 Nothing Do He power 3	MOTOR TI) PI WNSTREAM	UMP, CX MO . Sctween YDRAULIC O	TOR TO GEA IT AND THI R ELECTRIC	R TRAIN, IN F FUNCTION,	rerface in 15 unchang	Z H Z	884 84 84 84 84 84 84 84 84 84 84 84 84	DTOR - E	JALL SC IC LINI OWER 5, BALL	CREW EAR ACT. TRAIN (MAY INCLUDE .SCREWS, & GEARS)

Contraction of the second s

2.2.4 Environmental Control System - The environmental control system (ECS) in the aircraft provides proper conditions for crew, avionics, and missiles. The crew requires cockpit pressurization, heating, cooling, and ventilation in addition to windshield and canopy defog, windshield anti-ice and rain removal. The missiles require air for pre free flight conditioning and the avionics requires the removal of self generated heat. A breakdown of the maximum continuous heat loads imposed on the ECS are shown in Table 10.

In order to provide the necessary data for the initial electrical load analysis (see Appendix E) an ECS system, felt to be representative of the type which might result from this study, was hypothesized. The resulting system is shown in figure 10. The system derives most of its input power from shaft inputs at its various compressors, pumps, and blowers. These shaft inputs could be supplied either directly from the engine or via electric or hydraulic motors. A small amount of power (<10%) was derived from bleed air to pressurize the cockpit and provide makeup air.

Figure 10 shows that the maximum continuous power required per the electrical load analysis was 80.17 K.W. This loading was considered to exist during penetration and combat. During all other mission segments the maximum continuous load was considered to be 44.22 K.W. The five second peak loads were 136% of these values and were 109.03 K.W. and 60.14 K.W., respectively. Figure 10 also shows that heat is rejected to fuel and lists some of the salient features of the system.

2.2.5 <u>Aircraft Configuration</u> - The inboard profile for the aircraft selected as the baseline for the study is shown in figure 22. The figure also lists most of the major subsystem components used in the aircraft as it was originally conceived. Further discussion of this configuration will be found in paragraph 4.1.3.

2.2.6 <u>Armament Subsystem</u> - The gun carried by the aircraft was a GE430 four barrel, 30 mm Gatling type. The gun required 22 hp steady state while the linear linkless feed system (LLFS) required 25 hp. The total gun requirement was 47 hp.

Air-to-air (ATA) missiles were semisubmerged on the underside of the fuselage. The missiles were launched by forcible ejection; jettison mode was provided for emergency release. Electrical, hydraulic and environmental air conditioning lines run through the fuselage to connect with the missile for preflight conditioning and release.

Design mission air-to-ground weapons were conformal and were carried in tandem. Standard ejectors supported the weapons and provided forcible ejection for safe separation under all aircraft flight conditions.

2.2.7 Engine Starting Loads - Figure 11 shows the starting characteristics for the F404 GE 400 Engine. This was a 16000 1b S.L.S. thrust engine which required 125 starting horsepower on a Standard Day for a 35 second start. The F-18 Sundstrand Starter ATM08 used on the engine could generate 167 HP on a standard day.

TABLE 10. ECS HEAT LOAD

Subsystem	Heat Load KW	
Cockpit		3,51
Avionics Armament Communications Engine Information Mgmt. Sys. Navigation Target Acquisition Defensive Subsystem A/V Electrical System	2.11 1.93 .79 2.62 .59 14.67 21.50 .50	
25% Growth	44.76 <u>11.19</u> 55.95	55.95
		59.46
Cockpit Pressurization		4.27
*Composite circulation loop efficiences		63.73 .935
Required output of pumps & blowers		68.18

* 0.93 Freon Loop 0.98 Air Loops 0.95 Liquid Loops



:



NOTES:

(1) No customer air bleed during starting.

-50

-40

-30

150

(2) Power extraction permitted between smrter cutout and ground idle, up to 34 HP providing starter cut out is at least 5900 rpm \pm 200 rpm.

-20

-10

Q

AMBIENT TEMPERATURE (C)

10

20

30

40

50

(3) Minimum engine firing speed 3900 rpm.

Figure 11. Starting Torque and Speed Requirements (Sea Level Static Conditions) The ATS engine SLS thrust is 9912 lbs and, proportionately scaling starting H.P., would call for slightly more than 105 H.P. However, accessory drag will not decrease proportionately, therefore, for purposes of this study, starting horsepower was established at 120 H.P. (89.5 K.W.).

2.3 Historical Review

2.3.1 Data Revisions and Additions - Subsequent to the definition of the baseline aircraft, as discussed in paragraphs 2.1 an 2.2 certain baseline aircraft requirements data were revised or expanded and clarified. These data items are discussed in the following paragraphs:

2.3.1.1 <u>Ground Cooling Fuel Heat Sink Door</u> - The ground cooling fuel heat sink was originally included but was subsequently eliminated as an item in the trade study. This arose from the fact that the power requirement was so small that it would not have been practical to perform the function hydraulically. The smallest motor it is feasible to manufacture delivers approximately 700 times the power required for this application. A linear actuator sized to perform the function would have been so small that the port bosses necessary for providing extend and retract pressure would have represented more that 50% of the total volume of the actuator. These small power functions have historically been, and will continue to be operated electrically. Therefore, it was assumed that this function would be performed electrically on both baseline aircraft. This is shown in the "comments" column of Table 9.

2.3.1.2 Canopy - The canopy was also originally included and later eliminated as an actuation function. Extensive study indicated that the canopies should be opened and closed manually and thus should not be a utility actuation function. The decision to return to manual operation came during a rather detailed analysis, in terms of weight and reliability. of the actuation system (electrical or hydraulic) necessary to open or close the canopies. This analysis considered the requirements with the power supply system failed or during ground maintenance with no power on the power supply system. To meet these requirements the system became very complex and heavy, particularly when meeting the range of conditions which could reasonably be expected during normal operation. Complicated as it was such an actuation system would have been necessary had the canopy system consisted of one large canopy rather than two separate smaller canopies. However, since the canopy system was broken into two canopy units the airloads acting on each unit and the weight of each was considerably less. Thru the use of counter balancing, in the form of torsion bars or gas springs, the static deadweight loads could be nearly eliminated and each canopy unit could be handled easily under all but the severest gust load conditions. Since each canopy unit was automatically locked open (manual unlock to close) and manually locked closed, all functions were manual. Even though opening the canopy against the most severe adverse wind loads would take considerable physical effort, the fact that the unit locked open automatically meant that this level of effort had to be sustained for only a short period of time.

Manual operation of the canopy offered many benefits to the aircraft a few of which were as follows:

- A. Reduced aircraft weight
- B. Simplified power distirbution and utilization system whether electrical or hydraulic.
- C. Improved aircraft reliability under normal operating conditions.

- D. Improved maintainability arising from both the simplified actuation system and from improved access to the cockpit during routine maintenance (i.e. the canopy can be opened and closed an indefinite number of times without the need for batteries, accumulators or ground power.
- E. Improved emergency access or egress.

2.3.1.3 Utility Engine Actuation Functions - Two of the three major power users during the critical combat phase (i.e. the operational phase which determined power generation system size) were the plug-throat and the thrust reverser. Figure 12 shows the actuation mechanism which operates these functions plus the external flap and thrust vector vane. It will be noted that the plug-throat, the thrust reverser and thrust vector vane were all operated via various power trains driven by actuation devices mounted in the sidewall at the ends of the rectangular engine exhaust duct. The plugthroat was driven by a shaft running down the center of the intermediate sidewall cross tie while the thrust reverser was powered by a shaft mounted on the front face of the aft side wall cross tie. In this way the motors or linear actuator (i.e. the power transducers) could be mounted in the relatively cool sidewall area while the shafts, gearboxes and ballscrews, which were more temperature tolerant, were mounted in the hot (400°F with cooling-1000°F without) nozzle area.

Originally it had been hoped that the thrust reverser power transducing devices, for the electrical system or the hydraulic system or both, could be mounted in the hot area. This would have made possible the elimination of all, or a large part of, the existing power train in the hot area and thus would have reduced complexity and weight. Towards this end an envelope was given in Figure 2-12 defining thrust reverser installation requirements assuming all actuation devices were in the hot area. Subsequent studies however, indicated that the ductwork necessary for, and the induced drag increase associated with, attempting to cool the envelope area down to 400°F max would offset most of the potential weight savings and this approach was dropped. It was finally assumed that the thrust reverser and the plugthroat would both be operated by 10,000 RPM motors (electric or hydraulic) mounted in the relatively cool engine exhaust nozzle side wall.

2.3.1.4 Environmental Control System - As previously pointed out (paragraph 2.2.4) an ECS System was hypothesized to make possible the determination of ECS loads in case the ECS system became an integral part of the trade study. It was also pointed out that the system derived most of its input power from shaft inputs at its various compressors, pumps and blowers. These shaft inputs could have been supplied either directly from the engine or via electric hydraulic motors. A small amount of power (4.27 KW) was derived from bleed air to pressurize the cockpit and provide makeup air. The balance was supplied in the form of shaft power inputs.



Figure 12 Engine Actuation Functions

A STATE OF A

and the second se

and the second second

and an and a state and the second state of the

41

Figure 10 shows that the maximum continuous power required was 80.17 KW of which 9.27 KW must be carried by the electrical system. This loading was considered to exist during penetration and combat. During all other mission segments the maximum continuous load was 44.22 KW.

After considerable study it was determined that the ECS system for the aircraft which were to be traded (i.e., aircraft I and aircraft II, see paragraph 2.4.6 for definition) could, and should, be identical and thus should be dropped as an item in the trade study. The factors which led to this determination were as follows.

1. The heat loads seen by the ECS system in both aircraft were identical. This was true because the only heat load variations between aircraft were those generated by the actuation systems and all actuation systems, whether electro mechanical, hydraulic or integrated actuator package type, reject heat to their surrounding ambient (air or fuel) and not to the ECS system. The heat load represented by cockpit, avionics (air) and avionics (liquid), Figure 10, was unchanged for all study aircraft. Hence, the blowers, pumps and heat exchangers servicing these heat loads were unchanged. In effect the ECS dropped out as a trade study item except for the impact of its small load requirement on the two different types of electrical power generation systems used in each aircraft. The only additional factor which was considered was the diffences between actuation approaches with reference to their heat load impact on the fuel. For those approaches which tended to overheat the fuel, a weight/cost/reliability penalty was assessed in term of increased condenser size and/or the addition of auxiliary fuel cooling equipment.

2. The compressors were shaft driven. This approach was selected because direct shaft power extraction was much more efficient than having an intermediate hydraulic or electrical transmission link (97% shaft vs 72% with electrical or hydraulic). Since the compressor represented by far the major load imposed on the propulsion system by the ECS system, this increased efficiency was greatly to be desired. The desireability of direct shaft power extraction was further enhanced by the fact that the compressor should be close to the condenser, which must be in the final fuel inlet line to the engine, which, in turn, was adjacent to the AMAD (see Figures 17 and 22).

3. All blowers and circulating pumps were electric motor driven in both aircraft I and aircraft II. This decision was based on historic data and B-1 aircraft experience. Historically, on most aircraft, ECS blowers and ECS coolant circulating pumps have been electric motor driven and the results have been generally very satisfactory. In contrast, on the B-1 aircraft, several blowers and coolant pumps were hydraulic motor driven. The results have not been satisfactory. The motor ripple frequency interacting with vibrational frequencies generated by the coolant pumps and blowers have caused erratic unpredictable premature failures. This fact, plus the fact that the individual pump loads (.87 KW) were so small that they are well below the power capability of the smallest 8000 psi hydraulic motor it will be practical to manufacture in the 1990+ time period (i.e. practical minimum 4.0 KW), indicated quite clearly that these motors should be electric. The blower motor loads were also marginal but the deciding factor in this instance was the fact that the blowers were in pressurized compartments (avionics and cockpit) where it was desireable not to have fluid leaks.

2.3.1.5 Redundancy Definitions and Percent Output Load Requirements - A table was prepared (Table 11) to define what constituted fail safe for the various flight control actuation functions. The same table also more accurately defined the actual output load requirements for the various flight control functions in the presence of failures. This definition stated in essence that after first failure all actuation functions must retain 100% load capability. This was in line with the overall air vehicle requirement that the aircraft be able to complete its mission after any single power generation and/or distribution system failure (hydraulic or electric). The refined definition also conformed with air vehicle requirement by stating that after any second failure, each critical actuation function must retain sufficient power capability to allow the aircraft to recover from any manuver and return to base. For this purpose the residual power capability of each actuation function must range from 50 to 70%. The amplified and redefined load requirements are shown in Table 2-11. It can be seen in this table that the rudders required only 50% capability after two failures while the inboard (and midspan) flaps required 70%.

2.3.1.6 Installation Envelopes - In order to more accurately define the problem of installing actuation devices in the ATS aircraft three installation envelopes were provided as additional requirements data. These envelopes were for the outboard trailing edge (Figure 13), the inboard trailing edge (Figure 14), and the thrust reverser (Figure 12). The outboard trailing edge represented the smallest chordal thickness application, the inboard trailing edge represented the highest hingemoment and highest power application, and the thrust reverser represented the hottest operating environment in the flight control system. It was felt that, if satisfactory actuation devices could be provided for these applications, a satisfactory actuation device could be provided and defined for all flight control and utility functions on the aircraft without the need for detailed design in every instance.

2.3.1.7 <u>Baseline Utility Actuator Requirements</u> - In the course of the analysis leading to the preparation of Table 9 certain changes were made which impacted the basic study. The changes made were as follows:

- 1. It was decided that the brake system would remain hydraulic for both versions of the study aircraft (hydraulic and electrical). The reasons for this decision are discussed in paragraph 4.2.1.2.
- 2. It was decided that the standard (UARRSI) inflight refuel receptacle would be used in both aircraft I and II. The impact of this decision is discussed in paragraph 4.2.1.3.
- 3. As discussed in paragraph 2.3.1.1 the ground cooling fuel heat sink door actuation was eliminated as a trade study item. Table 2-8 reflects this fact by noting that this function is always electrically operated.
- 4. As discussed in paragraph 2.3.1.2 the canopy was eliminated as an actuation function.
- 5. Table 2-8 reflects the revised interface load requirements resulting from the decision relating to utility engine actuation functions discussed in paragraph 2.3.1.3.

TABLE 11. ACTUATOR OUTPUT LOAD REQUIREMENTS

REDUNDANCY REQUIREMENT	PERCENT OF OUTPUT LOAD (ESTAB. IN NA- 79-378-12, TABLE 3-3) REQ'D PER ACTUATOR	ACTUATION FUNCTIONS AFFECTED	ACTUATORS PER SURFACE
FAIL OPERATE- FAIL OPERATE- FAIL SAFE (FO ² -FS)	70%	(1) INBOARD FLAPS (1) MIDSPAN FLAPS	3 3
FAIL OPERATE- FAIL OPERATE- FAIL SAFE (FO ² -FS)	50%	(I) UPPER RUDDER (I) LOWER RUDDER \$ SPEED BRAKE	3 3
FAIL OPERATE- FAIL SAFE (FO-FS)	100%	(2) CANARD (1) AILERON	2 2
FAIL OPERATE-	⁽⁴⁾ 33½%	(3) LEADING EDGE FLAP	6

- FAIL SAFE DEFINED AS FOLLOWS:

(I) LOCKED IN TRAIL OR FLOATING WITH DAMPING

- (2) LOCKED IN FAILED POSITION
- (3) LOCKED IN BLOWBACK POSITION

- TABLE 3-3 (REF. NA 79-378-12) LOADS ARE CLARIFIED AS FOLLOWS:

(4) LOADS ARE THOSE FOR THE COMPLETE LEADING EDGE FLAP ON ONE WING (POWERED BY 6 ACTUATORS). AS AN ALTERNATE THE FLAPS ON EACH WING MAY BE BROKEN DOWN INTO 3 SEGMENTS POWERED BY 2 ACTUATORS PER SEGMENT





6. Table 9 defined the types of actuators which will be used in aircraft I (electrical) and aircraft II (hydraulic). The selection of these actuator types is based on the analysis made in paragraph 4.2.1.6.

2.3.1.8 <u>Baseline Flight Control Actuator Requirements</u> - Table 12 was created as an amplification of Table 8. It added data defining "Power at Interface" (explained in footnotes of Table 12), data on "Actuator Type" and additional functional information in the comments column which did not appear in Table 8. BASELINE FLIGHT CONTROL ACTUATOR REQUIREMENTS CABLE 12.

i

FOR (MBS), FOR (HLA) SEE (3) FOWER TRAIN EFF. =. 75 FOR (MLA) SEE (3) POWER TRAIN EFF. =-75 POWER TRAIN EFF. = .42 POWER TRAI .. EFE = 75 OWER TRAIN FFE =. 75 POWER TRAIN EFF =. 75 DOWER TRAIN EFF = .75 POWER TRAIN EFF.=.75 COMMENTS MBS HdW MBS LOAD CHARACTERISTICS DESIGN O INTERFACE FAILURE NO ACT ACTUATOR ΞE Han Hdw Hdw Hdw HAM HAM HYD LELEC HLA HUH HLA Hdw Hom Hom 2 S 4 ৩ 9 Ч Ч ف 4 ہ F02 F0² FOZ F0² FO FS ŝ ЪS 50 50 FS FS FS 10 ЪS Г € FS 0.52 KW (O. TOHP) 0.96 KW (1.2BHP) (DESIGN) (DESIGN) AT SURFACE POWER/ACT **3.02KW** (4.05HP) (0.74HP) 2.37 HP) 0.23KW (0.31HP) 14.87KW (919.93HP) 1.7.7 KW (1.49 HP) 0.55 KW (. [] KW 0.39KW (0.53HP) 0.75 KW (0.97 HP) STALL OPERATING LOAD RATE POWER ACT. ((4.94 HD) 2.27KW 0.83KW (0.56 HP) 1.00 HP) (0.23HP) 0.72 KW 0.41KW (3.04 HP) D.IJ KW 11. IS KW (9H 11-1) 2.4 %SEC ±2.1011/75,514 LB/64,017 LB/0.31 IPS 20°/SEC 20 /SEC 30 %SEC 50 %5EC 50 %5EC 2,40018 2000 LB 2.3 1PS 1N-L8 153,000 10,000 IN-LB 23,000 1N-LB 21,000 113,000 IN-LB 2,000 IN-LB IN-LB 9,000 30,000 IN-LB 1N-LB 13,000 73,000 26,000 N-LB 151,000 11-18 N-LB IN-LB 1 15.36N. ± 20° + 20° +45° ±25° ± 20° TRAVEL +45° - 30° - 30 ° 0 EDGE FLAP THRUST VECTOR VANE CANARD FUNCTION AILERON LEADING RUDDER MidSPAN RUDDER LOWER INBOARD UPPER FLAP FLAP

O POWER PER ACTUATOR DELIVERED AT THE SURFACE OR FUNCTION BEING OPERATED

POWER DELIVERED AT A MOTOR TO GEAR TRAIN INTERFACE WHERE THE GEAR TRAIN CONSISTS OF A GEARHEAD, A SHAFT AND A MECHANICAL HINGE FOR (MPH) AND OF A GEAR TRAIN AND A BAILSCREW FOR (MBS) Ð

USE STALL LOAD AND PRESSURE TO DETERMINE PISTON AREA AND USE RATE TIMES AREA TO DETERMINE POWER FOR (HLA) 1

ffo² = fail operate - fail operate

= FAIL OPERATE = FAIL SAFE **(b)** FO

F3

HLA - HYDRAULIC LINEAR ACTUATOR S. MPH - MOTOR - POWER HINGE MBS = MOTOR · BALL SCREW

2.4 Trade Study Ground Rules

2.4.1 <u>Hydraulic System Burst Factors</u> - Burst factor is essentially a safety factor applied to hydraulic system components and represents the ratio between the system's normal operating pressure and the minimum pressure at which any component in the system will burst. The higher the factor the thicker the walls of pressure containing components, such as valves, tubing, fittings etc., tend to become. For this reason burst factors are a major determinant of hydraulic system component size and weight. As discussed later (paragraph 4.2) 8000 PSI was selected as the normal operating pressure for the hydraulic system in aircraft II. Since 8000 PSI is considerably higher than the current standard operating pressure of 3000 PSI, it was felt that a study should be conducted to determine whether the burst factor for an 8000 PSI system should be higher, lower, or the same as the burst factor of 4.0 currently used for the design and test of 3000 PSI systems. Such a study was conducted.

The results of this study are summarized in Table 13 and show that a burst factor slightly less than 3 can be used in an 8000 PSI system (a burst factor of 4 is the standard requirement for conventional 3000 PSI hydraulic systems). The study was based on an aircraft pressure change duty cycle as follows:

- 1. 5000 system start-up, shut-down cycles
- 2. 200,000 rapid valve closures and openings
- 3. 1 X 10¹¹ pump ripple cycles

Item 1 above represented the number of times the system was pressurized and depressurized, both prior to and after flight and on the ground during ground servicing. Item 2 above represented a composite of all the wide ranging pressure excursions which occur in the system predominantly as a result of rapid opening and closing valves. The composite pressure changes (200,000 cycles) were based upon rapid valve closure in a line in which fluid was flowing at 25 ft per sec. It is interesting to note that the magnitude of the pressure perturbations caused by valve action (whether large or small) were primarily a function of fluid flow velocity and were relatively little effected by the increased bulk modulus and density which resulted from increasing the system's operating pressure from 3000 to 8000 PSI. Because flow velocity had no direct relationship to rated system operating pressure, it was assumed that flow velocities were equal (i.e. 25 ft/sec) for all system pressure levels considered in this study. As a result the pressure variations (and hence the stress cycling which occurs in the tubing) were of nearly constant amplitude independent of the system's rated pressure.

As an example; the value operation pressure variation cycle amplitude (See Table 13 for an 8000 PSI system was only 12% greater than that for a 3000 PSI system. The net effect of this was that the value operation pressure variations degraded the tubing much less (in terms of fatigue life) in the relatively heavy walled 8000 PSI tubing than they did in the 3000 PSI tubing.

Item 3 above represented a composite of the pump ripple induced pressure variations plus those which might have been induced by other sources such as hydraulic motors and servo valves. These had the same general characteristics as valve operation in that the magnitude of this type of pressure change did not increase in proportion to system pressure increase. As an example, in Table 13 when the rated pressure was increased from 3000 PSI to 8000 PSI (a 2.66 factor), the pump ripple amplitude, as measured in actual pump tests (Reference TABLE 13. BURST PRESSURE COMPARISON

Duty cycle based on ATS aircraft, Ref NA TFD-81-110

Duccesso	Burst		Pressure	\$ U.\$	r.s.	''R'' (])	Cyc1(2)	Actual	96	\$ Life	Plot
(PSI)	Factor	Condition	Range (PSI)	Mir	Max	Factor	Life K= 3 t	Duty Cycles	Life Used	Used Total	Point No
		Start Shutdown	0-3000	0	25.0	0	5x10 ⁵	5x103	1.0		$\overline{\mathbf{G}}$
3000	(12000)	Valve Operations	1864-4163	15.5	34.5	+0.45	4.5x10 ⁵	2x10 ⁵	¢4.4		5
	Psi	Pump Ripple	2825-3175	23.5	26.5	+0.89	8	1×10 ¹¹	0	45.4	3
	3.0	Start Shutdown	0-8000	0	33.0	0	9xi0 ⁴	5x10 ³	5.6		4
8000	(24000) Psi	Valve Operations	6712-9288	28.0	38.7	+0.72	10×10 ⁶	2x10 ⁵	2.0		S
		Pump Ripple	7750-8250	32.3	34.4	+0.94	8	1x10 ⁴¹	0	7.6	6
	(

(1) 'R' Factor, R = Minimum Stress Maximum Stress $\binom{2}{R.I}$ K = 3 (Notch Factor) Determines which Goodman diagram is used. Conservatively based on R.I. experience

.

3), increased from 350 PSI to only 500 PSI (a 1.43 factor). This also tended to reduce the fatigue life impact of this type of pressure variation on the tubing as rated system pressure increased.

It can be seen in Table 2-13 that, working with the duty cycle outlined above, the percent of tubing life used was 45.4% at a 4 burst factor in 3000 PSI system. When the system pressure was increased to 8000 PSI and the burst factor was simultaneously reduced to 3 the life used reduced to 7.6%. Actually the burst factor could be reduced to 2.5 before the tubing performance would be reduced to equal that in the 3000 PSI system. However, to err on the conservative side, and to allow for increases in flow velocity in some 8000 PSI subsystems, a burst factor of three was used in the study.

The data for the percent life used column in Table 13 was derived from the ratio of the cycles imposed on the tubing (ninth column in Table 13) to the life expectancy of the tubing under the type of cycles imposed (eighth column in Table 13). The actual life expectancy data was derived from the appropriate Goodman diagram (Figure 15), with the data point number shown on the Goodman diagram being the same as the "Plot Point Number" shown in the last column of Table 13. The Goodman diagram was based on 3AL 2.5V titanium with an ultimate tensile strength (F_{tu}) of 130,000 FSI and an assumed notch factor (K_t) of 3. This notch factor was representative of the stress risers which occur in tube to fitting joints and around oval crossection bends. These were the points where experience has shown that most tube failures occur.

2.4.2 Hydraulic System Design Pressures - Based largely on the results of the burst factor study a revised set of design pressure requirements, applicable to an 8000 PSI system, were created. These requirements are shown in Table 14 and are arranged in the same general format as that used in MIL-H-5540. Conventional 3000 PSI system pressures, from MIL-H-5540, are shown in the table for comparison purposes. It will be noted that the percent system pressure for an 8000 PSI system was in no cases greater, and in several cases less, than those used in a 3000 PSI system. This reduction in pressure ratio, where it occurs, was justified based on an extension of the burst factor data generated in paragraph 2.3.1.

2.4.3 <u>Hydraulic System Design Criteria</u> - The definition of the baseline hydraulic system for aircraft II was based on the following criteria:

- 1. System configuration was to be in accord with MIL-H-5440.
- 2. MIL-H-83282 hydraulic fluid was to be used where fluid soak temperatures would not be less than -20°F, MIL-H-5606 was to be used at lower temperatures.
- 3. Rated system pressure was to be 8000 PSI
- 4. Aircraft mission was to be as defined in paragraphs 2.1.1, 2.1.2, and 2.1.3.
- 5. Aircraft functional configuration was to be in accord with Figure 6 (Sheet 1 and 2) except as revised in this report.
- 6. Aircraft physical configuration (inboard profile) was to be as shown in Figure 22 except as modified elsewhere in this report.
- 7. Actuation loads were to be as defined in Table 8 and 9 and as expanded upon in Table 12.



. ...



	3000 PSI	SYSTEM	8000 PSI :	SYSTEM
CHARACTERISTICS	NOMINAL PRESSURE	% SYST. PRESSURE	NOMINAL PRESSURE	% SYST. PRESSURE
PUMP (VARIABLE VOLUME)		ل		
a. PUMP UNLOADING PRESSURE	3000 PSI		8000 PSI	
b. MAX. LIMIT OF FULL FLOW PRESSURE	2950 PSI	•	7900 PSI	
c. MAX. SYSTEM RELIEF VALVE SETTING	3850 PSI		8850 PSI	
AT MAX. SYSTEM FLOW				
TURBURL BELTER MALUE CETTING (MAY)				
INERMAL RELIEF VALVE SETTING (MAX.)	1	T		
a. EQUAL TO SYSTEM RELIEF VALVE	150 PSI		150 PSI	
SETTING PLUS VALUES NOTED				
PROOF PRESSURE (MIN.)			<u></u>	
a. LINES FITTINGS AND HOSES	6000 PSI	200 X	140C0 PSI	175 %
b. COMPONENTS CONTAINING GAS UNDER	6000 PSI	200 %	14000 PSI	175 %
PRESSURE				
C. PUMP SUCTION AND CASE DRAIN LINES	150 % OF		150 % OF	
COMPONENTS AND RESERVOIRS (BOOT STRAP	RESERVOIR		RESERVOIR	
TYPE)	PRESSURE		PRESSURE	
d. COMPONENTS UNDER SYST. PRESSURE ONLY	4500 PSI	150 %	12000 PSI	150 %
AND PRESS. CIRCUITS (INCLUDING LINES,		}		
FITTINGS, AND HOSES WHICH ARE A PART				
OF THE COMPONENT)				

TABLE 14. SYSTEM PRESSURES

I work was the show the states

1.0

a de maise présentes de commerces après : les présentes de la présente de la présente de la présente de la prés

10.00

l of 2

		3000 PS1	SYSTEM	8000 PSI	SYSTEM
	CHARACTERISTICS	NOMINAL PRESSURE	X SYST. PRESSURE	NOMINAL PRESSURE	% SYST. PRESSURE
PRO	OF PRESSURE (MIN.) (CONTINUED)				
e.	COMPONENTS UNDER RETURN PRESS. ONLY	2250 PSI	75 %	6000 PSI	75 %
	AND RETURN CIRCUITS (INCLUDING LINES,				
	FITTINGS, AND HOSES WHICH ARE A PART				
	OF THE COMPONENT)				
BUR	ST PRESSURE (MIN.)	· · · · · · · · · · · · · · · · · · ·	L	l	
a.	LINES FITTINGS AND HOSES	12000 PSI	400 %	24,000 PSI	300 %
ь.	COMPONENTS CONTAINING AIR AND FLUID	12000 PSI	400 %	24,000 PSI	300 %
l T	UNDER PRESSURE				
c.	PUMP SUCTION AND CASE DRAIN LINE	300 % OF		300 % OF	
	COMPONENTS AND RESERVOIR (BOOT STRAP	RESERVOIR OPERATING		RESERVOIR OPERATING	
	TYPE)	PRESSURE		PRESSURE	
d.	COMPONENTS UNDER SYST. PRESSURE ONLY	7500 PSI	250 %	20,000 PSI	250 %
	AND PRESS. CIRCUITS (INCLUDING LINES,				
	FITTINGS, AND HOSES WHICH ARE A PART				
	OF THE COMPONENT).				
e.	COMPONENTS UNDER RETURN PRESS. ONLY	4500 PSI	150 %	12,000 PSI	150 %
	AND RETURN CIRCUITS (INCLUDING LINES,				
	FITTINGS, AND HOSES WHICH ARE A PART				
	OF THE COMPONENT)				

TABLE 14. SYSTEM PRESSURES (CONCL)

- 8. Tubing material for all pressure return and suction tubing was to be 3AL 2.5V titanium.
- 9. The landing gear system was to meet operating requirements at -40°F and all other systems were to meet operating requirements at+20°F.

2.4.4 Peak Actuation Loads - It was assumed, based on experience, that during maximum actuation power demand for any mission segment no more than 2/3 of all actuators would "peak" simultaneously. It was, therefore, assumed that, for both aircraft I and II, the peak actuation power demand would be 2/3 of the theoretical sum of all short term actuation power demands occurring during a given mission segment.

2.4.5 Actuation Configuration - Where rotary shaft input power using motors was indicated as the best solution for aircraft II actuation functions, motors were also to be used on aircraft I. All elements downstream of the common mounting pad (interface) for these motors were to be identical between aircraft I and aircraft II. All actuation functions which use hydraulic linear actuators in aircraft II were to use linear ballscrew actuators attaching to the same end points in aircraft I.

2.4.6 <u>Trade Study Aircraft</u> - The trade study aircraft configurations were designated as aircraft I and aircraft II. Aircraft I was the "all electric" power-by-wire version in which essentially all secondary power on board the aircraft was generated and utilized electromechanically and was transmitted electrically. Aircraft II was the baseline aircraft which employed a conventional power split between hydraulic and electrical power generation, distribution and utilization. Figure 16 flowcharts the candidate configuration concepts. The basic concepts, which were traded, are represented by boxes 2 and 6 in Figure 16 with box 6 representing the baseline configuration. The other concepts shown in figure 16 (boxes 1, 3, 4, and 5) were examined in some depth but were gradually eliminated as the study progressed.

Figure 16. Candidate Configuration Concepts.



i

3.0 CONCLUSIONS AND RECOMMENDATIONS

3.1 <u>Conclusions</u> - Based on the study results it was concluded that the "All Electric" approach, to aircraft secondary power system design, did not provide a viable alternative to the more conventional Hydraulic-Electric when applied to aircraft of the 1990 + time period. The data presented in paragraphs 4.0 and 5.0 showed quite clearly that the "All Electric" approach was deficient in all major catagories (i.e., weight, maintainability, reliability and life cycle costs) relative to the advanced (8000 psi) version of the conventional Hydraulic-Electric approach.

During the conduct of the study, Rockwell was aware that the study results were probably not going to favor the application of the "All Electric" approach to the ATS aircraft. However it was felt by Rockwell personnel that, even though not satisfactory for a compact high performance aircraft such as the ATS, the all-electric approach might prove highly desireable for a large subsonic aircraft. This optimism stemmed from three factors as follows:

1. The belief that the Maintainability and Reliability advantages inherent to the single power type distribution system (Electrical) versus the dual power type distribution system (Hydraulic-Electrical) would more than offset some of the known Reliability deficiencies characteristic of certain power handling electrical components (switches and power control inverters).

2. The belief that the known weight penalties associated with the need to use power control inverters in the selected (270 VDC) electrical system could be more than offset, as the aircraft grew larger, through the supposed weight savings associated with transmitting power longer distances through wire as opposed to transmitting the same, or slightly greater, power through both hydraulic tubing and paralle; wire routings.

3. Given an assumed Maintainability/Reliability improvement and a weight reduction as the aircraft became larger, it was believed that, at some point (particular'y if the loads, and hence inverter weights, did not increase proportionately because the aircraft was no longer supersonic) the life cycle costs would cross over and favor the "All Electric" approach.

Unfortunately these optimistic presumptions proved to be incorrect. The basis for the foregoing statement was provided by a brief study conducted by Rockwell to roughly evaluate the presumed advantages of the large subsonic aircraft. This study assumed an aircraft having a gross takeoff weight (GTOW) more than ten times the GTOW of the ATS (i.e., 400,000 lb GTOW) and gross dimensions (i.e., length, height and wing span) four times those of the ATS. It was further assumed that, because the aircraft was subsonic (0.85M), the power generation (pumping system) and utilization systems (actuators) were essentially identical in power requirements to those of the ATS. The assumed aircraft was very similar to the Lockheed L1011 in all basic characteristics except that the power required for the L1011 was significantly less than that required for the ATS (387 hp for the L1011 versus 461 hp for the ATS). It was felt, however, that the higher power requirement was probably quite representative of an advanced 1990 + control configured large subsonic aircraft. Using these assumptions the only significant difference between the ATS study aircraft (Aircraft II) and the

large subsonic aircraft was the power distribution system. This system was assumed to carry the same power at the same transmission efficiency (pressure drop or voltage drop) from the source to the utilizing function through an average transmission distance which, for the large subsonic aircraft, was 4 times that of the ATS.

The weight of hydraulic plumbing (filled with fluid and transmitting power at equal efficiency) tends to increase by a factor of two when the transmission distance increases by a factor of four. This was derived from the pressure drop curves of reference 34. In contrast the weight of electrical wiring tends to increase by a factor of three under the same circumstances. This was derived from wire performance data contained in reference 35.

Using the ATS as a baseline, it can be seen in Table 38 and Paragraph 4.2.2.9 that the total weight of power transmission elements in Aircraft II is as follows:

Tubing, Fittings and Supports	116.8
Reservoir and Supports	44.3
	161.1

Power Wiring

23.0

Extrapolating these weights to the large subsonic aircraft results in the following:

161 ib X 4 (length factor) X 2 (hyd. factor) = 1288 lb 23 lb X 4 (length factor) X 3 (elect. factor) = $\frac{276 \text{ lb}}{1554 \text{ lb}}$

From Table 4-12 it can be seen that the total weight of power transmission is 120.3 lb for Aircraft I. Extrapolating this weight to a large subsonic aircraft results in the following:

120.3 lb X 4 (length factor) X 3 (elect. factor) = 1440 lb

From the above it can be seen that the weight saving in power transmission elements, through the use of the All Electric approach, was as follows:

For the large subsonic aircraft114.0 lbFor the ATS (Aircraft I versus II)63.8 lb Δ weight saving-large aircraft
versus small aircraft50.2 lb

Although there was an increased (Δ) weight saving, in going from the small to the large aircraft, the amount of weight saved was too small, by an order of magnitude, to effectively offset the adverse weight impact of the inverters. A weight saving of at least 500 lb would have been necessary to negate the effect of the inverters. From the evaluation of the large subsonic aircraft it was concluded that no significant weight

advantages could be expected from increased aircraft size. Essentially the Weight, Reliability, Maintainability, and Life Cycle Cost penalties associated with the "All Electric" approach would tend to remain the same whether the aircraft was large or small, or high performance or low performance.

3.2 <u>Recommendations</u> - Although this trade study does not indicate that the "All Electric" approach will be viable through the mid 1990s, it is still a very intriguing concept. For this reason it is recommended that work be continued on inverter development. If inverter weight could be halved while reliability was improved the "All Electric" approach would become viable in a wide range of applications.

The trade study indicates that power hinge type devices will be increasingly needed as a basic element in primary flight control surface actuation whether the actuation function is powered hydraulically or electrically. However, even though the power hinge seems a perfectly feasible device for this type of application, there is very little background based on actual operating experience, particularly, on a high output flight control duty cycle using long multi-slice small diameter power hinges. It is therefore recommended that multi-slice (at least 15 slices) small diameter power hinges (less than 1.5 in. dia) be developed and tested for thin wing trailing edge control surface applications. Additional characteristics which should be demonstrated during the course of development are as follows:

> 20 hz Frequency Response Stall Hinge Moment 25.000 in-1b Operating Hinge Moment 20.000 in-1b Operating (Design Load) Rate 50°/sec 100°/sec Maximum (No Load) Rate 5 X 10⁵ in-1b/rad Minimum Dynamic Stiffness Minimum Hinge Stack L/D 15 Operational Service Life 8000 hrs

Demonstrated ability to function in the presence of wing (hinge line) flexing.

Because the study indicates that the 8000 psi hydraulic system approach offers the greatest potential for low Life Cycle Cost power systems in the mid 1990 time period, it is recommended that development be pushed in at least two critical areas. These areas are as follows:

1. Small Motors - Small high frequency response 8000 psi servo-motors should be demonstrated. These servo-motors should have the following general characteristics:

Speed-max.20,000 rpmTorque-stall21 in-1bFrequency ResponseSuitable for
operating with power
hinge described above

Envelope

2.5" wide 1.5" high 3.5" long
2. Dynamic Seals - Although adequate data is available on short term seal life (up to 500 hr), more long term testing (up to 8000 hr) is needed to assure that 8000 psi systems will meet the projected reliability goals.

60

4.0 TRADE STUDY AIRCRAFT DESCRIPTION

4.1 <u>Aircraft I</u> - Aircraft I was the "All Electric" version of the baseline study aircraft. In this version all the power required for secondary functions was extracted from the primary power sources (engine or APU), in the form of rotary shaft power and was either utilized directly (i.e. as in the case of the ECS compressor mounted or the airframe mounted accessory gearbox) or was distributed to the various utilizing functions in the form of electrical power. There was no hydraulic or pneumatic power generation and distribution system on board the aircraft except for two integrated actuator package (IAP) type systems represented by the brakes and the inflight refuel receptacle.

4.1.1 <u>Power Generation and Starting</u> - Figure 4-1 shows the arrangement of the power generation and starting system for aircraft I. It shows that power for all secondary function was extracted from the engine through the PIO shaft with the exception of a very small amount of power (4.27 HP Max) which was withdrawn from the engine in the form of bleed air to maintain pressure in the cockpit and avionics compartments. Based on the data given in paragraph 2.2.6 the required power for starting was 120 HP (89.5 KW). The generator shown in Figure 17, having a 60 KW rating both as a generator and a starter, would meet this requirement because, operating within its normal 150% (90 KW) overload capability for 2 minutes, it would easily meet the 120 HP (89.5 KW) for 35 seconds starting requirement. (See further discussion in paragraph 4.1.2.2) Numerous power generation/starting approaches were examined before the approach shown in Figure 17 was selected as the baseline. The major potential approaches examined and compared were as follows:

- 1. Conventional 400 Hz AC power (115/200 V)
- 2. Double voltage 400 Hz AC power (230/400 V)
- 3. Integrated drive starter generator type of constant speed drives
- 4. 270 Volt DC power
- 5. Mixed 270 VDC and conventional 400 Hz AC power
- 6. 270 Volt DC starting with drain and fill torque converter
- 7. Generators integrated with and buried in the engines



Figure 17. Aircraft I Secondary Power Generation System.

62

4.1.1.1 <u>Conventional vs Double Voltage 400 HZ ZC Power Comparison</u> - A double voltage system was compared to a standard voltage system as a potential power generation system approach for use in the study aircraft (see items 1 and 2 above). The double voltage system was defined as a 230/400 volt, 400 Hz, 3-phase AC system. The double voltage approach was a recent development and its single major practical application has been on the B-1 Bomber aircraft. The standard voltage system was a 115/200 volt, 400 Hz, 3-phase, AC system. Standard voltage systems have been in service on aircraft for a number of years and were accepted as reliable and safe.

Subjects of concern considered during the comparison study were safety (personnel and aircraft), corona, electromagnetic interference, utilization equipment and electrical components (E.G., relays, connectors, wire, etc.) and weight.

4.1.1.1.1 <u>Safety</u> - When evaluating personnel safety, a reasonably safe "let go" voltage for man, assuming wet contacts, was considered to be between 10 and 21 volts. Since the potentials of the standard voltage system and the double voltage system were both considerably above the "let go" voltage, the hazards were not considered to be significantly different for either voltage.

With regard to aircraft safety, there was no reason to believe that a fault or short circuit should occur with any greater frequency on a double voltage system than on a standard voltage system. This assumed that the design incorporated terminal spacing, insulation characteristics, etc., commensurate with double voltage system requirements. As to the effect of a fault, the double voltage generator would limit the steady state fault current(beyond ½ cycle) to roughly half of that in a standard voltage system. The amount of power to the fault would be approximately equal to that in a standard voltage system, and therefore, the consequences, or result, of the fault should be approximately equal regardless of the system voltage used. In summary, from a safety standpoint, the precautions and procedures necessary for a double voltage system were not considered significantly different from those required with a standard voltage system. The hazards present in either system could be minimized by good engineering design and observance of good safety practices.

4.1.1.1.2 Corona - Corona would occur whenever the voltage gradients between electrically energized electrodes exceed a critical value. Its onset voltage would be a function of ambient pressure, temperature, insulation material and thickness, and time. Undesireable effects of corona would be insulation degradation, interference, and power loss. The altitudes where corona may become a problem would be above 50,000 feet, with the most critical altitudes being between 100,000 and 200,000 feet. Generally speaking, voltages of 300 volts (peak) would not cause corona onset even at altitudes of 100,000 feet. The maximum transient voltage on a double voltage system is 508 volts (peak) and would only exist a ing abnormal operation of the generating system. In summary, the potential for presence of corona existed with the double voltage system, however, the proper conditions occured only under random and rare circumstances.

4.1.1.1.3 <u>BMI</u> - The primary source of interference from the electrical power and distribution system was through electrostatic or a electomagnetic coupling to a susceptible signal circuit. Power input to a black box was assumed to be adequately filtered and not a factor. The magnitude of the electromagnetic coupling would be determined by the change in current. Since the circuit currents in a double voltage system would be approximately half of that in a standard voltage system, it was anticipated that there would be a decrease in electromagnetic coupling. The effectiveness of electrostatic coupling would be largely determined by the voltage. Therefore, electrostatic coupling might be greater in the double voltage system. This type of coupling could usually be adequately controlled by proper use of shields .

THE REPORT OF STREET AND ADDRESS OF STREET

In summary, it was felt that the total electromagnetic interference due to a double voltage system should be no greater than that of a standard voltage system.

4.1.1.1.4 Utilization Equipment and Electrical Components - New utilization equipment could be designed to operate directly from a double voltage input without any increase in weight or volume. There were felt to be no significant technical barriers to the development of this equipment.

Electrical components rated for operation at 230 volts presented no significant problems. Some equipment (switches, connectors, wire, etc.) could be uprated for double voltage operation without any change or penalties. Other equipments would require minor modifications.

4.1.1.1.5 Weight - During its design, numerous configurations of the electrical power generation and distribution system were studied to determine the system with the maximum weight advantage for the B-1 aircraft. These studies (Reference 15) showed that for a large aircraft (395,000 Lbs. take-off weight) with a large electrical system (345 KVA), reasonably large weight savings could be realized with a double voltage electrical system. Most of the weight saving was derived from the use of long runs of primary power feeder cables and resulted from less copper being required to carry the reduced current characteristic of the double voltage system.

The weight advantage which could be realized by utilizing a double voltage system, instead of a standard voltage system, on aircraft I was relatively small and could be determined by a detailed weight analysis of the major area bus feeders, transformers, and secondary power rectifiers. For an aircraft the size of aircraft I (36,043 Lbs. take-off weight), with a "worst-case" AC load of approximately 50 KVA the weight savings would not have been substantial enough to have been cost effective. In order to have been cost effective, all of the double voltage utilization equipment would have to have been available in 1990 for use in aircraft I as off-the-shelf equipment or with only minor modification. Since there was no discernable trend to the development of double voltage utilization equipment at the time of this study, a standard voltage electrical system was considered most acceptable.

4.1.1.1.6 Summary - The effects of certain technical considerations (i.e., safety, corona, EMI, utilization equipment and electrical components) did not legislate against the use of a double voltage electrical system. However, for air vehicles the size of aircraft I, the weight saving potential available, through the use of double voltage, did not appear sufficient to justify the cost of developing "new" utilization equipment. Therefore, further consideration of double voltage systems was dropped.

4.1.1.2 <u>Generators Integrated with Engines</u> - The possibility of integrating the generators with the engine was looked at in considerable depth. The engine used in the ATS aircraft is a 1995's type twin spool, low bypass ratio, variable cycle engine. Based on this design the apparent practical locations for the generators were considered to be confined either to installation in the engine inlet bullet nose or mounted as an external accessory on the engines' waist. A third possibility, which was integration of the generator with N₁ or N₂ spool shaft, did not appear practical primarily due to the extremely poor accessibility of such an installation.

The engine bullet nose location had been used in the past and had proved reasonably feasible for a single generator of small size. However, considering the fact that this engine installation required the mounting of two generators, and both of large size, the engine bullet nose approach appeared much less desireable. The engine bullet nose on the ATS engine had a useable inside diameter of 6.25 in. which was much too small to seriously consider side by side mounting of the generators. Even if the generators were mounted in tandem, their diameter (10 in.) would have increased the bullet diameter and, hence, the engine diameter thus offsetting one of the primary virtues of the generator's integration with the engine; i.e. reduced installed engine/inlet frontal area. The extreme length of a tandem installation would have futher amplified an already existing problem characteristic of previous bullet nose installations which was excessive cantilever vibration. As final factors legislating against selection of this approach, bullet nose installations had the following additional disadvantages:

- 1. Poor accessibility with engine installed
- 2. A significant increase in the possibility of engine FOD damage
- 3. Poor acess, as a starter generator, to the high speed (N₂) spool which is the spool which must be powered during starting.

An engine waist installation was considered as being essentially identical to an AMAD installation. The selection between the two would go to that approach which provided the best accessibility and the smallest increase in engine/inlet system frontal/wetted area.

Because the engine used in this application was a low bypass ratio type it tended to be barrel shaped and have very little 'waist". For this reason any acessories mounted on the engine even in the waist area, tended to increase the engine's frontal area by an amount at least equivalent to the frontal area of the item being mounted. It can be seen however, when looking at the ATS inboard profile (Figure 22) that, due to the nature of the inlet duct (above and outboard on the fuselage) and its relation to the wing (see station 560 crossection), a forward extending PTO and AMAD setup would allow the installation of two large generators, in place of the items shown as 4 and 8 on the inboard profile, with no significant increase in engine/inlet system frontal area or wetted area. As pointed out else where in this report, and as shown in Figure 17, the AMAD also drives the freon compressor. The ability to drive the freon compressor by direct mechanical drive, rather than by an intervening electrical link, was felt so important efficiency-wise that, even if the generators were mounted on the engines' waist, an AMAD would have had to have been provided for driving the compressor by itself or the three units (two generators and one compressor) would have had to have been mounted as a group on the engine's waist. This latter approach represented a prohibitive penalty because, in addition to the three components's large size, there was the fact that they had to be intermixed with other engine ancillary equipment, such as fuel controls, with the result that there was a large adverse impact on, not only the engine/inlet system frontal and wetted area, but also on accessibility and maintainability. Based on the foregoing considerations, the AMAD approach was selected as offering the best access and maintainability in combination with a minimum increase in frontal and wetted area.

4.1.1.3 <u>270 Volt DC Power Versus Mixed 270 VDC/400 HZ AC Power</u> - The primary source of power on almost all civil and military aircraft for the past 25 years has been based upon the distribution of 115/200 volts, 400 Hz AC power. With the passage of time, this system has been widely accepted and has shown a steady increase in reliability and specific power output. As will be seen later (paragraph 4.1.5.1), the primary control surfaces (in the study aircraftaircraft I) are activated by power hinge actuators utilizing 270 volt DC samarium-cobalt, brushless (permanent magnet) drive motors. Due to the importance of this system to aircraft I operation and because the actuation loads are a major portion of the aircraft's total continuous load (approximacely 66 percent during combat), it was decided that a 270 volt DC power generation and distribution system would provide the most efficient power source for this type of equipment.

Numerous analyses have been performed relative to the impact on aircraft weight, reliability, safety, and cost if a 270 VDC system were substituted 100% for the conventional 400 Hz AC system (Reference 9, 12, and 13). The conclusions arrived at, from the various studies, were that equipment in the form of brushless DC motors, power semiconductors, solid state switchgear, aircraft cables, and inverters were available for a 270 VDC system and that, in large aircraft, a lighter weight and lower maintenance cost system could be expected through the use of 270 VDC power. (Reference 17, 18 and 19). However, the situation with respect to small high performance aircraft (7.33 G-M2 2) was not made clear. In this type of aircraft the weight fraction of the distribution system tends to become small and the weight fraction of 270 VDC power had been made for the primary control surfaces (modulated) electric motors, the selection of the best power for the unidirectional and/or non modulated electric motors used on the aircraft was open to further examination.

4.1.1.3.1 Non Modulated Motors - Studies done by others were targeted at the use of 270 VDC power for large non modulated motor loads, such as fuel pumps and blowers, that were normally operated by AC induction motors (Reference 13). The two alternatives available were (1) retain the continuously running AC induction motors and drive them with dedicated inverters that convert the 270 VDC to souare wave AC power or, (2) use brushless DC motors; i.e., synchronous motors with permanent magnet rotors and armature windings, controlled through solid stale circuits (inverter) to provide commutation and run at adjustable speeds. In either event an inverter was required. At the time these (Reference 13) studies were conducted brushless DC motors were considered to have marginal advantages but had only been developed in sizes of approximately one kilowatt. Recent development in brushless DC motors have made their performance better than that of induction motors, with higher efficiencies and better torque/speed characteristics. In addition, several companies have produced prototype DC motors in ratings up to 12 KW.

Estimates made during this study program suggested that the brushless DC motor, with its inverter, would be lighter and smaller than an electronically (inverter) controlled inductions motor of the same rating. There was a third alternative however, and this was the possibility that the overall system would be lighter and more reliable if 400 Hz AC power was supplied for all non-modulated electric motors. As pointed out earlier both of the above alternatives (1 and 2) for operating motors from a 270 volt DC bus require solid state electronics (inverters) for commutation and control. Due to losses developed during transistor switching and conduction, power inverters require cooling and, since the size and weight of the inverter is usually determined by the type of cooling employed, the inverter and its cooling method become an integral part of the motor evaluation.

والمراجع بالتصبير ليرب ستنقط تقروا الحاجات يصبغ والعار

To form an initial assessment on the size of the inverters required, if 270 VDC power/was to be used for all motor loads, the study for the design of the surface control systems (paragraph 4.1.5.1) was reviewed. In this study, inverter cooling requirements were determined to be in one of two categories:

- Inverters rated at 25 amperes (6.75 KW) or smaller may be cooled by natural radiation and convection. A common design for all inverters in this category was sized at 231 cubic inches (11" x 7" x 5") and weighed 10 lbs.
- (2) Inverters rated larger than 25 amperes (6.75 KW) required alternate cooling techniques of which evaporative cooling was selected.

Cooling techniques based on natural radiation and convection were preferred, of course, because of their simplicity and low cost.

Continuous motor loads on the study aircraft included two fuel boost pumps (1.98 KW each during cruise operation, and 2.36 KW each with afterburner), ten fuel transfer pumps (0.64 KW each during cruise operation and 0.74 KW each with afterburner), and two blowers (3.2 KW each) and two pumps (0.87 KW each) in the environmental control system. The weight of a common inverter to support operation of these motors from a 270 volt DC bus was estimated to weigh 6.9 lbs $\left(\sqrt{\frac{j}{6.75}}\times_{1C}:6.9\right)$ by comparing its power rating with the power rating of the surface control actuation system inverter. As a result of the above analysis, a total inverter weight of 110.4 lbs. (6.9 x 16 = 110.4) was concluded to be necessary to operate the motors from the 270 volt DC bus.

In addition to the inverters required for the continous load function enumerated above, inverters would also have been required for all the low power (6.75 KW) utility functions listed in Table 2-1. This would have required at least 10 more inverters involving an additional 69 lbs weight penalty.

If on the other hand 115/220 V 400 Hz AC power could be supplied to these motors the inverters could be eliminated, and simple on-off or extend retract switches substituted, allowing a net weight saving of at least 127 lbs based on an assumed average switch weight of 2 lbs. Although the use of 270 VDC power for servicing these loads would have reduced duplication and the weight of the power feeder cables (see following paragraph 4.1.1.3.3) it was not felt that the saving in an aircraft of the small size of the study aircraft would have been great enough to offset the 127 lbs inverter weight penalty. Therefore, in consideration of this fact, plus the fact that the continued use and availability of 400 Hz power would be convenient for use in avionics black boxes and lighting equipment and would give added flexibility and adaptability to an advanced aircraft of the 1990's a dual power output (270 VDC and 115/200V 400 Hz AC) power generation system was selected for use in the baseline aircraft I.

4.1.1.3.2 <u>Aircraft Cables</u> - It was apparent that the use of a 270 VDC system had the obvious advantage over the conventional AC system of reducing the aircraft's total cable length. This would have been true because the number of power feeder cables required from the generator to the distribution center, as a minimum, decreased from four to two. Earlier estimates on distribution wiring (Reference 20) assumed that high voltage DC power could be supplied by a single cable from each generator with aircraft structure providing a ground return. In comparison with a conventional 3-phase machine, this meant three cables could be replaced by one. With the increased impact of composite structures, it appeared unwise to assume a structural ground return to be practical. Accordingly, for the purpose of this study, the DC feeder system between each generator and the distribution center was assumed to consist of a positive and a negative cable and the AC feeder system was assumed to consist of 3 positive cables and a ground return cable.

Using the four wire (3 phase AC power) versus 2 wire (DC power) as a basis, a rough study was made of the comparative weights involved in distributing power to the 16 continuous motor loads discussed in the previous paragraph (paragraph 4.1.1.3.1). It was assumed that the motor loads were distributed approximately evenly between the four power systems. It was further assumed that the power distribution center was located three feet from the generators and that the various motors were located as shown on the aircraft inboard profile (Figure 22).

Based on this study the wire weight saving which would have resulted from using a 270 VDC distribution system, rather than 115/200 V 400 Hz AC system, was approximately 4 lbs. Although the total wire weight saving which might have been expected in this aircraft, through. using 100% DC power versus using a system approaching 100% AC power, might exceed this value by an order of magniture (i.e. 40 lbs), the rough study indicated quite clearly that, on an aircraft this small, the wire bundle weight saving potential of this approach was small.

In recent years the properties of, and insulation for, wires and cables have improved substantially. At present Kapton and Tefzel are the lightest available materials. It was assumed that these insulation materials were representative of the types which would be used in a 1990 + aircraft and therefore wires insulated with these materials were used for all weight studies for general airframe application. Studies with respect to the effect of a double voltage system on aircraft parameters such as aircraft and personnel safety, corona, EMI and wire have been discussed previously in this report. The conclusion was that precautions and procedures necessary for a double voltage system were not significantly different from those required with a standard voltage system. The hazards present in either system are minimized by good engineering design and observance of good safety practices.

The selection of a 270 volt value for the high voltage DC portion of the system resulted from the ease with which this voltage could be obtained when full wave rectification was applied to a conventional 115/200 volt 400 Hz 3-phase AC generator. As a result there was little doubt that any thin-wall cable acceptable for operation with a double voltage system (such as that used on the B-1 aircraft) would have had adequate margin for operation at 270 VDC.

4.1.1.3.3 <u>Switch Gear</u> - In todays aircraft, electromechanical devices such as circuit breakers, contactors and relays are used for switching and protection in feeder and distribution systems. For an AC powered electrical system this is no problem. During any mechanical switching action, the arc which occurs across the gap tends to extinguish itself at the "current zero" point of the sine wave. With a 270 volt DC system, however, the arc tends to be self-sustaining (at least 3 seconds) and if not detected and extinguished very quickly by some form of forced commutation could result in severe damage.

Earlier studies recognized the danger of arcing and flashover to be very real. They also showed, by laboratory tests, that under certain fault conditions a self-sustaining arc condition might occur that could not be extinguished by the mechanical switchgear then available As a result of this, even today a reluctance exists on the part of designers to accept the 270 VDC electrical system. Adequate protection against ground faults is dependent upon rapid detection and isolation of the fault condition. Although this cannot be accomplished with the thermal circuit breaker, there was reason for confidence that high rates or rise in current could be detected and transistor switches developed to break a high current fault before it had risen to extreme levels. In addition to opening the circuit, backup safety could be provided by the capability to de-excite or mechanically decouple the generator very rapidly.

Recent developments in high powered, solid state switching devices were considered sufficiently well advanced that, although they had not replaced conventional components in the AC and DC distribution systems of modern aircraft, they were in good position to do so. The use of solid state technologies was felt to be inevitable and crucial in the 270 VDC system to replace functions previously performed by contactors and thermal circuit breakers in the AC system. As always, considerations for cooling the solid state devices were an added problem.

In order for a sustained arc to occur, two faults must coexist; (1) the fault must occur and (2) the protection system (overcurrent sensing) must fail. The improbability of a double fault, along with the confidence that can be placed in the protection system as a result of advances in semiconductor technology, made it reasonable to accept the 270 VDC system in the study aircraft.

For the purpose of this study, solid state power controllers (SSPC's) were utilized to perform AC and DC load switching and protection functions. Each unit provided the function of a circuit breaker and remotely controlled switch (contactor) in a common module. For small load currents the SSPC was considered to be an "all solid state" device. For SSPC's in the 10 to 400 ampere current range a hybrid (solid state switching plus electromechanical contactor) unit was utilized. All units provided for their own heat rejection without the need for additional cooling. 4.1.2 Power Generation and Distribution - The power generation and distribution system is shown schematically in Figure 18 and the terms used in the schematic are defined in Page 2 of the Figure. The most significant characteristic of this system, and the one which represented the greatest departure from previous electrical system philosophy, was the fact that the total power system consists of four completely independent power channels. There were no bus tie contactors and each generator was dedicated solely to a particular channel. This approach was used to allow the electrical system to satisfy reliability and redundancy requirements (See Tables 4 and 11). Because the electrical system in this aircraft provided actuation power for the flight control and utility systems as well as acting as a source of control signal power for both, it must have a higher order of reliability than that characteristic of electrical systems of the past such as typified by the electrical system of aircraft II. This arose from the fact that, in picking up the actuation functions which have historically been accomplished hydraulically, the electrical system must duplicate or exceed the hydraulic power system's redundancy. Through long and sometimes bitter experience, it has been found that hydraulic systems, in their historic power supplying functions, must have redundant and absolutely isolated systems. It was found that it must not be possible for a failure in one system to propagate into another system because, if it is possible, it will happen.

Conventional electrical systems using bus tie contactors have always been subject to, and have frequently experienced, "cascade" type failures in which a single failure in one generating system has propagated through all systems wiping out all generated power on the aircraft. Such a system approach would not meet the fail operate, fail safe requirements imposed on the flight control system of the study aircraft. This continued to be true even when using stored energy in the form of batteries or APU's as an emergency power source. To meet the failure and reliability requirements of aircraft I it was felt imperative that at least 3 completely independent dedicated power systems be provided. This would match the redundancy of the hydraulic system used in aircraft II.

After more detailed study it was determined that, although 3 independent power systems were adequate, 4 independent systems more closely approached the optinum for the reasons listed below:

1. The use of four systems was the only practical way to balance engine power extraction loads, generator size, and bus loads so that each was simultaneouly reasonably uniform.

2. Four systems fitted well with the 5 channel (4 channel plus model channol) philosophy used in the aircraft I fly-by-wire control system.

3. The use of four dedicated systems required smaller generators and a lighter weight generating system than that of the conventional "bus tie contactor" approach.

With reference to reason #1 above, it will be seen later that system '5 second" and "continuous" loads could be distributed among the 4 channels so that they vary no more than ± 24% from the mean. This could be done while meeting the load and redundancy requirements of Table 11. It was also true that the minimum loads imposed on the PTO shaft by the electrical power system were determined by the generator's size and its fault clearing capability (approximately 250% of continuous load rating). It was also highly desireable that the PTO extraction loads be essentially identical between the two engines. If they were not, the PTO power train in the engine and AMAD must be designed to the highest load seen by either engine. This was necessary to maintain engine interchangeability. Thus, if the loads applied were allowed to get seriously out of balance, the overall PTO system became unnecessarily heavy.



Figure 18. Aircraft I Electrical Power Generation and Distribution System Schematic (sheet 1 of 2)

71

GLOSSARY OF ABBREVIATIONS AND ACRONYMS

	<u>A</u>	E
AC	- Alternating Current	EPM - External Power Monitor
AEBR	- AC Essential Bus Relay	ESS Essential
AEPC	- AC External Power Contactor	EXT External
ALC	- AC Line Contactor	G
AMAD	- Airframe Mounted Acessory Drive	= G - Generator
APU	- Auxiliary Power Unit	GCU - Generator Control Unit
	B	н
BAT	- Battery	
BC	- Battery Charger	nz - nertz (cycres per second)
BCR	- Battery Charger Relay	<u>R</u>
BR	- Battery Relay	RSCR - Reversible Silicon
BS	- Battery Switch	controlled Rectiller
	C C C C C C C C C C C C C C C C C C C	<u>S</u>
CCV	- Cyclo Converter	SCA,B,C&D - Starter Contactor A,B,C,D, Etc.
	D	SI - Starter Inverter
DC	- Direct Current	SR - Starter Relay
DEBR	~ DC Essential Bus Relay	<u>r</u>
DEPC	- DC External Power Contactor	TRF - Transformer, Rectifier,
DLC	- DC Line Contactor	Filter
		<u>v</u>
		VDC - Voltage, DC

Figure 18. Aircraft I Electrical Power Generation and Distribution System Schematic (Sheet 2 of 2)

PTO system power extraction loads could have been balanced using four equal sized generators feeding three systems. However, since the aircraft had two engines this almost inevitably lead to two one generator systems, each generator driven by a different engine, and one two generator system with each of its generators driven by one of the two engines. If it was assumed that all four generators were to be of equal size, for logistics and interchangeability reasons, and were to be of minimum weight, it meant that the two generator system had to have assigned to it twice the continuous load of the other two systems. With the aircraft configuration used for this study, this always led to trouble in the reliability survivability area. In attempting to distribute the loads, to properly load the two generator system, it usually worked out that the engine actuation functions (plug throat, external flaps, etc.) for both engines had to be on the two generator power channel. This meant that a single failure could lead to degrading the whole propulsion system to a marginally fail safe condition. In contrast when each engine was powered by its own system (as is true in the four system approach) only one engine's output was degraded to marginally fail, safe after a single failure and it took two failures to achieve the same level of propulsion system degradation. This represented a serious reduction in aircraft reliability and survivability. Attempts to redistribute the loads in other ways, while having each engine's actuation functions powered by separate systems, always seriously impacted the redundancy of power distribution to the flight control system or seriously unbalanced the loads between generators. In either event, a significant negative impact on weight or survibability/reliability always occured when attempting to use a 3 channel power system rather than a four channel approach.

With reference to reason #2 above, it is pointed out elsewhere (see figure 35) that a four signal channel plus model channel fly-by-wire (and light) control was selected for the study aircraft. Using a battery to power the 5th (model) channel, the four power systems approach fitted nicely with the four signal channels and at the same time allowed absolute system separation (power and signal) to be maintained. A three channel power system did not fit so well. At some point a single power system must power two signal channels and thus, a single failure would have eventually lead to the failure of two signal channels.

With reference to reason #3 above, it was found, rather surprisingly, that one of the benefits of a 4 channel dedicated power generation system was that it was significantly smaller and lighter than a conventional 4 channel split-parallel system using bus tie contactors. In the early studies on this program the latter system was thoroughly studied and the results of that study are included in this report for comparison purposes as follows:

"The electrical power generation and distribution systems (EPGDS), shown schematically in Figure 19 is designed to provide power during conditions of normal and emergency aircraft operation. The aircraft I system, shown in Figure 19 consists of an engine-driven 270 volt DC starter-generating system, and APU-driven backup generator, converted 115/200 volt 400-Hz power, an emergency battery, and provisions for use of an external power source.

The maximum average load demand of Aircraft I, for a 15-minute operating condition, is estimated to be 223 KW. The loads which establish this maximum are those listed in appendix "A" of NA 79-492 (Interim Report Electrical Load Analysis Reference 12) except that items 405 and 406 page A-16 are deleted. The deletion arises from the fact that, unlike the original assumption, the freon compressor is no longer powered electrically but is powered directly by the engine. To supply this mission load the primary electrical system utilizes four 270 volt DC generators, each with a canacity rating of 115 KW. At this rating, the primary electrical power generation system provides a 100 percent



Figure 19. Aircraft I EPGDS Schematic

74

reserve capacity. The four generators, mounted two-per-engine, supply four main DC buses to support multiredundancy requirements of the flight control system. During normal flight operation bus-tie contactor #2 (BTC 2) is open while BTC 1 and BTC 3 are closed. In this mode the system operates in a split-parallel configuration. Generators #1 and #2 operate in parallel to supply main buses #1 and #2, isolated from main buses #3 and #4, and generators #3 and #4 operate in parallel to supply main buses #3 and #4. In the event of a generator failure, or single engine flameout, BTC 2 will be closed and the remaining generators operated in parallel to ensure an uninterrupted supply of power to the four main DC buses. This arrangement (split/parallel) provides fault isolation by preventing disturbances on one-half of the electrical system from affecting equipment on the other half. In addition to supplying primary electrical power, all 4 generators on aircraft I provide power for starting the main propulsion engines. Aircraft II utilizes a conventioanl pneumatic start system for the engines and the generators supply only primary electrical power.

In configuring the electrical subsystem it is anticipated that 270 HVDC will not always be the most efficient power source. Lighting, instrumentation, avionics, engine controls, and motors (where rapid response is not a critical requirement) are design areas that can fall in this category. A conventional 115/200 volt, 3-phase, 400 Hz electrical conversion system is provided for those subsystems that can conveniently utilize such power. For baseline system sizing considerations all motor loads such as surface control actuators, fuel pumps, ECS fans and pumps, etc., are regarded as powered by 270 HVDC. All housekeeping (non-actuator) loads, except motor loads, are considered to require conventional 400 Hz power. The total 400 Hz load requirement is 44 KW and two 45 KW static inverters provide redundant sources for this amount of power. Of the total capacity available from the 4 generators approximately 110 KW (55 per redundant channel) is allocated for static conversion to a conventional 115/200 volt, 400 Hz power system. This total includes losses in the power conversion devices and provides for a 100-percent reserve capability over estimated load requirements. The remainder of the generating system capacity (175 KW) is reservered for distribution as 270 volt DC power".

From the above discussion of the conventional approach, two critical factors stand out; the size of the generators and the use of 3 bus tie contactors. It will be noted from the above extract that four 115 KW rated generators were going to be used in the conventional system while in the dedicated power channel approach the generator capacity could be reduced to 60/70 KW rated. Since the difference in weight between a 60 and a 115 KW generator of this type was approximately 50 lbs the total weight penalty, associated with the generators, for the conventional system was approximately 200 lbs. It will also be remembered that bus tie contactors were not required in the dedicated power channel approach so an additional weight penalty of 16.5 lb (3 units at 5.5 lb each) would have to be added giving a total weight penalty of 216.5 lb for the use of a conventional system arrangement.

The basic reason that the conventional approach suffered a weight penalty was the fact that, as generators failed, the remaining generators picked up the full system load. After two failures the remaining generators were carrying double their normal continuous load. In contrast the generators in the dedicated approach never carried more than their normal load. In effect there was automatic and weight free load monitoring in the dedicated approach. As generators failed the loads serviced by those generators no longer received power. The inbuilt redundancy of the power using functions, however, insured that the function would continue to operate at its required level of output in the face of the required number of failures. The conventional system could have reduced its required generator system size, including bus tie contactors, to a weight value approaching that of the dedicated system through the use of a rather elaborate load monitoring system. It was felt, however, that this could only be accomplished at a further reduction in system reliability additive to the relatively poor reliability which already existed in the conventional system due to its propensity for "cascade" type failure. Therefore, consideration of 4 generator/3 power system and/or split bus parallel arrangements were abandoned.

4.1.2.1 <u>Power Generation Definition</u> - Four 60/70 KW generators (two per engine) provided the electrical power required by the study aircraft. As shown in Figure 18, each generator divided its power output to provide both 270 volt DC and 115/200 volt, 400 Hz, 3-phase AC power. The larger portion (approximately 2/3) of each generator system's output was 270 volt DC and was predominantly that power provided for operation of the electromechanical surface control actuators. To obtain 270 volt DC, part of the wild frequency AC output of the alternator was rectified in the early stages of the VSCF power conditioning.

Conventional 115/200 volt, 400 Hz AC power was provided for avionics and other conventional AC loads such as induction motors (without the use of inverters), lighting, and heating. Although an advanced IDG design offered some advantages over the VSCF system in the areas of size and weight (where all output power was 400 Hz) a VSCF generator was selected because electronic power conversion was more suitable where different types of power output were required and where a majority of the raquirement was DC. The generator was a 10-pole, wound rotor unit driven over a speed range of 13750 to 27500 RPM in the power generator systems, all four generators (alternator plus cyclo- converter) were designed to be interchangeable for ease of maintainability and inventory preposes. The reason for selection of the generator size and the split between AC and DC loads is discussed at greater length in paragraphs 4.1.4 and 4.1.2.2.

4.1.2.2 <u>270 Volt DC Start Capability</u> - The engine start requirement was 89.5 KW of power applied to the generator output terminals for 35 seconds. To meet this requirement, the generator must be sized as at least a 60 KW unit. At this rating the generator would be capable of providing 90 KW (150-percent) for 2 minutes. The general arrangement of the starting system is shown on Figure 18.

In the starting mode, one synchronous generator on each engine functioned as a brushless DC machine with variable frequency AC power supplied to it. This was made possible throught the use of onboard, dedicated static power inverters ("SI" in Figure 18) to provide programmed voltage and frequency power supply for the starter generators. Each starter-inverter could be powered from an onboard APU or from external power as shown in Figure 18. System operation was analogous to a DC shunt machine supplied by a phase-controlled rectifier. The inverter operated as a phase controlled rectifier to adjust the voltage level and to switch the current among the armature windings. A position sensor on the machine informed the inverter which winding must be supplied with current. The use of the inverters represented a major weight penalty but was more than offset by the elimination of the need for air compressors and air turbine starters such as used on aircraft II (See Figure 45). In addition the need for high power ground air supplies was eliminated since, through the dual functional use of the generator made possible by the inverter, only electrical power was needed for the starting function.

A DESCRIPTION OF THE PARTY OF

In the start mode, a unique application of the starter-inverter and the 270 VDC rectifier portion of the VSCF power conditioner were combined with a torque converter to crank the engine through an aircraft mounted accessory drive (AMAD) pad.

As shown in Figure 20, upon initiation 270 volt DC power was applied through the 15 KVA inverter to the alternator (as AC power) to bring it up to a programmed synchronous speed of 10,000 RPM. Programmed firing of the inverter transistors was timed, by feeding back rotor speed and position to the inverter, to accelerate the alternator. To protect the system from drawing excessive power from the 270 VDC start bus, the absolute power output from the inverter was sensed and the current limited by adjusting the transistor firing angle. In Figure 18, engine start from the APU was initiated by closing contactors SRI and SR2. Closure of contactors SCC and SCB routed power through the starter-inverter (SI) to the alternator.

In Figure 20, when the alternator rotor speed (N_2) reached 10,000 RPM the system logic of the controller sent a reversing signal to the SCR's of the 270 VDC full wave bridge rectifier. Concurrently, starter contactors (SC) C and B were opened and starter contactor A was closed. This function transfered the alternator power source from the starter inverter to the reversed SCR's of the 270 volt DC rectifier. The commutation of the main power conditioner SCR's was accomplished by the presence of alternator back EMF. At 10,000 RPM the peak line-to-line voltage of the synchronous motor was momentarily higher than the 270 volt DC bus voltage necessary for SCR commutation. The closure of contactor A also directed the opening of the torque converter fill valve.

At this point, approximately one second transpired since start initiation. The alternator was at full speed, but starter output shaft rotation downstream of the torque converter had not begun. As the torque converter filled, torque from the alternator shaft, controlled to maintain 10,000 RPM, was transferred through the torque converter to the AMAD pad. Only at this time was the full torque requirement of engine starting reflected electrically to the 270 volt DC bus.

Starter output torque followed a curve similar to Figure 21 as the engine accelerated to starter cutout speed. When the AMAD pad speed (N_1) reached starter cutout, the controller logic inhibited the reversing mode signal to the rectifier SCR's and drained the torque converter by removing the fill valve driver signal. As the engine continued its acceleration up to idle speed, and the starter generator tended to slow due to removal of inverter SCR power, the torque flow reversed. Utilizing this torque flow reversal, a set of overrunning clutches was provided which disengaged the torque converter and simultaneously direct coupled the engine speed range. The full wave bridge rectifier and filter provided 270 volt DC power for the aircraft primary bus system. As shown in Figure 18, 270 volt DC power and 115/200 volt, 400 Hz AC power were available to the aircraft by closing contactors DLC and ALC.

4.1.2.3 <u>Power Distribution Bus Arrangement</u> - Figure 18 shows the general arrangement of the power distribution system up through the various busses. As previously indicated, the output of the generator was wild frequency AC power. This power was processed in a power conditioner ("PC" in Figure 18) which consisted basically of a voltage regulator and a cyclo converter. The voltage regulator delivered up to 1/3 of the generator's rated capacity, in the form of voltage controlled wild frequency AC power, to the cyclo converter for conversion to constant frequency 400 Hz AC power. The balance of the voltage regulator's output was delivered to a transformer-rectifier-filter unit (TRF" in Figure 18) for conversion to 270 VDC power. The generator control unit ("GCU" in Figure 18) sensed that the generator was out of starting mode, was up to speed and was ready to sustain load. When this occurred it closed the AC and DC contactors.

Figure 20. Aircraft I 270 VIX: Starting Arrangement.

e 15 es



78

•



Figure 21. Aircraft I Engine Starting Torque Converter Characteristics

("ALC" and "DLC" for the various systems in Figure 18) and delivered power to all the busses attached to that particular system. From Figure 18 it can be seen that systems No. 2 and No. 3 each powered a primary AC and DC bus while system No. 4 powered an AC and DC essential bus in addition to the primary busses. In common with system No. 4, system No. 1 powered AC and DC primary and essential busses but added to them a battery bus powered through a diode. A large battery was provided. This battery was sufficient to provide a minimum of 4 minutes of power for essential emergency flight control actuation functions as well as for other emergency power requirements (central computer, emergency lighting etc.), occuring at the same time. This battery was charged from system No. 2 via a battery charger and relay (BCR" and "BC" respectively in Figure 18). Normally system No. 1 powered the battery bus and the various emergency functions attached thereto. However, if all systems failed the battery charger relay (BCR) switched to the position shown in Figure 18 and the battery powered the emergency functions. The diode, shown in Figure 18, prevented the battery from delivering power to functions not essential in an emergency such as those attached to the DC essential bus and the primary bus. If power output was re-established, in one or more of the primary or essential busses (say by flying the aircraft down to 20,000 ft and starting the APU) the battery charging relay (BCR) returned to the charging position. The battery also powered the APU start bus. This bus provided power to the APU start motor to bring the APU to self sustaining speed after an engine start had been initiated. The function of the battery switch (BS) and battery relay (BR) was to make it possible to remove all power from all busses while the aircraft was parked and inactive.

The rules governing the assignment of functions to the various busses were as follows:

1. Those functions, and only those function, necessary for recovering from a maneuver and maintaining level or descending flight as well as those functions necessary for towing and parking were assigned to the battery bus.

2. Those functions necessary for a safe return to base and landing were assigned to the various AC or DC essential busses.

3. All other functions were assigned to primary AC or DC busses.

The actual assignment of functions is covered in more detail in paragraph 4.1.4.

As shown in Figure 17 and again in Figure 18 the APU mounted and powered two generators. These were sized based on the starting load requirement discussed in paragraph 4.1.2.2 and were rated at 45 KW each. Since the two generators operated in parallel during engine starts this rating gave a starting system 2 minute rating of 135 KW (45 KW x 2 x 1.5). This would meet the 40 sec start requirement of 90 KW with enough left over to meet essential bus loads of 45 KW. The maximum continuous bus loads which occured on either of the two essential busses during the starting sequence were 34.68 KW (see table 16 sheet 11) and the maximum 5 sec loads were 50.28 KW (see table 16 sheet 12). Based on these figures it was assumed that the maximum 2 minute loads would not exceed 42 KW which was a value within the limits of the 45 KW available.

The APU generators generated and delivered AL and DC power in a manner identical to that already described for the main system generators. Assuming it was designed to start engine No. 1 through generator No. 1 (B1 in Figure 18) the following status would exist initially. In Figure 18 SR1 and SR2 would be closed, AEBR₂ and DEBR₂ would be open on the starter side, AEBR, and DEBR, closed on the starter side and, as already described in paragraph 4.1.2.2, SCB, and SCC, would be closed while SCD and SCA, would be open. In this way power would be directed to the starter/generator (G1) through the starter inverters (I) to initiate the starting cycle and at the same time power would be directed to AC ESS. bus 1, DC FSS. bus 1 and to the BAT bus. The starting sequence would then proceed as previously described in paragraph 4.1.2.2.

During normal operation the APU would deliver power to the essential busses of system No. 1, or system No. 4 or both dependent upon which system, or systems, had failed. In this instance starter relays SR_1 and SR_2 were open and the APU generators operate independently. The use of two generators in this manner was another attempt to maintain the absolute system separation which was the goal of this system arrangement. While the interlinking involved in the starter system did defeat the absolute purity of system separation sought to a certain extent, it did maintain this separation during all normal and APU powered flight operations. There was also some potential interlinking involved in the battery chargin circuit which could not be avoided. However, even though systems No.1 and No. 2 had some interlinking and thus some possibility of a cascade failure occuring between the two, systems No. 2 and No. 3 maintained their absolute isolation.

4.1.3 <u>Aircraft Inboard Profile</u> - Figure 22 is the inboard profile of the baseline study aircraft. Figure 22 shows the general location of those items which are powered by the electrical system and which are common to study aircraft I and II. Typical of these general locations are the forward, intermediate, and aft avionics compartment and the various fuel pumps. As will be seen elsewhere (paragraph 4.1.1.3.1) it was the general study philosophy that, along with all avionics components, all functions powered by unidirectional motors or nonmodulated bidirectional motors employ conventional 400 Hz AC power. Therefore, these components (avionic "black boxes", fuel pumps, emergency door actuators, radar drives etc.) were considered to be unchanged between aircraft I and aircraft 1. and thus did not enter into the trade study except for the impacts their loads had on overall generation system sizing and the constraints their locations placed on wire bundle routing.

A REAL PROPERTY OF

Figure 23 is a specialized version of the inboard profile of figure 22 which provides a schematic of the "all electric" airplane (i.e. aircraft I). As such it defines the relative location of all the major power using equipment items on board the aircraft. Figure 23 also defines the power system or systems servicing each function as well as the general routing of the distribution elements to each function.

Several items stand out in Figure 23. The first is the fact that all four systems run to all parts of the airplane and the second is the rather large volume taken up by inverters, particularly those in the fuel tanks. It had been hoped that only two power supply systems would have to run fore and aft in the fuselage since, with the exception of the rudders, all functions along the fuselage centerline (plug throat, canard, nose gear, etc.) were basically one or two power system units. Had this been possible it would have reduced the wiring system's weight and, to a certain extent, its vulnerability. However, the reguirements for load balancing and the need for nose micro-processors (See Figure 37), located in the forward avionics bay, to have at least three independent power sources dictated otherwise. Because there were three large power users in the nose of the aircraft (the gun, the radar, and the defensive subsystems items 1151, 901, and 1001 repectively on the electrical load analysis) which tended to operate simultaneously during combat, the problem of power balancing these loads, plus the triple power source needs of the micro-processors, dictated that four power supply systems were required in the nose of the aircraft. The same problem was encountered in the aft end of the airplane. Here the high loads were represented by the plug throat and the thrust vector vane. Of the two the thrust vector vane had the highest redundancy requirements (fail operate - fail safe). To meet this requirement, each thrust vector vane needed to have two power supply channels, each of which was preferably powered by its engine. Since there were two engines and two thrust vector vanes, one for each engine, it followed that there was a need for four power supply systems at the aft end of the two engines.

The location of the various inverters required for the various actuation functions can be seen Figure 23. It will be noted that a majority of the inverters, and all of the large liquid cooled types were located in the fuselage. A few of the smaller ambient air cooled units, we \Rightarrow located in the wings. The largest inverters were located in the fuselage fuel tanks in the wing carry through area. An end view of these units can be seen in the FS 560 crossection shown in Figure 21. Although the function, size and the weight of the inverters is discussed later in paragraph 4.1.5.1.2, it can be seen that these inverters occupy a considerable volume (41.6 gal) and thus displace a significant weight of fuel (332 lb). Since the total useable fuel capacity of the aircraft was 18,000 lbs this represents a 1.8% fuel capacity reduction and an equivalent range reduction.





ST OF FOULPHIEN T 2 (2) 1995 ADVANCED VARIABLE CYCLE ENGINE 2 ACCESSCOY DRIVE SEAREDLY 4 PTO DRIVE SALET S FIR TURE INE STAETER S. AUXICIALY PONEE UNIT S AFU LOX THICK SECONDARY POLIER SYSTEM SELECTIONAL, & GENERATOR 50 LVA 15/200 VAL, 150 (LESS) & GENERATOR 7 KVA. . * TEANSFORMER RECTIFIER UNIT ECONNELEESD) WINVERTER ISCOVA IG & VOLTAGE REGULATOR (APU SYS.) S BATTERI'S 40 AMP HE NI- AD (SEEDD) POWER CONTRATCE HERETO SECONDARY POWER OVETEM (-YORAJLIC FOOD FOI) & PRIMARY PUMINE 28 6PM (4 EFGD) E APU PUNIP 28 574 & PEIMARY RESERICIZ SEMERGELSY EESEEVOIR & ACCUMU. 4TG2 FUEL SYSTEM & BOUSTER PUMPS (2 REDD.) E RANSFEP PUMPS (10 REC'L.) & EJECTOR PUNIPS (4 REQD.) 🛎 FUEL / OI: HEAT EXCHANGER (GEN & ADG CIL) W HEAT EXCHANGER FUMP & GROUND L'EFUEL RECEPTALLE 🕾 LEVEL CENTERL VALVES (G LEC'D.) 🕲 INFLIGHT PEREL RECEPTALLE G ENVIRONMENTEL CONTEDL SYSTEM BOOTSTRAP TURES COMPRESSOR & INTERCOULER SEAT EXCHANGER SWATER SEPAEATCR S CABIN SAFETY VALVE S CABIN PEESSLEE REGULATOR S PRECOOLER & S.E.TOR D PRESSURE REGULATORS (ENG. MTD) NOT SHOWN 6 FLIGHT CONTROL SISTEM CANARD ACTUATSES TANDEM UNBALANCED 2 RECD. 👻 WING L.E. LEVICE, Nº CH. HINGE ROTARY ACTUATORS 12 REQD CAILERON'S, MECHANICAL 4 EEQ0. .. © ELEVONS INBERED NIECH -GEEQU. 🕲 ELEVONS OUTECHE: 🕐 4000 BEUDDER ! SPEED BEAKE, MECH HINGE ROTARY ACT. 6 REDD. © ENGINE THPUST IECTOR ACTUATORS TANDEM UNBALANCED, 2 REDU ø AVIONICS SYSTEM 922 LBS; SO FT'SPACE PROVISIONS FOP: INFORMATION MANAGEMENT SYS. COMMUNICATIONS SYS NAVIGATION 515 TARGET ACQUISITION SYS. DEFENSIVE SYS. FLIGHT CONTROL SYS. FURNISHINGS Q. EJECTION SEATS, STENCEL S-IIIS 2 REGO. & OXYGEN SYSTEM - SLITER LOX C RELIEF PROVISICIJS - HORN & BOTTLE (NOT SHOWN)

the second second second

120

8106

ធិភិ

Figure 22. ATS Inboard Profile

A STATE AND A STATE AND A STATE AND A STATE AND AND A STATE AND AND A STATE AND AND A STATE AND A STA

83/84







4.1.4 <u>Electrical System Load Analysis</u> - An electrical load analysis was included as a part of the first interim technical report (Reference 8). Although this load analysis was roughly representative of the loads seen by the electrical system, several changes in system philosophy and refinements in system definition occured after the load analysis was originally issued. These changes were of sufficient magnitude, and had a sufficient impact on the loads reflected back to the generators, that it was felt, that not only was a revision to the load analysis needed, but a new approach to the presentation of the loads was required. The more important of these changes are listed as follows:

- 1. Revised flight control actuation requirements as discussed in paragraph 2.3.1.5.
- 2. Revised utility actuator requirements as discussed in paragraph 2.3.1.
- 3. The deletion of the ground cooling fuel heat sink door as discussed in paragraph 2.3.1.1.
- 4. The deletion of the canopy as an actuation function per paragragh 2.3.1.2.
- 5. The revision in approach to utility engine actuation functions as outlined in paragraph 2.3.1.3.
- 6. The deletion of major portions of the environmental control system (ECS) as a trade study item (See paragraph 2.3.1.4).
- 7. The adoption of the four independent channel approach to electrical power distribution disscussed in paragraph 4.1.2.

The last item (Item 7) was the major determinant of the need for a new method of presenting electrical loads for the purposes of this study. The computerized load analysis system used at Rockwell was eminently satisfactory for presenting and summing the loads on a conventional electrical system using bus tie contactors (i.e. where any pair of generators can eventually see the loads normally carried by four generators). However, the load analysis technique, as constitued at the time of this report, did not gracefully handle the identification and apportioning of various loads among the various busses, the effects of load monitoring, nor the impact of four dedicated systems (as opposed to a bus tie contactor system). Any one or all of these items could have been handled by the computerized load analysis system through extensive revision, however, it was not felt that the required expenditure of time nor the increased complexity of the final readout justified the effort required for this program. Therefore, a revised approach using manual tabulation was devised. The results are shown in Tables 15, 16 (12 pages) and 17. Table 15 is an

1111 INBOARD TRAILING EDGE FLAP MIDSPAN TRAILING EDGE FLAP OUTBOARD TEALING EDGE FLAP (AILERON) 1112 1113 1114 UPPER RUDDER 1115 LOWER RUDDER 1116 LEADING EDGE FLAP 1117 CANARD 1118 THRUST VECTOR VANE EXTERNAL FLAP 1121 1122 PLUG THROAT 1123 THRUST REVERSER 1131 NOSE GEAR 1132 NOSE GEAR DOORS 1133 MAIN GEAR 1134 MAIN GEAR DOORS NOSE GEAR STEERING 1135 1136 MAIN GEAR BRAKES RAM AIR SCOOP (RIGHT) 1141 1142 RAM AIR SCOOP (LEFT) 1143 EXHAUST DOOR-EQUIPMENT BAY GROUND COOLING FUEL HEAT SINK 1144 1151 30 MM GUN DRIVE 1161 REFUEL RECEPTACLE

1162 APU START MOTOR

TABLE 16. COMBAT	- !	5 SEC	LOADS	4	CHANNELS	OPERATIVE	(SHEET	1	0F	12)
------------------	-----	-------	-------	---	----------	-----------	--------	---	----	----	---

ITEM	115/2	00 V	4001	HZ AC	BUS	১ হ্য	270	VDC	805	SES			
NO		2	3	4	EI	E 4	1	2	3	4	E١	E 4	BAT
101	0.216												
107				0.216									
10.5	0.050			Ι.									
104				·									
105	ó. 243						İ						
106				C 243									
107	0.297			1		í							
108				0.297									
109	a243			2249		ļ							
						1	Ì		•	•	•	-	
						ł	ĺ					_	
201	a 837			ł			cc	MB	AT -	55	EC I	.OAC	DS
202	0.270	l		0.270	~	~	1	4	'HAN'	، دیخن کا ا	-iera	TIVE	1
203	0.010							1	1	}	1	1	-
204			ļ	12010				}	2			!	
			1	1.050							1	İ	
302	0.054		Ì	0.054									
303	0.027			0.061									
304	0.054		{ 1	0.037		-							
305	0.027			0.027	-	-							
314	0.027	ļ			1.		1						
315			i	0.027	1	-			}				
3/6	0.027	İ		1	-	1		l	1				
317	[0.027	'	1		1	1				
318	1		1	0.027	1	-	1				1		
	1			12 220									
401	3.220		1	6.A70									ĺ
403	0.010	G		6.054	4								1
1 70/											1		i i
501	-	-		1									
502	1-	· •											
503	-												
504	1-	-					}						
505		1	1					I		1	1		
500	7.3<7	0	0	4.54			0	1-	0	0	-	-	-

ITEM	115/2	200 V	400	HZ AC	BUS	SES	270 VDC BUSSES							
NO	1	2	3	4	EI	E4	1	2	3	4	E.I	E 4	BAT	
509	v													
510		~								-			_	
511	~												-	
512		~										}		
513	~			i i										
514	ļ	~	1											
515	~					1					-			
516	0.135	1		1							i			
517		0.108	ł				}							
518						Í				•	•	•		
101	5012	1				ł]							
607	1- <u>2</u> .912			5 9/2	5	1	co	MA	Δ 	5 5	50			
603	1.863			1.5.5				4	CHAN	NELSO	HERA	TIVE		
604			{ ·	1.863		~				[
605	0.999	[~] -								
606	[ļ	0. 999		~					[
607	0.999	1	1		~									
603	1	1	,	0.999							ļ			
609	1	}			5						_			
610	}	}				~		 ·-		1 }				
}		ļ] '							
701_	1		ł				0.486			0.243		-	~	
702	0.837	1	·			1		} .						
703	}		1				1.296	ļ]	0.679		1	-	
ani	ł					1				-				
907	1						0.210			0.155				
823			·			1	2005		-	0.0-4				
804	1			1		1	0.040	{		0,00.9				
								1						
901			12.744]		1	-					
902		ļ	1.755	ł	l									
903		:	6.162	}		ļ	ļ							
· -		~		1	1		•••	-	-	.				
1001_	·	1 <u>9.79/</u>	·			_				L				
1002			1.701	Į	1								 	
				+										
PG. TOTAL	q i 0.746	19.899	16.362	9774	-		2.269	0	0	1.182	- 1		-	

li

TABLE 16. COMBAT - 5 SEC LOADS 4 CHANNELS OPERATIVE (SHEET 2 OF 12)

NO	1	1	2	Δ	E.	E A		2			E .		DA-
NO	 		3		EI			2	3	4	EI	E 4	
////.		Į .				·· →	7722	8 861	8.861	17.722			Ļ.,
112_		!					3.544	1.772	1.772	3.544			
113					ļ		0.708	0.208	2.758	2 703		1	
114			-				0.912	0.956	0.456	0.912			~
115	L	i		i			0.066	0.433	0.433	0.866			~
116							3.315	3.315	315	3.315			_
117		ľ						0.283	0.293				
118]					1.200			1.200			
	ļ	i											
121]	}	1			13.767			13.767			
22	[1						19910	19 912		. —		
22	ł	1						20 40	20412				— —
\sim		ł						w.o.;	29.013	ł • •			
121		in					ł						
170			~	•			ł						
132	ļ	1	-									~~~) E
33	-				÷	ĺ		MR	- 17	5 30			5
34				-				م ا			PERAI		
35			i -				<u>↓</u>					 	
36	—					_ · - ·	-	· · -	-				
-													
41	-												·
42		1											¥
43_]	i l			1							~
42	Ľ	1											
		1											
15/									*	73.86-			
							Ì						
161	~										••		
162								• •					~
	*A C7	VAL A	CTUAT	TON A	C:VER	COR,	RECT	ED 70	30	-sen	MED	TOTAL	
	BASE	p on	THE .	95 <i>51/</i> /	1PTIC	N Th	AT AL	L ACT	VATIC	W FC	NCTTO	WSR	nz-
	107	PEAK	SIM	ULTAN	EOUS	LY				Í			
OTAL		•						~ ~ ~					_ ~
13 (766		1					۲2.U34	67.327	67.355	~.90			
CRASTO	• -												
total Is phie							29.323	58.239	- 3.22	57.267			
1 100 1 1	19.103	12000	1632	16 321			30.291	58.33	#a 740	5 4 455			

TABLE 16, COMBAT - 5 SEC LOADS 4 CHANNELS OPERATIVE (SHELT 5 OF 12)

91

ſ	ITEM	115/2	200 V	4001	HZ AC	BU	5555	270 VDC BUSSES							
	NO	1	2	3	4	EI	E 4	1	2	3	4	E١	E 4	BAT	
1	1111						ſ	31.837	19000	-				~	
	1112							6.380	2.650	7					
	1113							1.4/7	1.417				-		
	1114							1.366	2683						
	1115							1.300	0.65C	-	-				
	1116		Í					6. 63 0	6.630						
	1117						[•]		0.546	Y C					
	1118							2.400			Ľ			 	
	. –													1	
	112L					.		13.767			ľ				
	1122								42.916		Í			i	
	1123						ĮĮ		23.613	5					
		-						[]				I			
	1131							I							
	1132		~~						MBAT	-55	iec. e	MER	GENC	Y]	
	1/33			/				Į	2 CH	ANNE	ELS 0	PERA	TIVE		
	1134				_				L			Ļ			
	1135		-												
	1136	_													
	-	_													
	1141													-	
	1142													~	
	1143														
	1199	Ľ_													
				-									_		
	1151										·			_	
	1161	٢	▲ 1	SUME	5 50	EFACE	POWZ	RED I	5Y 5//	VGLE	ACTU	TOR			
	11.62		A	ancu A	CHES	STALL	FOR	5 320	ane			•			
		[_									
		₩ AC7	UAL	CUA	TION	POWE	R COA	RECT	20 70	643	0F56	ME	דסד ש	AZ	
				THE A	SCMI	PTION	7 749 7	ALL A	CTUAT	ian r	ENET?	ะพรา	VIL 2 1	07	
		PEAK	31-7	21700	0030	8									
	THIS PAGE	_						3 5 .697	12.135						
7	TOTAL							43.32	55.000	-3/	23.139	-39)+	39		
	HIS PAGE														
	ALL PAGE	10.103	:9 999					45.64	\$5.090						
	TOTAL		{	1											

TABLE 10. COMBAT - 5 SEC EMERGENCY 2 CHANNELS OPERATIVE (SHELT 4 OF 12)

ITEM	115/2	00 V	400	HE AC	: 803	SES	270 VOC BUSSES							
NO	11	2	3	4	EI	E4	1	2	3	4	E١	E4	BAT	
101	6.216													
107				6.216							i			
103	0.054													
104				0.054										
105	6.243													
106				0.243										
107	a297											1		
108				0.297										
109	0.243		1	6243								1		
110	1 1			0.243	~			t I			i	ł		
///						}								
201	6.037							OM			NITI)		19	
202	0.270			0.270	~	~		.061	A CHAR		OPERA	TIVE		
203	0.8.0					}							~	
204				0.510										
									i					
302	0054	i		0.054	~					 !		1		
303	0.0Z7			0.027								1		
304	0.054			0.054				1			[
305	0027			0.027		•								
306	0.027			0027										
3/4	0.027							Ì						
313	0 177			0027	~									
3/7	0.021			0.027		-		ł						
318				0017										
						1	}	}						
401	2 310			2.310			ł							
403	0.520			0.520		1	}							
407	0054			0.054				1				ł		
							ł					Ì		
501	Ľ													
502				1	ł	1		Į						
503	2			i	1	1								
505	~	~			1			}						
506		~		l	{									
PG TOTAL	6.097	0	0	5.197										

TABLE 16. COMBAT - CONTINUOUS 4 CUANNELS OPERATIVE (SHEET 5 OF 12)
ITEM	115/	200 V	4001	HZ AC	BUS	55ES	270	VDC	BUS	SES	, ,		
NO	1	2	3	4	EI	E4	1	2	3	4	E١	E 4	BAT
509	~												
510						[
511	~												
512	Į	~			ĺ								
513	-												
514	ł			İ	1				1				
515	r.				ł	1					1		
516	6.135			1						[{	
517	1	0.108			l					{			
518	í						l i			{			
601	2.133			1									
602				2.133			l cc	MB	∖ T ~	CON	TIN	non	15
603	2675					~	i i	40	HANN	EC2 01	(ERAT	ve 	
604			:	0.675	~	-				}			
605	p.297				_					l	}		
606				0297	~	ľ						i i	
607_	p.297		1			~							
605		:		0.297						1			
609				-	ľ	~						·	
610					ļ			-		ļ	,		
701			1]		1 44			1201		-	~
102	Sa	7		Į		}	0		- 1	0.270			
203	P [.] <i>O</i> , <i>j</i>		•	1			1.296			0.40		~	~
					}	}				[
801							0.270			0135		~	~
802	1			ļ	1	}	0.108	L		0.054		2	
803							0.068			0.068	•		
804		ļ	}		Í		C040			0040		ļ	
_		·	-	-) }
901	i		12.744	4	1	1	1	1	1			 	
902		i	7.755	1	{	1				' ا		ļ	[
303]	1	<i>p.162</i>		l			Į		i		 -	
· -		-		!	1		1	j			-		
1001		1 <u>9.79/</u>		-						}	ļ		
1.002			1.701					·		· · · ·			
1	1	1000	1621	2 42			1 20			1,100		<u> </u>	
1.002 PG TOTA	4.33	19.899	1.701 V6.362	3.424			2.263	 C	·	1.185			

TABLE 16. COMBAT - CONTINUOUS 4 CHANNELS OPERATIVE (SHEET 6 OF 12)

. • ``

ITEM	115	200 V	400	HE A	BUS	SES	270	DVDC	EU:	SSES	5		
NO	1	2	3	4	EI	E4	1	2	3	4	EI	E4	BAT
1111							8861	4.43/	4.431	8.861			0
1112_]					1.772	0.890	0.086	1.772			~
1113		1]			0.35	0.35	0.354	0.354			
1114			1	1			0.456	0228	C. 228	0.456			
1115			Ì	1		[043E	0.217	0.217	0433	1.		
1116_			1	1		ļ	1.82	1.82	1,823	1.823			
1117	L	1				ļ .	~	0.142	0.102	,			
1118	1	}				· · · · ·	0.600	2 _		0.600			
	1	1							1				
ILZL.]	1					6.88			6.884	-		
1122	ł		1	1				21.459	21459	/	L		
1123		Ì	1	1				1-	1 · ·	ļ			
{	 		~	{ .			-						
1131	ł	5	~				-						
1132	l	ſ	~				C	OMP	AT-	CON	TIN	UOU	s
1133	. .	[1				4	CHANN	ELS C	PERAT	IVE	-
11.34		- 1	i	1 1				·					
11.35			-			• • ••••						~	~
126	- 1				+		-		-				
line			í						}				1-1
1137					{								/
1143	<u> </u>	{						ł			-	i	
1197	~							İ	}		-	· -	
	- · ·	Į	i -			•	•	1					
1151		1						1		~			
		1						1			1		
1161	1					1		1					·
11.62		D APU	STAL	BUS	[Ŧ			, 1	•		
					ĺ			1				j	
		}								1			{
·													{
TOTAL							-			1		1	
THIS PAGE		•					21./83	29.540	29.54	21.183			
CAR PORTON	• •									1			
TOTIL										_			
											_		
ALL PAGE	10,000	aear	1 an	0 6 7 1		Į	12 11-	20 - 4	الم زير م		. 1	- 1	
	10.733	13053	Y 0.366	0.021		k	w 43	<i>⊾9.94</i> 0	29.340	2237/		1	1

TABLE 16. COMBAT - 5 CONTINUOUS 4 CHANNELS OPERATIVE (SHEET 7 OF 12)

ITEM	115/2	200 V	4001	HZ AC	- 80	SES	270	VDC	BU:	SSES			
NO	I	2	3	4	EI	E4	1	2	3	4	E١	E 4	BAT
1111							15.919	6.615					-
1117							3.190	1.329		2			
1113							0.709	a709	P	[5	
1114							0.683	2342					
1115					-]	0 650	0.325	-	5			~
1116].	3.3/5	3.315			-		
1117				-		ļ		0.283	~				
1118							1.200						
1						ĺ						i	
1121							6.884			·			
1122						[{	21.549		1			
1123	-							Ľ		Ì			
							l						-
1131								40 4 1		AL 7 16.11	10.10	5 .45	00
1132								1001		NIN	2002		K0
1133								אר א י - יי			- KAT	ME	
1134		~		·									
1135					~			-					
1136												-	
											-		
1141		• • •					'					Í	
1142					-								
1143_	-												
11.99	+	· •									ļ		
								•		~		-	
115/							ļ					-	
1161	ŀ	a				ļ					•		
11.62		- 76	J 5TA	RT BU	s								
ł									-				
						[[
	[\vdash			l	{						
TOTAL		-	• •				87 650				-		
WHIS PASE		1				1	002.30	J#.377					
CARACTED													
THIS PREE	1		•				· -			· · ·		····	
Le an		-		· ••									
TOTAL	19.103	19.959					32 55	34 377				+	

TABLE 16. COMBAT - CONTINUOUS EMERGENCY 2 CHANNELS OPERATIVE (SHEET 8 OF 12)

ITEM	115/2	200 V	400	HZ AC	80	SES	270	VDC	BUS	SES			
NO	1	2	3	4	EI	E4	1	2	3	4	EI	E4	BAT
11/1							13.212			13.212			
1112	}					·	2.659			2.659			
1113							. 709			.709			
1110			;				684			.684			
1115			1	·			.650			.650			
1116				İ			335			3.3/5			
1117		-	1			}	1						
1118	r - 1	-	1										
1121							i						
1122						1	l c	омв	AT -	CON	ITIN	uou	s
1123								EME	RGE	NCY	API	J	-
~~~	<b> </b> -							CHANI	NELS	I AN	040	PERA	TIVE
1131	[												
1132			···										
1/33			[				[						
1134			l										
1135			i :	-									
1136							· · · · ·					1	
	<b>-</b>				-								
1141									:				
1142													
1143.			i										
1199												i	
												1	
115/					1								
ł		ł											
1161	;	ļ					-				_		
1162						ļ					•		
		1				1	]						
						l							
		1 .	Į			1							
		ł	ļ			1							
THIS PAGE		1				1	21.229			21.22:			
		l		1 !		ĺ							
TOTAL	]					1	21.229			21.225			
HIS MOL	-				-		+2 268			·',eg			
ALL PAGE	\$,725			4.75Z			23,497			22.417			
TOTAL	1	1	1	5457	EM TO	TAL	28.222			27165			

TABLE 16, COMBAT - CONTINUOUS EMERGENCY APU CHANNELS 1 AND 4 OPERATIVE (SHEET 9 OF 12)

• #. -

ITEM	1157	200 V	400	HZ A	C 811	SSES	270	NUC	- ALI				
NO	1	2	3	4	EI	E4	1	2	13	A	EI	E 4	BAT
1111	<b> </b>	†	+	+	+	+	26.50		+	26.583			
1112		1	1	· [	1	1	5.317	·†	+	5.3/7	<u> </u>	<b>[</b>	
1113		<u> </u>	1	1	1		1.417	1	1	1.417			
1114_			1	T	<b>[</b>	1	1.36			1.367	1		
1115	[		1		1	1	1.300	, 		1.300			
1/16_	<b> </b>		<b>L</b> .		1	1	6.630			6.630			
1117	┣	_	1	4		<b>!</b>				<u> </u>	<b> </b>		
1118					┥──-	╄	ł	↓	<u> </u>	┟		i	
	{	• ·		ł			·	I	1	l	ł	<u> </u>	
1/2/_			1										
11.66			ł	1			c	OMB		CONT		SUS	-
123	-		1	· ·	}		ci	HANNE	RUER ELS I	AND 4	H PU L OPE	RATIN	1e -
1131		t ·	<u>}</u>	<u>†</u> ·	<u>†</u>	†	†						
1132				<b>†</b>			<b>†</b>						{
1133			1	i				t	f				
1134			 			L		·					
1135							L						
1136						ļ. <u>.</u>							
					ļ		i	}					]
1141			}.										]
1192			1										]
1173					·· .	ł	1						
1273		• -		·	-			• • • •	- <u></u> - :	h			
1151		ł								¦.		+	
												ł.	-
1161						•					·	··	
1162											• 1		
										•	ч. Ч	·	
	-	·.	- ·						. 1				
TOTAL		-	-										
THIS PAGE	j						42614			42.6/4			
COLEGITED							28.00		•••				]
TOTAL THIS PAGE							<i>CO</i> ,409		·}	28.409			]
ALL PAR	11.097		11.12.0	•••			30.67			74 500			
TOTAL	r		· · <b>-</b> 7	5757	EM T	OTAL	41.774			40.72/	-		

TABLE 16. COMBAT - CONTINUOUS EMERGENCY AND CHANNELS 1 AND 4 OPERATIVE (SHEET 10 OF 12)

The second rule is a

are a record against the second

ITEM	115/2	00 V	4001	HZ AC	BUS	SES	270	VDC	BU	SES	5		
NO	1	2	3	4	EI	E4	1	2	13	4	EI	E4	BAT
1111							18.608						
1112			-		[		3.722		· · ·				
1113							.709						
1114			}				.684						
1115						ļ	.650						
1116							3.915					L	
1117										·			
1118			;			┣				<u> </u>	ļ		
11.21_ 11.22 11.23	-						c	OMB En CH	AT - AERG		ITINI Y AF		5
11.31											<b> </b>		
1132									L		ļ		
1133										1			
1134									}				
1135			- ·	× ، ، ، س									
1136	-		ļ		• •	·•• <del>·</del>							
1141 11 <u>4</u> 2										-			
1193_													
1/44		·	·- ,	• • • • •					ł		ł		
1,51	——					-	-						
11.51									ł	1			-
1161												t	
1167								• • •	Ì				
						ļ					·		
						1				· —			
						1							
TOTAL THIS PAGE							27.689						
					}					. –			
TOTAL			i i			<b>.</b> .	+ 2.268	_					
					l		29.954						
ALL PAGE TOTAL	<del>\$</del> .725						+9.785	· • •		· •			
10176	1		t i	Jrs	VEM 7.	DTAL	19.60/		1				

TABLE 16. COMBAT - CONTINUOUS EMURGENCY APU CHANNEL 1 OPERATIVE (SHEET 11 OF 12)

74 C

.

l

......

99

-----

ITEM	115/2	200 V	400	HZ AC	BUS	SES	270	VDC	BUS	SES	5		
NO	1	2	3	4	EI	E4	1	2	3	4	E١	E 4	BAT
1111							37.216						
1112							7.444				· ·		
1113							1.417						
1114			r				1.367						
1115							1.300						
1116							6.630				<u> </u>		
1117						L							
1118			!			L.					<u> </u>		
											¦		
112L										-			
11.22											i		
1123	_		(			L_	•						
				L			ļo	OMB	AT -	CON	TINU	ous	
1131								E٢	AERG	ENC	APL	5	-
1132								CHA	NNEL	LOF	ERAT	IVE	
1133	_				I	- 1							
1134			 								 		
1135			·			ĺ							
1136		_											
{ .						i .							
1141													
1192					<b>.</b> .					• •			
1143_													
1199										-			
		-							-	-			
1151	-												• •
										-	- · · <b></b> ·	<b></b>	i
1161										··	• •		
11.62										_			
{	·		-						.				
					•	1						]	
·	-	-	L										
TOTAL			·						-				
THISPACE		-					55.374						
CORRETTO									-		·		
TOTAL HIS PACE							36.916				~		
		-					+2.248				~		
TOTAL	11.097				TELA	here	29.104	·					
1.0.0				223	IST I	DIAL	PU.20/		1			- i	

TABLE 16. COMBAT - CONTINUOUS EMERGENCY APU CHANNEL 1 OPERATIVE (SHEET 12 OF 12)

extension, and further breakdown, of the item number breakdown used in the original load analysis (reference 8, Appendix A pages A-11, A-10, A-21, A-26, and A-31). Basically Table 15 is a functional breakdown of item 1100 on page A-31 of Appendix A. Table 15 assigns an item number to each function in the eleven hundred series instead of breaking down the actuation functions by mission segments as was done on page A-31. This brought the eleven hundred series breakdown in line with that used for the rest of the original electrical load analysis.

Table 16 itemizes the loads occuring during combat for the 5 second and continuous load condition. The combat mission segment was selected for detailed breakdown because it represented the highest loads imposed on the generator and on the AMAD and engine PTO system. The first column on Table 16 lists the functional item numbers as taken from Appendix A plus the revised breakdown of the eleven hundred series.

The next thirteen columns in Table 16 list the various busses used in the 400 Hz AC and 270 VDC systems of aircraft I. They match the busses shown in the aircraft I power generation and distribution system schematic of Figure 16. The bus (or busses) to which the functional load, represented by an item number, was attached was indicated by a check mark. The magnitude of the load was entered as a numerical value in the column representing the AC or DC system (No 1, 2, 3 or 4) in which the load ultimately appeared. The location of the check mark and the load numerical value did not necessarily coincide. This occured only when the load was attached to the system's primary bus. Where no numerical load value was entered for a particular item number in the table, it indicated that no power was provided to that function during the mission segment under consideration (combat) or that the time under load was so short that it appeared only under 5 sec. loads but not under continuous loads (item 1151 -30 MM gun drive for example). Sheets 1, 2 and 3 of Table 16 tabulate the loads appearing on the AC and DC busses of power systems 1, 2, 3, and 4 under the "combat 5 sec. load" condition with 4 channels (systems) operative. Combat 5 sec. loads were a basis for determining the overload requirements of the generator system. Sheet 4 of Table 16 shows the "combat 5 sec. emergency loads" with only 2 systems operative. Sheets 5, 6 and 7 show the "combat-continuous load" condition with 4 channels (systems) operative. Combat-continuous loads were the basis for determining the basic rating requirements of the generator system. Sheet 8 of Table 16 shows the "combat-continuous emergency loads" with only 2 systems operative. In both the case of sheet 4 and sheet 8 "emergency" loads, the loads for item numbers less than item 1111 were not included because the loads for these lesser item numbers were identical to the values already listed on sheets 1 and 2 (vis a vis page 4) and sheets 5 and 6 (vis à vis sheet 8). The item loads for each bus were totaled sheet by sheet for each condition (5 second or continuous) and a grand total was accumulated for each bus on the third sheet. It will be noted, on sheet 3 and 4 on Table 16, that only 2/3 of the actuation loads were used for determining the total load for the 5 second condition. This was in consonance with the ground rule established in paragraph 2.3.4.

In addition to load determinations, the level of power source redundancy could be approximated from Table 16. The number of columns in which load entries (or check marks) appear for a given item number indicate the level of power source redundancy for that function. As an example of the extremes of power source redundancy which were incorporated in the actuation functions of aircraft I, consider the inboard flaps (item 1111) and the main landing gear (item 1133). The inboard flaps had access to 5 power fources (4 generators plus a battery) while the main gear bod access to only one. Actually, when considering primary power sources, the disparity between the two was not as great as it would at first appear. The mid span flaps had access to 3 primary power sources (2 engines and a battery) while the main gear hal access to 2 (an engine and free fall). In both instances the APU was not considered a primary power source because it could only be started below 20,000 feet.

It will be noted in Table 16 that the loads on an individual operating system increase as other systems become inoperative. As an example, consider DC busses No. 1 and No. 2 on sheets 3 and 4 respectively of Table 16. Sheet 3 sums the DC loads for all four busses at the bottom of the sheet (all page total) for the condition where all 4 channels (systems) are fully operative. sheet 4 sums the loads in a similar manner for systems No. 1 and No. 2 with the assumption that systems No. 3 and No. 4 have failed. This set of circumstances could occur, as an example, if high altitude battle damage had been experienced in which system No. 3 and No. 4 had been wiped out, and in which violent evasive maneuvers were in progress, and the APU could not be started because of the high altitude. Table 17 provides a summation of the AC, DC, and total loads accumulated on sheets 1 through 8 of Table 16. It can be seen from Table 17 that the generator size was determined from the loads on power system (channel) No. 2 under the headings "combat continuous emergency loads" and "combat 5 sec. load emergency" with 2 channels operative". From the summation it can be seen that the selected generator ratings were satisfactory. While the continuous rating of 70 KW selected for the generator left an apparent 29% margin for growth, the 120 KW 5 sec rating had a much smaller 4.4% margin but was still satisfactory. (See paragraph 4.1.6.)

The loads used in the preparation of Table 16 were taken from two sources. The first source was the electrical load analysis, reference 8, Appendix A pages 1 through 46. This was used for determining all loads associated with functions through item No. 1002. The second source was the utility and flight control function loads included as Figure 7 and Figure 8 in this report. These tables give output loads at the surface or function being powered. In order to convert these output loads to loads at the generator terminals, it was necessary to provide system loss data. This loss data for flight control functions is provided by Figure 24. The figure shows that the losses were broken down into three major categories; (1) power hinge losses, (2) motor losses, and (3) distribution losses. As will be seen later (paragraph 4.1.5) these losses, in terms of percent rated loads, were nearly identical for all TABLE 17. AIRCEAT J + ELECTRICAL SYSTEM LOAD SUMMATION

Theory and

POWER DESCRIPTION	POWE	R SYSTE	m nume	JER
		2	<b></b>	4
COMBAT CONTINUOUS LOA	DS - 4 C	HANNELS	5 OPERA	TIVE
AC POWER (KW)	10.493	19.899	16.362	8.621
PC POWER (KW)	23.415	29.540	29.540	175.25
TOTAL SYSTEM POWER(KW)	33.908	49.439	45.902	30.992
COMBAT CONT. EMERGENC	Y LOADS-	2 CHANNI	ELS OFE	<b>ZAT IVE</b>
AC POWER (KW)	18.103	19.899		
DC POWER (KW)	32.550	34.377		
TOTAL SYSTEM POWER (KW)	50.653	54.276		
COMBAT 5 SEC. LOADS	- 4 CHA	NNELS	OPERAT	IVE
AC POWER (KW)	18. 103	19.899	16.362	16.362
DC POWER (KW)	30.219	58.239	59.239	58.455
TOTAL SYSTEM POWER (KW)	48.394	78.138	74.601	74.266
COMBAT 5SEC. LOAD EMER	GENCY -	2 CHANN	IELS OPE	RATIVE
AC POWER (KW)	18.103	19.899		
DC POWER (KW)	45.666	95.090		
TOTAL SYSTEM POWER(KW)	63.769	114.989		





Figure 24. Flight Control Actuator Load Characteristics During Combat.

flight control actuation functions except the leading edge flap. The losses in the leading edge flap system were purposely made high to give it "no back" characteristics in the face of a power system failure. Because the losses in most of the flight control system functions were nearly identical (and for this study were treated as being identical) they were plotted by category as the solid lines in Figure 24. Only the limits for the leading edge flap function, i.e. the output power at the surface and the generator terminal power, were plotted in Figure 24 to avoid complicating the figure.

The power hinge losses, as used in Figure 24, consisted of the losses in the power hinge itself (or, in the case of the canard function, in the ball screw) and the losses in the gearhead. The motor losses consisted of the losses in the motor itself plus the losses in the inverter. The distribution losses consisted of all losses in the distribution system between the generator terminals and the inverter terminals. It can be seen in Figure 24 that the typical flight control actuation function was 60% power efficient at rated conditions and that the leading edge flap system was only 34% efficient. The individual efficiencies which provided the basis for these values are discussed later in more detail (paragraph 4.1.5). It can also be seen in Figure 24 that the power required at the generator terminals for "continuous" operation was 50% (57% for the leading edge flaps) of that required at rated power design conditions (5 sec loads). It is also interesting to note that, even with no load on the output, 25% of rated power is required (36% for the leading edge flaps) at the generator terminals to achieve rated rates of motion. The data from Figure 24 was used to assist in determining the load entries for item numbers 1111 through 1123 in Table 16.

Another factor which had to be considered in generating the flight control function entries in Table 16 was the fact that, when more than one actuator was powering a surface or control function, the actuators shared load and the individual actuator loads were reduced. In contrast, when various systems were rendered inoperative to the point that only one actuator powered a particular surface, or control function, that actuator attempted to carry full load but could only do so up to the limits of its design rated load limit capability. Calculations were made based on design rated (5 sec load) conditions to take this into account. These calculations are as follows:

### 5 SECOND LOAD ACTUATOR POWER COMPUTATIONS

Pds	=	Design power per surface, at design load and rate, required at surface
P _{DA}	-	Design power per actuator, at design load and rate, required at surface
Pœ	=	Power required at generator terminals per surface
Pga	=	Power required at generator terminals per actuator
Ra	=	Percent of design power $P_{DS}$ or $P_{GS}$ required per actuator x 0.01
ήн	=	Efficiency - motor mounting interface to surface being actuated
ήm	7	Efficiency - motor at design power
クヮ	=	Efficiency - distribution system at design power
P _{GS}	=	$\frac{P_{DS}}{\eta + \chi \eta + \chi \eta} = P_{GA} = P_{GS} \chi R_A$
Pgap	=	Power required per actuator at generator terminals when operated in parallel with other actuators on the same surface.
Nas	Ξ	Number of actuators operating per surface
PGAP	2	P GS NAS

## DESIGN POINT POWER INPUT (5 SEC) LOADS

INBOARD F	LAP (I	TEM NO.	<u>1111)</u>				
P	DS	2	15.95 KW (	1	Λm	Ξ	0.850
R	RA	=	0.70 (2)		ηD	=	0.939
へ	Н	2	0.752 (3	)	,		
P	്ട	=	15.950	v 0 020	PGA	=	26.546 x 0.70
		U./:	2 X U.85U	X 0.939			
		2	26.583 KW			2	18.608 KW
P	GAP	3	<u>26.583</u> 3	= 8.861	KW (N _{AS}	= 3)	
P	ĜAP	=	<u>26.583</u> 2	= 13.229	KW (N _{AS}	<b>≖</b> 2)	
P	GAP	2	P _{GA} = 18.	.608 KW	(NAS	= 1)	
MTOSDAN S			_				
MUSTAN I	LAP (I	TEM NO.	<u>1112)</u>				
P	'DS	<u>=</u>	<u>1112)</u> 3.19 кж (	D	ήм	2	0.580
P	'DS RA	<u>=</u> =	<u>1112)</u> 3.19 KN ( 0.70 (2)	Ð	ጎ m ⁄ ነ o	-	0.580 0.940
P P R F	<u>-LAP (1</u> Pos Ra } H	<u>TEM NO.</u> = = =	<u>1112)</u> 3.19 KH ( 0.70 (2) .752 (3)	〕 )	ጎ m ⁄ ነ D	=	0.580 0.940
P P R F P	<u>-сар (1</u> ² ds Ra } н ?gs	<u>TEM NO.</u> = = =	1112) 3.19 KH ( 0.70 (2) .752 (3) <u>3.190</u>		ጎ м ሲ D Р _{GA}	=	0.580 0.940 5.317 x 0.70
P R F P	<u>-LAP (1</u> 2DS RA } H 2GS	<u>TEM NO.</u> = = =	1112) 3.19 KW ( 0.70 (2) .752 (3) <u>3.190</u> 0.600	) )	ήм ηρ Р _{GA}	# #	0.580 0.940 5.317 x 0.70 3.722 KW
P R F P	<u>-сар (1</u> 2ds Ra } н ?gs	<u>TEM NO.</u> = = = =	1112) 3.19 KW ( 0.70 (2) .752 (3) <u>3.190</u> 0.600 5.317 KW		ጎ m	8 9 9	0.580 0.940 5.317 x 0.70 3.722 KW
P P P P P	- <u>LAP (1</u> 'DS RA } H 'GS	<u>TEM NO.</u>	1112) 3.19 KW ( 0.70 (2) .752 (3) <u>3.190</u> 0.600 5.317 KW <u>5.317</u> 3	) = 1.772	<b>八</b> m 人 D P _{GA} KW (N _A	= = = s= 3)	0.580 0.940 5.317 x 0.70 3.722 KW
P R P P P	- <u>LAP (1</u> ² DS ² A ² GAP ² GAP	<u>TEM NO.</u>	1112) 3.19 KW ( 0.70 (2) .752 (3) <u>3.190</u> 0.600 5.317 KW <u>5.317</u> 3 <u>5.317</u> 2	<ul> <li>1)</li> <li>= 1.772</li> <li>= 2.658</li> </ul>	<b>八</b> m 人 D P _G A KW (N _A KW (NA	= = s= 3) s= 2)	0.580 0.940 5.317 x 0.70 3.722 KW

### OUTBOARD TRAILING EDGE (ITEM NO. 1113)

Pos	=	0.85 KW 🚺				れ	М	*	0.8	50
R _A	z	1.00 ②				ላ	D	=	0.9	39
ήн	£	.752 (3)								
PGS	2	0.850		P	GA	=		1.41	17 х 17 К	1.00 W
	a	1.417 KW								
PGAP	=	$\frac{1.417}{2}$ =	.708	ĸw	(NAS	æ	2)			
PGAP	=	PGA =	1.417	ĸw	(N _{AS}	3	1)			

#### UPPER RUDDER (ITEM NO. 1114)

Pds	2	0.82 KW (1)	<mark>ήм =</mark> 0.850
R _A	=	0.50 2	Λ _D = 0.939
ήн	2	0.752 3	
P GS	=	0.820 0.600 1.367 KW	P _{GA} ≕ 1.367 x 0.500 = 0.683 KW
PGAP	=	$\frac{1.367}{3}$ = 0.456	KW (N _{AF} = 3)
PGAP	Ξ.	$\frac{1.367}{2} = 0.683$	Kw (N _{AF} = 2)
P _{GAP}	=	P _{GA} = 0.683	$(N_{AF} = 1)$

### LOWER RUDDER (ITEM NO. 1115)

PDS	2	0.780 KW 🕕	ήм	=	0.850
R _A	=	0.500 (2)	ΛD	=	0.939
ήн	=	0.752 3			
PGS	2	<u>0.780</u> .600 1.300 KW	P _{GA}	5	1.300 x 0.500 .650 KW
PGAP	-	$\frac{1.300}{3}$ = 0.433 K	W (N _{AF}	= 3)	
PGAP	=	<u>1.300</u> = 0.650 KW 2	(N _{AF}	= 2)	
PGAP	Ξ	PGA = 0.650 KW	(NAF	= 1)	

STATE AND A

LEADING EDGE FLAP (ITEM NO. 1116)

Pos	=	2.24/3 = 0.747 KW / M	=	0.850
R _A	#	1.000 ② 内口	=	0.939
<b>Л</b> н	=	0.424 (3)		
PGS	Ξ	0.747 PGA 0.424 x 0.850 x 0.939	2	2.210 x 1.000
	z	2.210 KW	-	2.210 84
PGAP	2	$\frac{2.210}{2}$ = 1.105 KW (N _{AF}	= 2)	
PGAP	=	$P_{GA} = 2.210 \text{ KW} (N_{AF})$	= 1)	

CANARD	(ITEM N	0. 1117)	-							
	PDS	=	0.170 4	(W ()	ŀ		ήм	5	0.850	
	R _A	=	1.000 (	2			れヮ	2	0.940	
	ήн	=	0.752	3						
	PGS	=	0.170				PGA	2	0.283 0.283	x 1.000 KW
		2	0.283	ĸw						
	PGAP	=	0.283	=	0.142	ĸw	(N _A	F=	2)	
	PCAP	=	PGA	=	0.283	KW	(NA	F =	1)	

THRUST VECTOR VANE (ITEM NO. 1118)

Pos	=	0.72 אא 🚺	)		Лн	2	0.850
R _A	=	1.000 2			hο	=	0.939
ήн	=	0.752 (3)					
٩œ	Ħ	0.720			PGA	3	1.200 x 1.000 1.200 KW
	2	1.200 KW					
PGAP	z	<u>1,200</u> 2	2	0.600 KW	(N _{AF}	3	2)
PGAP	-	PGA	Ħ	1.200 KW	(N _{AF}	=	1)

## EXTERNAL FLAP (ITEM NO. 1121)

PDS	=	4.130 KN 👍	∕\н =	0.850
R _A	=	1.000	\□ =	0.939
Λн	2	0.752 ③		
P _{GS}	ŧ	<u>4.130</u> 0.600	PGA =	6.883 x 1.000
	E	6.883 Ki	=	6.883 KJ
PGAP	2	P _{GA} ⇒ 6.883 KN	(N _{AF} =	1)

# PLUG THROAT (ITEM NO. 1122)

PDS	2	25.520 KN (4)	ስ M	= 0.850
R _A	=	1.000	/ D =	* 0.939
<b>⁄</b> { н	=	0.745 (5)		
PGS	. \$	25.520 0.745 x 0.850 x 0.939	P GA =	42.918 x 1.000
	=	42.918 KW	=	42.918 KW
P GAP	*	PGA = 42.918 KW		

#### THRUST REVERSER (ITEM NO. 1123)

Pos	3	19.640 KW 🕘	ήм	=	0.850
R _A	2	1.000	<b>Λ</b> D	=	0.939
hн	=	0.860 (5)			
PGS	=	19.640 0.860 x 0.850 x 0.939	PGA	=	28.613 x 1.000 28.613 KW
	2	28.613 KW			
PGAP	2	P _{GA} = 28.613 KW			

- See Table 8
- 2 See Table 11
- 3 See Tables 18 and 19
- 4 See Table 9
- 5 See Table 9

2 -----

Another factor considered in making the load entries in Table 16 was the fact that the flight control functions (item 1111 through 1118) and the engine (flight control type) functions (items 1121 through 1123) were motor loads. This meant that, when approaching an output stall condition, these functions could increase the apparent load at the generator terminals by nearly a factor of 2.5. This phenomena is discussed in more detail in paragraph 4.1.S.1.1. To account for this, the load analysis of Table 16 assumed that, during an emergency (5 sec loads), at least one actuator approached stall for a short period of time in recovering from the maneuver which the aircraft found itself in at the time of the emergency. This is illustrated in Table 16 page 4.

Four additional pages (sheets 9 through 12) were added to the Table 16 load analysis. These were added to cover emergency loads experienced during APU operation and were used to help size the APU and the two APU generators. The sizing of the APU generators has already been discussed in paragraph 4.1.2.3 and, as pointed out, the sizing used the data from Table 16 sheets 11 and 12. Sheets 9 and 10 of Table 16 were included to provide the data for sizing the APU itself. The maximum load on the APU was that resulting from operating two systems simultaneously or the sum of the two generator loads. From sheet 9 it can be seen that the maximum continuous APU load requirement was 28.222 KW + 27.169 KW = 55.391 KW (74.25H.P.) and from sheet 10 it can be seen that the maximum design load (5 sec) was 41.774 KW + 40.721 KW = 82.495 KW (110.58 H.P.).

4.1.5 Power Utilization - For the purpose of this portion of the actuation trade study the utilization functions on the aircraft were divided into three general categories as follows:

- 1. Flight control actuation
- 2. Utility actuation
- 3. Other power consuming systems

The first two categories had a major impact on all aspects of the trade study while the impact of the third category was largely confined to it's affects on generator sizing.

4.1.5.1 <u>Flight Control Actuation</u> - The design and definition of the major flight control actuators was subcontracted to Airesearch because of their extensive experience in the development of electro mechanical actuation systems (Reference 17 and 20). Using the flight control actuation requirements listed in Table 8 Airesearch submitted a comprehensive set of preliminary design parameters. These are shown as Tables 18, 19, and 20. Airesearch also submitted envelope and weight data which are shown in Figures 25 through 31. It can be seen that both the inboard flap actuation system (Figure 25) and the airleron actuation system (Figure 27) met the envelope requirements established in Figures 14 and 15 respectively even though the fit in both instances was very tight.

	11 - 12			Drives	Type**
Actuator	l nad	Ratio*	η _{GR} *	per	Dianieter x Length
	FUGUE			Surface	Weight
					Epicyclic
Inboard Flap	151,000 in-lb	1320	0.80	1-3	3.5 x 9.4 in
•					11.1 lb
					Epicyclic
Midspan Flap	dl-ni 000,00	1320	0.80	1-3	1.6 x 9.1 in
					2.9 lb
					Epicyclic
Aileron	dl-ni 000,05	2640	0.80	1-2	1.4 x 11.8 in
		č			3.2 lb
					Epicyclic
Upper Rudder	dl-ni 000,0	3300	0.80	1-3	1.4 x 4.1 in
					1.1 lb
					Epicyclic
Lower Rudder	13,000 in-lb	1650	0.80	1-3	<b>1.6 x 4.0 in</b>
					1.3 lb
					Epicyclic
L.E. Flap	175,000 in-lb	13,200	0.45	1-2	4.0 x 7.2 in
					12.8 lb
		1 0E 10-2			Ball Screw ^{###}
Canard	2,400 lb		0.80	1-2	1.0(2.0) x 19.9 in
		111-11GV			11.6 lb
* Overall, u	using motor dat	a from calc.	No. 1-002.	See attach	ed procedure
** Selected	tõ meet crvelo	ne & perform	ance *	(00) 01 **	ХĽ

TABLE 18. APPROXIMATE DRIVE SIZINC

TABLE 19. APPROXIMATE GEARIBEAD SIZING

ŝ

Actuator	Input Stail Toroue	Ratio*	η _{GH} *	Qty per Surface	Diameter x Length Weight **
Inboard Flap	141.8 in-lb	06~	46.0	e	~5.5 x 5.0 in ~15.0 lb
Midspan Flap	28.4 in-lb	06~	0.94	e	3.0 x 3.5 in 6.0 lb
Aileron	12.6 in-lb	~180	0.94	7	2.5 x 3.0 in 2.5 lb
Upper Rudder	3.2 in-lb	~220	0.94	3	1.5 x 1.8 in 1.0 lb
Lower Rudder	6.3 in-lþ	~110	0.94	ß	2.0 x 2.0 in 2.0 lb
L.E. Flap	15.8 in-lb	~ 90	0.94	2	2.5 x 3.0 in 3.0 lb
Canard	3.2 in-lb	~20	46.0	3	1.5 x 1.8 in 1.0 lb
° Forgea °° Selecte	rhead only. Siz d to satisfy inpu	ed by uso of It torque req	data from uirements	Air Report N	0. 80-17284

TABLE 20. APPRONIMATE MOTOR SIZING*

Actuator	Calc'd Power	Motor Power	η _{GR}	Motors për Surface	speed** Diameter x Length Weight
Inboard Flap	44.93 hp 33.52 kw	45.0 lıp 33.6 kw	0.80	<b>m</b> .	20,000 rpm 3.5 X 10.6 in 12 lb
Midspan Flap	8.93 hp 6.66 kw	9.0 hp 6.7 kw	0.80	3	20,000 rpm 2.3 x 7.7 in 2.3 lb
Aileron	3.87 hp 2.89 kw	4.0 lıp 3.0 kw	0.80	2	20,000 rpm 1.9 x 6.7 in 1.0 lb
Upper Rudder	1.07 hp 0.80 kw	1.0 hp 0.75 kw	0.80	e	20,000 rpm 1.9 x 5.1 in 0.3 lb
Lower Rucher	1.55 hp 1.15 kw	2.0 hp 1.5 kw	0.80	ę	10,000 rpm 2.3 x 6.0 in 2.4 lb
L.E. Flap	5.15 hp 3.84 kw	5.0 hp 3.7 kw	0.45	2	10,000 rpm 2.3 x 8.1 in 2.6 lb
Canard	0.94 hp 0.70 kw	1.0 hp 0.75 kw	0.80	7	10,000 rpm 2.3 x 5.3 in 0.5 lb
» Assu «** Sele	imes drive effic cted to meet en	iency of $\eta_{\mathrm{Er}}$ velope & per	. Sized by formance	attached pr	ocedure

116

-----

------

-







Name L

· · · · · · · · · · · ·

î,

Solution to address



-23.6-

 $\sim 15\,\mathrm{lb}$ 

ì

ę



Figure 28. Upper Rudder installation



Figure 29. Lower Rudder Installation.

21 lb

121

i i



20 lb ( $\sim$ 240 lb for 12 actr's)

Figure 30. L.E. Flap Installation



;



Figure 31. Canard Installation.

í,

1 . It can also be seen that in all actuation applications, exc pt that for the canard, the actuation system was built around hingeline act ators. In the case of the canard a ball screw was used as the final output los device. Studies such as those for the B-1 rudder have consistently shown that, where the input is derived from a rotary power device (hydraulic or electrical motors) and the output is to a surface having a long slender hingeline, the best power transducer is a hingeline compound planetary gear type drive (power hinge). This device provides the greatest hinge moment capabilities and highest stiffness at minimum weight.

Two general approaches are shown in Figures 25 through 31. One uses discrete power hinges and electric motors for each power input channel. The other combines the output of the three discrete motors in a single adapter gearbox powering a single longer power hinge. The first approach, called the 'hingeline installation" in the illustrations, had the highest potential reliability because a "disconnect" type failure downstream of the clutch (i.e. in the gearhead or in the power hinge) would not cause loss of control of the surface. The second approach, called the "PDU installation" in the illustrations, would fail destructively in the presence of a disconnect. In spite of this fact the "PDU installation" approach was preferred. This arose from the fact that, for equivalent reliability, the "PDU" approach could be smaller, lighter, and more adaptable for installation. As an example, considering the inboard flap, each power hinge and gearhead when used in the "hingeline" approach must be capable of 70% of rated output load (Reference Figure 2-5). This meant that the three power hinges and gearheads attached to each surface must have a total capacity, considered as a unit, of 210%. In other words this approach, though safe, was larger and heavier than it needed to be by a factor approaching 2.1. In contrast the single power hinge used in the "PDU" approach could theoretically be sized to 100% capacity. In actuality, by sizing it at 150%, the reliability of the unit would closely approach 1.0 and would equal or exceed that of the "hingeline" approach. On this basis the relative weights of power hinges and gearheads for the two approaches would be in the ratio of 1.5/2.1 or the "PDU" approach would weigh 28% less than the "hingeline" approach.

The projected weights of the various installations are shown in the lower left hand corner of the illustrations in Figures 25 through 31. In each instance (except for the installation in Figure 31) the weight quoted was that for the "hingeline" installation. The weight for the "PDU" installation was approximately 20% less, and where applicable, was quoted as the second entry on the Figure. The motor powers listed in Table 20 were consistently higher than the power per surface requirements given in Table 8. This arose from the fact, illustrated in Figure 32, that the motor was current (torque) limited. A motor which would meet the stall torque ( $\theta$ =0) without overheating had excess power at design load/rate conditions.

One of the most important conclusions drawn from the various installation illustrations shown in Figures 25 through 31 was the fact the "PDU" approach would fit within the installation envelope for the outboard trailing edge (aileron). It had been felt that, because of its very shallow chord, it would be impossible to install an actuation system in this area without using chordwise blisters. If this had been necessary it would have imposed a significant drag/weight penalty.

The design data given in Tables 18, 19, and 20 and in Figure 25 through 31 was developed using the Airesearch T1-59 "RAATS" program. A typical example of the analytical procedures used and the computer printouts developed are shown in Appendix A. The particular example used in Appendix A represented the first cut at sizing the inboard flap actuation system. In this case each of the three motors, gearheads, and hingeline drives making up the complete actuation system were being sized as 100% units (i.e. any one of the threepower trains attached to the surface could meet 100% of the stall hinge moment requirement of the surface) rather than as 70% units, which was the value later established as the basic requirement for this surface (See Table 11).

4.1.5.1.1 Actuation System Detail Design - In order to lend credibility to the weight, performance and envelope projections made in Tables 18, 19 and • 20 and Figures 25 through 31, it was decided that a detail design of at least one of the actuation functions should be made. It was felt that, if the resulting design matched the projected weight and envelope of a critical function within a reasonable degree (+ 5%), it could be expected that the other projections were probably equally accurate. The inboard trailing edge surface (inboard flap) actuation system was selected as the function around which the design would be based. This function was selected because it represented the most severe combination of design requirements, exhibited by any of the flight control actuation functions, in terms of power, load, frequency response and failure mode. The Airesearch report covering this design is included in this report as Appendix b.

As can be seen in the report (page 2, 5, and 6 of Appendix D) the design met envelope and performance requirements. However, in the report a question was raised (comments and recommendations page 5 of Appendix D) regarding the need to design to the force summed stalled torque of three motors. Although it was probable that the design torque for the hingeline gearbox could be reduced to a value, closely approaching the 216 x  $10^3$  in-1b value mentioned in the discussion, it was not done for the following reasons:



Figure 32. Inboard Flap Motor Characteristics.

- 1. Since there was no redundancy downstream of the motors (i.e. in the PDU, the reduction gearbox, the two torque tubes or the hingeline gearbox) the downstream items needed a reliability very closely approaching one (at least .9999). To achieve this the design needed to be considerably over designed by normal standards. The over design represented by  $453 \times 10^3$  in-1b capability (210% of required stall torque) versus an actual maximum applied load of 216 x  $10^3$  in-1b (100%) appeared reasonable.
- 2. Even though, during normal operation, the inboard trailing edge surface would never hit a stop (the electronic controls would determine travel limits) there was always the possibility that, during maintenance operations or during some special in flight failure mode, the surface would inadvertently bottom out. To avoid costly damage the unit should be designed to withstand this condition.
- 3. Even though it was extra heavy, the gcar train and all items down stream of the motor would be identical for Aircraft I and Aircraft II. Therefore, the extra weight did not represent a delta weight for comparison purposes in the study. (Refer to paragraphs 4.2.1.6 and 2.3.5).

The detail design of the inboard flap actuator is shown in Airesearch drawings 2022794, 796, 798, and 824 included as part of Appendix D. Drawing 2022824 represents the overall actuator system and shows that the total weight was 100 lbs. Of this 45 lbs represented the weight of the power drive unit (PDU) shown in drawing 2022798, the electric motors constituted a major share of the PDU weight at 36 lbs (12 lbs each). Page 2 of Appendix D lists the total weight of the inboard trailing edge (flap) actuation system as 220 lbs per surface. Since there were two such surfaces, the total system weight is 440 lbs of which 300 lbs are components subject to replacement by equivalent hydraulic components during the trade study. The weight of these replaceable components was arrived at as follows:

6	inverte	rs	at	38 lbs each	=	228	lbs
6	motors	at	12	lbs each	=	72	lbs
						300	lbs

The electric motors were rated at 45 HP. This was considerably above the 21.38 HP(15.95 KW) shown in the table of basic flight control actuator requirements Table 8. The reason for this is shown in Figure 32. The figure shows that the motor power capability was determined by the stall torque requirement. To prevent overheating, electric motors of this type are current limited. The current limit required for this application is shown on Figure 32 as 140 AMPS. At this current the motor would generate 142.88 in-1b of torque which, when appearing at the control surface after going through the transmission elements.

was equivalent to 150,886 in-1b. Each of the three motors used for the inboard flap actuation was capable of handling 70% of the required stall torque and hence was commonly thought of as a 70% actuator.

Figure 32 shows that there was a 2000 RPM speed drop in going from no load to full load. At this speed (20,000 RPM), and operating at the current limit, the motor delivered 45 HP. However, the design point load capability requirement was only 18.7 HP at 50° rated speed. This translated to 14.96 HP at the control surface, due to the inefficiencies of the intervening transmission elements, and represented 70° of the 21.38 HP (15.95 KW) required in the basic requirements table (Table 8). It is interesting to note that the motor would very nearly equal the 100° output requirement (19.94 HP versus the required 21.38 HP) at the design point. Thus the motor, although rated as a 70° motor based on stall capability, was actually a 100° motor at operating speeds slightly above the design point. This meant that, even in the face of two failures, the output performance of the actuation system would be essentially unimpaired in most of the practical areas of the flight prefile.

4.1.5.1.1.1 Power Drive Unit Design - The power drive unit (PDU) is shown on drawing 2022798 in Appendix D. The drawing showed that the unit was powered by three 270 VDC permanent magnet brushless motors mounted on a torque summing spur gear type gearbox. The motors drove the gearbox through a dog clutch which could be disconnected (but not reconnected) while the motor was rotating and while it was transmitting torque. The dog clutch was normally maintained engaged by a spring and could only be disengaged by energizing a declutching coil (shown on the drawing). Although not shown on the drawing, provisions were made so that the clutch, once disengaged, could only be re-engaged manually with the whole PDU inoperative. This eliminated the possibility that the failure of the coil from overheating while holding the dog clutch disengaged, or from a failure due to wire breakage or other electrical interruption, could allow the dog clutch to attempt to reengage when in motion. This could lead to failure of all three channels in the PDU. The drawing also showed a rotor position sensor, used for commutation, at the anti-drive end of the motor. The maximum output speed of the PDU was 22,000 RPM which was the same as the motor unloaded speed.

4.1.5.1.1.2 <u>Gearbox Design</u> - The gearbox was shown in drawing 2022796 Appendix D. This unit performed the combined function of a right angle gearbox and a speed reducer coaxial with the control hingeline. As shown the speed reduction ratio of the unit was 88:1 via a compound planetary gear train at 91.3% efficiency. By having the first significant gear reduction in this unit, rather than in the PDU, the relatively long shaft between the PDU and the gearbox was small, high speed, low torque, and lightweight. The relatively large, high torque output shaft was short, hence, its weight impact too was minimal. 4.1.5.1.1.3 <u>Power Hinge Design</u> - The power hinge design is shown on drawing 2022794 of Appendix D. It was a classic power hinge design of a type with which Airesearch has had considerable experience. The only unique feature was the relatively large number of "slices" used. A "slice" consisted of all the elements of a power hinge (i.e. two stationary ring gears and mounting lugs, one moving ring gear and its mounting lug, a set of planets with two radial loading rings and a sun gear) and this design used 14 of them. The gear reduction ratio was a relatively modest 15:1 at 88.4% efficiency. The drawing also showed the very high stiffness of the power hinge which was 32 times the stiffness requirement established in Table 8.

4.1.5.1.2 <u>Power-By-Wire/Fly-By-Wire Control System Definition</u> - The basic power control system was defined by Aireseach as a part of their subcontract effort on this program and is included in this report as Appendix B. The discussion in Appendix B showed that the two basic elements of the power control system, for modulated actuators, were the inverter and the controller.

4.1.5.1.2.1 <u>Inverter Description</u> - The functioning of the inverter is described in Appendix B. Essentially the inverter chops and pulse width modulates the 270 VDC power supplied by the electrical power system to cause the actuator's permanent magnet motors to operate bi-directionally at infinitely variable speeds in response to command signals received from the controller. The detail functioning of the inverter is shown on pages 2 through 8 in Appendix B.

4.1.5.1.2.2 <u>Controller Description</u> - The functioning of the controller is also described in Appendix B. Essentially the 'ontroller's function was to monitor feedback from the electro-mechanical actuator's output and, using this information, modify and reprocess the flight control system's input signals so that the resulting signals could be used to properly control inverter power to achieve the desired actuator output. The detail functioning of the controller is discussed and illustrated on pages 1 through 3 in Appendix B.

4.1.5.1.2.3 <u>Inverter Design</u> - The detail design of the inverter is outlined in Appendix C. It can be seen in this appendix that the inverters were supplied in three basic sizes to cover the actuator load requirements assigned to Airesearch for their study. Actually a fourth and much larger size was subsequently found to be necessary to meet the requirements of the plug throat. Although the size and weight of the total complement of inverters used in the aircraft could have been reduced by tailoring each inverter to its load application, or at least by increasing the number of sizes, it was felt that 4 sizes represented the optimum compromise between volume/weight versus logistics/ maintainability in terms of life cycle costs. The four inverter sizes and some of their critical characteristics are shown in Table 21. This table repeats the data already given in Tables T-1 and T-2 of Appendix C and uses the data for the 3 original inverter types shown therein as a basis for extrapolating the fourth inverter type (i.e. the plug throat inverter).
TABLE 21. INVERTER CHARACTERISTICS

		APPLICAT	ION	
CHARACTERISTIC	PLUG THROAT	INBCARD FLAP	EXTERNAL FLAP MIDSPAN FLAP	THRIST VICTOR ZANE AILERON CANARD LAIDING SOGE FLAP UF PERCEWER RUDDER
CURRENT RATING	350 AMP	150 AMP	50 AMP	25 AMP
POWER RATING	94.5 KW	40.5 KW	13.5 KW	6.75 KW
DIMENSIONS DIAMETER WIDTH DEPTH	9.0 M	7.2 (N	5.0 IN	7.0 IN 3.0 IN
LENGTH	23.0 M	10.7 IN	10.4 IN	11.01N
VOLUME	1500.0IN ³	761.4 IN3	204.2 IN ³	231.0 IN ³
WEIGHT	67.5 LB	38.0 LB	12.5LB	ÍO.O LB
COOLING	EC	EC	EC	NC

* NC = NATURAL RADIATION / CONVECTION COOLING EC = EVAPORATIVE COOLING - FINNED OUTER SURFACE WITH FORCED CONVECTION, 130°F, 30 CFM / FT OF LENGTH, 2 IN H20 AP, OR IMMERSION IN FLUID HEAT SINK

4.1.5.1.2.4 <u>Inverter Cooling</u> - The 54 inverters used in aircraft I will, as a group, typically reject 4.03 kW in the form of heat during longest high output sustained duty cycle (terrain following - 32 Min). Heat rejection of this magnitude was felt to be a potentially serious problem especially when considering the fact that a large proportion of the heat would be rejected from very small components (i.e. the field effects transistors "FETs" used in the inverters). For this reason both Rockwell and Airesearch studied the problem and came to generally the same conclusions. These conclusions were that inverters rated at less than 25 amps could be cooled by natural conduction and convection and that inverters rated at 50 amps and above must be evaporatively cooled. Airesearch's analysis of the subject is contained in Appendices D-1, D-2 and D-3 to Appendix C. Rockwell's analysis is discussed in the following paragraphs.

Ideally the electro-mechanical actuator should reject its internally generated heat (i.e. that resulting from motor or control system inefficiencies) to its immediate surroundings. By so doing, a system using electro-mechanical actuators could avoid the need for auxiliary cooling ducting or numerous liquid cooling lines spreading out through the aircraft to service each actuator. If such a spiderweb of lines and/or ducts were to prove necessary, it was felt it would offset a large portion of the advantages derived from deleting the hydraulic system. Tests conducted and reported in reference 17 show quite conclusively that the electric motor/power hinge portion of an electro-mechanical actuation system can reject it's self generated heat to its immediate surroundings. However the analysis made in Appendix C showed that auxiliary cooling aids were required where large sized inverters were a part of the actuation system.

Table 22 shows the heat rejection characteristics of the inverters for all the various flight control (continuous duty type) actuators and two engine actuation functions. These two engine actuation functions were included because, even though classified as utility functions, they were modulated and had continuous duty characteristics. The other utility actuators were not included in the table because, in general, heat rejection was not a problem for this type of actuator. In most instances this was because they are not continuous duty, and therefore their operations were infrequent, and their operating times were short. Continuous duty elements in the environmental control system, such as pumps, and blowers, were also not included because they did not employ inverters and because they were an integral part of, and could reject such heat as they did generate to the ECS system. This, plus the fact pointed out earlier, that the ECS system for all study aircraft would be essentially the same, justified their elimination. Table 22 is an expansion of the data given in the table on page D-1-1 of Appendix C and the new additions were an extrapolation from the data on which that table is based,

TABLE 22. INVERTER LOSSES FOR 25% DUITY CYCLE*

. . . . .

	MOTOR.	INVE	RER	25%	LOSSES PER	TOTAL NO. OF	TOTAL
ACTUATOR	CURRENT (AMPS)	ж ж З4үт	RATING (AMPS)	CURRENT (Arips)	INVERTER (WATTS)	INVERTERS	LOSSES (WATTS)
INBOARD FLAP	156	EC	150	390	316 9	9	1061
MIDSPAN FLAP	31	EC	50	7.8	56.9	\$	341
AILERON	14	NC	25	3.5	6.6	4	26
UPPER RUDDER	3.5	NC	25	6.0	1.6	9	2
LOWER RUDDER	6.9	NC	25	1:7	3.0	9	18
L. E. FLAP	17	NC	25	4.3	8.7	12	104
CANARD	3.5	NC	25	0.9	1.5	4	و
THRUST VECTOR VANE	10.2	NC	25	2.6	4.7	4	61
EXTERNAL FLAP	59	EC	50	14.8	113.5	4	454
PLUG THROAT	364.6	EC	350	91.1	0.107	2	1402
* DATA FOR 50% 51	DEEN				W	L055E5	4281
** COOLING METHOD		EVAPOR	L (COND	COOLING/CONTINE/CO	ONVECTIVE) (	DOLING	

Of the actuation functions listed in Table 22 only the first and last two rejected significant amounts of heat. For the other types of actuators the heat rejection was low enough so that all the heat generated could be rejected to ambient air directly or through the actuator's mounting pads and thence via structure to ambient air.

The high heat rejection functions involved the 18 actuators used for powering the inboard and midspan flaps on the left and right side and the two engine functions (plug throat and external flap). These actuators required a more exotic cooling method. The method which appeared best was some form of evaporative cooling based on the nucleate boiling of a dense, inert, low viscosity fluid.

Figure 33 illustrates the mechanism of nucleate boiling. It shows that accomplishing component cooling in phases 1, 2 and 3 (Figure 33) was the most effective and indicated that the component surface temperature should never be more than the peak value, shown at "a", above the coolant's temperature. The Figure 33 data showed, in both the main figure and in block 3 in the pictorial illustrations of boiling in the upper part of the figure, that the peak  $\Delta T$  was 65°F for water. The equivalent value for a typical inert cooling fluid, such as freon 113, was approximately 35°F. The difference resulted largely from the cooling fluid's specific heat. Evaporative cooling for electronic components involved a regenerative cycle which consisted of nucleate boiling followed by vapor condensation in a closed system. Figure 34 illustrates this general approach. It shows three circuit boards, mounting high output electronic devices, installed in a sealed housing and immersed in a coolant fluid. The electronic devices were rejecting sufficient heat so that the fluid was boiling at a relatively high rate (phase 3 in Figure 33) which was sufficient to cause the vapor bubbles to rise through the liquid and escape to the vapor zone. In the meantime cooling airflow, or a heatsink fluid, was circulated over the finned outer surface of the housing and cooling it sufficiently to cause the vapor to recondense. When cooling airflow was used it was induced by convection when the aircraft was on the ground and by ram effects or forced cooling air in flight.

A reasonably well designed forced convection air cooling system would remove 0.05 watts of rejected heat for each square inch of cooling surface per degree centigrade differential temperature (0.05 watts/in²/°C). A comparable figure for an evaporative cooling fluid was 1.5 watts/in²/°C). In other words cooling fluids were 30 times as effective at removing heat from a surface as was air. It was, therefore, reasonable to assume that, when air cooled, the area of the outer finned surfaces of a unit, such as Figure 34, should be at least 30 times the effective heat transfer area of the heat generating electronic components themselves. This assumed that  $\Delta T$  between the electronic component and the fluid approximately equaled the  $\Delta T$  between the vapor and the cooling air.



Typical boiling data for a wire heated electrically in a pool of water at atmospheric pressure. (Extracted from "Heat Transfer to Water Boiling Under Pressure," by E. A. Farber and R. L. Scorah, published in Trans. ASME, Vol. 79, 1948, with permission of the publishers. The American Society of Mechanical Engineers)

- 1 Free convection heating (no boiling).
- 2 Nucleate boiling Bubbles condense in liquid.
- 3 Nucleate boiling Bubbles rise through liquid and escape into vapor zone.
- 4 Partial film boiling Bubbles are formed so fast on the heating surface that part of the heating surface is covered with a vapor film. This wapor film insulates the heating surface, decreasing the heat flux.
- 5 Film boiling The heating surface is completely covered with a vapor film.
- 6 Radiation Radiation heat transfer dominates the film boiling.

Figure 33. Heat Transfer with Change in Phase.



TO STRUCTURE



TABLE 23. EVAPORATIVE COMUNCIFICIED PROPERTIES

	CAN	IDIDATE FLUI	0
CHARACTERISTIC	FREDN 113 ()	FC 78 ()	FC 15 @
SPECIFIC HEAT @ 73.F (Cp) (BTU/LB/F)	.214	.240	.245
HEAT DE VARORIZATION @ 13 F (hf3) (BIU/LB)	99	41	36
SPECIFIC GRAVITY @ 73 F (G)	1.57	1.70	1.76
DENSITY @ 73°F (V) (L8/FT3)	96	901	011
VISCOSITY @ 73 F(4) (CENTIPOISE)	.64	.40	18.
DUILING POINT @ I ATM (°F)	130	122.	216
VAPOR PRESSURE @ 73.F (B) (mmHg)	310	260	35
THERMAL CONDUCTIVITY (BUU/HR/FT2/+F1)	.043	.036	.037
COEFFICIENT OF EXPANSION (FT)/FT3 °F)	NO DATA	6000.	6000.
POUR POINT (°F)	- 31	- 135	-135
MATERIAL COMPATIBILITY	GOOD	EXCELLENT	EXCELLENT
0 DU PONT DE NEMOURS & CO. 0 MINNESOTA MINING & MFG. CO.			

.....

.

Table 23 lists the basic characteristics of three potential evaporative cooling fluids. Of the three, Freon 113 appeared to offer the best balance of properties. It had the highest heat of vaporization (i.e. about 40% greater than that of the other two fluids, but still only 7% of that of water), and the highest thermal conductivity. It was also the lowest in weight (i.e. roughly 10% less than the other two fluids but still nearly 60% heavier than water). The only areas where Freon 113 appeared to be deficient, relative to the other two fluids, was in pour point and material compatibility. It was desirable, but not mandatory, that the fluid have a pour point below the minimum operating temperature (-65°F). However, in a passive system, such as that shown in Figure 34 where fluid circulation is not required at low temperatures, the only adverse impact of a high pour point is the possibility, that at lower operating temperatures, the fluid will freeze in a damaging manner deforming encapsulated components. It is not believed that, at -65°F, Freon 113 would freeze solid enough to damage components.

Freon 113 was also somewhat deficient, with respect to the other two fluids, as regards its long term inertness relative to some materials of construction (specifically silicone compounds) commonly used in electronic hardware. Freon contained both flourine and chlorine in its molecular structure. Chlorine would tend to attack some silicone compounds as well as some highly stressed metals under certain conditions. The "FC" fluids contained only flourine as the halogen in its molecular makeup. For this reason the "FC" fluids were almost perfectly inert to all materials of construction.

The transistors, used in all continuous duty cycle (flight control type) inverters, had a maximum continuous junction temperature limit of  $125^{\circ}C$  ( $256^{\circ}F$ ) and a short time maximum junction temperature limit of  $150^{\circ}C$  ( $302^{\circ}F$ ). Cooling these transistors would be no problem except under mach 2.2 flight conditions where the cooling air temperature (ram ambient air) was  $117^{\circ}C$  ( $242^{\circ}F$ ). Since this condition could exist for as long as 22 minutes continuously on a single flight (see Figure 1 and Table 3), thermal lag could not be counted on. The maximum continuous transistor junction temperature limit was only  $8^{\circ}C$  ( $14^{\circ}F$ ) above the cooling air temperature. This was an impractically small differential temperature for achieving any significant heat transfer.

There were three possible solutions to this problem.

- 1. Use higher boiling temperature fluid, such as FC-75 or FC-43, in an evacuated sealed housing.
- 2. Provide cooling air from the ECS system to the inverters during high speed flight.
- 3. Immerse the inverters in fuel.

The use of a higher boiling temperature fluid in an evacuated housing would make it possible to maintain relatively low boiling temperatures and hence, relatively low junction temperatures when the available ambient cooling air temperature, at a relatively constant  $\Delta T$ , up to the maximum temperature encountered. This is illustrated in Table 24 which shows that, with cruise at 32,000 ft, the ram ambient cooling air was -6°F and the corresponding transistor junction temperature was a chilly +39°F. At the other end of the spectrum the available cooling air reached a maximum of 242°F during mach 2.2 cruise and the corresponding junction temperature became 298°F. This was 4°F under the short time maximum limit of 150°C (302°F). Although the junction temperatures shown in Table 24 were not necessarily accurate they were in the right ballpark and were probably accurate within  $\pm 2\%$  for a typical transistor outer housing surface.

This approach (using a sealed housing and allowing the junction temperature to follow the cooling air temperature) largely avoided the problems associated with using a fluid in a constant pressure (and hence constant boiling temperature) housing. If a low temperature boiling fluid had been used, in the constant pressure approach the fluid would have tended to turn completely into a large and unmanageable volume of gas during the high temperature (mach 2.2) portion of this mission.

In affect the cooling would thus have been occurring in the stable film boiling or radiation cooling range shown as zones 5 and 6 in Figure 33. Under these conditions the junction would have been in the  $1000^{\circ}F^+$  range and would have immediately failed. If a high temperature boiling fluid ( $262^{\circ}F$  at 1 ATM) were used, one which would still be in the nucleate boiling range (zone 3 Figure 33) at the high cooling air temperatures ( $242^{\circ}F$ ) associated with mach 2.2 flight, the transistor junction temperatures would have tended to be around or above the allowable maximum continuous junction temperature ( $256^{\circ}F$ ) for a greater portion of the transistor's operational service life. Since transistor life was an inverse exponental function of junction temperature this could have had a serious adverse impact on life and reliability.

As an example of conditions existing under other circumstances the junction temperature of a constant pressure boiling system would have been about 272°F during low level terrain following whereas that for a sealed variable pressure approach under the same circumstances would have been as indicated in Table 24. (i.e. 198° F). On an average the transistor junction temperature for the sealed variable pressure approach would have been at least 80°F less than that for the constant pressure boiling system under the operating conditions and flight times logged by the ATS aircraft.

Based on the preceding discussion, the solution offered by item 1 above might have been marginally satisfactory, particularly if a 10 to 20 degree further

TABLE 24. EVAPORATIVE COOLING PRESS/TEMP CONDITIONS

100 Sec. 31

	FLIGHT S	SEGMENT OR	OPERATION	
PARAMETER 3	CRUISE	TOM LEVEL	SUPERSONIC	FILLING
	32,000F1: 0.9M	TERRAIN FOLLOWING	CRUI9E 2.2 M	HOUSING DURING FAB
JUNCTION TEMP.	+39°F	198°F	296°F	241°F0
BOILING TEMP.	+ 	163°F	262°F	206°F
BOILING PRESS.	.06 PSiA	6 PSIA	38.7 PSIA	AI297.41
CONDENSATION TEMP.	+ 4 °F	153°F	252°F	0 7.961
COOLING AIR TEMP.	- 6 • F	143°F	242°F	186°F ()

Ζ,

() A POINT CONDITION PASSED THRU ON THE WAY TO AND FROM SUPERSONIC CRUISE .

(2) ASSUMES USE OF MINNESOTA MINING \$ MFG. CO. COOLING FLUID FC-75.

***

increase in allowable transistor junction temperature could have been assumed for the 1990+ time frame. However, even if this approach had been usable, the weight represented by the ducting and equipment necessary to direct the proper amount of cooling air to all 18 of the inverters requiring ram air cooling plus the ram drag rise associated with extracting air from the airstream indicated that it was not one of the better choices.

As indicated in item 2 above, the second approach to cooling the actuators would have been to provide ECS cooling. In this approach the same basic evaporative cooling techniques would still be used. However, in this instance, a low temperature boiling fluid at constant pressure would have been the cooling medium. Evaporative cooling would have been used in preference to liquid cold plate cooling or direct air cooling because evaporative cooling was so much more effective at extracting the heat from the hot spot and transporting it for dissipation over a large surface area. The large surface area would then have been cooled by the ECS system either by air or liquid (coolanol). However, this approach was subject to the same general objections as those cited for item 1 above. This approach would still require either, complex ducting if air cooled, or complex piping if coolanol cooled.

In either event the ultimate heat sink would have been the aircraft's fuel (see Figure 10). This fact lead to the conclusion that the solution offered by item 3 above was the best approach.

Item 3 envisioned immersing the inverters in the fuel tank. This appeared to be a reasonable approach based on the following:

- 1. The inverters should be close to the generators supplying them (3 to 6 ft).
- 2. The generators were surrounded by sump tanks (see tanks #3 and #4 in Figure 22).
- 3. The inverters could be installed in the sump tanks and be within 6 ft of the generators.
- 4. Rejection of heat to fluid was approximately 30 times as effective as rejecting to air, therefore, by using fuel as a heatsink, the inverter housing could be of minimum size and weight.
- 5. The aircraft was equipped with an air to fuel heat exchanger in the fuel recirculation loop for alert status ground cooling. This heat exchanger, and its ground cooling fuel heat sink door (see Table 9), could be used for subsonic inflight cooling of the fuel before and after the M 2.2 portions of the mission to ensure a low sump tank fuel temperature ( $<70^{\circ}$ F).

- 6. The limit temperature for the fuel was 150°F and the sump tank fuel capacities were 2500 lbs. Under these conditions the sump tanks had a heat absorption capacity of 100,000 BTU (2500 lb x 0.5 BTU/LB/°F for fuel x 80°F = 100,000 BTU), in going from 70°F to 150°F during M 2.2 operations.
- 7. The heat rejected to the fuel during the 22 minutes maximum of mach 2.2 operations was 5128 BTU, which was only 5.1% of the 100,000 BTU capacity, and indicated that, even with the other heat sink demands placed on the fuel, of which the 74,407 BTU placed on it by the ECS system (see Reference 12, page 3-20) is the major item, the heat sink capacity is adequate.

The heat rejection value used in item 7 above was derived from Table 22. This table listed the losses of all the inverters used in the aircraft when operated on a 25% duty cycle. This duty cycle was felt to be representative of the mean loads which would be encountered over the 22 minutes mach 2.2 operation encountered during penetration and combat (see load analysis Table 16, sheet 3). The table included the losses for both the air cooled (NC) inverters and the fuel cooled (EC) inverters. 4098 watts represented the losses for the fuel cooled inverters, out of a grand total of 4281 watts for all inverters, and was the value used to determine the 5128 BTU heat rejection figure used in item 7 above.

4.1.5.1.2.5 Power-By-Wire/Fly-By-Wire Control System Arrangement - Figure 35 represents the general arragnement used for power control in Aircraft I. The arrangement attempted to take maximum advantage of the four indepentdent electrical systems by combining them with triple redundant actuators and five channel flight control inputs (4 control channels plus a model channel) to obtain maximum reliability. The use of five channels in the "Fly-By-Wire" flight control inputs made possible voting in the face of a third failure. This practically eliminated the possibility of a "hard over" third failure and simplified the achievement of a "fail safe" condition after the third failure. The micro processors in the various systems (system #1 through #4) exchange data and voting information via optical interties. In this way absolute separation of the four power systems was maintained in that there were no electrical interconnections either for power transfer or signal interchange. Optical interties were also used exclusively between the various microprocessors and their respective inverters, motors, and actuators as well as for the fly-by-wire (fly-by-light) flight control inputs. Through the use of this approach the potential adverse impact of electro magnetic interference (EMI) generated in the inverters and elsewhere was minimized. The use of optical interties, in the manner indicated in Figure 35, also gave the power control system a high degree of resistance to electro magnetic pluse (EMP) effects such as would be associated with lightening strikes or nuclear blasts.

Figure 35 is an expansion and elaboration of the basic Airesearch block diagrams shown as Figure C-4, page 291, Figure C-5 page 293, and Table C-4 page 294, of Appendix C. In the event of failure of a given inverter or actuator motor in a given system failure was detected in the system's micro processor based upon data fed back from the defective inverter or motor. This information was then sent electro-optically to the appropriate neighboring microprocessor where a signal was generated to cause the appropriate solid state relay to energize and disconnect the motor clutch in the malfunctioning motor. As a specific example (see Figure 35), if the failure were in the inverter (INV) for the right hand (R.H.) inboard trailing edge (ITE) surface for system #1, the system #1 microprocessor would sense the failure via electro optical feed back from the inverter and send the failure intelligence to system #2 and system #4 microprocessors. These microprocessors would process the information and send a signal to their respective solid state relay banks which would cause the appropriate relay in each bank to energize its declutching coil in the motor for ITE INV R.H. #1. As can be seen in the power drive unit drawing (Airesearch drawings No. 2022798 -Appendix D) the dog clutch must be electrically energized to disengage. Although not clearly shown in drawing No. 2022798, the clutch actuating coil was actually a dual coil powered, in this case, by system nos 2 and 4. Either system by itself was capable of declutching the unit.

The system could have been designed so that loss of power could have caused it to declutch (i.e., spring loaded to declutch). However, this approach



Figure 35. Wing Trailing Edge Surface Power Control

was discarded because it would have meant that all the coils in all actuators would have to be energized at all times during normal operation. This appeared to be an unacceptable heat load and power drain. In contrast, considering the selected design, only those motors which have failed or are part of a system which has failed were energized. In the event of a dual system failure this might have meant that as many as 8 motors (out of 16 in the trailing edge flap system) might have had energized clutches. However, this would have occurred relatively infrequently and since, with the selected design, the coil was never energized while the motor was running the motor heat load was not additive to the coil heat load. In contrast, with the spring loaded to declutch design, the two heat loads would have been continuously additive and would have created a major cooling problem.

As implied above a complete system power failure (generator out) or microprocessor malfunction in system #1 would have caused system nos. 2 and 4 microprocessors to cause all clutches on all motors in system #1 to declutch. The reason for two declutching coils was to meet the two fail operate requirement. If system #1's failure had been preceeded by a failure in the #2 power system, power from system #4 would have been necessary to accomplish the required declutching.

It should be noted, when examining Figure 35, that an electro optical converter was not shown at the inverter end of the microprocessor/inverter signal transmission line. In the interest of avoiding further complexity in Figure 35 it was assumed that this electro optical converter along with its power supply was built into the inverter.

Figure 36 shows the declutching connections for the wing trailing edge actuation system. The reason for the unbalance in the number of connections (i.e. 11 in systems #1 and #4, and 5 in systems #2 and #3, versus an ideal of 8 in each system) was as a result of the unbalanced distribution of flight control actuation functions needed to adapt dual and triple channel actuators to a four channel power system. In effect, as can be seen by examining the electrical system load summation Table 17 the distribution of actuation functions between power systems succeeded in balancing power demand to  $40.22 \pm 9.22$  KW continuous and  $63.27 \pm 14.87$  KW 5 sec loads during combat.

Although the schematic of Figure 35 covered only the control of the power for the actuation systems on the wing trailing edge, it was representative of the power control approach which was used for all flight control actuation functions (i.e. all those listed in Table 8). Figure 35 was limited to this coverage to avoid excess complexity in the presentation and thus to avoid confusion which such complexity was likely to generate.

Figure 37 shows the general arrangement of the signal and power hockups between power systems, flight data computers, and the various microprocessors.



OTE = OUTBOARD TRAILING EDGE MSTE = MIDSPAN TRAILING EDGE ITE = INBOARD TRAILING EDGE Figure 36. Wing Trailing Edge System Actuator Declutching Connections



and the product of the street of the second second second

Figure 37. Signal and Power Hookups

The figure shows that there are two redundant flight data computers. The first one, powered by system No. 1, was located in the aft avionics compartment (see Figure 22) and the second one, powered by system No. 4 was located in the intermediate avionics compartment. This gave wide system separation between the two computers and greatly reduced the possibility that battle damage would incapacitate the two units simultaneously. It can be seen in Figure 37 that the microprocessors were divided into 3 geographical groups, nose, wing and tail, which indicates the general physical location of the microprocessors in the aircraft as well as the actuation functions which they serve. W #1 microprocessor and W #2 microprocessor in Figure 37 were the same as system #1 microprocessor and system #2 microprocessor W #5 microprocessor in Figure 37 was the respectively in Figure 35. "model" microprocessor indicated but not shown in Figure 35. The signal transmission interconnects shown in Figure 37 between the #1 and #4 flight data computers and the wing (W) grouping of microprocessors were the same interconnects as those indicated as "flight control input channels" on It can also be seen in Figure 37 that the "wing" and "tail" Figure 35. groups contained full five channel capability with five microprocessors each whereas the nose group contained only three microprocessors. The reason for this was the fact that the wing and tail microprocessor groups service all the flight critical (2 fail operate-fail safe) functions on the aircraft. These were also the functions which could not tolerate a hard over signal after the third failure nor could they tolerate being locked in any position other than trail after a third failure. To accomplish this it was necessary to have five signal channel capability so that a defective signal channel could be voted out of action as a result of its being the third failure.

In the case of the nose grouping, however, circumstances were considerably different. Here only 3 functions required the services of microprocessor controlled inverters. These were the gun, the nose gear steering and the canard. Of the three only the canard had greater than a fail safe requirement. The canard was single fail operate-fail safe, however, because the nature of the canard was such that it could and would, aerodynamically blow back and lock when disconnected. The disagreement betweeen channels which would exist after a second failure could be used to trigger a disconnect and, therefore, no more than 3 channels were required to meet fail safe requirements.

4.1.6 <u>Starter/Generator System Definition</u> - A more detailed definition of the starter/generator system, was prepared which expanded on that given in paragraphs 4.1.1 and 4.1.2. This improved definition was derived from the additional data generated in paragraphs 4.1.4 and 4.1.5. The general characteristics of the starter/generator system thus defined were as follows:

Generator Rating:	60/70 KW AC	/DC output	
AC Output:	115/200 V 4 per MIL-STD	00 Hz 3 pha -704A	ise power
DC Output:	270 VDC per	MIL-STD-70	)4A
Speed (Generating):	Range Minimum Maximum	= 2:1 = 13750 = 27500	) RPM ) RPM
(Starting):	Minimum Maximum	= 0 RPN = 10,00	1 )0 RPM
Cycloconverter Rating:	20 KW (25 K	WA at 0.8 H	PF)
Transformer-Rectifier-Filter Rat	ing:	47.5	KW
Cooling:	Conduction ( cooling-cool AMAD (i.e. s	[cold plate] . oil suppli shared syste	oil ied by em)
Weight:			
ITEM	ITEM CODE (Figure 4.2)	WEIGHT LBS	REF CODE
General Equipment			
Generator Cycloconverter Transformer-Rectifier-Filter Generator Control Unit	(G) (CCV) (TRF) (GCU)	59.2 30.7 17.1 8.0	- 1054
Generator System Total		115.0	
Starting System Equipment (For S	tarter/Generator	r Only)	
Drain and Fill Torque Conver 90 KW Reverse SCRs and Contro 15 KW Inverter Unit 90 KVA AC Start Contactor 15 KVA AC Start Contactor 15 KW DC Start Contactor 90 KW DC Start Controctor Starting System Equipment To	ter ol (RSCR) (SI) (SCA) (SCB) (SCC) (SCD) tal	6.1 21.5 19.8 6.5 0.7 8.0 9.0 71.6	6
Generating System Total Starter/Generator System Tota	al	<u>115.0</u> 186.6	

ł

1.1.1

The generator, cycloconverter, and transformer-rectifier-filter ratings listed above were derived from the electrical load analysis summation, listed in Table 4-3, using the following logic. From Table 4-3, assuming emergency conditions with two systems failed (2 channels operative) and considering the most highly loaded of the two remaining channels (i.e. channel 2), the maximum required outputs of the generating system were as follows: (Table 4-3 values rounded off)

AC Output	(continuous load)	20	ΚW	(25	KVA	at	0.8	PF)
DC Output	(continuous load)	35	KW					
AC Output	( 5 second load)	20	KW	(25	KVA	at	0.8	PF)
DC Output	( 5 second load)	95	KW					

From these values the following ratings of the cycloconverter and transformerrectifier-filter were derived.

Cycloconverter (based on continuous loads)	20 KW (25 KVA at 0.8 PF)
Transformer-rectifier-filter (based on 50% of 5 sec load)	47.5 KW

From the foregoing ratings the generator ratings were derived as follows:

Continuous load rating 20 + 47.5 = 67.5 KW rounded off for growth = 70 KW 5 second load rating

95 + 20 = 115 KW rounded off to 120 KW X 50% = 60 KW

Hence the generator was assigned a 60/70 KW rating.

The weights for the various starter/generator system elements were derived from various data source references as indicated by the "REF CODES" used in the right hand column of the weight tabulation above and listed below.

- (1) Average of the data from two sources i.e., reference 11 page 54 and reference 14 page 39.
- (2) Data from reference 14 reduced by 10% to account for advances in the 1990+ time period
- (3) Derived from a curve plotted from data from reference 11 page 38

(4) Data from reference 11 page 38 through 41

(5) Data from reference 11 page 39

To arrive at the generating system defined above, three generating system approaches were considered all of which had the following characteristics in common.

- 1. Each was rated at 60/70 KW output
- 2. Two of the four generating systems, one on each engine, had engine start capabilities.
- 3. Each was capable of delivering 20 KW (25 KVA at 0.8 PF) of continuous AC power and 47.5 KW of continuous DC power. The three generating system approaches considered were:
  - 1. Integrated starter generator (ISD) employing a constant speed drive.
  - 2. VSCF starter/generator system employing a cycloconverter for AC output.
  - 3. VSCF starter/generator system employing a DC link for AC output.

Although the starting system based upon the ISD type starter/generator was the lightest by approximately 58 lbs, it was dropped from consideration because its full load and cruise load efficiency was poorer than that of the other two approaches by approximately 14% (Reference 28 page 52) and because of its relatively poor reliability. The poor cruise efficiency effectively cancelled a large portion of its weight advantage (i.e. 47 lbs) as shown in the following analysis:

> The sum of the average powers delivered at the four busses is 80 KW during a typical mission. Based on this, the power extracted from the 2 jet engines combined is:

> > For ISD system  $\frac{80}{0.71 \text{EFF}}$  = 112.68 KW or 151.05 H.P.

For VSCF system  $\frac{80}{0.85 \text{ EFF}} = 94.12 \text{ KW or } 126.16 \text{ H.P.}$ 

Therefore the ISD system extracts 24.89 HP more than the VSCF system.

Assuming a specific fuel consumption of 0.7 lb fuel/BHP-HR for the jet engines and using the 2.7 HR maximum unrefueled mission time of the ATS aircraft (page 6), the total

extra fuel required for the aircraft using the ISD is:

24.89 HP X 0.7 1b/HP-HR X 2.7 HR = 47.04 1b

The ISD's relatively poor reliability has been recognized for years and has been the driving force behind the development of the VSCF approach. In view of it's small potential weight saving and it's poor record in maintainability and reliability, the ISD was not considered a viable candidate for the 1990+ time period and was dropped from further consideration.

There was very little to choose between the cycloconverter and DC link approaches to VSCF generator design. At the size (60/70 KW) and speed range 2:1) characteristic of this application it was projected that the cycloconverter would be 5 to 10% lighter than the DC link (reference 28) for the same power output. However, the DC link full load efficiency would be 5-9% greater than that of the cycloconverter and its temperature tolerance would be greater (120°C vs 80°C continuous input cooling oil temperature limit). Balanced against this the cycloconverter's part load effeciency was 1-2% better than the DC link (reference 18). Because cooling heat critical components would not be a problem in this aircraft (evaporative cooling) the heat tolerance advantage of the DC link was considered offset. Therefore, because the generator operated at part load most of the time and because weight was critical, the cycloconverter approach was selected.

4.1.7 <u>APU Driven Generator Sizing</u> - As shown in Figure 17 two generators, rated at 45KW each, are driven by the APU. This size selection was justified as part of the discussions in paragraph 4.1.2.3. Based on this power rating each generator including its generator control unit (GCU) plus its 45 KW transformer rectifier (TRF) and 7 KVA cycloconverter (CCV), weighed 83.9 lbs.

4.1.8 <u>APU Sizing</u> - The APU for aircraft II was rated at 485 HP sea level static and weighed 245 lbs including all peripherals such as a starter, fuel control, blade containment provisions, reduction gearbox with generator mounting pads, lube oil, oil tank, etc.. The APU was a free turbine unit with an annular inlet, three-stage axial and single-stage centrifugal compressor, annular combustion chamber, single-stage compressor-turbine and counter rotating power turbine and is similar in functional arrangement to the Hamilton Standard ST6L-73 APU.

The power rating determination was based on the following computation:

89.5 KW required at primary generator shaft (Paragraph 2.2.7)

 $\div 0.85$  Primary generator efficiency 105.29 KW required at primary generator terminals

÷ 0.97 Transmission efficiency (voltage drop)

108.55 KW required at APU generator output terminals for starting load (See comparative value at the end of paragraph 4.1.4)
+50.28 KW electrical system 5 sec essential loads from Table 16 sht 12
158.83 KW total required at APU generator output terminals during starting
÷ 0.85 APU generator efficiency
186.86 KW at APU generator shaft
÷ 0.94 APU adapter gearbox efficiency
198.79 KW required APU output at 20,000 ft
÷ 0.55 Sea level correction factor (reference 25 page 12)
361.43 KW sea level static rating
÷ 0.746 KW to HP conversion factor
484.49 HP (use 485 HP rating)

The APU had a 0.610 lb/HP-HR specific fuel consumption during typical starting and emergency return duty cycles and fitted in a rectangular compartment whose dimensions were 18 X 18 X 40 in.

### 4.1.9 AIRCRAFT I ELECTRICAL SYSTEM WEIGHT ANALYSIS-

The Aircraft I weight analysis considered all those elements of the electrical system which were unique to, or were otherwise impacted by, the change from the more conventional secondary power system arrangement used in Aircraft II (see paragraph 4.2) to the "ALL ELECTRIC" approach used in Aircraft I. In effect, therefore, Aircraft II became the "BASELINE" aircraft against which all other variants were measured. The major elements considered in arriving at Aircraft I's relative weight were as follows:

- 1. All the electrically powered actuation functions which were hydraulically powered in Aircraft II.
- 2. All the power distribution elements which service the actuation functions of item (1) above.
- 3. All the elements in the electrical power generation system.
- 4. All the components making up the auxiliary power and starting system.
- 5. The impacts on the fuel system resulting from changed heat rejection and fuel displaced by inverters.

Structural impacts were not considered since all actuators, for both aircraft, used the same tie off points and reacted the same loads. Minor impacts due to differences in actuator envelopes, actuator weights, bulkhead penetration points for electrical cables vs hydraulic lines, component weights and component envelopes were ignored as being so small as to be within the "Noise Level"." Component weight and envelopes were, of course, considered in terms of growth factors and fuel displaced respectively.

Environmental control system impacts were not considered since, as discussed in paragraph 2.1.6, the systems, and heat loads they must service, were essentially identical between Aircraft I and Aircraft II.

4.1.9.1 <u>Electrically Powered Actuation Functions</u> - Table 25 lists the actuation functions outlined in item 1 above and shows the weight chargeable to each of these functions. The weight of the various components (i.e. motors, ballscrews, inverters, relays, etc.) making up each actuation subsystem, as shown in Table 25 was derived from data included as figures 38 and 39 and as derived or extrapolated from Table 21. Table 25 shows that the total weight of the actuation subsystems is 1667.1 lb. The calculations used to determine the weight entries for the various ball screw actuator entries in Table 25 are presented as follows:

EXTERNAL FLAP

Stall Load	=	8600 lb	(1)
Stroke	=	10 <b>inc</b> h	
Motor Power	=	8.80 kw	(2)

A CTUMATOR INDUCT         MOTOR			TABLE 25		REAL ACH	and NOTIVI	NUTION WILL	SS DC	TOTAL	NIMBER	TOTAL
ACTUARDRIMENT TO OUT         MOTOR         NOTOR         NOTOR </td <td>_</td> <td>RATED P</td> <td>OWER (</td> <td>(KW)</td> <td>MOTOR</td> <td>IFM</td> <td>INVER I EN</td> <td></td> <td></td> <td></td> <td>MENCHT</td>	_	RATED P	OWER (	(KW)	MOTOR	IFM	INVER I EN				MENCHT
OUT         MOTOR         POWER         (MMES)         (LB)		CTUATOR	NPUT TO	MOTOR	TUPUT	WEIGHT	WEIGHT	WEIGHT	WE JOH	01 1/C/1	
TER CONTROLLED       3357       66.8       12.0       38.0       -       50.0       6       300.0         11.16       10.66       3357       66.8       12.0       38.0       -       14.8       6       300.0         2.27       3.72       6.71       13.8       2.3       1.0       10.0       -       11.0       4       44.0         0.085       1.42       2.52       2.5       0.3       10.0       -       10.3       6       61.8         0.39       0.66       1.50       2.4       2.4       10.0       -       12.4       6.1.8         0.33       0.65       1.50       2.4       2.4       10.0       -       12.4       6       74.4         0.33       20.64       1.01       2.1       1.4       10.0       -       12.4       6       74.4         0.43       0.55       1.01       2.1       1.4       10.0       12.6       12.1       11.7         AV CONTROLLED       7.4       10.0       -       12.6       12       11.7         0.43       0.55       2.191       3.74       0.16       2.1       1.4       11.7         <		DUT	MOTOR	POWER	(AMPS)	(11)	(BJ)	(18)		PER VC	757.70
11.16       16.60       33.57       66.8       12.0       38.0       -       50.0       6       30.0         2.27       3.72       6.71       13.8       2.3       1.0       10.0       -       11.0       4       44.0         0.05       1.42       2.52       5.3       1.0       10.0       -       10.3       6       1.44         0.05       1.42       2.52       2.53       1.0       10.0       -       10.3       6       61.8         0.39       0.66       1.50       2.4       2.4       10.0       -       12.4       6       74.4         0.370       7.8       2.6       10.0       -       12.6       12       151.2         0.41       0.66       1.50       2.4       2.6       10.0       -       12.4       6       74.4         0.370       2.6.5       3.73       60.86       12.4       2       11.7       11.7       11.7       11.7       11.7       11.7       11.7       11.7       11.7       11.7       11.7       11.7       11.7       11.7       11.7       11.7       11.7       11.7       11.7       11.7       11.7       11.7	Ē	2 CONTR	SOLLED							,	0000
11.16       10.00       1.01       1.00       1.00       1.00       4       44.0         0.05       1.42       2.52       5.3       1.0       10.0       10.3       6       61.6         0.05       1.42       2.52       5.3       1.0       10.0       12.4       6.16       44.0         0.39       0.65       1.50       2.4       2.4       10.0       12.6       12.1       151.2         0.75       2.10       3.70       7.8       2.6       10.0       12.6       151.2       151.2         0.75       2.10       3.70       7.8       2.6       10.0       12.6       151.2       151.2         0.75       2.10       3.70       7.8       2.6       1.01       2.1       44.0         AV CONTROLED       7.8       2.6       1.4       10.0       11.7       12       151.2         AV CONTROLED       3.74       6.77       13.65       2.3       3.1       6.75       11.7       11.7         AV CONTROLED       0.72       1.3.74       0.56       17.4       10.0       25.0       11.7       11.7       11.7       11.7       11.7       11.7       11.7	i		10,01	2267	68.8	12 0	38.0	1	50.0	و	300.0
2.27       3.72       6.71       13.8       6.3       10.0       -       11.0       4       44.0         0.041       0.66       1.50       2.4       2.4       2.4       10.0       -       12.4       6       13.4         0.39       0.65       1.50       2.4       2.4       2.4       10.0       -       12.4       6       174.4         0.75       2.10       3.70       7.8       2.6       0.00       -       12.6       12       151.2         0.75       2.10       7.8       2.6       10.0       -       12.6       12       151.2         0.73       2.101       2.1       1.4       10.0       -       12.6       12       151.2         Ar CONTROLED       2.1.91       3.74       6.77       13.65       2.3       3.1       2       11.7         Ar CONTROLED       2.1.91       3.74       6.77       13.85       2.3       1.1.6       1.1.7       11.7         Ar CONTROLED       3.74       6.77       13.85       2.3       1.1.8       1.1.7       2       36.4       149.2         Ar CONTROLED       3.74       19.6       1.2.8       1.1.6 <td></td> <td>11.16</td> <td>09.01</td> <td>, c.Cc</td> <td></td> <td><u>,</u></td> <td>1</td> <td>(</td> <td>14.8</td> <td>2</td> <td>8.86</td>		11.16	09.01	, c.Cc		<u>,</u>	1	(	14.8	2	8.86
0.05       1.42       2.52       5.3       1.0       10.0       -       10.3       6       61.8         0.41       0.66       0.75       2.5       0.3       10.0       -       12.4       51.2       151.2         0.39       0.65       1.50       2.4       2.4       2.4       10.0       -       12.6       12       151.2         0.43       0.56       1.01       7.8       2.6       10.0       -       12.6       12       151.2         25.52       33.73       60.88       12.4.9       21.8       67.5       -       89.3       2       178.6         AY CONTROLED       0.43       0.56       1.01       2.1       1.4       10.0       -       12.6       12       11.7         AY CONTROLED       3.74       6.77       13.85       2.3       -       0.8       3.1       2       6.2       38.4         AY CONTROLED       3.74       6.75       14.6       1.4       10.0       -       2.8       6.2       4.2       6.2       4.2       4.49       4.49       2       4.13       4.49.5       4.49       4.49       4.49       4.49       4.49       4.49		2.27	3.72	6.71	2.61	c.,				ন ব	44.0
0.41       0.66       0.75       2.5       0.3       10.0       12.4       6       74.4         0.39       0.65       1.50       2.4       2.4       10.0       12.6       12       151.2         0.75       2.10       3.70       7.8       2.4       10.0       12.6       12       151.2         25.52       33.73       60.68       1.50       2.1       1.4       10.0       11.7       1       11.7         AV CONTROLLED       0.43       0.55       1.01       2.1       1.4       10.0       11.7       1       11.7         AV CONTROLLED       3.74       6.77       13.85       2.3       7       0.8       3.1       2       6.2         AV CONTROLLED       3.74       6.77       13.85       2.3       7       11.7       11.7       11.7         AV CONTROLLED       3.74       6.77       13.85       2.3       7       14.92       11.7         AV CONTROLLED       3.74       6.77       13.85       2.3       17.7       2       36.4         AV CONTROLLED       3.74       6.77       13.85       2.3       1.77       2       36.4         AVIORS-DC		0.85	1.42	2.52	r, r,	0.	0.01	\ 	- <u> </u>		813
0.39       0.65       1.50       2.4       2.4       2.4       10.0       -       12.4       13.12       151.2         25.52       33.73       60.86       124.9       21.6       10.0       -       12.6       12       151.2         25.52       33.73       60.86       124.9       21.6       6.75       -       89.3       2       175.6         AV CONTROLLED       0.43       0.56       1.01       2.1       1.4       10.0       -       12.6       12       11.7         AV CONTROLLED       0.101       2.1       1.3.85       2.3       95.2       17.4       10.0       -       26.6       26.6         R       19.64       26.87       48.63       99.52       17.4       -       11.7       2       66.2         R       19.64       26.87       48.63       99.52       17.4       -       12.7       2       6.2         ANDRS-DC MOTOR - INVERTEK CONTROLLED       0.17       6.27       1.00       -       25.0       2       30.4       149.2         ANIE       0.72       1.00       1.00       15.0       10.0       -       25.0       2       50.0		0.41	0.68	0.75	2.5 2	0.3	10.0		0.0	۰ د	
0.75       2:10       3.70       7.8       2.6       10.0       -       12.6       12.6       12.1         25.52       33.73       60.88       12.1       21.8       67.5       -       89.3       2       178.6         AV CONTROLLED       0.43       0.56       1.01       2.1       1.4       10.0       -       11.7       1       11.7         AV CONTROLLED       0.43       0.56       1.01       2.1       1.4       10.0       -       11.7       1       11.7         AV CONTROLLED       0.43       0.52       17.4       -       0.8       3.1       2       6.2       36.4         R       19.64       26.87       48.63       99.52       17.4       -       1.6       19.2       2       6.2         ANTORS-DC MOTOR - INVERTEK CONTROLLED       0.1       0.0       1.66.7       1.00       25.0       25.0       2       36.4         ANE       0.72       1.00       15.0       10.0       -       25.0       2       36.4       149.2         ANE       0.72       1.00       15.0       10.0       -       25.0       2       55.0       2       50.0		0.39	0.65	1.50	2.4	2.4	0.01	١	12.4	2 9	t t
25.52       33.73       60.68       124.9       21.8       67.5       -       89.5       2       10.0         AV CONTROLLED       AV CONTROLLED       -       1.01       2.1       1.4       10.0       -       11.7       1       11.7         AV CONTROLLED       -       0.43       0.56       1.01       2.1       1.4       10.0       -       11.7       1       11.7         AV CONTROLLED       -       0.11       3.74       6.77       13.85       2.3       -       0.6       3.1       2       6.2       36.4         AN CONTROLLED       -       0.13       6.77       13.85       2.3       -       0.6       3.1       2       6.2       36.4         AN CONTROLLED       0.17       0.27       1.00       15.0       10.0       -       2       37.3       4       149.2         ANE       0.72       1.00       15.0       10.0       -       2       35.4       149.2         ANE       0.72       1.00       15.0       10.0       -       2       37.3       2       355.4         ANE       0.72       1.03       1.67.7       10.0       1.5	; ,	0.75	2.10	3.70	7.8	2.6	10.0	1	12.6	20	7.00
AY CONTROLLED       0.43       0.56       1.01       2.1       1.4       10.0       -       11.7       1       11.1         AY CONTROLLED       ES       1.91       3.74       6.77       13.65       2.3       -       0.6       3.1       2       6.2         R       19.64       26.87       46.63       99.52       17.4       -       1.6       19.2       2       36.4         ANTORS-DC MOTOR - INVERTEK CONTROLLED       0.17       0.27       1.00       1.00       15.0       10.0       -       25.0       2       36.4         ANTORS-DC MOTOR - INVERTEK CONTROLLED       0.17       0.27       1.00       15.0       10.0       -       25.0       2       36.4         ANTORS-DC MOTOR - RELAY CONTROLLED       24.8       12.5       -       37.3       4       149.2         ANTORS - AC MOTOR - RELAY CONTROLLED       24.6       167.7       10.0       -       25.0       2       355.4         ANE       0.72       1.13       2.05       4.19       167.7       10.0       177.7       2       355.4         ANTORS - AC MOTOR - RELAY CONTROLLED       0.03       1.00       1.5       0.1       0.6       35.0		25.52	33.73	60.88	124.9	21.8	67.5	١	89.3	· ·	9. r 
AY CONTROLLED       AY CONTROLLED       0.8       3.1       2       6.2         R       19.64       26.87       46.63       99.52       17.4       -       1.6       19.2       2       36.4         IATORS-DC MOTOR - INVERTEK CONTROLLED       19.64       26.87       48.63       99.52       17.4       -       1.6       19.2       2       56.5         IATORS-DC MOTOR - INVERTEK CONTROLLED       0.17       0.27       1.00       15.0       10.0       -       25.0       2       36.4         ALI3       4.13       4.66       0.00       1.00       15.0       10.0       -       25.0       2       35.3       4       149.2         ANE       0.72       1.13       2.05       4.19       167.7       10.0       -       25.0       2       355.4         ANE       0.72       1.13       2.05       4.19       167.7       10.0       -       25.0       2       355.4         ANE       0.72       1.13       2.05       4.104       6.9       -       0.5       9.4       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6 <td></td> <td>0.43</td> <td>0.56</td> <td>1.01</td> <td>2.1</td> <td>4. –</td> <td>10.0</td> <td>1</td> <td>1.1</td> <td>-  -  -+</td> <td></td>		0.43	0.56	1.01	2.1	4. –	10.0	1	1.1	-  -  -+	
TS       1.91       3.74       6.77       13.85       2.3       -       0.6       3.1       2       6.2         R       19.64       26.87       48.63       99.52       17.4       -       1.8       19.2       2       38.4         IATORS-DC MOTOR - INVERTEK CONTROLLED       0.17       0.27       1.00       15.0       10.0       -       25.0       2       50.0         ANE       0.72       1.00       10.00       15.0       10.0       -       25.0       2       37.3       4       149.2         ANE       0.72       1.13       2.05       4.19       167.7       10.0       -       25.0       2       35.4         ANE       0.72       1.13       2.05       4.19       167.7       10.0       -       25.0       2       35.4         ATOR5-AC MOTOR- RELAY CONTROLLED       74.1       6.77       10.0       -       27.7       2       35.4       1.667.7       2       35.4         OP       0.01       1.05       1.67.7       10.0       -       25.0       2       35.5.4       1.667.7       2       35.5.4         OP       0.03       1.09       1.97		CONTROL	LED								•
R         19.64         26.87         46.63         99.52         17.4         —         1.6         19.2         2         30.4           JATORS-DC MOTOR - INVERTEK CONTROLLED         0.17         0.27         1.00         15.0         10.0         25.0         2         50.0           ANE         0.17         0.27         1.00         15.0         10.0         -         25.0         2         50.0           ANE         0.17         0.27         1.00         18.00         24.8         12.5         -         37.3         4         149.2           ANE         0.72         1.13         2.05         4.19         167.7         10.0         -         177.7         2         355.4           ATORS - AC MOTOR - RELAY CONTROLLED         2.05         4.19         167.7         10.0         -         177.7         2         355.4           OP         0.33         1.09         1.97         4.04         6.9         -         0.4         9.4         1         9.4           OP         0.93         1.09         1.97         4.04         6.9         -         0.1         1.6         1         1.6         1.6         1.6         1.6		191	3 74	6.77	13.85	2.3	1	0.8	 	~	- 0 0 0
MTORS-DC MOTOR - INVERTEK CONTROLLED       15.0       10.0       25.0       2       50.0         MATORS-DC MOTOR - INVERTEK CONTROLLED       1.00       15.0       10.0       25.0       2       50.0         ANE       0.17       0.27       1.00       18.00       24.8       12.5       -       37.3       4       149.2         ANE       0.72       1.13       2.05       4.19       167.7       10.0       -       177.7       2       355.4         ANE       0.72       1.13       2.05       4.19       167.7       10.0       -       177.7       2       355.4         ATORS - AC MOTOR - RELAY CONTROLLED       0.01       1.5       -       0.5       9.4       1       9.4         0P       0.03       1.09       1.97       4.04       6.9       -       0.5       9.4       1       9.4         0P       0.01       .02       0.04       0.07       1.5       -       0.1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6	<u>}</u>	19.64	26.87	46.63	99.52	17.4	1	9.1	19.2	2	30.4
MILAD TOCHON INVERSION 15.0       10.0       15.0       10.0       5.0       25.0       2       50.0         ANE       0.17       0.27       1.00       15.0       10.0       -       25.0       2       50.0         ANE       0.17       0.27       1.00       16.00       24.8       12.5       -       37.3       4       149.2         ANE       0.72       1.13       2.05       4.19       167.7       10.0       -       177.7       2       355.4         ATORS-AC MOTOR-RELAY CONTROLLED       0.93       1.97       4.04       6.9       -       0.5       9.4       1       9.4         0P       0.03       1.09       1.97       4.04       6.9       -       0.5       9.4       1       9.4         0P       0.01       .02       0.04       0.07       1.5       -       0.1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6 <t< td=""><td></td><td></td><td></td><td>NUFDTE</td><td>CONTE</td><td></td><td></td><td></td><td></td><td>_</td><td></td></t<>				NUFDTE	CONTE					_	
0.17       0.27       1.00       15.0       10.0       5.0       10.0       5.0       37.3       4       149.2         ANE       0.72       1.13       2.05       4.19       167.7       10.0       -       177.7       2       355.4         ATORS-AC MOTOR-RELAY CONTROLLED       16.00       24.8       12.5       -       37.3       4       149.2         ATORS-AC MOTOR-RELAY CONTROLLED       0.03       1.97       4.04       6.9       -       0.5       9.4       1       9.4         0P       0.03       1.09       1.97       4.04       6.9       -       0.5       9.4       1       9.4         0P       0.01       .02       0.04       0.07       1.5       -       0.1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1       1       1.6       1       1       1.6       1       1       1       1       1       1       1 <td< td=""><td></td><td>277-02</td><td>L NOID</td><td></td><td></td><td></td><td></td><td></td><td>0 2 0</td><td>2</td><td>50.0</td></td<>		277-02	L NOID						0 2 0	2	50.0
A.I.3       4.66       8.80       18.00       24.8       12.5       -       37.3       2       355.4         ANIE       0.72       1.13       2.05       4.19       167.7       10.0       -       177.7       2       355.4         ATOR5 - AC MOTOR - RELAY CONTROLLED       0.03       1.97       4.04       6.9       -       0.5       9.4       1       9.4         OP       0.01       .02       0.04       0.07       1.5       -       0.1       1.6       1       1.6         OP       0.01       .02       0.04       0.07       1.5       -       0.1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1       1.6       1.6       1.6       1       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6       1.6 <t< td=""><td></td><td>0.17</td><td>0.27</td><td>1.00</td><td>00.1</td><td>15.0</td><td>0.01</td><td> </td><td></td><td></td><td>1 4 9 2</td></t<>		0.17	0.27	1.00	00.1	15.0	0.01				1 4 9 2
ANE     0.72     1.13     2.05     4.19     167.7     10.0      177.7     2     0.00.7       ATORS - AC MOTOR - RELAY CONTROLLED     ATORS - AC MOTOR - RELAY CONTROLLED     0.9     0.5     9.4     1     9.4       0P     0.03     1.09     1.97     4.04     6.9      0.5     9.4     1     9.4       0P     0.01     1.05     0.07     1.5      0.1     1.6     1       0P     0.01     1.65     2.99     6.11     34.4      0.6     35.0     1     35.0       1.40     1.65     2.99     6.11     34.4      0.6     55.7     2     111.4       2.71     3.19     5.77     11.61     54.9      0.6     55.7     2     111.4	0	4.13	4.66	8.80	18.00	24.8	12.5	{	51.5	+ r	1 7 7 V
ATOR5 - AC MOTOR - RELAY CONTROLLED       0.97       4.04       6.9       -       0.5       9.4       1       9.4         0P       0.93       1.09       1.97       4.04       6.9       -       0.5       9.4       1       1.6       1       1.6         0P       0.01       .02       0.04       0.07       1.5       -       0.1       1.6       1       1.6         0ATORS - DC MOTOR - RELAY CONTROLLED       1.5       -       0.6       35.0       1       35.0         1.40       1.65       2.99       6.11       34.4       -       0.6       35.7       2       111.4         2.71       3.19       5.77       11.61       54.9       -       0.6       55.7       2       111.4	ANE	0.72	1.13	2.05	4.19	167.7	0.0	1	7.11	J     	
0     0.93     1.09     1.97     4.04     6.9     -     0.5     9.4     1     9.4       0     0.01     .02     0.04     0.07     1.5     -     0.1     1.6     1     1.6       0     0.01     .02     0.04     0.07     1.5     -     0.1     1.6     1     1.6       0     1.40     1.65     2.99     6.11     34.4     -     0.6     35.0     1     35.0       2.71     3.19     5.77     11.81     54.9     -     0.6     55.7     2     111.4	ATOF	25 - AC M	OTOR - R	ELAY CO	NTROLL	ED		-			
0         0.01         .02         0.04         0.07         1.5         -         0.1         1.6         1         1.0           UATORS-DC MOTOR - RELAY CONTROLLED         0.6         35.0         1         35.0         1         35.0           1.40         1.65         2.99         6.11         34.4         -         0.6         35.0         1         35.0           2.71         3.19         5.77         11.61         54.9         -         0.6         55.7         2         111.4		0.93	60.1	1.97	4.04	6.9		0.5	4.6		9.6
UATORS-DC MOTOR - RELAY CONTROLLED 1.40 1.40 1.65 2.99 6.11 34.4 - 0.6 35.0 1 35.0 1 35.0 1 35.0 1 11.4 11.61 54.9 - 0.8 55.7 2 111.4 1667.1	o o	0.0	.02	0.04	0.07	- 22	1	 0	9 	-	
1.40     1.65     2.99     6.11     34.4     -     0.6     35.0     1     35.0       2.71     3.19     5.77     11.81     54.9     -     0.0     55.7     2     111.4       2.71     3.19     5.77     11.81     54.9     -     0.6     55.7     2     111.4	LINATO	RS-DC P	40TOR -	RELAY	CONTRO	LLED					
2.71 3.19 5.77 11.81 54.9 - 0.8 55.7 2 111.4 707AL 1667.1			1 12	1 2 99	11.9	34.4		0.6	35.0		0.00
1667.1 2017 JULY 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1 2017 1		 		Г Г	181	54.9		0.0	55.7	2	111
		1.1	1.C							TOTAL	1667.1

----





4.1.9.1 (Cont.)

Weight

Zero Stroke Weight	= 16.2 lb (3)
Stroke Weight (10 x .53)	= 5.3 lb 🕉
Notor	= 3.3 lb (4)
Total	= 24.8 lb

CANARD

Stall Load	$= 6400 \ 1b \ (5)$
Stroke	= 5.75 in (5)
Motor Power Weight	= 1.00  kw (2)

Iero Stroke Weight		=	11.8 1	o (3)
Stroke Weight (5.75 x	.39)	=	2.2 11	5 (J)
Motor = $(2 \times 0.5)$	•	=	1.0 1	⇒ <b>(</b> 4)
	Total	=	15.0 1	5

THRUST VECTOR VANE

Stall Load	= 75,314 1	b 🌀
Stroke	= 5.20 i	n (6)
Motor Power	= 2.05 k	<u>w</u> (2)
Weight		

Zero Stroke Weight		=	128.0	TD	Ŵ
Stroke Weight (5.2 x	4.59)	=	23.9	1Ъ	Ø
Motor = $(2 \times 2.6)$		=	5.2	1b	(4)
	Total	=	167.7	1b	

RIGHT RAM AIR SCOOP

 Stall Load
 = 4100 lb (1)

 Stroke
 = 2.52 in (1)

 Notor Power
 = 1.97 kw (2)

 Weight
 = 1.97 kw (2)

Zero Stroke Weight	$= 7.5 \ 1b \ (3)$
Stroke Weight $(2.52 \times .25)$ Motor Weight $(1.0 \times 0.8)$	$= 0.6 \ 1b \ (3)$ = 0.8 lb (4)
Total	= 8.9 lb

LEFT RAM SCOOP

Stall Load Stroke Motor Power Weight	= 1200 lb (1) = 1.00 in (1) = .04 kw (2)	
Zero Stroke Stroke Weig Motor (1.0 :	Weight ht (1.0 x .08) x 0.2) Total	$ \begin{array}{c} = 1.2 \text{ lb } (3) \\ = 0.1 \text{ lb } (3) \\ = 0.2 \text{ lb } (4) \\ = 1.5 \text{ lb } \end{array} $

4.1.9.1 (Cont.) NOSE GEAR = 14,900 lb (1 Stall Load 5.0 in Struke 2.99 kw (2) Motor Power = Weight Lerc Stroke Weight = 27.4 lb Stroke Weight  $(5.0 \times .91) = 4.6$  lb = 2.4 lbMotor Weight (2 x 1.2) Total = 34.4 lb MAIN GEAR = 24,000 1b (1 Stall Load 6.00 in Ð Stroke 5.77 kw Motor Power Ξ Weight = 44.2 1b (7 Zero Stroke Weight  $\overline{7}$ Stroke Weight  $(6 \times 1.44) = 8.6 \text{ lb}$ Motor Weight  $(1 \times 2.1) = 2.1 \text{ lb}$ Total = 54.9 lb5) See Figure 48 and Paragraph 4.2.1.6 See Table 9 See Table 12 See Table ?5 Extrapolated from Figure 30 See Figure 39 See Figure 38

4.1.9.2 Electrical Power Distribution Elements - Figure 40 is a plain view of the aircraft showing the general location of the major components constituting system No. 1 power generation distribution and utilization elements. It was used as a basis for determining the wire lengths and sizes of the wiring used in the bus feeder and power distribution portions of system No. 1. It was assumed that system No. 1's lengths, sizes, and routings were sufficiently like the other three systems so that it could be considered a representative average of the other three. Thus, the total system's wiring weight was determined by multiplying the weight determined for system No. 1 by four. Table 26 is a detailed listing of the feeder and power distribution wiring for Aircraft I. It shows that the total weight of the power wiring including supports, harness, shielding and connectors was 120.3 lb. It should be remembered that this was the weight for 270 volt power distribution wiring and did not include 400 Hz AC power wiring, avionics equipment wiring, 28 VDC wiring, or the wiring for the fly-by-wire/fly-by-light system. In these latter four instances, as has already been discussed, Aircraft I and Aircraft II were considered essentially identical and thus these elements did not enter into the trade study.

4.1.9.3 <u>Auxiliary Power and Starting System</u> - The APU has already been defined in paragraph 4.1.8 and the starting system in paragraph 4.1.6. However, a major element of the auxiliary power system, not yet considered, was the battery which was provided primarily to supply power for a descent to 20,000 from any higher altitude in the event of an emergency. As pointed out in paragraph 4.1.2.3, this battery was to be of sufficient size to provide at least 4 minutes of power in an emergency descent mode. The continuous load,



ą 1.14

## TABLE 26. MELGHE - ELECTRICAL POWER WIRING (PAGE 1 OF 2 PACES)

	r			Y				
ROUT-	SEG-	LEN-	RAT-	AWG	WT.		ROUT-	BEG-
ING	MENT	GTH	ING	NO.WIRES			ING	MEN
NO.	NO,	FT.	AMP.	-GAGE	LBS		NO.	NÔ.
1	1-1	2	130	6 - IZ	.28		23	23-34
2	2-2	2	130	6-12	.28			23-37
3	3-3	6	16	1 -14	.09		24	24-38
	3-4	6	16	1-14	.09		25	25-39
	3-5	S	16	1-14	.07			25-40
4	4-6	6	6	1-20	.03		26	26-41
	4-7	6	6	1-20	.03		27	27-42
	4-8	6	6	1 - ZO	.03		28	28-43
5	5-9	5	130	6-12	.70		29	29-44
	5-10	4	130	6-12	.56		30	30-45
6	6-11	2	130	6-12	.28			30-46
7	7-12	2	130	6-12	.28		31	31 - 47
8	8-13	2	88	6-14	.18			31-48
9	9-14	6	3	1-24	.02		32	32-49
	9-15	5	3	1-24	.01		33	33-50
10	10-16	6	5	1-22	.02		34	34-51
	10-17	6	5	1-22	.02		35	35-52
l ti -	11-18	6	5	1-22	.02		36	36-53
	11-19	4	5	1-22	.02		37	57 -54
12	12-20	6	5	1-22	.02		38	38-55
	12-21	6	5	1-22	.OZ		39	39-56
	12-22	6	5	1-22	۶Q		40	40.57
	12-23	6	5	1-22	DZ		41	41-58
13	13-24	3	5	1-22	.01		42	42-55
14	14-25	6	Э	1-24	.02			42-60
	14-26	4	Э	1-24	.01			42-61
15	15-27	6	5	1-22	.02			42-62
	15-28	3	5	1-22	.01		43	43-69
16	16-29	5	5	1-22	.02			43-64
17	17-30	6	50	5-18	.24			43-65
18	18-31	3	30	2-14	.09		44	44-66
19	19-32	6	30	2-14	.18		45	45-67
20	20-33	3	6	1-20	.02			45-68
21	21-34	6	6	1-20	.03		46	46 - 69
22	22-35	6	8	1-18	.05			46-70
TOTA	LWE	GHT	THIS	COL.	3,79	Т	OTAL	WEI

ţ

4 3 1-24 .01 6 6 1-20 .03 5 16 1-14 .07 5 3 1-24 .01 6 36 3-16 -18 3 30 2-14 .09 6 30 2-14 -18 3 6 1-20 .02 6 6 1-20 .03 6 130 6-12 .81 6 158 7-12 .95 6 9 1-18 .05 6 9 1-18 .05 6 9 1-18 .05 9 7 1 -18 .06 3 3 1-24 .01 З 1-24 .02 6 З 1-24 6 .02 3 1-24 8 .02 6 З 1-24 .02 6 3 1-24 .02 6 5 1-22 .02 8 5 1-22 .03 GHT THIS COLUMN 4.96 8.75

LEN-RAT-

88

88

88

5

5

30

88

10

10

З

З

З

GTH

FT.

6

5

6

6

4

5

1,

6

4

4

4

5

AWG

AMP - GAGE LBS

6 - 14

6-14

6-14

1-22

1-22

2-14

6 - 14

1-16

1-24

1-24

1-16 .06

1-24 .01

ING NO. WIRES

WT.

.54

.45

·54

.02

.02

.15

. 30

.04

.01

.01

TOTAL WEIGHT THIS PAGE

TABLE 26. WEIGHT - FEETRICAL POWER WIRING (PAGE 2 OF 2 PAGES)

ROUT-	SEG-	LEN-	RAT-	AWG	WT.		
ING	MENT	GTH	ING	no wires	ţ		
NO.	NO.	FT.	AMP	-GAGE	LBS.		
47	47-71	6	3	1-24	.02		
	47-72	7	3	1-24	.02		
48	48-73	6	3	1-24	.02		
	48-74	5	3	1-24	.0Z		
49	49-75	5	3	1-24	.02		
50	50-76	8	3	1-24	.02		
	50-77	10	3	1-24	.05		
	50-78	10	3	1-24	.05		
	50-79	10	3	1-24	.03		
	50-80	12	3	1-24	.03		
	50-81	10	3	1-24	.03		
	50-82	8	3	1-24	.02		
51	51-83	Э	3	1-24	.01		
	51-84	6	3	1-24	.02		
	51-85	6	3	1-24	.02		
15	15-86	2	150	6-12	.28		
25	25-87	5	130	6-12	07.		
35	35-88	6	130	6-12	.80		
	35-89	6	130	6-12	.80		
	35-90	6	130	6-12	.80		
45	45-91	6	130	6-12	.80		
l	45-92	7	130	6-12	1.00		
A1	IA-93	3	160	7-12	.50		
2A	2A-94	6	160	7-12	90.		
IA	1A-95	3	160	7-12	.50		
2A	22.96	6	160	7-12	.90		
TOTAL WEIGHT THIS COLUMN 8.32							

WEIGHT SUMMARY	
TOTAL WIRE WEIGHT	* 68.28
SUPPORTS, HARNESS & SHIELD	)ING = 10.24
CONNECTORS .95/SYS AND	
380/AC - AVERAGE WT.	
PER CONHECTOR = 1.1 LO.	
380 x 1	-1 =41.80

TOTAL WIRING SYSTEM WEIGHT = 120.32

TOTAL WIRE WEIGHT IN AIR-CRAFT POWER SYSTEM (4 SYSTEMS)=4 × 17.07 = 68.28

TOTAL SYSTEM #I WIRE WT. 17.07

existing during an emergency descent, was determined as follows:

20.442 KW (See Table 16, PG 8, BAT BUS)
.810 KW (See Table 16, PG 5, ITEM 203)
2.268 KW (See Table 16, PG 6, ITEMS 701 through 804, BAT BUS

At the time this report was written, current state of the art NI-CAD batteries exhibited specific weights of 5.8 Watt-Hr/lb. This specific weight included such items, necessary for a practical installation, as the battery case, shrouding, thermistors and thermal switches. Because of the rapid advances which it was felt would have occured in battery technology by the 1990 time period, it was assumed that an equivalent battery specific weight at that time would be 11.4 watt-Hr/lb. From this, the battery weight was determined as follows:

23.520 Watts			4 Min							
11.5 Watt-Hr		Х	6 <u>0 Min</u>	Х	1.5	Operating	Margin	=	204.52	Lb.
LB	•		HR							

Use 205 Lb

4.1.9.4 <u>System Weight Summary</u> - Table 27 summarizes all the weights subject to trade in the Aircraft I electrical system. As can be seen in Table 27 the total weight was 2817.0 Lbs.

# TABLE 27. AURCRAFT I FLECTRICAL SYSTEM WEIGHT SUMMARY

	QUAN	TITY	UNIT	TOTAL
FOUIPMENT ITEM ()		PER	WEIGHT	WEIGHT
	· · · ·	A/c	(LB)	(LB)
G - PRIMARY GENERATING SYSTEM (2)	)	4	115.0	460.0
GE - EMERGENCY GENERATOR 3	,	2	83 5	167.8
APU - AUXILIARY POWER UNIT (3)			245.0	245.0
STARTING SYSTEM EQUIPMENT (2)		2	716	143.2
AFPC - AC EXTERNAL POWER CONTACTO	R	4	19	7.6
ALC - ACLINE CONTACTOR		4	19	7.6
DEPC - DC EXTERNAL POWER CONTACT	DR	4	25	10.0
DIC - DCLINE CONTACTOR		Δ	29	11 6
AFRP - AC ESSENTIAL PUS RELAY		2	2.5	4 0
BCR - BATTERY CHARGER RELAY			2.0 0.4	0.4
IRD - BATTERY RELAY			2 0	20
IRS - RATTERY SWITCH				2.0
$ \mathbf{A}  = \mathbf{A} \mathbf{T} \mathbf{T} \mathbf{E} \mathbf{R} \mathbf{Y}$			0.E	205.0
DERD-DESCENTIAL BUS PELAY		2	203.0	205.0
DEDR - DUESSEN HAL DUS NEUT	(40)	C	3.0	6.0
EMM - EXI. FOWER MUNITUR & RECFI.			3.0	3.0
EPM - EXI. FOWER MUNITUR & RECHT.	(DC)		4.0	4.0
SR - STARIER KELAY		Z	3.8	7.6
BC - BATTERY CHARGER	• •		5.0	5.0
ACT - ELECTRICAL ACTUATION FUNCTION	NG	-		1167.1
AMAD - AIRFRAME MOUNTED ACESSORY (	ORNE	Z	90.0	180.0
WIRE, SUPPORTS & HARNESS (3)		-		78.5
CONNECTORS 3		-	<del>نگر</del> وکه	41.8
COMPONENT SUPPORTS AND MISC.		-		59.4
Тс	TAL	WEI	GHT	2817.0

() ACRONYMS HEADING ITEM TITLES CORRELATE WITH FIG.

SEE PAGE SEE PAGE 4 SEE TABLE

3 SEE TABLE

4.2 <u>Aircraft II</u> - Aircraft II represented the more conventional approach to secondary power generation, distribution, and utilization in that the aircraft used hydraulic power to power those components which have historically been powered electrically on advanced military aircraft of the immediate past. Thus aircraft II had a conventional power split between hydraulics and electrical but departed from the conventional by using an advanced (8000 PSI rated pressure) hydraulic system. The selection of 8000 PSI as the hydraulic system's rated pressure was based on extensive study programs and hardware development programs, as well several flight tests conducted at Rockwell's Columbus Division. This series of programs has been conducted over the last 15 years and has been documented in references 1 through 11. These programs indicated that 8000 PSI was very close to an optimum system pressure for advanced hydraulic systems given current and near future materials of construction. A listing of the major advantages and disadvantages of 8000 PSI rated system pressure, versus the current conventional 3000 PSI, as derived from these programs follows:

### **ADVANTAGES**

- 1. Projected weight saving 30%
- 2. Projected installed volume saving of 40%
- 3. Survivability gains due to:
  - A. Less Volume
  - B. Less projected area
  - C. Heavier walled components
- 4. Lower component costs due to:
  - A. Less material used
    - B. Less machining costs due to heavier walls largely off setting the higher costs of slightly tighter tolerances

#### DISADVANTAGES

- 1. New ground test equipment required
- 2. Snear stable fluids required
- 3. Adverse effects of actuator stiffness

Table 28 represents a generic weight breakdown of the hydraulic system in the baseline study aircraft and showed the expected system weight at three rated operating pressures (3000, 4000 and 8000 PSI). This figure showed a projected weight saving of 584 lbs by using an 8000 PSI system in preference to a 3000 PSI system. Using a figure c. 2.7 lb of gross take off weight saved per lb of direct weight saving the gross take off weight could be reduced by 1577 lb through the use of 800 PSI and was considered a very significant figure.

Figure 41 is a plot of the data from Table 28 and showed that 8000 PST was very close to an optimum weight i.e., that increases in operating pressure above 8000 PSI would not achieve significant further reductions in weight. This further verified that 8000 PSI was probably the proper pressure selection.

As will be seen elsewhere (paragraph 4.2.1.6) all flutter (stiffness) critical actuators on the study aircraft (i.e. the upper rudder, the aileron and the inboard flap) were powered through mechanical hinges using hydraulic motors. Since the gear reduction between motor and surface in all instances, was

	SVSTEM PRESSURE				
	3000 PS/	4000.551	ECCO PSI		
HYCRAULIC SYSTEM	[793]	[70.9]	[487]		
POWER GENERATION	(358)	(319)	(251)		
PUMPS	98	38	65		
RESERVOIRS	117	103	75		
SUPPORTS	2	6	4		
MISCELLANEOUS	/36	122	:06		
POWER DISTRIBUTION	(435)	(3.90)	(236)		
FLUMBING & FITTINGS	263	235	142		
FLUID	172	155	94		
FLIGHT CONTROLS	[266]	[774]	[682]		
ACTUATION DEVICES	(490)	(438)	(399)		
CYUNDERS É ROTARY ACT.	454	406	377		
MECHANISM & SUPPORTS	36	32	22		
CONTROL MODULES	(275)	(247)	(228)		
SERVO VALVES ETC.	275	247	228		
POWER DISTRIBUTION	(101)	(89)	(55)		
PLUMBING & FITTINGS	68	60	37		
FLUID	33	29	18		
LANDING GEAR & OCCRS	[/28]	[114]	[84]		
ACTUATION DEVICES	(51)	(45)	(41)		
CYLINDERS ETC.	43	38	36		
MECHANISM & SUPPORTS	8	7	. 5		
POWER DISTRIBUTION	(77)	(69)	(43)		
PUMBING & FITTINGS	66	59	36		
FLUID	11	10	7		
MISCELLANEOUS SYSTEMS	[/59]	[/43]	[109]		
ACTUATION DEVICES	(92)	(82)	(74)		
CYLINDERS, MOTORS ETC.	84	75	69		
MECHANISM & SUPPORTS	5 8	7	. 5		
POWER DISTRIBUTION	(67)	(61)	(35)		
PLUMBING & FTT ? INGS	50	46	26		
FLUID	17	15	9		
TOTAL	1946	1740	1362		
POWER GENERATION	358	319	251		
POWER DISTRIBUTION	630	609	369		
ACTUATION					
ACTUATORS	633	<u></u> 15	514		
CONTROL MODULES	275		27A		

TABLE 28. ALS HYDRAULIC SYSTEM WEIGHTS AT ARTOUS PRESSURES.

165


Figure 41. AIS Weight Trends at Various System Pressures

greater than 5000:1 (see Table 18 & 19), stiffness in the hydraulic circuit was no problem. With stiffness eliminated as a consideration the "advantages" listed above so far outweighed the "disadvantages" that 8000 PSI was selected as the design pressure.

A double voltage approach was also considered as a means to update and modernize the electrical power system on aircraft II in a manner similar to the update of the hydraulic system. However, for the reasons already given in paragraph 4.1.1, this approach did not seem to offer any advantage so a conventional 115/200V 400 HZ AC power system was retained for aircraft II.

4.2.1 <u>Hydraulic System Description</u> - The baseline hydraulic power generation, distribution, and utilization system is shown in block diagram form in Figure 42. Figure 43 is a more detailed schematic showing all major components making up the total hydraulic system, and Figure 44 shows the spatial prrangement of these components. The system had two equal authority hydraulic systems and a third emergency system. The third system powered only those functions necessary to recover from a maneuver and maintain level flight. The primary hydraulic system pumps were rated at 8000 psi. They were driven by airframe mounted accessory drives (AMADs), two pumps to a system, each powered by the same engine, and incorporating the master slave concept. A fifth pump, also rated at 28 GPM and 8000 PSI, was driven by the APU/EPU which acted as a third (emergency) power source. The AMAD was driven by the engines via a power take off shaft (PTO) or as an alternate by the auxiliary/emergency power unit (APU/EPU) shown in Figure 45. The APU/EPU was started by a hydraulic motor using stored energy from an accumulator.

4.2.1.1 <u>APU/EPU Operation</u> - The APU/EPU operation under various conditions was as follows:

#### A. APU/EPU Start - Aircraft on Ground - Engines Inoperative

APU/EPU startup is initiated by a cockpit switch. This switch, using battery power, actuates a solenoid valve which ports high pressure hydraulic fluid, stored in an accumulator, to the APU/EPU start hydraulic motor (SM in Figure 45). The motor accelerates the APU compressor - turbine to light-off speed. Using the combustion of a mixture of jet fuel and air in the conventional manner as an energy source, the compressor turbine accelerates to operating speed. At full speed the surplus energy from combustion (i.e. the energy over and above that necessary to power the compressor turbine) drives a free turbine which in turn, through suitable gearing, powers a 28 GPM (131 H.P.) "emergency" hydraulic pump, a 7 KW (9.4 H.P.) emergency generator and a 180 KW 242.0 H.P.) load compressor. It is capable of starting at all airport altitudes up to 5000 ft. (1524 M) as well as achieving inflight starts at all altitudes up to 20000 ft (6096M).

### B. Engine Start - Aircraft on the Ground

1. Normal Start Using APU/EPU

With the APU/EPU running, the load compressor (see Figure 45) delivers pneumatic power to the air turbine start motor (ATS/M) associated with the engine selected for starting. The load



Figure 42. Hydraulic System Block Diagram















 •

-----





Figure 45. Aircraft II Secondary Power Generation System

compressors are limited in capability to starting one engine at a time. The ATS/M, working through the accessory gearbox and power take off (PTO) shaft, accelerates up to self sustaining speed. Start time to ground idle is nominally 35 seconds.

2. Normal Start Using Ground Cart

For engine starts using a ground cart the APU/EPU is normally not operating. Ground cart high pressure air is supplied at the ground connection; (see Figure 45) from this point the starting sequence is like B.1 above.

3. Accelerated Start

If rapid response requirements dictate, simultaneous engine starts may be achieved in either of two ways. The first is the most rapid but the least fuel efficient. In this case the APU portion of the APU/EPU is started in the normal way, after which one of the engines is started. Simultaneously the EPU gas generator is started, using LOX and jet fuel, and the second engine is started from this source. Using this approach the LOX required for high altitude engine start is largely used up and, assuming the reason for needing simultaneous engine starts is the need to "scramble" rapidly, there is no time to refuel with LOX prior to take off.

The second method for achieving simultaneous engine starts is to start pneumatically from a ground source. In this instance there is adequate power to start both engines simultaneously without the need to start up the APU/EPU. Under some circumstaries this method can be faster than method No. 1 above. If, for example, air is drawn from a centralized air base air supply and is immediately available the time necessary to start up the APU and EPU in method No. 1 is avoided. The only time required is the time necessary to connect and disconnect the air hoses and to open and close the air supply valve. In those instances where the aircraft is held in alert status, lined up with the runway, with the air hose already connected, and has a tear away type of disconnect at its hose to fuselage connection, the time to start engines and to break ground is considerably reduced over that required for method No. 1.

The normal method for starting engines, however, will be to start each engine sequentially with the APU. The second engine will be either started at the ramp or while the aircraft is taxiing out to take off position.

#### C. APU/EPU Operation - Low Altitude

Below 20,000 ft (6096 M) altitude there is sufficient atmospheric oxygen so that the APU portion of the APU/EPU can be started and will operate continuously meeting its load demands. In this regime the APU will supply duty cycled loads up to 70.3 H.P. for the system No. 3 (emergency) hydraulic pump and 9.8 H.P. for the emergency generator while delivering sufficient air to air start an engine. In the event that an engine cannot be restarted the APU will deliver sufficient pneumatic power to the associated ATS/M so that, with the PTO shaft uncoupled (see Decoupler Figure 45), the hydraulic pump and generator on that gearbox (AMAD) can be driven at duty cycled power levels up to 101 H.P.. This same power level or more is available for driving either of the two AMADS on the ground for use in maintainance and checkout operations.

#### D. APU/EPU Operation - High Altitude

Above approximately 20,000 ft. (6096 M) altitude there is insufficient oxygen to start or maintain operation of the air breathing APU. If any emergency (such as a two engine flameout) occurs above this altitude which requires power the EPU will be turned on. Since the EPU burns LOX-JP4 (both stored on the aircraft) it can be started at any altitude. The LOX-JP4 mixture is burned in a catalytic combustor (hence nearly instantaneous light off) and the gaseous products of combustion are directed to the APU's free turbine. The free turbine, through the APU's output gear train, drives the emergency pump and generator, however, the load compressor is unloaded. The emergency pump and generator supply the electrical and hydraulic power necessary to fly, or glide, the aircraft down to an altitude at which the APU can be started and engine start attempts can be made.

4.2.1.2 Brake System - It was decided that the brake system would remain hydraulic for both versions of the study aircraft (hydraulic and electrical). In each instance the brake system would use the newly developed chlorotrifluorethylene type non flammable hydraulic fluid and would be a small separate system independent of the main hydraulic or electrical power generation and distribution systems. This decision was made based upon consideration of several factors:

- A. If aircraft of the mid 1990's retain hydraulics as a prime element of power generation and distribution in their secondary power systems, they would almost certainly use a separate non-flammable fluid subsystem for their braking system. This was felt to be true because of the many aircraft losses which have been traceable to brake fires fed by the currently used flammable hydraulic fluids.
- B. Brake systems for future high performance military aircraft would have an advanced version of a fly-by-wire type of antiskid. In this approach the pilot's brake input and the incipient skid sens would both be transmitted as an electrical signal and would be mixed electronically to provide a modified output signal. This output signal would be sent to an amplifier in the form of a metering valve for the hydraulic system or an inverter for the electrical system. The four servo controlled metering valves required for the hydraulic approach would weigh approximately 2 lbs while the 4 ambient air cooled inverters for the electrical approach will weigh at least 12 lbs.

C. Electrical brake systems have been under development for several years, however all work to date, has been analytical; no hardware has been built and no components have been tested. Based on the analysis and some optimism it was felt that electric brakes might be developed to the point where they were nearly equivalent in weight and reliability to current hydraulic brakes. There was, however, continuing doubt that they would meet the wet and dry runway anti-skid performance requirements currently met by hydraulics.

The electric brake's primary item of desireability was the fact that it would eliminate brake fires. A secondary advantage was the fact that electric brakes offered the possibility of reduced routine maintenance requirements relative to a hydraulic system and particularly to a separate hydraulic system (actually two small separate hydraulic systems with two separate reservoirs) using non-flammable fluid, such as planned for the ATS aircraft.

Considering items A through C it was apparent that the scales were tipped in favor of the hydraulic approach. The fact that the prime virtue of the electric brake (no brake fires) was offset by the use of non-flammable fluid in the hydraulic approach was instrumental in shifting the balance radically. In effect, based on what was known at the time of this report, reascnable extrapolation of the state of the art to the mid 1990's would still favor the hydraulic approach in the following areas:

> Weight Steady State (stalled) Power Demand Heat Rejection During Braking Reliability Unscheduled Maintenance Performance (anti-skid capability)

Only in the area of scheduled maintenance would the electric brakes have had a clear superiority. Because the non-flammable fluid brake subsystem could have been in both baseline aircraft without impairing the "all electric" power generation and distribution characteristics of the aircraft, it was decided that this was the approach to use.

As previously indicated, the brake system became an arrangement consisting primarily of two small compact hydraulic systems employing non-flammable hydraulic fluid; each having its own pump, reservoir, filters, control values and actuators. Each hydraulic system powered brakes on both the right and left hand main gear wheel with one system acting as a backup for the other. One system had an accumulator which could supply limited emergency braking in addition to its basic function of providing parking brake capability. The pressure compensated variable delivery pump for each hydraulic system was driven by a hydraulic motor on the hydraulic aircraft (Aircraft II) and by an electric motor on the all electric aircraft (Aircraft I). A schematic showing the brake system is included as part of Figure 43.

4.2.1.3 In-Flight Refueling - It was decided that the standard (UARRSI) inflight refuel receptacle would be used in both vehicles. This decision was based on Rockwell experience in attempting to use a non standard refuel receptacle in the B-1 aircraft. Even though the change was modest; substituting 4000 psi actuators, valves, and plumbing for the standard receptacle's 3000 psi components and saving a little weight in the process, it was not bought. The Air Force felt that it was of such overriding importance that the standard receptacle be used, and thus be warehoused and available for use interchangeably on any Air Force aircraft, that they were willing to give up the potential weight saving and suffer an additional weight penalty to avoid the use of a non-standard receptacle. The additional weight penalty was that represented by the pressure reducers and pressure relief equipment which was necessary to adapt the B-1's 4000 psi system to the 3000 psi receptacle. Since the standard unit is even more widely used now than it was at the time of the B-1 decision, and considering that its use will be even more extensive during the 1990's, it was felt a standard receptacle was nearly mandatory unless some truly compelling reason legislated to the contrary. There appeared to be no compelling reason.

Both the electric and hydraulic baseline aircraft could provide 3000 psi hydraulic power for the standard receptacle thru the use of power transfer unit (PTU). Essentially a PTU was a specialized version of a motor pump. Such a device (PTU) was mandatory for the "all electric" baseline aircraft if a standard (UARRSI) receptacle was to be used. Theoretically a pressure reducer (rather than a PTU) could have been used in the 8000 psi hydraulic baseline aircraft to adapt to the standard receptacle. However, return line pressures near the receptacle in an 8000 psi system would often exceed 3000 psi. Since a pressure reducer would have been referenced to return pressure near the receptacle, the pressure in the receptacle components would always be higher than return pressure and often much higher than 5000 psi. These high pressure return conditions could be largely avoided by running a dedicated return line 30 ft back to the reservoir. However, this would involve an added weight penalty of 4 lbs and still leave a serious doubt as to whether the allowable return pressures would not intermittently be exceeded at low temperatures. For these reasons use of a pressure reducer for this application was considered unacceptable. Therefore, as indicated above, a PTU was used for the hydraulic baseline as well as for the electric. In the hydraulic baseline case the PTU consisted of a 3000 psi pressure compensated pump supplying the receptacle driven by an 8000 psi constant displacement hydraulic motor. In the case of the "all electric" airplane, the hydraulic motor would be replaced by an electric motor but the 3000 psi pump would remain unchanged. In affect everything downstream (on the receptacle side) of the hydraulic pump mounting flange (interface point) was identical for both the hydraulic and electrical baselines. The "downstream" items were the pump (1.9 lb), a reservoir module (3.8 lb) containing pressure, case drain and return filters, relief valves, fluid level gages etc., plumbing and fittings (0.3 lb), and the receptacle itself.

Under these circumstances an interface was created which, once its transmitted power and rotational speed were defined, could be driven either by hydraulic or electric motors. The power at the interface and the rotational speed is given in Table 9. 4.2.1.4 Landing Gear - The nose gear was extended and retracted by both main hydraulic systems (See Figure 42) hence a tandem linear actuator was used. The main gear was retracted and extended by system #2 only. Upon loss of system #2, emergency extension was obtained by free fall, with the weight of the gear the prime mover. Conventional up and down locks, and fairing door actuation was employed. Run-around valving allow for gear operation with the aircraft on blocks.

4.2.1.5 <u>Pump Sizing</u> - The peak hydraulic system flow demands, and the ones which sized the system, were those which occured during combat. The magnitude of the flow demands were derived from the loads shown in Tables 9 and 12 and were tabulated in Table 29 to show the loads in each system which apply during combat. In a great many aircraft the peak flow demand, in relation to pump capacity, occurred during landing flare-out when flight control and landing gear demands were high and the pump capacity was low because the engine was at idle RPM. This was not true of this aircraft, however, since the flow demands of gun, thrust reverser, and plug throat operations during combat far exceeded flare-out demand. Therefore, the pumps for all three systems were sized based on the total flow demand load (in KW) shown in Table 29.

The maximum flow rating of the pump (in GPM) was determined using the following formulas and assumptions:

Formulas:

Flow (GPM)	=	Power (HP) X 1714 Pressure (PSIP)
and		
Power (HP)	=	Power (KW)

.746

Assumptions:

The actual maximum power demand was 2/3 of the the theoretical (summed) power demand, since in a group of actuators such as those listed in Table 29, not all actuators would be operating at a given time, and of those operating, not all would be operating at their peak demand capability. This was a restatement of a basic "ground rule" already given in paragraph 2.4.4.

The effective pressure at the load was 4667 PSID in an 8000 PSI system based on the following:

8000 PSID System pressure Less 500 PSID Lost in supply line Less 500 PSID Lost in return line Less 2333 PSID Lost in valving (1/3 X 7000 PSID) -----4667 PSID Net at operating load and rate TABLE 29. AIRCRAFT IT LOADS DURING COMBAT (5 SEC. LOADS)

.....

# • POWERED BY SYSTEM NA NOT OPERATED DURING - NOT POWERED BY SYSTEM THIS MISSION PHASE

اليا ديم المحمد وسايد البن

ſ <u></u>		SY	STE	M	LOAD (KW)	/АСТ.	NO.OF	LOAD	SYS(	KW)
FUNCTION	-	#I	₹2	<b>#</b> 3	AT INTERFACE	AT ACT. PORTS	ACT/	<b>#</b> 1	#2	<b>*</b> 3
INBOARD FLA	P	•	0	•	14.87	17.49	2	35.0	35.0	35.0
MIDSPAN FLA	P	•	•	۲	3.02	3.55	2	7.1	7.1	7.1
AILERON		•	•	1	1.11	1.30	2	2.6	2.6	—
UPPER RUDDE	ER	٠	•	•	0.55	0-65	2	1.3	1.3	1.3
LOWER RUDDE	ER	•	•	•	0.52	0.61	2	1.2	1-2	1.2
LEADING EDGE		٠	•		1-77	2.08	6	12.5	12.5	—
	EFT	•	•		0.23	0.27	2	0.2	0.3	—
THRUST	GHT	•	•		0.72	0.72	2	0.3	0.4	
EXTERNAL LI	GHT	•	•	-	4.13	4.13	5	8.3	8.3	
PLUG U	IGHT	•	-	-	28.67	33.73	1	33.7	33.7	
THRUST LE	EFT   KHT	•		Ξ	22.84	26.87	1	26.9	26.9	
NOSE GEAR		٠	•	—	1.40	1.40	1	N/A	N/A	
MAIN GEAR ACTUATION			•		2.71	2.71	2		N/A	
NOSEGEAR		٠	-	-	0.48	0.56	1	N/A	-	—
MAIN GEAR BRAKES		۲	•		3.18	3.74	2	N/A	N/A	
RIGHT RAM		٠	-	-	0.93	0.93	1	N/A		-
LEFT RAM AIR SCOCP		•	-	-	0.01	0.01	1	N/A	-	_
ECHAUST DOOR	2 VY	•		-	0-01	0.01	1	N/A	—	
ARMAMENT	•	٠	-	-	35.05	41.24	1	41.2		—
REFUELING		٠	-	-	0.83	0.98	1	N/A		—
ULOADS FROM TABLES 7 IND 12 1109.1 44.6 TOTAL										

Using the preceding formulas and assumptions the required flow from the pump at military (combat) power became:

Flow (GPM) = 
$$\frac{2 \text{ Power } (KW)}{5 \text{ X} .746}$$
 (Pressure PSID)  
=  $\frac{2 \text{ X} (KW) \text{ X} 1714}{3 \text{ X} .746}$  = .328 X (KW)

Therefore the required flow for the three systems using the load demands from Table 9 was as follows:

(System No. 1).328 X 171.0= 56.09 GPM(System No. 2).328 X 129.7= 42.54 GPM(System No. 3).328 X 44.6= 14.6 X 3/2 = 21.9 GPM

The 3/2 factor was added to system 3's determination of required flow to cancel out the effect of the 2/3 simultaneous flow assumption. This was done because the emergency system (i.e. system 3), unlike the primary systems, had relatively few actuators all of which were very likely to be working at maximum power simultaneously during an emergency.

The foregoing statement might seem to be in conflict with the approach used in handling emergency loads in aircraft I however it actually was not. Electrical power systems differ from hydraulic in at least two vital areas as follows:

1. Electrical system load demands tend to follow and be proportional to the torque demands of its actuators, whereas, hydraulic system load demands tend to follow and be proportional to the output velocity or rate of its actuators. On this basis, as pointed out and discussed in paragraph 4.1.5.1.1, electrical systems would exhibit a peak load demand on the generating system at stall, whereas hydraulic systems would impose their peak demand at maximum surface rate-no load conditions. 2. Aircraft hydraulic power generating systems are power limited. Once the maximum displacement of the pump (or pump) is achieved at a given speed an absolute limit in power generation capability is reached which cannot be exceeded. Electrical power generation systems, in contrast, can exceed their continuous power rating by 250% for short periods.

Based on these differences it was felt that the 2/3 rule should be suspended when considering emergency conditions and hydraulic power systems in this type of aircraft. Although it was highly improbable that an aircraft would find more than one of its major emergency control surfaces approaching stall while 2/3 of its remaining surfaces were at design load and rate when recovering from a maneuver after a combat induced emergency, it was highly probable that under the same conditions all emergency surfaces, even though experiencing low aerodynamic loading, could be asked to move simultaneously at maximum rate for short periods of time. Since the hydraulic system was load limited in terms of rate and thus did not have the forgiveness of an electrical system, it was decided that all emergency control surfaces should be able to meet their maximum rate requirements simultaneously. Based on the preceding data a pump size of 28 GPM was selected. This meant that two pumps would be required for systems No. 1 and No. 2 while one would be more than adequate for system No. 5. It could be seen that, with this size selection, pump capacity for systems No. 2 and No. 3 was oversize (13.46 GPM for system No. 2 and 6.1 GPM for system No. 5). Thumps could have been selected which were exactly sized for each of the three systems, however, this would have tripled the number of nump types in the logistic pipeline. The negative impact on life cycle arising from the increased maintenance and stocking problems, would more than offset any positive impact from the exactly sized pumps reduced weight.

In addition to the standardization advantages resulting from using five uniformly sized pumps, and assuming one exactly sized pump for each system as the alternate, the standardized pump approach allowed more installation flexibility in system No. 1 and No. 2 in that their smaller size made it easier to tuck the pump into available spaces without the danger of bumping the mold line or causing major structural revisions. The individual pump units weighed less than half as much as their larger sized alternates, therefore, they were much easier to handle during maintenance operations. It should be remembered also that, even though the selected pumps for systems No. 2 and No. 3 were oversized, the plumbing systems were sized for the actual design flow and, not the rated flow of the pumps. For this reason the plumbing system remains unchanged and does not grow as would be expected if it were sized to meet pump ratings. Therefore, the weight chargeable to the use of multiple standardized, but oversized, pumps was nearly negligible.

4.2.1.6 Actuator Installation Design - It was recognized that the lightest weight and most efficient actuator design for most of the actuation functions on the aircraft would be a conventional piston type linear actuator if it could be made to fit inside the aircraft mold line. Although widely used for most actuation functions in the past, this approach had encountered difficulties, when applied to the flight control functions of more modern aircraft, because of the higher loads and thinner wings, characteristic of these aircraft. The wings on the ATS aircraft tended to be even thinner and the loads higher. However, in spite of this fact, it was felt that, because of the potential actuator size reduction and load output increase resulting from the use of an 8000 psi system, a conventional linear actuator might still be useable. To verify this a study was made to determine the viability of attempting to install a linear actuator to perform the aileron actuation function. This function was selected because it represented one of the tightest installation envelopes (See Figure 13 and paragraph 2.3.1.6) on the aircraft.

A COLUMN TWO IS NOT

After examining several approaches, such as remote located actuators working through a series of bellcranks to power the surface, an approach involving direct actuation was settled upon. This approach is shown in Figure 46 and 47. In order to fit the actuator into the wing it was necessary to make each actuator as flat as possible. To do this each actuator was designed to consist of three balanced pistons in parallel. This helped to reduce the rod and piston sizes to a point where a flat pancake like housing could be created which would fit inside the wing mold line between the rear spar and hingeline.



Figure 46. Aileron Actuator Instl.



As can be seen in Figure 46 three direct acting (single stage) electrohydraulic servo valves (EHSV) were used for each actuator and two actuators were used for each surface. The reason for the use of 3 EHS valves was two fold; first, a single large direct acting valve, capable of handling the flow requirements of the three parallel pistons in a given actuator, did not fit within the mold line of the wing whereas, three smaller valves would fit; and second, the use of three valves was very compatible with the 4 channel (three control channels plus model channel) fly-by-wire system used on the aircraft. In order to save space and keep weight to a minimum, the actuator housing was designed to transfer the control surface reaction loads generated by the hingeline to the rear spar.

Figure 47 shows a crossection view of one of the actuators. It shows that, in spite of the heroic measures taken to keep all elements of the actuator within the wing mold line and within the fore and aft envelope, it none-theless violated both. One end of the balanced piston penetrated the rear spar, which was undesireable and constituted a weight penalty, and the hingeline was displaced 5/8 inch below the lower wing mold line. This meant that at least 4 fairings 1-1/2 inches deep by 6 inches long by 1-1/2 inches wide would have been required to cover the hinge mechanism. Even though the drag induced by these fairings might have been considered acceptable, the design was unacceptable for other reasons. The performance data shown on Figure 47 shows that the actuator would meet hinge moment and rate goals, however, it shows that the actuator was deficient in resolution. In order to meet the .050 in/deg resolution requirement the hingeline pivot ann would have had to have been increased from the 1.517 inches shown to 2.92 inches. This would have doubled the size of the fairings, doubled the stroke and, through doubling the stroke, would have increased the length of the actuator by several inches thus wiping out the rear spar. This was considered entirely unacceptable. After failure to meet the aileron envelope and performance requirements this same general approach was cursorily examined for application to the midspan and inboard flaps. Here too it could not meet requirements and further consideration. of this approach was dropped.

At or about the time this decision was being reached the data discussed in paragraph 4.1.5.1 was received from Airesearch and it became apparent that the best possible solution for most of the control surfaces was to use a mechanical power hinge and drive it with a motor (electrical motor for the electrical system and hydraulic motor for the hydraulic system). From this was formulated the general ground rule (see paragraph 2.4.5) which was made a basic part of the program, and which stated in effect that, where the hydraulic system (aircraft II) uses motors, the electrical system (aircraft I) shall use motors, and vice versa, and also stated that, where the hydraulic system uses linear actuators, the electrical system shall use the electrical verision of a linear actuator- the ballscrew.

The ground rule of paragraph 2.4.5 evolved gradually as the program progressed. Epicyclic geartrain (power hinge type) actuators were looked at as a possible means for actuating the landing gear and landing gear doors on aircraft I. However, after extensive consideration of the difference in load paths between a power hinge, which reacts its loads in the immediate area of the trunion or hinge line, and a typical linear hydraulic actuator, which reacts its loads to remote structure, it was decided that the errors in assessing weight impact which could arise from incorrect assumptions about structural load paths, and the nature of the structure to which either type of actuator attached, could far outweigh any inaccuracies or unfair weight penalties which arose from adopting the ground rules of paragraph 2.4.5. Examination of other actuation functions, such as the canard lead to the same general conclusion and when it became apparent that 11 other flight control functions would be power hinge operated whether, hydraulic or electric, the ground rules of paragraph 2.4.5 were adopted.

As indicated above the canard actuator was examined in some detail since it was considered representative of those applications where a linear hydraulic actuator could be used advantageously. Figure 48 shows the canard actuator as it finally evolved. It can be seen immediately that the stroke was reduced from that shown in Table 3 and also in Table 12. The Table 8 stroke was 15.36 in. and that given in Figure 48 for the final hydraulic system actuator design, was 5.75 in. or slightly more than 1/3 of the original requirement. The reason for this change was two fold; first the 24 in, installed (retracted) length assigned to this actuation function on the aircraft would not accommodate a tandem cylinder design employing a stroke of 15.36 in. and second, even if it had, the resulting cylinder (particularly in an 8000 psi configuration) would have been so long and slender that it would have been unstable as a column in compression. Therefore the actuator was relocated to a point closer to the pivot point of the canard surface such that a 5.75 in. stroke with a 6400 lb stall capability and a 5,333 lb design operating load capability would meet the canard surface's requirements. The output design power requirements of course, remained unchanged at 0.17 KW. Although the electric version of the actuator (see Figure 31) was not restricted by the tandem actuator requirement of the hydraulic version and, therefore, could meet the original retracted length and stroke requirement, it too was dangerously close to column instability and could profit from meeting the new shorter installed length. Therefore, it was assumed, for the purposes of aircraft I definition, that the installed length of the ball screw actuator was 19.25 in. Instead of the 24 in. shown in Figure 31 and that, in common with it's aircraft II hydraulic counterpart, it's stroke was 5.75 in. and its stall load was 6400 lb. The weight of the ball screw unit, nowever, remained essentially unchanged at 15 lb since the weight increase associated with the increased load offset the weight decreased associated with decreased installed length.

It is interesting to note that the ball screw unit with its two motors and torque summing gearing plus brakes actually weighed less than the 8000 psi hydraulic actuator. This resulted from the fact that the hydraulic version of the canard actuator, shown in Figure 47, included two EHSV valves which were the power switching and control devices equivalent to an inverter in the electrical approach. When the weight of the two inverters required for the electrical approach (10 lb each per Table 21), were included the total weight of the electrical actuation function became 39 lbs versus 21.25 lbs for the hydraulic.



Figure 48. Canard Actuator Design

4.2.1.7 Actuator Sizing - The basic characteristics of the hydraulic linear actuators used on aircraft II are listed in Table 30. All of these actuators, with the exception of the canard actuator, performed utility functions. Three of the actuators (thrust vector vane, nose gear, and canard) were tandem unbalanced, and the rest were single unbalanced. All the actuators were checked for column stability and found to have adequate margin. The retract lengths were verified as being adequate for the alternate ball screw type electromechanical actuators used on aircraft I. The 3000 psi weight was determined based upon swept volume and the data given in Reference 22. This weight represented an actuator which had 3/8 of the required power output (i.e. the ratio of 3000 psi to 8000 psi). At the same swept volume, but operating at 8000 psi and hence 100% of required power output, it was assumed that the actuator would weigh 8/3 times the 3000 psi actuator's weight times 83% giving a weight correction factor of 2.21 in going from 3000 psi to 8000 psi. The 83%, used in determining the correction factor, arose from the consistent 15 to 17% weight reduction reported in Reference 7. These two analyses represented two independent and rather detailed design analyses for substituting various 8000 psi actuators for 3000 psi actuators in two aricraft (F-14 and A-7). In each instance these actuators performed the same actuation function at the same installed length.

A plot from the actuator weight data presented in Reference 22 was made and is shown as the solid (3000 psi) lines in Figure 49. The solid lines nes represented a mean value of the scatterband of weight values presented in Reference 22, for a tandem unbalanced utility type actuator (lower solid line). The upper and lower dashed lines represent corresponding data for 3000 psi actuators of the same swept volume and were determined by applying the 2.21 correction factor, previously mentioned, to the 3000 psi data. The validity of the resulting curves was further verified by comparing them to extensive in-house data on 4000 psi equipment. In each case the weight data presented in Figure 49 represents that of the complete (dry) actuator ready for installation. For utility actuators this was considered to include typical plumbing and support brackets mounted on the actuator prior to installation and for flight control actuators it was considered to include, not only plumbing and brackets, b the direct acting electro hydraulic servo valve (EHSV) as well. The weight of an actuator filled with fluid was considered to be 110% of the dry weight for 3000 psi actuators and 102% for 8000 psi units.

4.2.1.8 Accumulator Sizing - It should be noted in Figure 45 that an APU/EPU start accumulator was required for aircraft II and it could further be seen in Figure 42 that this accumulator was plumbed into system #2 and was the only accumulator used on the aircraft. Since accumulators tend to have a "double barreled" impact on the system in which they are used by virtue of the fact that, not only is their size and weight directly additive to the system, but they have a parallel and proportional size and weight impact on the reservoir servicing that system. This arises from the fact that the reservoir must be increased in size to accept the fluid discharged from the accumulator and must be further increased to provide additional

ACTUATORS	
LINEAR	
HN/DRAUL/IC	
30.	
TABLE	

•••

a di 191

	JROKE	RETRACT	DESIGN	ROD(IN)	BORE (IN.)	NET	EXCH.	SWEPT	3000PS!	CORR.	BOODPSI
ACTUATOR	(IN)	(IN)	(LBS.)	AREA(IN?)	AREA(N?)	ARCA (IN ² )		(i.v.)	WT.(LB)	0	WT.(LB)
EXTERNAL	10.00	14.50	8600 (E)	0.625 (.307)	(1.188 (1.108)(E)	0.801(E)	3.07	11.08	4.7	2.21	10.39
THRUST (1) VECTOR VANE	5.20	26.00	75,314 (E)(R)	(2.074) (2.074) (2.074)	3.938 (12.120) 3.875 (11.793)	11.793(E) 9.719(R) 10.106(E) 9.419(R)	14.36	113.87	57.0	=	125.97
NOSE (T) GEAR	5.00	16.75	14, <i>9</i> 00 (E)	1.250 (1.227) 1.000 (0.785)	(1.875 (2.761) (1.995) (1.995)	1.995 (E) 1.976 (E) 1.976 (E) 1.489 (R)	6.14	19.85	J6. J	3	35.58
MAIN GEAR	6.00	13.13	24,000 (R)	1.250 (1.227)	2.375 (4.430)	3.203(R)	7.36	26.58	<b>B</b> .9	:	19.67
RIGHT RAM AIR SCOUP	2.52	5.33	4,100 (R)	0.437 (150)	0.938 (.691)	0.541(R)	0.38	1.74	1.2	:	2.65
LEFT RAM AIR SCOOP	1.00	2.88	1,200 (E) (R)	0.375 (011.)	0.625 (.306)	(x)961.0	0.11	0.31	0.35	=	0.77
EXH. DOOR EQUIP. BAY	2.66	5.06	1,300 (R)	0.375 (011.)	0.625 (.306)	0.196m	0,29	0.81	0.72	2.21	1.59
CANARD		EE FIG	INRE								
(R)= RETRACT (E)= EXTEND (T)= TANDEM (C)	INBALAN	CED AC	TUATOR	0 BASI © Corr	ED ON SW	VEPT VO	= BOOO	(REFERE /3000 ×	NCE 27) 83 % = 2		



Figure 49. Hydraulic Actuator Weight

volume and/or make up fluid to offset the thermal expansion and contraction of the accumulator's "charged" volume of fluid. If the accumulator involved was relatively large, this could have a significant and serious negative impact on overall system size. For this reason the elements leading to accurate sizing of the accumulator were looked at in considerable detail.

The total energy necessary to start the aircraft II APU/EPU was judged to be about the same as that necessary to start the APU for the F-18 aircraft. Therefore the F-18 aircraft was used as a guide. Tabulated below is the basic data relative to the F-18 APU start motor and its accumulator. The tabulation also includes the comparable known data for aircraft II as well as the items to be determined (TBD) in subsequent calculations:

	F-18	Aicraft II
Fluid	MIL-H-83282	MIL-H-83282
System Pressure (P)	3000 psi	8000 psi
Accum. Precharge Press/Temp $(P_{GN})/(T_N)$	1950 psig/75°F	(TBD)/75°F
Accum. Oil Volume Max (VM)	143 IN ³	(TBD)
Accum. Oil Volume (VN)	(TBD)	(TBD)
Nitrogen Gas Volume (V _{GN} )	290 IN ³	(TBD)
Motor Displacement (△)	0.364 IN ³	(TBD)
Speed (S)	14,000 RPM	14,000 RPM
Flow (Q)	22.06 GPM	(TBD)
Torque (T)	174 IN-LB	174 IN-LB
Power (HP)	(1BD)	(TBD)

Aircraft II APU Motor Displacement

_ ___

مستحرب بواجه متهومين والداري والد

$$P_1$$
  
 $P_2$  $X$   
 $P_2$  $\Delta_1 = \Delta_2$ WHERE: $P_1$   
=Rated absolute pressure (PSIA)  
F-18 motor $\Delta_2$ = $\frac{3014.7}{8014.7}$   
PSIA $X$  $0.365$  $IN^3/REV$  $P_2$   
Aircraft II motor $\Delta_2$ = $0.137$  $IN^3/REV$  $\Delta_1$   
=Displacement IN^3/REV

Aircraft II APU Motor Flow

$\frac{\Delta 2 \times S 2}{231} = Q_2$	WHERE: S 2 = Speed (RPM) Aircraft II motor
0.137 IN ³ /REV 14.000 RPM	$Q_2 = Flow GPM Aircraft II motor$
$231 \text{ IN}^3/\text{GAL x } 0.85 = Q_2$	2 = Motor Efficiency = 0.85

## Aircraft II APU Motor Power

 $\frac{Q_2 \times P_2}{1714} = HP_2$ WHERE:  $HP_2 = Max.$  instantaneous power  $(h_p)$ Aircraft II motor  $\frac{HP_2 = 9.77 \text{ GPM X 8014.7 PSIA}}{1714}$   $HP_1 = Max.$  instantaneous power  $(h_p)$ F-18 motor  $HP_1 = \frac{22.06 \text{ GPM X 3014.7 PSIA}}{1714 \times 0.85}$ 

 $HP_1 = 45.64 h_p$ 

Both accumulators are required to deliver the same energy to their respective APU start motors

Consider the F-18 accumulator:



VGc is obtained from:

PGN VGN = PGC VGr

WHERE:  $P_{GN}$  = Precharged accumulator pressure (PSIA)

- $V_{GC} = \frac{1964.7 \text{ PSIA X } 290 \text{ IN}^3}{3014.7 \text{ PSIA}}$   $P_{GC} = Charged accumulator pressure (PSIA)$
- $V_N = V_{G_N} V_{G_C} = 290 \text{ IN}^3 189 \text{ IN}^3$ = 101 IN³ (at 75°F)

= 188.99 IN³ use 189 IN³

To determine the energy available from the F-18 accumulator consider the gas expansion ( $V_{C}$ ) as an adiabatic process because the accumulator will be depleted within 1.75 seconds during a normal APU start. The adiabatic relationship is:

 $P_{GN_{e}} = \frac{(3014.7)(189)^{1.4}}{(290)^{1.4}}$  WHERE: n = 1.4  $P_{GN_{e}} = \frac{(3014.7)(189)^{1.4}}{(290)^{1.4}}$   $P_{GN_{e}} = \frac{P_{GN_{e}}}{discharge to V_{GN} volume}$ 

= 1655.52 PSIA

Therefore from the relationship:

$$\frac{T_{GN_{e}}}{T_{GC}} \approx \left( \frac{V_{GC}}{V_{GN}} \right)^{n-1}$$
WHERE:  $T_{GC} \approx Absolute gas temperatureof charged accumulator(°R)$ 

$$T_{GN_e} = (460 + 75) \frac{189}{290} \quad T_{GN_e}$$
  
= 450.8 °_R (-9.2°F)

- 0

 Absolute gas temperature of adiabatically discharged accumulator (°R)

and the second second showing the

Both the aircraft II and F-18 accumulators are required to deliver the same amount of useable energy. The work of expansion is described by the formula:

WORK = 
$$\int_{1}^{2} \frac{PdV = P_2 V_2 - P_1 V_1}{1.0 - n} = \frac{P_{GNe}V_{GN} - P_{GC} V_{GC}}{1.0 - n}$$

For the adiabatic process, (see above) at or below 3000 psi, n approximates 1.4. Therefore work = (1655.43) (290) - (3014.7) (189)

$$1.0 - 1.4$$
  
 $= 480074.7 - 569778.3$   
 $- 0.4$   
 $= 224.259 \text{ in-lbs}$ 

To deliver this amount of energy at pressures high enough to be useable by the 8000 psi hydraulic motor it is necessary that the ratio of charge pressure (Pgc) (8014.7 PSIA) to pressure after adiabatic expansion (PgNe) be equal to, cr higher than, that of the 3000 PSI F-18 accumulator. Therefore: PGNe =  $\frac{1655.43}{3014.7}$  x 8014.7 = 4401.03 PSIA for aircraft II. To accurately compare

the 3000 PSI F-18 accumulator and the aircraft II accumulator, compressibility must be taken into account. Figure 50 plots typical compressibility characteristics for air in the pressure-temperature regime which will be experienced by the accumulator. From the figure it ca., be seen that the following relationship exists:

$$\frac{P_1V_1}{Z_1T_1} = WR = \frac{P_2V_2}{Z_2T_2} \quad \text{Where } R = 639.6 \quad \frac{IN-LB_f}{LB_n \circ R}$$

Therefore if the weight (W) of gas in the 3000 PSI accumulator is:

 $W = \frac{P_1V_1}{R Z_1 T_1} = \frac{3014.7 \times 189}{639.6 \times 1.049 \times 535} = 1.587 \text{ LB}$ 

the charged volume  $(V_2)$  is:

- ..

$$V_2 = \frac{W R Z_2 T_2}{P_2} = \frac{1.582 \times 639.6 \times 1.381 \times 535}{8014.7}$$
$$= 93.59 \text{ IN}^3$$

 $V_2$  presumeably has excessive stored energy by the ratio of the pressures 8014.7 PSIA vs 3014.7 PSIA. Therefore:

$$V_{GC} = V_2 \text{ ratioed} = 93.59 \times \frac{3014.7}{8014.7} = 35.20 \text{ IN}^3$$

 $W = 1.587 \text{ LB } \times \frac{3014.7}{8014.7} = .5975 \text{ LB}$ (8000 PSI) 8014.7



Figure 50. Compression Factor  $(\Xi)$ 

Using the relationship PGC  $V_{GC}^n \neq P_{GNe} V_{GN}^n$  and rearronging:

$$V_{\text{CN}} = \int_{-0.4}^{n} \int_{-0.4}^{P_{\text{GN}e}} \int_{-0.4}^{P_{\text{GN}e}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{4401.03}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{4401.03}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{4401.03}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{-0.4}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{-0.4}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{-0.4}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{-0.4}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{-0.4}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{-0.4}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{-0.4}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{-0.4}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{-0.4}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{-0.4}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{-0.4}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{-0.4}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{-0.4}} \int_{-0.4}^{1.4} \int_{-0.4}^{\frac{8014.7 \times (35.20)^{1.4}}{-0.4}} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0.4}^{1.4} \int_{-0$$

Increase V_{GC} and W by the ratio of required energy to actual energy  $\frac{224,259}{111,049} = 2.02$ 

Then:

.....

$$V_{GC} = 71.12 \text{ IN}^{3}$$

$$W = 3.206 \text{ LB}$$

$$V_{GN} = \frac{1.4}{\sqrt{\frac{8014.7 (71.12) 1.4}{4401.03}}}$$

$$= 109.13 \text{ IN}^{3}$$

$$WORK = \frac{4401.03 \times (109.13) - 8014.7 (71.12)}{-0.4}$$

$$= 224,303 \text{ IN-LB} (\leq \text{the required} \\ 224,259 \text{ IN-LB})$$

$$V_{n} = V_{GN} - V_{GC} = 109.13 - 71.12 = 38.01 \text{ IN}^{3}$$

$$= \text{Fluid expelled}$$

$$P_{GN} = 8000 \text{ PSI accumulator precharge pressure}$$

$$= \frac{P_{GC} V_{GC}}{V_{GN}} = \frac{8014.7 \times 71.12}{109.13}$$

= 5223.18 PSIA

Ņ

The weight of the 8000 PSI precharged APU start accumulator was determined as follows:

WACC = 15 LB (from Figure 51) WAIR = W = 1.207 LB Total weight = 16.21 LBS



Figure 51. Accumulator Weight



4.2.1.9 APU/EPU Sizing - The APU/EPU performance data used for sizing the APU/EPU, as it was used on aircraft II, was based upon computations made on the following pages and upon the summation of this data as shown in Table Most of the column headings shown on Table 31 are self-explanatory 31. except for the columns headed No. 1 through No. 11. These represent power transmission interfaces between components as shown by the coded ballons on Figure 45. The coded number series is generally arranged in reverse order of power flow from source to final output. Thus column No. 11 in Table 31 is the required output of the unit when encountering the various operating conditions listed under the function column. The first two function listings represent operation as an emergency power unit (EPU) above 20,000 ft altitude and the next four listings represent various conditions of operation as an auxiliary power unit (APU) below 20,000 ft. For a more detailed discussion of several of these operating conditions (functions) see paragraph 4.2.1.1. However, several factors having an impact on APU size, were not exhaustively discussed in paragraph 4.2.1.1 and are expanded here. These factors were:

- When operating as an EPU the load compressor (LC in Figure 45) was always unloaded therefore, as an EPU, the unit was required to supply only drag load power for the load compressor and either drag load or duty cycle load power for the emergency pump and generator. The reason the freon compressor was unloaded was the fact that, during an emergency, the ram air scoop doors were open to provide the required cooling.
- 2. When attempting "ground" or "in-flight" engine starts all units drawing power from the AMAD were unloaded (i.e. drag power only) except for the engine PTO.
- 3. During "ground" engine starts the emergency pump and generator were unloaded, however, during "in-flight" engine starts both units were loaded and remained so until the APU/EPU was shutdown in response to a signal that both primary hydraulic and electrical systems were once again on line and functioning properly.
- 4. During "ground" functional checkouts both the two AMADs and the AGB could be powered simultaneously using ground air supplied through the ground connection shown on Figure 45. The AMADs were driven by their respective ATS/M and the AGB was driven by partial arc admission of air to the free turbine of the APU/EPU. The ground air system was sized so that, with the engines decoupled, the system could meet the ground checkout simultaneous duty cycled load requirements of all components mounted on both AMADs and the AGB. However, when power for checkout purposes was supplied by the APU the simultaneous load requirements must be reduced. Only one AMAD, with its mounted components, and an AGB, with its emergency generator de-excited, could be driven on a duty cycled checkout load as seen by those components. Using the APU as a power source, all elements of the electrical system could be checked out at 100% of the duty cycled checkout load as could all elements of the hydraulic system except main landing gear actuation functions. The main landing gear could be operated through a complete checkour cycle only if AMAD #2 was being powered by the APU. Also, with the APU as a power source, only one freen compressor could be operated. However, even though this reduced the ECS cooling capacity by 50%, it was adequate to meet all cooling requirements encountered during checkout.
TABLE 31. APU/APU POWER REQUIRINGNIS DEFINITION

1

.

:

化氯化氯化 化分子 医鼻骨炎 计分子数字 化分子数 化化合物 化合金属

1.....

	APU/ED	a Aviani ha	DPERMING	POWE	R DEL	NER	GH) QE	) AT F	DINT	0000	ED IN	<b>FIGU</b>	RE 4.2	4	
FUNCTION	PPERATINE	ALTINDE	MURATION	-	2	ςΩ	4	ß	و	2	B	6	õ	:-¢	REMARKS
APU/EPU START	GAS GEN.	62,300	< 5 SEC	0	0	0	0	0	0	0	<u>≜</u> 13.48	≙ 0.56	<b>6.49</b>	23.96	1 DRAG POWER COMUTED ON PRGE 199 2 BASED ON IS MAI EMERG- ENCY HOUSEKEEPING 10AD3 ELECT LOND ANAL, REF. &
HIGH ALTITUDE EMERGENCY	GAS GEN	62,300	4 MIN.	Q	0	0	0	O	0	0	 13.46	<u>ک</u> 9.84	A 70.34	3366	A COMPLED AS SHOWN ON PAGE 202 A BASED ON FULL RATE OF HIGHEST DEMAND SURFACE PUD 10% FULL RATE ALL OTHERS
GROUND ENGINE START	GAS TURBINE	5000	40 SEC.	2.6I	4.02	<b>8</b> 6,31	\$ \$	<u>لا</u>	A 179.70	(702) 07.971	ه ه 13359	@ 0.56	<b>∂.</b> 49	(304) 264.51	名」NFLIGHT START POWER = 50% GROUND START RUNER 1405 TO WINDMILL ASSIST 6 ATS/M EFF = 0.75
GROUND FUNCTIONAL CHECKS [APU]	GAS TURBINE	5000	4 HR	48.20	43.64	<b>▲</b> 76.47	A 20.671	র্	210.65	(242) 2:0.45;	280,56	∆ 0.56	<b>A</b> 4820	(403) 35966	A BASED ON 5 SEC CRUISE HOUSEKEEPING LOADS WITH ECS LOADS DELETED ; FROM FLECT. LOAD ANALYSIS REF. 6
IN FLIGHT ENGINE START	GAS TURBINE	20,000	60550	2.61 2.61	4.02	∆ 16.98	A A 2698	\$0.00	<b>A</b> 104.64	(130) 104.64	<u>گ</u> 139.52	<u>ک</u> 9.84	₫ 70.34	(425) 233.73	A LOAD COMPRESSOR (LC) EFF. = 0.75 AMAD AND AGE GEAR- BOX EFF. = 0.94 A FROM FIGURE 2-4
EMERGENCY RETURN (APU)	GAS TURBINE	20,000	1.8 HR	S.6	<b>≜</b> 24.87	A 76.47	\$ 10.59	ৰণ	13010	(236)	173.47	0.56	<b>≜</b> 46.20	(430) 236.41	( ) SEA LEVEL EQUIV- ALENT POWER PER REF - PAGE 12 - PAGE 12 - REOM AMAD

....

ł

4.2.1.9.1 (Cont.)

5. In flight engine start requirements were slightly more than 50% of the ground engine start requirements (see column No. 6 on Table 4.2-4). This resulted from the fact that, even though the type of engines used in the study aircraft were incapable of achieving a windmilling engine start in flight, there was a significant windmill assist, during all flight conditions, which could be used to cut the starting power demanded of the APU.

The computations used in the preparation of Table 31 are as follows:

• •

Drag ⁽¹⁾ Power	at API	J/EPU Gear Box Pads		
Pump Drag	=	Rated Power X 0.065		
D ₁₀	-	130.7 HP X 0.065	=	<u>8.49 HP</u>
Gen. Drag	=	Rated Power X 0.05		
D ₉	=	7 KW X 0.06 .746 HP/KW	=	0.56 HP
Load Comp. Drag _D 8	2	Rated Output Power X	0.064	
0	=	210.65 HP X 0.064	=	<u>13.48 HP</u>
Drag ⁽¹⁾ Power	at AM/	AD Pads		
Pump Drag		$D_{10} = D_3 = \underline{8.49 \text{ HP}}$		
Gen. Drag	z	Rated Power X 0.06		
D ₂	=	50 KW X 0.06 .746 HP/KW	=	4.02 HP
Freon Comp. Drag	=	Rated Output Power X	0.064	
D ₁	=	0.56 KW X 0.064 .746 HP/KW	=	2.61 HP

Air Turbine = Rated Output Power X 0.064 Start Motor Drag  $D_4 = \frac{152.98 \times 0.064}{.746 \text{ HP/KW}} = \frac{13.12 \text{ HP}}{.13.12 \text{ HP}}$ 

(1) Power loss with units unloaded or de-excited (see Figure 45)

4.2.1.9.1 APU/EPU Power Requirement for Engine Starting

(Gro	ound	- 5000 Ft. Altitude)
P5	=	AMAD power input to engine (HP) (Code 5 Figure 45)
	*	89.5  kW/.746  kW/HP = 119.97  HP
P4	#	Air Turbine Start Motor (ATS/M) input to AMAD (HP) (Code 4 Figure 45)
	=	$2 D_{10} + D_9 + D_8 + P_5 = 16.98 + 4.02 + 2.61 + 119.97$
		AMAD Eff94
	=	<u>152.74 HP</u>
P8	=	P4
·		ATS/ _M Eff. x LC Eff. = $\frac{152.74}{0.85 \times 0.75}$ = $\frac{239.59 \text{ HP}}{239.59 \text{ HP}}$
P ₁₁	2	APU/EPU Starting Power Requirement
	2	$\frac{P_8 + D_9 + D_{10}}{AGB Eff.} = \frac{239.59 + 0.56 + 8.49}{0.94}$
	=	264.51 HP

(2) From page 3-22 reference 8

# 4.2.1.9.2 Ground Service Power at AMAD Pads

ļ

(at pump ports)	2	Based on full rate on highest demand surface plus 10% of full rate on all others (2 pumps)
Po	2	(1.71-35) 0.1 + 35 = 48.6 KW = 65 HP
Pump Input = 1 (at AMAD pad)	°3 =	$P_0/Pump Eff. = 65/0.85 = 76.47 HP$
Freon Compressor Output (at compre Ports)	essor	$P_{10} = 30.56 \text{ KW}/.746 \text{ HP}/\text{KW} = 40.97 \text{ HP}$
Freon Compressor Input (at AMAD pa	ad)	40.97/Comp. Eff. ≠ 40.97/0.85
1	P1 =	48.20 HP
Generator Output (at Gen. Terminal	n Ls)	Based on 'warm up and take off- 5 sec. load" from load analysis (5)
1	P ₂₀ =	27.67  KW/.746  HP/KW = 37.09  HP
Generator Input	2	$P_{20}/0.85 = 37.09/0.85$
I	P ₂ =	43.64 IIP
AMAD Input	a	$\frac{P_1 + P_2 + P_3}{AMAD Eff.} = \frac{48.20 + 43.64 + 76.47}{0.94}$
	P4 =	$\frac{168.31}{0.94} = \frac{179.05 \text{ HP}}{100000000000000000000000000000000000$

3 From Table 29

(4) From Figure 10

(5) From Reference 8 with ECS compressor loads deleted

4.2.1.9.3 Emergency Power at APU/EPU Gearbox (AGB) Pads

Pump Power = Required output power/pump eff. = 44.0 KW/.746 X .85 = 70.34 HP

① See Table 29

From Table 31 column No. 11 it can be seen that the EPU power delivery requirement was 93.66 HP and that it was based on the high altitude emergency condition. This same power delivery requirement applied to the partial arc air admission when conducting ground checkouts using ground air supplies at altitudes to 5000 ft. Column 11 also shows that the APU was rated at 430 HP sea level static. This was ratioed from the 236.41 HP it must deliver at 20,000 ft. Based on these ratings the APU/EPU unit weighed 399 lbs. This weight included the hydraulic start motor, permanent magent generator,  $GN_2$  pressurized LOX tank, and the required LOX, the gas generator, and the accessory gearbox shown as "SM", "PMG", "LOX", "GG", and "AGB", respectively on Figure 45.

Other system components such as the load compressor (LC) and the air turbine start motor (ATS/M) were also sized from the data given in Table 31. Column 7 provided the rated output power (sea level static) requirement of the load compressor. This proved to be 242 HP based on the ground functional checkout requirement at a 5000 ft. altitude air base. Column 4 provided the rated output power requirement of the ATS/M. This proved to be 179 HP also based on the ground functional checkout requirement. Based on these load ratings and a 1990 + time frame, the weight of the LC was determined as 75 LB and that of the ATS/M as 32 LB. 4.2.1.10 <u>Plumbing System Sizing</u> - The basic characteristics of the plumbing system are shown in Tables 32 and 33. Most of Table 32 is selfexplanatory except for the three columns labelled "Rating at  $-20^{\circ}$ F". These columns represent the flow,  $\Delta$ P, and velocity ratings, used as a guide for sizing both the trunk and subsystem lines in the aircraft. They were based on  $-20^{\circ}$ F instead of the  $-40^{\circ}$ F temperature, at which the landing gear system must meet full performance, or the  $+20^{\circ}$ F temperature at which all flight control subsystems must also meet full performance. The reason for this was the fact that, designing for  $-20^{\circ}$ F, represented the optimum compromise between the excessively high fluid flow velocities which would result from rating the lines at  $+20^{\circ}$ F, wherein rated flow velocities would approach 90 ft/sec, and the excessively large lines which would result from rating the lines at  $-40^{\circ}$ F. It will be noted from Table 32 that the mean flow velocity in the line never exceeded 50 ft/sec and, in the smaller lines where most of the valves were located, was considerably less.

searchest the second street, where the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second

Table 34 shows the main landing gear system designed at -20°F and assumed the tubing's design rated flow existed in each tube in the system (i.e., branch and trunkline pressure, return and suction tubing). Table 34 also assumed that, since increased pressure represents increased pressure drop, the mean pressure in the system was 4000 psi (i.e., 1/2 of 8000 psi). It can be seen in Table 34 that the total pressure drop in the system, with the line diameters and line lengths indicated, was 2563 psi. This is less than 1/3 of the available system pressure (2666 psi) and was very satisfactory for full performance of the landing gear. However, the branch lines of the system must actually operate at -40°F and will actually flow at some flow less than design rated flow. The actual flow required for the main landing was derived from the "power at operating load-rate" column in Figure 8. The derivation assumed a very conservative effective pressure across the actuator of 4667 psi (see pump sizing paragraph 4.2.1.5) and used the 3.63 HP found in Figure 8. From this the required flow rate per main landing gear retract actuator was found to be 1.33 GPM. This determination used the following formula:

$$GPM = \frac{HP \times 1714}{PSI}$$

Where:

HP = 3.63PSI = 4667

This flow rate was considerably less than design flow (i.e., 5.5 GPM for 3/8 inch diameter return lines and 5.0 GPM for 3/8 inch pressure lines per Table 32). Therefore, even though the pressure drop at a given flow at  $-40^{\circ}$ F increased by roughly a factor of three, the actual pressure drop in the system was considerably less than that shown in Table 34 because of the reduced actual flow. Using an actual flow of 1.33 GPM at  $-40^{\circ}$ F, in the branch lines and design rated flow at  $+20^{\circ}$ F in the trunk lines, the actual pressure

TABLE 32. 3AL-2.5V TITANIUM TUBING DATA

(ENI) TON 8.58 130.2 12.81 108.4 120.4 113.9 133.0 139.9 35.3 TOTAL 2.64|145.6 1126.2 VELOCITYLENGTH WEIGHT FLUID 16.29|147.7 10L **ð**1.1 8.78 8.83 16.65 4.85 101.53 DE VELOCI I LENVILLE 4.07 4.71 LINE LINE 887 114 58 362 202 44 ١ 121 63 0 2 2 20.05 33.67 49.65 47.89 28.48 10.38 23.64 31.09 49.01 49.11 48.61 -20°F TOTAL 4 S 2 S 2 S 43 22 22 6 4 5 0 5 20 ഗ COMBINED RATING AT FLOW 280 550 0 0 0 0 0 0 0 60.0 6PM 39.0 30.0 50.0 75.0 60.0 17.0 0 10 10 -FOR TITANI!'M WEIGHT LB/FT. .128 102. 202. 264 .045 .287 555 .362 .47 FLUID LB/FT. LB/FT. .075 .135 FOR TUBE .033 .044 601. 212. .159 .220 298 .222 .407 ... C 0.0309 LB/IN3 LB/iN3 LBYFT. FLUID 090. .093 .048 .033 .182 .075 .135 601. .012 .251 191. 0,168 FLUID VOLUME IN SFT. L 1.944 1.548 3.024 2.412 4.788 4.356 5.892 1.056 4.560 3.528 .876 6.192 8112 408 .034 AREA (IN?) .085 011. 202 WALL .022 .054 610. 016 .148 <u>109</u> 105 190. Zo Zo BASED BASED AREA .363 .294 .399 .088 RET .162 PRESS .129 .380 FUNG-FLOW ,033 .516 .252 .676 (2NI PRE54.073 .201 .491 RET. ¢ TION PRESS PRESS SUCT. PRESS PRESS. RET. RET. SUCT RET. SUCT ତ୍ର .069 .047 .029 059 .023 .035 1.042 **WALL** .020 .035 .036 .023 .032 .027 180. SIZE (IN.) DIA. . 19.00 19.00 24 800 200 3/4 70 ° 3/4 4/4

TABLE 33. FITTING DATA

;

į

-					_				-
10 IAL LINE	SUPPORT WEIGHT LEAV	3) 1.18	1.54	1.52	1.79	.47	.36	.35	
TOTAL FITTING	WEIGHT LB/AV	7.25	8.72	9.02	10.25	21.2	2.36	2.37	46.63
NO. UF	FITTINGS REQUAN	2) 145	96	55	41	80	5	4	0 0 7
6)	B000 PSI FITING	.050	.065	630.	.164	.250	.340	.593	TOTAL-
ihT (LE	BURST PRESS. RATIO	× 24,000 =	•					× 24.000 =	
WEIG	4000P51 FITTING	ر) روعع	,043	.059	601.	.167	.227	.395	
Ш	TUBE DIA(IN)	74	3/8	2/1	5/8	3/4	8/L	_	
ZIS	DASH NO.	- 4	- 6	81	-10	- 12	-14	-16	

- BASED ON TITANIUM ELBOW FITTING AS REPRESENTATIVE OF AVERAGE FITTING WEIGHT I.E., UNIONS AND BULKHEAD UNIONS SLIGHTLY LESS AND TEES AND CROSSES SLIGHTLY MORE. Θ
- BASED ON JNE FITTING FOR EVERY 2.5 FT. OF LINE (REF. TABLE 32)  $\odot$
- (3) BASED ON 5% OF COMBINED LINE AND FITTING WEIGHT

TABLE 34. LANDING GEAP △P AT RATED FLOW AND -20°

<u>ئ</u>لت

F I

Adda a' suffer and an a su

. . . . . . . . . . .

officially the s

FITTINGSYS. AP AP AT RATED (PSI) FLOW (PSI) 71.0 787.2 1007.2 177.8 302.0 TOTAL 218.1 | FITTING TOTAL 34.8 497.3 97.2 132.0 26.0 92.1 | | ł (PS/FTG) 34.4 23.0 12.3 12.3 80.0 44.0 17.4 40.0 33.0 15.0 30.7 26.4 Q 10 10 10 10 ΔP 3 FTGS. ЗР ЗР 6 Į 2 4 60 N TOTAL LINE AP (PSL) 2066 143 672 910 40 20 126 I LINE AP AT LENGTHRATED SYSTEM TOTALS+ 202 64 65 45 52 29 29 51 21 25 5 44 ف S ł l ł ----5 ł FUNC- RATED TION FLOW (GPM) 60.0 50.0 60.0 60.0 28.0 55.0 45.0 12:0 39.0 30.0 υ Ο Ο RET. PRESS. SUCT. RET. PRESS. PRESS. RET. PRESS RET. PRESS. SUCT. RET. PRESS. Suct. RET. .029 .027 .032 .042 .061 MALL (IN) .023 .035 .023 .036 020. 690. **FUBE SIZE** (in) (in) 22 50 0 0 0 200 880 22 DASH ÖZ 00 222 - 14 44 ര ര -16 0 00 1 1 1 1

drop was found to be 1281.9 psi. From this it would appear that the branch lines could have been reduced to 1/4 inch diameter even though they would have been operating slightly overrated. However, in so doing, the system pressure drop would have jumped by a factor of nearly six and would have become 7345 psi which was unacceptable.

Surge magnitude is a direct function of fluid flow velocity and is associated with fast closing valves. In early aircraft hydraulic system designs, flow velocities were held to 15 ft/sec to control surge induced pressure pulses. In recent years this has been increased to 25 ft/sec. It will be noticed in Table 32 that rated flows in small sized tubes (1/4 in. and 3/8 in. diameters) were held below 25 ft/sec. The larger sized lines, which do not contain fast closing valves and which are little affected by fast closing valve action in the small sized lines branching off of them, were allowed to approach 50 ft/sec fluid flow velocity. In this way the somewhat conflicting requirements of good low temperature operation, high normal temperature transmission efficiency, and low surge pressure generation potential, have been met in an optimum manner for aircraft II.

From Tables 32 and 33 it can be seen that the total weight of the nydiaulic plumbing system, complete with fluid, fittings and line supports, is 151.43 lb. Also from Table 32, the fluid volume in the tubing is 1126. in³ to which was added 89.6 in³ for the fluid volume contained in the fittings to give a total of 1215.8 in³.

4,2,1.11 Motor Sizing - The motors were sized using existing 3000 PSI motors as a base point. It was assumed that the output section of the motor (i.e., shoe bearing plate, thrust bearing, output shaft, etc.) would be essentially unchanged, whether the motor was a 3000 PSI unit or an 8000 PSI unit, except those changes which would result from being able to operate at slightly ) higher speed. It was felt that this would be true because the output ( would be the same for either the 3000 or 8000 PSI unit. The input sec-DCtion including porting valve plate, block, block bearing and piston diameters were all reduced in size. The block diameter was not reduced as much as might > thought at first giance, however, because the piston shoe circle, and hence the piston bore circle, would reduce only slightly (i.e. as a function of t' ~ slight speed increase). None-the-less, the small block diameter reduction due to the piston diameter decrease allowed for a slightly reduced size for the block bearing and hence justified the slight speed increase already mentioned. Based on these considerations it was projected that any advanced (1990 + time frame) 8000 PSI motor would weigh approximately 85% of its 3000 PSI counterpart (i.e. same power output) and would be rated at a 17% higher speed. Table 35 is a tabularization of data comparing existing 3000 PSI motors to projected 8000 PSI units and Figure 52 plots the 8000 PSI motor data from the table. It can be seen in Figure 52 that, although the larger sized motors showed an almost constant weight/power ratio (i.e. 0.136 lb/output HP). the smaller sized units tended to have a higher ratio. This resulted from

TABLE 35. MOTOR CHARCTERISTICS AND COMPARISONS

	CONT-	FLUID VOL	1.18	2.59	5.18	10.36	17.76	27.16	35.54	48.40
	MOTOR	(ସେ)	1.20	1.73	2.63	4.09	5.98	8.15	9.80	12.94
DTOR	DISPL- ACEMENT	IN ³ KEV	0.016	0.035	0.070	0.140	0.240	0.367	0.479	0.654
PS1 M	SPEED	(મ્પ્સ)	16,000	14,095	11,746	9,866	B,457	7,517	6,930	6,695
8000	THEO.	FLOW	1.08	2.16	3.57	6. <u>0</u> 0	8.77	11.95	14.36	18.96
	ACTUAL	FLOW	1.16	2.32	3.84	6.45	9.43	12.85	15.44	20.40
	POWER	(GH)	5.42	10.83	1.92	30.11	44.00	59.96	72.05	95.20
$\geq$	POWER	(dH)	4.60	9.21	15.23	25.59	37.40	50.97	61.28	80.92
ļ	POWER	© (dy	ł	10.83	17.92	30.11	44.00	59.96	22.09	95.20
DTOR	ACTUAL	FLOW (Cervi)		6.19	10.24	17.20	25.14	34.26	41,19	54.39
M ISOC	THEO.	FLOW (GPM)	•	5.76	9.52	16.00	23.38	31.86	38.31	50.58
300	SPEED	(RPM)		12,400	10,000	8,400	7,200	6,400	5,900	5,700
	MOTOR	(LBS)		2.00	3.11	5.02	7.26	9.80	11.56	14,45
	DISPL-	IN3/ NI		0.11	0.22	0.44	0.75	1.15	1.50	2.05

- U VOLUMETRIC EFFECIENCY = 0.93
  OVERALL EFFICIENCY = 0.85
  HIGH CASE PRESSURE MOTORS (3000FSI)
- () NO EQUIVALENT 3000 PSI MOTOR FRAME SIZE WHICH IS THIS SMALL (B) BASED ON .136 "PHP EXCEPT FOR THE 3 SMALL EST SIZED MOTORS (SEE FIGURE S2 )



Figure 52. Motor Sizing

ì

the fact that the motor tended to become of "watchwork" size and was impractical to build as small as it could theoretically be. Added to this was the fact that the porting could not reduce proportionately since port sizes for this airplane are not allowed to be less than -4.

4.2.1.12 <u>Reservoir Sizing</u> - The reservoir was sized using MIL-R-8931 paragraph 3.4.1 as a guide. The sizing process followed steps (a) thru (h) shown in paragraph 3.4.1 of the MIL Specification and is listed in approximate order as follows:

(a) An amount of fluid sufficient to ensure that the hydraulic pump inlet pressurization and satisfactory circulation is maintained.

System #1	System #2	System #3
23 IN ³	20 IN ³	10 IN ³

(b) A fluid volume equivalent to 100 percent of the possible net depletion caused by actuator volumetric changes during operation. This data is tabularized in Table 36 and its summation is listed here as follows:

System #1	System #2	System #3
32.72	46,54	0.0

(c) A fluid volume equivalent to 100 percent of the reservoir fluid volumetric change caused by charging all accumulators.

System #1	System #2	System #3
0.0	*38.01	0.0

* See Paragraph 4.2.1.8

- (d) A fluid volume equivalent to 130 percent of the volumetric capacity of the largest quantity-measuring type of hydraulic fuse in the system.
  - No fuses used.
- (e) A fluid volume equivalent to the maximum thermal contraction which was expected to occure when the entire fluid content of a recirculating system was exposed to a temperature decrease from 70°F down to -40°F.

		#1	SYSTEM	
		<b>π L</b>	10 <b>Z</b>	#3
SUMP VOLUME	(a)	23.00 IN ³	20.00 IN ³	10.00 IN ³
ACTUATOR VOL. CHANGES	(b)	32.72 IN ³	46.54 IN ³	- <u></u>
ACCUMULATOR VOLUME	(c)		38.01 IN ³	<u> </u>

VOLUMES
EXCLANCE
(INV
THEMS
ACTUATOR
36.
TABLE

	SV5TF			575	# Σ 1	6	<b>3YSTEN</b>	1 # 3
			10.11			1111		FYCH AND
ACTUATOR	NO.0F	M S I S I S I	E (INJ)	NO. 01	N N N N N N		5 52	
	ACTUATORS			ACTUATORS			ACTUATORS	SWEPT
	REQ/SYS	EXCH.	SWEPT	REQ/SYS	EXCH.	SWEPT	REQ/SYS	YOLUME
CANDO	2×15 M	7.06	9.89	2 × 1/5 (T)	7.06	9.89	1	
THOUST VECTOR VANE	5×12 (E)	14.36	113.87	2×1/2(T)	14.36	113.87	1	
EXTERNAL FLAD		6.14	22.18	2	6.14	22.18	ł	
NOSE GEAR	1×12(T)	3.93	9.93	1 × ½(T)	3.07	9.93	Į	
NO CEARDOWN LOCK	_	04	.12	-	.04	. 12	1	
NOSE GEAR UP LOCK		40	.12		<b>6</b> 0.	.12	1	
NOSE GEAR DOOR	-	.33	1.00	-	.33	1.00	I	
NOSE GEAR DOOR LOCK		Ş	.12	-	.04	.12	1	- C
MAIN GEAR	ł	1	İ	2	14.72	53.16	I	) -
MAIN GEAR DOWN LOCK	I	l		2	.07	12.	1	
MAIN GEAR UP LOCK	I		I	~	.07	<u>5</u>	1	
MAIN GEAR DOOR	1	١	I	~	.53	1.60	ł	
MAIN GEAR DOOR LOCK	1	1	l	~	.07	.2	1	
EXHAUST DOOR		.29	õ	1		1	I	
RIGHT RAM AIR SCOOP	_	.38	1.74	!			1	
LEFT RAM AIR SCOOP		. 11	.31	1	1	1	1	-
		32.72	160.09		46.54	212.62		0
(T) - TANDEM ACTUATO	R							

FUSI	e volume	(d)						
FLUI SYST CONS	ID RECIRCULATING TEM VOLUME SISTING OF:	(e)						
	ACTUATOR SWEPT VOL.	160.09	9 IN ³	212.62 IN ³				
2,	COMPONENT VOLUMES	320.3	5 IN ³	304.03 IN ³	$127.77 \text{ IN}^3$			
3.)	TUBING VOLUME	489.5	) IN ³	470.50 IN ³	166.23 IN ³			
<b>(</b> 4 <b>.</b> )	FITTING VOLUME	40.20	<u>) IN³</u>	<u>36.20</u> IN ³	<u>13.24</u> IN ³			
Ŭ	TOTAL	1065.8	6 IN ³	1127.90	317.24			
(1.)	See Table 36.							
2.	See Table 37 which is a weights and contained vo other than actuators, to	master equi olumes of al ubing and fi	pment li 1 compone ttings.	ist (MEL) which tabulate ents making up the Hydra	es the mulic System			
3.	See Table 32							
<u>(4</u> ,	See Table 33							
The coe The	The differential temperature ( $\Delta T$ ) in varying from 70°F to -40°F is 110°F and the coefficient of thermal expansion (e) for MIL-H-B2382 fluid is 4.6 x 10 ⁻⁴ IN ³ /IN ³ /°F. Therefore the differential volume ( $\Delta V$ )							
	V=	$\Delta T \times C \times S$	YSTEM VO	LUNE				
	For SYSTEM #1 V=	110 x .0004	6 x 1065	.86 = 53.93 IN ³				

For SYSTEM #2  $\triangle V = 110 \times .00046 \times 1127.90 = 57.07 \text{ IN}^3$ For SYSTEM #3  $\triangle v = 110 \times .00046 \times 307.24 = 15.55 \text{ IN}^3$ 

f) A fluid volume equivalent to not less than 5% of the entire system fluid volume, including the reservoir, of a recirculating system in order to minimize the frequency of refilling.

		55.99 IN ⁵	59.25 IN ⁵	$16.64 \text{ IN}^3$
	TIMES 5%	.05	.05	.05
Volume based on a) through e) above		1119.79 IN ³	1184.97 IN ³	332.79
		SYSTEM #1	SYSTEM #2	SYSIEM #5

## TABLE 57. MASTLK LQUIPMENT LIST (PAGE 1 OF 2 PAGES)

-----

HYD. SYST.	MASTER EQUIPME	NT CO	MPONENT TITLE	PORT	DRY WT.	FLUID VOL.
NO.	LIST NUME	BER		NO.	(LBS)	(IN ³ )
1	291001	DIS	CONNECT-QUICK, PRESS	-12	3.60	1.50
	<u>1</u> Do a		CONNECT-QUICK, SUCTION	- 16	1. <b>90</b>	3.5C
	IFOI	FIL	TER, RETURN	-14	1020	33.00
	IFO	FIL	TER, PRESSURE	-12	20.41	35.00
	1F0.3	J FIL	TER, CASE DRAIN	-6	3.23	<u>6.9</u> 3
	THO	Γ HE	AT EXCHANGER	-6	3.30	21.20
	_LMO	I_TR	NSDUCER, PRESSURE	-4	0.51	0.03
	IPOI	PU	MP, HYDRAULIC	-10	26.00	6.20
	I POZ	2 PU	MP. HYDRAULIC	-10	26.00	6.20
			·			
┝╌┝╌╴	IRO I	RE	SERVOIR, HYDRAULIC	-14	*	
┠┼━					0 12	· 0/2
	110		VE, CHECK STST TRETUKN	-14	0.42	0.65
<u>├</u>			KESTRICIOR CHELK, SUCI. Q. D	- 10	0.19	0.40
}		2	AVENES PREVIOUS	-14	0.23	<u> </u>
$\vdash$		7	DIMAD, REIVEN FILLER			
- +			IVERBANDO DE LEE RESERV	-10	0.42	ORC
		7	BI FED RESERVOIR	-4		
	110	8	CHECK CASE DRAIN	-6	0.05	0.12
	I IVO	9	CHECK PUMP PRESSURE	-10	0.22	0.34
	IVIC	5	CHECK, PUMP PRESSURE	-10	0.22	0.34
	111		CHECK, PRESSURE Q.D.	-12	a 30	0.52
	LVI	2 VAL	VE, RELIEF, PRESSURE	-12	1.3	1.20
						l
	11/2		VE, CONTROL, N.G. STEERING	-4	0.60	0.80
	1/2	2	CONTROL, ARMAMENT	-10	4.80	0.60
	1 1/2	3	SHUTOFF, ARMAMENT	-10	2.12	0.32
	29 112	4 IVA	VE CHECK ARMAMENT	1-10	0.14	0.40

HYD	MASTER		PORT	DRY	FUID
BYST.	EQUIPMENT	COMPONENT TITLE	DASH	WT.	VOL.
NO.	LIST NUMBER		NO.	(185)	(IN ³ )
1	29125	VALVE, SELECTOR, REFUEL RECEPT.	-4	0.50	0.05
4	1V26	SELECTOR, L.H. RAM SCOOP	-4	0.35	0.04
	1 1 1 2 7	SELECTOR, R.H.RAM SCOOP	-4	0.35	0.04
	1728	CHECK, SYS, #1, NOSE GEAR	-4	0.08	0.04
	_1 <u>v</u> 29_	VALVE, SELECTOR, NOSE GEAR	-4	0.55	0.06
	<u> </u>	VALVE, 2-WAY RESTRICTOR, N.G. UP LOCK	-4	0.08	0.04
	11132	2-WAY RESTRICTOR, N.G. CYL.	-4	0.08	0.04
	<u>IV 33</u>	, 2-WAY RESTRICTOR, N.G. DNLOCK	-4	0.08	0.04
	1V34	VALVE SELECTOR, N.G. DOOR	-4	0.30	0.05
	IV59	VALVE, SHUTOFF, BRAKE MOTOR	-6	0.71	0.11
	1460	CONTROL, R.H. L'D'G EDGE FLAP	-6	1.51	0.20
	1461	L-H. L'D'G EDGE FLAP	-6	1.51	0.20
	1V62	R. H. INBOARD FLAP	-8	1.82	0.24
	1763	L. H. INBOARD FLAP	-8	1.82	0.24
	1164	R.H. MIDSPAN FLAP	-6	1.51	0.20
	1/65	L. H. MIDSPAN FLAP	-6	1.51	0.20
<b> </b>	LV66	R.H.ALERON	-4	0.65	0.06
	1467	L. H. AILERON	<u>  -4</u> .	0.65	0.06
	1168	, B.H. LOWER RUDDER	-4	0.55	0.06
<b>├</b>	1769	L.H. LOWIER RUDDER	-4	0.55	0.06
	1170	RH-UPPER RUDDER	-4	0.55	0.06
		CONTROL, L.H. UPPER RUDDER	-4	<u>ass</u>	0.06
	IV12	, SELECTOR, EXHAUST DOOR CYL.	-4_	0.35	0.04
	1273	CONTROL, L.H. EXTERNAL FLAP	-4-	0.93	0.09
$\vdash$	LV 74	R.H. THRUST VECTOR	-4	0.55	0.06
	IV.75	L.H. THRUST VECTOR	- 4	0.55	0.06
<b> _</b>	14.76	CONTROL, L. H. PLUG THROAT	-8	2.41	0.27
$\vdash$		VALVE, SELECTOR, L.H. THRUST REV.	1-0	1.82	0.24
	<b>↓</b>		l		
	I IMHOL	MOTOR, HYDRAULIC, N.G. STEERING	1- <u>4</u>	1.20	1.18
	29 IMH02	MOTOR, HYDRAULIC, ARMAMENT	1-10	8.15	27.16

### TABLE 37. MASTER EQUIPMENT LIST (PAGE 2 OF 9 PAGES)

.....

## TABLE 57. MASTER EQUIPMENT LIST (PAGE 5 OF 9 PAGES)

. . .

and a construction of the second second standard and the second development of the second of the second of the

to the second of the

•• • • •

HYD	MASTER		PORT	DRY	FLUID
BYST.	EQUIPMENT	COMPONENT TITLE	DASH	WT.	VOL.
NO.	LIST NUMBER		NO.	(LBS)	(IN ³ )
1	29 IMH03	MOTOR, HYDRAULIC, REFUEL RECEPT	-4	1.20	1.18
	4				
	IMH08	MOTOR HYDRAULIC, BRAKE	-6	1.73	2.59
	IMHO9	A RH LEADING EDGE FLAP	-4	1.73	2.59
	IMHIO	LHLEADING EDGE FLAP	-4	1.73	2.59
	IMHII	RH INBOARD FLAP	-8	4.09	10.36
	IMH12	LH INBOARD FLAP	-8	4.09	10.36
	IMH13	, RH MIDSPAN FLAP	-6	1.73	2.59
	IMHI4	LH MIDSPAN FLAP	-6	1.73	2.59
	IMH15	RH AILERON	-4	1.20	1.18
	IMH16	, LH AILERON	-4	1.20	1.18
	IMH 17	, RHLOWER RUDDER	-4	1.20	1.18
	IMH 18	, LH LOWER RUDDER	- 4	1.20	1.18
	IMH 19	, RH UPPER RUDDER	-4	1.20	1.18
	IMH 20	, RH LOWER RUDDER	- 4	1.20	1.18
	1 IMH25	1 , LH PLUG THROAT	-8	8.15	27.16
	29 IMH26	MOTOR, HYDRAULIC, L.H. THRUST REV.	-8	5.98	17.76
l		· · · · · · · · · · · · · · · · · · ·	ļ		
				l	
L					
				l	
			Í		
			L		
			<u> </u>		ļ
				L	
ļ					
			L	<u> </u>	
			I	<b></b>	
	1	<u> </u>	L	<b> </b>	
<b>*</b> T(	DTAL LESS RESE	RVOIR TOTAL/F	AGE	39.36	86.85
		SYSTEM	TOTAL	178.49	320.35

.

HYD.	MASTER		PORT	DRY	FLUID
5757.	EQUIPMENT	COMPONENT TITLE	DASH	WT.	VOL.
NO.	LIST NUMBER		NO.	(LBS)	(IN ³ )
2	29 2001	DISCONNECT-QUICK, PRESS.	-12	3.60	1.50
4	t 2002	DISCONNECT-QUICK, SUCTION_	-16	1.90	3.50
	2F01	FILTER, RETURN	-14	10.20	33.00
	2F02	FILTER, PRESSURE	- 12	20.41	35.00
	2F03	FILTER, CASE DRAIN	- 6	3.23	5.93
	2 401	HEAT EXCHANGER	-6	3.30	21.20
		······································			
	2M01	TRANSDUCER, PRESSURE	-4	0.5	0.03
	2M02	GAGE, N2 PRESSURE, SYS * 3 APU	-4		
	2P01	PUMP, HYDRAULIC	-10	26.00	46.20
	2092	PUMP, HYDRAULIC	-10	26.00	<u>46.20</u>
<b> </b>					
┠━╁╼╍	2R01	RESERVOIR, HYDRAULIC FLUID	-14	*	#
	2R02	ACCUMULATOR, SYS#3 APU	-8	*	*
┠╼┥╴╍	l			<u> </u>	
<b> </b> _↓_	2001	VALVE, CHECK, SYS * 2 RETURN	-14	0.42	0.63
<b> </b> }_	2002	RESTRICTOR CHECK, SUCT. Q.D.	-10	0.19	0.40
$\vdash \downarrow$	2v03	CHECK, SUCTION G.D.	-16	0.29	0.75
	2004	BYPASS, RETURN FILTER	-14	0.30	
┣_┥	2005	CHECK, RESERVOIR RETURN	-14	0.42	0.63
<b>├</b>	2106	OVERBOARD RELIEF, RESERVOIR	<u>-14</u>	0.11	0.85
┣-┥—	207	BLEED, RESERVOIR	-4	0.07	0.04
┣-┿	2708	, CHECK, CASE DRAIN	-6	0.05	0.12
┣-∔	2009	CHECK, PUMP PRESSURE	1-10	0.22	0.34
┣-┿	210	CHECK, PUMP PRESSURE	1-10	0.22	0.34
	<u>2VII</u>	CHECK, PRESSURE Q.D.	1-12	0.30	0.52
┣-┼	21/2	VALVE, RELIEF, PRESSURE	-12	1.31	1.20
H-					<b> </b>
	<u>1 2v24</u>	VALVE, N2 FILL, APU ACCUMULATOR	-4	0.07	
2	1292725	IVALVE, SHUTOFF, APU MOTOR	<u>-4</u>	0.43	0.04
		IUTAL/P/	10E	199.55	1200.53

HYD	MASTER		PORT	DRY	FLUID
SYST.	EQUIPMENT	COMPONENT TITLE	DASH	WT.	VOL.
NO.	LIST NUMBER		NO.	(LBS)	(IN ³ )
2	29 2V26	VALVE, DUMP, EPU MOTOR	-4	0.23	0.05
	<u> </u>	CHECK, EPU MOTOR	-8	0.18	0.19
	2 28	CHECK, SYS #2 NOSE GEAR	-4	0.08	0.04
	2729	SELECTOR, NOSE GEAR	-4	0.55	0.0e
	2√31	2 WAY RESTRICT. N.G. CYL UPLOCK	-4	0.08	0.04
	2732	ZWAY RESTRICTOR N.G. CYL.	-4	0.08	0.04
	2v33	2 WAY RESTRICT N.G. CYL DN LOCK	-4	0.08	0.04
	2134	SELECTOR, NOSE GEAR DOOR	-4	0.30	0.05
	2736	ZWAY RESTRICT. N.G. DOOR LOCK	-4	0.08	0.04
	2137	ZWAY RESTRICTOR NG DOOR CYL.	- 4	0.08	0.04
	2V38	CONTROL, R.H. CANARD CYL.	-4	0.65	0.06
	2v 39	CONTROL L.H. CANARD CYL.	-4	0.65	0.06
	2y 40	CHECK, SYSTEM #2 MAIN GEAR	- 6	0.09	0.09
	2741	SELECTOR, SYST. "I MAIN GEAR	- 6	0.79	0.08
	242	EMERGENCY DUMP, MAIN GEAR	-4	0.23	0.05
	2:√ 43	2 WAY RESTRICT, R.H. MAIN GR. CYL	-6	0.09	0.09
	2.144	ZWAY RESTRICT, R.H. MAIN GR. CYL	-6	0.09	0.09
	2.V 45	ZWAY RESTRICT, R.H. M.GUPLOCK	-4	0.08	0.04
	2V 46	ZWAY RESTRICT, R.H. M.G. DNLOCK	-4	0.08	0.04
	247	ZWAY RESTRICT, LH. M.G. DNLOCK	-4	0.08	0.04
	2148	2WAY RESTRICT, LH.M.GUPLOCK	-4	0.08	0.04
	2v 49	ZWAY RESTRICT, LH. MAIN GR CYL	-6	0.09	0.09
	2150	2WAY RESTRICT, LH MAIN GR.CH	-6	0.09	0.09
	2151	SELECTOR, MAIN GEAR DOOR	-6	0.79	0.08
	2152	EMERGENCY DUMP, M.G. DOOR	-4	0.23	0.05
	2153	2WAY RESTRICT. R.H. M.G. DOOR	-4	0.08	0.04
	2154	ZWAY RESTRICT., R.H. M.G. DOOR	-4	0.08	0.04
	2155	2WAY RESTRICT, R.H. M.G. DOORLOCK	-4	C 08	0.04
	2V 56	ZWAY RESTRICT, L.H. M.G. DOOR LOCK	1 - 4	0.08	0.04
	2157	2 WAY RESTRICT, LH. M.G. DOOR	-4	0.08	0.04
2	29 24 58	VALVE, 2WAY RESTRICT, L.H. M.G. DOOR	-4	0.08	0.04
		TOTAL/P	AGE	6.33	1.82

217

......

HYD	MASTER		PORT	DRY	FLUID
SYST.	EQUIPMENT	COMPONENT TITLE	DASH	WT.	VOL.
NO.	LIST NUMBER		NO.	(185)	(IN ³ )
2	292 1 59	VALVE, SHUTOFF, BRAKE MOTOR	-6	0.71	0.11
4	2160	VALVE, CONTROL, RH LEADING EDGE FLAP	-6	1.51	0.20
	2161	LH LEADING EDGE FLAP	-6	1.51	0.20
_	2162	RH INBOARD FLAP	-8	1.82	0.24
	2163	LH INBOARD FLAP	-3	1.82	0.24
	2764	RH MIDSPAN FLAP	-6	1.51	0.20
	2165	LH MIDSPAN FLAP	-6	1.51	0.20
	2166	, RH AILERON	-4	0.65	_0.06
	2767	LH AILERON	-4	0.65	0.06
	2468	RH LOWER RUDDER	-4	0.55	0.06
	2769	LH LOWER RUDDER	-4	0.55	0.06
	2 4 70	, RH UPPER RUDDER	-4	0.55	0.06
	271	LH UPPER RUDDER	-4	0.55	0.06
	2173	RH EXTERNAL FLAPS	-4	0.93	0.09
	2174	RH THRUST VECTOR	-4	0.55	0.06
	275	LH THRUST VECTOR	-4	0.55	0.06
	2176	RH PLUG THROAT	-8	2.41	0.27
	2 177	VALVE CONTROL RH THRUST REVERSER	-8	1.82	0.24
				<b></b>	
	2 MH05	MOTOR, HYD, SYSTEM #3 APU		4.09	10.36
			·		
	2 MH08	BRAKE	-6	1.73	2.59
	2 MH09	RH LEADING EDGE FLAP	-6	1.73	2.59
	2 MH10	LH LEADING EDGE FLAP	-6	1.73	2.59
	2MHII	RH INBOARD FLAP	-8	4.09	10.36
$\square$	2 MHIZ	, LH INBOARD FLAP	-8	4.09	10.36
	2MHI3	, RH MIDSPAN FLAP	-6	1.73	2.59
	2MHI4	LH MIDSPAN FLAP	-6	1.73	2.59
	2MH15	RH AILERON	-4	1.20	1.18
	2 MHIG	, LH AILERON	-4	1.20	1.18
	2MH17	, RH LOWER RUDDER	-4	1.20	1.18
2	292 MH18	MOTOR HYD., LH LOWER RUDDER	-4	1.20	1.18
		TOTAL/	PAGE	45.87	51.22

TABLE 37. MASTER EQUIPMENT LIST (PAGE 7 OF 9 PAGES)

F

í

HYD	MASTER		PORT	DRY	FUID
BYST.	EQUIPMENT	COMPONENT TITLE	DASH	WT.	VOL.
NO.	LISI NUMDER	MATAR LAND CHUNCER DURACE	NU.		UNY
2	292MH19	MOTOR HYD, RH UPPER RUDDER	-4	1.20	1.10
	- 2 MH20	ILH UPPER RUDDER	-4	1.20	1.18
	2MH25	,RH PLUG THROAT	-8	8.15	27.16
2	292 MH26	MOTOR HYD, RH THRUST REVERSER	-8	5.98	17.76
ļ		-			
		·· · · · - ·			·· <del></del>
[			1		·
<b> </b>			t	[	
<b> </b>			1		
			1		
L				1	
			l	J	
	<b></b>				
			<b></b>		
<b> </b>	<u> </u>		ł		
<b> </b>			<u>† – – – – – – – – – – – – – – – – – – –</u>	t ——	<b>h</b>
					<b>1</b> - · · · ·
				1	1
			<b>_</b>	ļ	<u> </u>
	]		<u> </u>		
# TOT	TAL LESS RESER	VOIR TOTAL/	AGE	16.53	47.28
	A UCRASCOMICAN	- SYSTEM TO	JIAL	16228	1304,03

# TABLE 37. MASTER EQUIPMENT LIST (PAGE 8 OF 9 PAGES)

. . . .

, ~

SYST.       EQUIPMENT NO. LIST NUMBER       COMPONENT TITLE NO. (LBS)       DASH (N3)       WT.       VOL. NO. (LBS)         3       293 DO1       DISCONNECT- QUICK, PRESSURE DISCONNECT- QUICK, SUCTION       -10       2.61       1.04         4       4       3 DO2       DISCONNECT- QUICK, SUCTION       -12       1.25       2.09         3       FO1       FILTER, RETURN       -10       6.54       23.10         3       FO2       FILTER, RETURN       -10       6.54       23.10         3       MO1       TRANSDUCER, PRESSURE       -4       0.51       0.03         3       P01       PUMP, HYDRAULIC       -10       26.00       46.20         3       NO1       RESERVOIR, HYDRAULIC       -10       26.00       46.20         3       VO1       VALVE, CHECK, RETURN       -10       0.30       0.40         3       VO2       ARSTRICTOR GHECK, SUCTION Q.D.       -12       0.23       0.35         3       VO3       BYPASS, RETURN FILTER       -10       0.25       0.80         3       VO3       BYPASS, RETURN FILTER       -10       0.22       0.34         3       VO4       OVERBOARD RELIEF, RESERVOIR       -4       0.07	HYD	MASTER		PORT	DRY	FUID
NO.         LIST         NUMBER         NO.         (LBS)         ((N3)           3         29 3 D 0 1         DISCONNECT- QUICK, PRESSURE         -10         2.61         1.64           4         3 D 0 2         DISCONNECT- QUICK, SUCTION         -12         1.25         2.09           3         FO1         FILTER, RETURN         -10         6.54         23.10           3         FO2         FILTER, RETURN         -10         15.96         19.60           3         M01         TRANSDUCER, PRESSURE         -10         15.96         19.60           3         M01         TRANSDUCER, PRESSURE         -4         0.51         0.03           3         P01         PUMP, HYDRAULIC         -10         26.00         46.20           3         NO1         RESERVOIR, HYDRAULIC         -10         0.30         0.40           3         VO1         VALVE, CHECK, RETURN         -10         0.30         0.40           3         VO2         RESERVOIR, HYDRAULIC         FLUER         -12         0.23         0.35           3         VO3         GYPASS, RETURN FILTER         -10         0.22         0.34           3         VO4         OVERBOARD R	דכים.	EQUIPMENT	COMPONENT TITLE	dash	WT.	VOL.
3       29 3 D 01       DISCONNECT- QUICK, PRESSURE       -10       2.61       1.04         4       3 D 02       DISCONNECT- QUICK, SUCTION       -12       1.25       2.09         3       FOI       FILTER, RETURN       -10       6.54       23.10         3       FOI       FILTER, PRESSURE       -10       15.96       19.60         3       MOI       TRANSDUCER, PRESSURE       -4       0.51       0.03         3       POI       PUMP, HYDRAULIC       -10       26.00       46.20         3       ROI       RESERVOIR, HYDRAULIC       -10       26.00       46.20         3       ROI       RESERVOIR, HYDRAULIC       -10       4       *         3       VOI       VALVE, CHECK, RETURN       -10       0.30       0.40         3       VOI       VALVE, CHECK, RETURN       -10       0.22       0.35         3       VO3       ,BYPASS, RETURN FLITER       -10       0.22       0.35         3       VO4       ,OVERBOARD RELIEF, RESERVOIR       -10       0.22       0.34         3       VO5       BLEED, RESERVOIR       -4       0.07       0.04         3       VO6       ,CHECK, PRESSURE<	NO.	LIST NUMBER		NO.	(LBS)	(IN ³ )
1       1       3 DO2       DISCONNECT - QUICK, SUCTION       -12       1.25       2.09         3       FOI       FILTER, RETURN       -10       6.54       23.10         3       FO2       FILTER, PRESSURE       -10       15.96       19.60         3       MOI       TRANSDUCER, PRESSURE       -4       0.51       0.03         3       PO1       PUMP, HYDRAULIC       -10       26.00       46.20         3       ROI       RESERVOIR, HYDRAULIC       -10       4       *         3       VOI       VALVE, CHECK, RETURN       -10       0.30       0.40         3       VOI       VALVE, CHECK, RETURN       -10       0.30       0.40         3       VO2       RESTRICTOR, CHECK, SUCTION Q.D12       0.23       0.35         3       VO3       ,BYPASS, RETURN FLITER       -10       0.25       0.80         3       VO4       ,OVERBOARD RELIEF, RESERVOIR       -10       0.25       0.30         3       VO4       ,OVERBOARD RELIEF, RESERVOIR       -10       0.22       0.34         3       VO6       ,CHECK, PRESSURE       -10       0.22       0.34         3       VO6       ,CHECK, PR	3	293001	DISCONNECT- QUICK, PRESSURE _	-10	2.61	1.04
3F01       FILTER, RETURN       -10       6.54       23.10         3F02       FILTER, PRESSURE       -10       15.96       19.60         3M01       TRANSDUCER, PRESSURE       -4       0.51       0.03         3P01       PUMP, HYDRAULIC       -10       26.00       46.20         3R01       RESERVOIR, HYDRAULIC       -10       26.00       46.20         3V01       VALVE, CHECK, RETURN       -10       0.30       0.40         3V02       RESERVOIR, HYDRAULIC       -10       0.30       0.40         3V03       ,BYPASS, RETURN       -10       0.23       0.35         3V04       ,OVERBOARD RELIEF, RESERVOIR       -10       0.25       0.80         3V05       ,BLEED, RESERVOIR       -4       0.07       0.04         3V06       ,CHECK, CASE DRAIN       -6       0.05       0.04         3V07       ,CHECK, PRESSURE       -10       0.22       0.34         3V08       ,CHECK, PRESSURE       -10       1.07       0.96         3V08       ,CHECK, PRESSURE       -10       1.07       0.96         3V08       ,CHECK, PRESSURE       -10       1.07       0.96         3V63       , LH INBOARD F		1 3 DO2	DISCONNECT-QUICK, SUCTION	-12	1.25	2.09
3 FOI       FILTER, RETURN       -10       6.54/23.10         3 FO2       FILTER, PRESSURE       -10       15.96       19.60         3 MOI       TRANSDUCER, PRESSURE       -4       0.51       0.03         3 POI       PUMP, HYDRAULIC       -10       26.00       46.20         3 ROI       RESERVOIR, HYDRAULIC       -10       26.00       46.20         3 ROI       RESERVOIR, HYDRAULIC       -10       0.30       0.40         3 VOI       VALVE, CHECK, RETURN       -10       0.30       0.40         3 VO2       A RESTRICTOR, CHECK, SUCTION Q.D.       -12       0.23       0.35         3 VO3       BYPASS, RETURN FILTER       -10       0.25       0.80         3 VO4       OVERBOARD RELIEF, RESERVOIR       -10       0.25       0.80         3 VO5       BLEED, RESERVOIR       -4       0.07       0.04         3 VO6       CHECK, PARESSURE       -10       0.22       0.34         3 V06       CHECK, PRESSURE       -10       0.22       0.34         3 V07       CHECK, PRESSURE       -10       0.22       0.34         3 V08       CHECK, PRESSURE       -10       1.07       0.96         3 V62	┟╍╾┿╸╸					
3 F02       FILTER, PRESSURE       -10       15.92       19.60         3 M01       TRANSDUCER, PRESSURE       -4       0.51       0.03         3 P01       PUMP, HYDRAULIC       -10       26.00       46.20         3 R01       RESERVOIR, HYDRAULIC       -10       26.00       46.20         3 R01       RESERVOIR, HYDRAULIC       FLUID       -10       *       *         3 V01       VALVE, CHECK, RETURN       -10       0.30       0.40         3 V02       RESTRICTOR, CHECK, SUCTION Q.D.       -12       0.23       0.35         3 V03       GYPASS, RETURN FILTER       -10       0.25       0.80         3 V04       OVERBOARD RELIEF, RESERVOIR       -10       0.22       0.34         3 V05       BLEED, RESERVOIR       -10       0.22       0.34         3 V06       CHECK, CASE DRAIN       -6       0.05       0.04         3 V08       CHECK, PRESSURE       -10       0.22       0.34         3 V08       CHECK, PRESSURE       -10       0.22       0.34         3 V08       CHECK, PRESSURE       -10       1.07       0.96         3 V08       CHECK, PRESSURE       -10       1.07       0.96 <tr< td=""><td> </td><td>3501</td><td>FILTER, RETURN</td><td>-10</td><td>6.54</td><td>23.10</td></tr<>		3501	FILTER, RETURN	-10	6.54	23.10
3 MOI       TRANSDUCER, PRESSURE       -4       0.51       0.03         3 POI       PUMP, HYDRAULIC       -10       26.00       46.20         3 ROI       RESERVOIR, HYDRAULIC       -10       76.00       46.20         3 ROI       RESERVOIR, HYDRAULIC       FLUID       -10       *         3 VOI       VALVE, CHECK, RETURN       -10       0.30       0.40         3 VO2       , RESTRICTOR, CHECK, SUCTION Q.D.       -12       0.23       0.35         3 VO3       , BYPASS, RETURN FILTER       -10       0.25       0.80         3 VO4       , OVERBOARD RELIEF, RESERVOIR       -10       0.22       0.34         3 VO5       , BLEED, RESERVOIR       -4       0.07       0.04         3 VO6       , CHECK, CASE DRAIN       -6       0.05       0.04         3 VO6       , CHECK, PRESSURE       -10       0.22       0.34         3 V07       , CHECK, PRESSURE       -10       0.022       0.34         3 V08       , CHECK, PRESSURE       -10       1.07       0.96         3 V08       , CHECK, PRESSURE       -10       1.07       0.96         3 V62       , CONTROL, PH INBOARD FLAP       -8       1.82       0.24		3102	FILTER, PRESSURE	-10	15.96	19.60
3 P01       PUMP, HYDRAULIC       -10       26.00       46.20         3 R01       RESERVOIR, HYDRAULIC FLUID       -10       26.00       46.20         3 R01       RESERVOIR, HYDRAULIC FLUID       -10       *       *         3 V01       VALVE, CHECK, RETURN       -10       0.30       0.40         3 V02       , RESTRICTOR, GHECK, SUCTION Q.D.       -12       0.23       0.35         3 V03       , 6YPASS, RETURN FILTER       -10       0.25       0.80         3 V04       , 0VERBOARD RELIEF, RESERVOIR       -10       0.90       0.70         3 V05       , BLEED, RESERVOIR       -4       0.07       0.04         3 V06       , CHECK, CASE DRAIN       -6       0.05       0.04         3 V06       , CHECK, PRESSURE       -10       0.22       0.34         3 V07       , CHECK, PRESSURE       -10       0.22       0.34         3 V08       , CHECK, PRESSURE       -10       0.22       0.34         3 V08       , CHECK, PRESSURE       -10       1.07       0.96         3 V62       , CONTROL, RH INBOARD FLAP       -8       1.82       0.24         3 V63       , LH INBOARD FLAP       -6       1.51       0.20     <	<u>├</u> ── <u></u> ┼──	3 MOI	TRANSMICED PRESSURE		0.51	0.03
3 P01       PUMP, HYDRAULIC       -10       26.00       46.20         3 R01       RESERVOIR, HYDRAULIC FLUID       -10       4       *         3 V01       VALVE, CHECK, RETURN       -10       0.30       0.40         3 V02       A RESTRICTOR, GHECK, SUCTION Q.D12       0.23       0.35         3 V03       ,6YPASS, RETURN       FILTER       -10       0.25       0.80         3 V04       ,0VERBOARD RELIEF, RESERVOIR       -10       0.20       0.70         3 V05       ,BLEED, RESERVOIR       -10       0.22       0.34         3 V06       ,CHECK, CASE DRAIN       -6       0.05       0.04         3 V06       ,CHECK, PRESSURE       -10       0.22       0.34         3 V08       ,CHECK, PRESSURE       -10       0.22       0.34         3 V08       ,CHECK, PRESSURE       -10       0.02       0.34         3 V08       ,CHECK, PRESSURE       -10       1.07       0.96         3 V08       ,CHECK, PRESSURE       -10       1.07       0.96         3 V62       ,CONTROL, RH INBOARD FLAP       -8       1.82       0.24         3 V63       , LH INBOARD FLAP       -6       1.51       0.20	! · <del> </del> ·		I KANSDUCER, FRESSURE		0.31	
3 ROI       RESERVOIR, HYDRAULIC FLUID       -10       *         3 VOI       VALVE, CHECK, RETURN       -10       0.30       0.40         3 VO2       , RESTRICTOR, CHECK, SUCTION Q.D.       -12       0.23       0.35         3 VO3       , BYPASS, RETURN FILTER       -10       0.25       0.80         3 VO3       , BYPASS, RETURN FILTER       -10       0.25       0.80         3 VO3       , BYPASS, RETURN FILTER       -10       0.25       0.80         3 VO4       , OVERBOARD RELIEF, RESERVOIR       -10       0.25       0.80         3 VO5       BLEED, RESERVOIR       -4       0.07       0.04         3 VO6       , CHECK, CASE DRAIN       -6       0.05       0.04         3 VO6       , CHECK, PUMP PRESSURE       -10       0.22       0.34         3 VO8       , CHECK, PRESSURE Q. D.       -10       0.22       0.34         3 V08       , CHECK, PRESSURE Q. D.       -10       0.22       0.34         3 V09       , RELIEF, PRESSURE       -10       1.07       0.96         3 V62       , CONTROL, PH INBOARD FLAP       -8       1.82       0.24         3 V63       , LH INBOARD FLAP       -6       1.51       0.20     <		3 201	PUMP, HYDRAULIC	-10	26.00	46.70
3 ROI       RESERVOIR, HYDRAULIC FLUID       -10       *         3 VOI       VALVE, CHECK, RETURN       -10       0.30       0.40         3 VO2       RESTRICTOR, CHECK, SUCTION Q.D12       0.23       0.35         3 VO3       BYPASS, RETURN FILTER       -10       0.25       0.80         3 VO3       BYPASS, RETURN FILTER       -10       0.25       0.80         3 VO4       OVERBOARD RELIEF, RESERVOIR       -10       0.90       0.70         3 VO5       BLEED, RESERVOIR       -4       0.07       0.04         3 VO5       BLEED, RESERVOIR       -4       0.07       0.04         3 V05       CHECK, CASE DRAIN       -6       0.05       0.04         3 V06       .CHECK, PUMP PRESSURE       -10       0.22       0.34         3 V08       .CHECK, PRESSURE Q. D.       -10       0.22       0.34         3 V09       .RELIEF, PRESSURE       -10       1.07       0.96         3 V09       .RELIEF, PRESSURE       -10       1.07       0.96         3 V62       .CONTROL, RH INBOARD FLAP       -8       1.82       0.24         3 V63       .LH INBOARD FLAP       -8       1.51       0.20         3 V64 <td< th=""><th></th><th></th><th></th><th></th><th></th><th>10,00</th></td<>						10,00
3 VOI       VALVE, CHECK, RETURN       -10       0.30       0.40         3 VO2       , RESTRICTOR, CHECK, SUCTION Q.D12       0.23       0.35         3 VO3       , BYPASS, RETURN FILTER       -10       0.25       0.80         3 VO4       , OVERBOARD RELIEF, RESERVOIR       -10       0.90       0.70         3 VO5       BLEED, RESERVOIR       -4       0.07       0.04         3 VO6       , CHECK, CASE DRA!N       -6       0.05       0.04         3 VO6       , CHECK, PESSURE       -10       0.22       0.34         3 VO8       , CHECK, PESSURE       -10       0.22       0.34         3 V08       , CHECK, PRESSURE       -10       0.22       0.34         3 V09       , RELIEF, PRESSURE       -10       0.22       0.34         3 V09       , RELIEF, PRESSURE       -10       1.07       0.96         3 V62       , CONTROL, PH INBOARD FLAP       -8       1.82       0.24         3 V63       , LH INBOARD FLAP       -8       1.51       0.20         3 V64       , RH MIDSPAN FLAP       -6       1.51       0.20         3 V65       , LH MIDSPAN FLAP       -6       1.51       0.20         3 V69 <td></td> <td>3 R01</td> <td>RESERVOIR, HYDRAULIC FLUID</td> <td>-10</td> <td>*</td> <td>*</td>		3 R01	RESERVOIR, HYDRAULIC FLUID	-10	*	*
3 VOI       VALVE, CHECK, RETURN       -10       0.30       0.40         3 VO2       , RESTRICTOR, CHECK, SUCTION Q.D.       -12       0.23       0.35         3 VO3       , BYPASS, RETURN FILTER       -10       0.25       0.80         3 VO4       , OVERBOARD RELIEF, RESERVOIR       -10       0.90       0.70         3 VO4       , OVERBOARD RELIEF, RESERVOIR       -10       0.90       0.70         3 VO5       , BLEED, RESERVOIR       -4       0.07       0.04         3 VO6       , CHECK, CASE DRAIN       -6       0.05       0.04         3 VO6       , CHECK, PUMP PRESSURE       -10       0.22       0.34         3 V08       , CHECK, PRESSURE       -10       0.22       0.34         3 V09       , RELIEF, PRESSURE       -10       0.07       0.96         3 V09       , RELIEF, PRESSURE       -10       1.07       0.96         3 V63       , LH INBOARD FLAP       -8       1.82       0.24         3 V64       , RH MIDSPAN FLAP       -6       1.51       0.20         3 V65       , LH MIDSPAN FLAP       -6       1.51       0.20         3 V63       , LH MIDSPAN FLAP       -6       1.51       0.20						
3 V02       , RESTRICTOR, CHECK, SUCTION Q.D12       0.23       0.35         3 V03       , BYPASS, RETURN FILTER       -10       0.25       0.80         3 V04       , OVERBOARD RELIEF, RESERVOIR       -10       0.90       0.70         3 V05       , BLEED, RESERVOIR       -10       0.90       0.70         3 V06       , CHECK, CASE DRAIN       -6       0.05       0.04         3 V06       , CHECK, PUMP PRESSURE       -10       0.22       0.34         3 V08       , CHECK, PRESSURE       -10       0.22       0.34         3 V08       , CHECK, PRESSURE       -10       1.07       0.96         3 V09       , RELIEF, PRESSURE       -10       1.07       0.96         3 V62       , CONTROL, PH INBOARD FLAP       -8       1.82       0.24         3 V63       , LH INBOARD FLAP       -8       1.82       0.24         3 V64       , RH MIDSPAN FLAP       -6       1.51       0.20         3 V65       , LH MIDSPAN FLAP       -6       1.51       0.20         3 V64       , RH LOWER RUDDER       -4       0.55       0.06         3 V69       , LH LOWER RUDDER       -4       0.55       0.06       0.06		3 101	VALVE, CHECK, RETURN	-10	0.30	0.40
3 V03       ,6YPASS, RETURN FILTER       -10       0.25       0.80         3 V04       ,0VERBOARD RELIEF, RESERVOIR       -10       0.90       0.70         3 V05       ,BLEED, RESERVOIR       -4       0.07       0.04         3 V06       ,CHECK, CASE DRAIN       -6       0.05       0.04         3 V06       ,CHECK, CASE DRAIN       -6       0.05       0.04         3 V07       ,CHECK, PUMP PRESSURE       -10       0.22       0.34         3 V08       ,CHECK, PRESSURE Q. D.       -10       0.22       0.34         3 V09       ,RELIEF, PRESSURE       -10       1.07       0.96         3 V62       ,CONTROL,PH INBOARD FLAP       -8       1.82       0.24         3 V63       ,LH INBOARD FLAP       -8       1.82       0.24         3 V64       ,RH MIDSPAN FLAP       -6       1.51       0.20         3 V65       ,LH MIDSPAN FLAP       -6       1.51       0.20         3 V68       ,RH LOWER RUDDER       -4       0.55       0.06         3 V69       ,LH LOWER RUDDER       -4       0.55       0.06         3 V69       ,LH LOWER RUDDER       -4       0.55       0.06         3 29 3 V71		3 V02	RESTRICTOR CHECK, SUCTION Q.D.	-12	0.23	0.35
3 V04       ,0VERBOARD RELIEF, RESERVOIR       -10       0.90       0.70         3 V05       BLEED, RESERVOIR       -4       0.07       0.04         3 V06       ,CHECK, CASE DRA:N       -6       0.05       0.04         3 V07       ,CHECK, PUMP PRESSURE       -10       0.22       0.34         3 V08       ,CHECK, PRESSURE       -10       0.22       0.34         3 V08       ,CHECK, PRESSURE       -10       1.07       0.96         3 V09       ,RELIEF, PRESSURE       -10       1.07       0.96         3 V62       ,CONTROL, RH INBOARD FLAP       -8       1.82       0.24         3 V63       , LH INBOARD FLAP       -8       1.82       0.24         3 V64       , BH MIDSPAN FLAP       -6       1.51       0.20         3 V65       , LH MIDSPAN FLAP       -6       1.51       0.20         3 V68       , RH LOWER RUDDER       -4       0.55       0.06         3 V69       , LH LOWER RUDDER       -4       0.55       0.06         3 V69       , LH LOWER RUDDER       -4       0.55       0.06         3 V69       , LH LOWER RUDDER       -4       0.55       0.06         3 V69       , L		3 V03	, BYPASS, RETURN FILTER	-10	0.25	0.80
3V05       BLEED, RESERVO(R       -4       0.07       0.04         3V06       CHECK, CASE DRA!N       -6       0.05       0.04         3V07       CHECK, PUMP PRESSURE       -10       0.22       0.34         3V08       CHECK, PRESSURE Q. D.       -10       0.22       0.34         3V09       RELIEF, PRESSURE       -10       1.07       0.96         3V62       CONTROL, RH INBOARD FLAP       -8       1.82       0.24         3V63       LH INBOARD FLAP       -8       1.82       0.24         3V64       RH MIDSPAN FLAP       -6       1.51       0.20         3V65       LH MIDSPAN FLAP       -6       1.51       0.20         3V65       LH MIDSPAN FLAP       -6       1.51       0.20         3V69       LH LOWER RUDDER       -4       0.55       0.06         3V69       LH LOWER RUDDER       -4 <td< td=""><td>ļ</td><td>3 104</td><td>OVERBOARD RELIEF, RESERVOIR</td><td><u>-10</u></td><td>0.90</td><td>0.70</td></td<>	ļ	3 104	OVERBOARD RELIEF, RESERVOIR	<u>-10</u>	0.90	0.70
3 V06       , CHECK, CASE DRA!N       -6       0.05       0.04         3 V07       , CHECK, PUMP PRESSURE       -10       0.22       0.34         3 V08       , CHECK, PRESSURE Q. D.       -10       0.22       0.34         3 V09       , RELIEF, PRESSURE       -10       1.07       0.96         3 V62       , CONTROL, PH INBOARD FLAP       -8       1.82       0.24         3 V63       , LH INBOARD FLAP       -8       1.82       0.24         3 V63       , LH INBOARD FLAP       -8       1.82       0.24         3 V64       , RH MIDSPAN FLAP       -6       1.51       0.20         3 V65       , LH MIDSPAN FLAP       -6       1.51       0.20         3 V63       , RH LOWER RUDDER       -4       0.55       0.06         3 V63       , RH LOWER RUDDER       -4       0.55       0.06         3 V69       , LH LOWER RUDDER       -4       0.55       0.06         3 29 3 V 71       VALVE, CONTROL, LH UPPER RUDDER       -4       0.55       0.06	<b> </b>	3V05	,BLEED, RESERVOIR	- 4	0.07	0.04
3V07       ,CHECK, POMP PRESSURE       -10       0.22       0.34         3V08       ,CHECK, PRESSURE Q.D.       -10       0.22       0.34         3V09       ,RELIEF, PRESSURE       -10       1.07       0.96         3V62       ,CONTROL, RH INBOARD FLAP       -8       1.82       0.24         3V63       , LH INBOARD FLAP       -8       1.82       0.24         3V64       , LH INBOARD FLAP       -8       1.82       0.24         3V65       , LH INBOARD FLAP       -6       1.51       0.20         3V65       , LH MIDSPAN FLAP       -6       1.51       0.20         3V65       , LH MIDSPAN FLAP       -6       1.51       0.20         3V68       , RH LOWER RUDDER       -4       0.55       0.06         3V69       , LH LOWER RUDDER       -4       0.55       0.06         3V69       , LH LOWER RUDDER       -4       0.55       0.06         3293V71       YALVE, CONTROL, LH UPPER RUDDER       -4       0.55       0.06		3 106	CHECK, CASE DRA!N	-6	0.05	0.04
3 V08       ,CHELK, PRESSURE G. U.       -10       0.22       0.34         3 V09       ,RELIEF, PRESSURE       -10       1.07       0.96         3 V62       ,CONTROL, PH INBOARD FLAP       -8       1.82       0.24         3 V63       ,LH INBOARD FLAP       -8       1.82       0.24         3 V63       ,LH INBOARD FLAP       -8       1.82       0.24         3 V64       ,RH MIDSPAN FLAP       -6       1.51       0.20         3 V65       ,LH MIDSPAN FLAP       -6       1.51       0.20         3 V68       ,RH LOWER RUDDER       -4       0.55       0.06         3 V69       ,LH LOWER RUDDER       -4       0.55       0.06         3 V69       ,LH LOWER RUDDER       -4       0.55       0.06         3 V69       ,LH LOWER RUDDER       -4       0.55       0.06         3 29 3 V71       VALVE, CONTROL, LH UPPER RUDDER       -4       0.55       0.06	┝╾┼╴	307	CHECK, PUMP PRESSURE	1-10	0.22	_0.34
3 V09       ,RELIEF, PRESSURE       -10       1.07       0.96         3 V62       ,CONTROL, RH INBOARD FLAP       -8       1.82       0.24         3 V63       , LH INBOARD FLAP       -8       1.82       0.24         3 V64       , RH MIDSPAN FLAP       -6       1.51       0.20         3 V65       , LH MIDSPAN FLAP       -6       1.51       0.20         3 V63       , RH LOWER RUDDER       -4       0.55       0.06         3 V68       , RH LOWER RUDDER       -4       0.55       0.06         3 V69       , LH LOWER RUDDER       -4       0.55       0.06         3 V69       , LH LOWER RUDDER       -4       0.55       0.06         3 V69       , LH LOWER RUDDER       -4       0.55       0.06         3 V69       , LH LOWER RUDDER       -4       0.55       0.06         3 29 3 V 71       VALVE, CONTROL, LH UPPER RUDDER       -4       0.55       0.06	<b> </b>	3408	CHELK, PRESSURE Q. U.	-10-	0.22	0.34
3 V62       CONTROL, RH INBOARD FLAP       -8       1.82       0.24         3 V63       I.H INBOARD FLAP       -8       1.82       0.24         3 V64       RH MIDSPAN FLAP       -6       1.51       0.20         3 V65       I.H MIDSPAN FLAP       -6       1.51       0.20         3 V65       I.H MIDSPAN FLAP       -6       1.51       0.20         3 V68       RH LOWER RUDDER       -4       0.55       0.06         3 V69       I.H LOWER RUDDER       -4       0.55       0.06         3 V69       RH UPPER RUDDER       -4       0.55       0.06         3 V69       I.H LOWER RUDDER       -4       0.55       0.06         3 V69       I.H UPPER RUDDER       -4       0.55       0.06         3 V69       I.H UPPER RUDDER       -4       0.55       0.06         3 29 3 V71       VALVE, CONTROL, LH UPPER RUDDER       -4       0.55       0.06	-+	3409	RELIEF, PRESSURE	-10	1.01	0.36
3 V63       1.11 INOOARD FLAP       -8       1.82       0.24         3 V64       RH MIDSPAN FLAP       -6       1.51       0.20         3 V65       LH MIDSPAN FLAP       -6       1.51       0.20         3 V63       RH LOWER RUDDER       -4       0.55       0.06         3 V68       RH LOWER RUDDER       -4       0.55       0.06         3 V69       LH LOWER RUDDER       -4       0.55       0.06         3 V69       LH LOWER RUDDER       -4       0.55       0.06         3 V69       LH LOWER RUDDER       -4       0.55       0.06         3 V69       LH LOWER RUDDER       -4       0.55       0.06         3 V69       LH LOWER RUDDER       -4       0.55       0.06         3 V69       LH UPPER RUDDER       -4       0.55       0.06         3 29 3 V71       VALVE, CONTROL, LH UPPER RUDDER       -4       0.55       0.06		3 162	MNTPOL RH INPOACO ELAP	- 8	1.82	024
3 V64       , RH MIDSPAN FLAP       - 6       1.51       0.20         3 V65       , LH MIDSPAN FLAP       - 6       1.51       0.20         3 V65       , LH MIDSPAN FLAP       - 6       1.51       0.20         3 V68       , RH LOWER RUDDER       - 4       0.55       0.06         3 V69       , LH LOWER RUDDER       - 4       0.55       0.06         3 V69       , LH LOWER RUDDER       - 4       0.55       0.06         3 V69       , LH LOWER RUDDER       - 4       0.55       0.06         3 V69       , LH LOWER RUDDER       - 4       0.55       0.06         3 V69       , LH LOWER RUDDER       - 4       0.55       0.06         3 29 3 V71       VALVE,CONTROL, LH UPPER RUDDER       - 4       0.55       0.06		3763	I H INBOARD FLAP	- 8	1.82	0.24
3V65       , LH MIDSPAN FLAP       -6       1.51       0.20         3V68       , RH LOWER RUDDER       -4       0.55       0.06         3V69       , LH LOWER RUDDER       -4       0.55       0.06         3V69       , LH LOWER RUDDER       -4       0.55       0.06         3V70       RH UPPER RUDDER       -4       0.55       0.06         3293V71       VALVE, CONTROL, LH UPPER RUDDER       -4       0.55       0.06		3 V64	RH MIDSPAN FLAP	-6	1.51	0.20
3 V68       RH LOWER RUDDER       -4       0.55       0.06         3 V69       LH LOWER RUDDER       -4       0.55       0.06         1 3 V70       RH UPPER RUDDER       -4       0.55       0.06         3 29 3 V71       VALVE CONTROL, LH UPPER RUDDER       -4       0.55       0.06		3765	LH MIDSPAN FLAP	- 6	1.51	0.20
3 V68       RH LOWER RUDDER       -4       0.55       0.06         3 V69       LH LOWER RUDDER       -4       0.55       0.06         1 3 V70       RH UPPER RUDDER       -4       0.55       0.06         3 29 3 V71       VALVE, CONTROL, LH UPPER RUDDER       -4       0.55       0.06						
3 V69       LH LOWER RUDDER       -4       0.55       0.06         1 3 V70       RH UPPER RUDDER       -4       0.55       0.06         3 29 3 V71       VALVE, CONTROL, LH UPPER RUDDER       -4       0.55       0.06		3768	RH LOWER RUDDER	-4	0.55	0.06
1         3         293V70         +         RH UPPER RUDDER         -4         0.55         0.06           3         293V71         VALVE CONTROL, LH UPPER RUDDER         -4         0.55         0.06	<b> </b>	3 169	LH LOWER RUDDER	-4	0.55	0.06
3 29 3 V71 VALVE, CONTROL, LH UPPER RUDDER -4 0.55 0.06		1 3 170	RH UPPER RUDDER	-4	0.55	0.06
	3	29 3 171	VALVE, CONTROL, LH UPPER RUDDER	-4	0.55	0.06
				┟	<b> </b>	
	L	1	ITOTAL /D		65 00	9715

i

HYD BYST. NO.	MASTER EQUIPMENT LIST NUMBER	COMPONENT TITLE	PORT DASH NO.	DRY WT. (LBS)	FLUID VOL. (IN ³ )
3	293MH11	MOTOR, HYD. RH INBOARD FLAP	-8	4.09	10.36
ī	1 3MHIZ	HYD LH INBOARD FLAP	-8	4.09	10.36
	3MH13	HYD. RH MIDSPAN FLAP	- 6	1.73	2.59
$\square$	3MH14	HYD LH MIDSPAN FLAP	-6	1.73	2.59
	L				
	3MH17	, HYD RH LOWER RUDDER	- 4	1.20	1.18
	3MH18	HYD. LH_LOWER RUDDER	- 4	1.20	<u> </u>
	+ 3MH19	+ , HYD. RH UPPER RUDDER	-4	1.20	1.18
3	2932420	MOTOR, HYD. LH. UPPER RUDDER	-4	į.20	1.18
		·	{	·	
<u> </u>					
<b> </b>	····		┣───		
		· · · · · · · · · · · · · · · · · · ·		<b> </b>	
	<u> </u>	······································	┨────		
			╉───	<u> </u>	
			<u> </u>	t	
}	<u> </u>		t	l	
<u>}</u>				<b></b>	
			<u>†</u>	t	
<b> </b>					
			1		<b></b>
			<b>_</b>		
			ļ		L
			<b>_</b>		<b>i</b>
	<u> </u>		<b> </b>	<b> </b>	<b></b>
<b> </b>	h		┢	Į	<b></b>
<b> </b>			<b> </b>	<b> </b>	<b> </b>
	<u> </u>		┫───	<b> </b>	╂
				11000	20/0
# TOT	ING LEDD KESER	SYSTEM T	MGE Otal	814	1 30.62

(g) A fluid volume equivalent to that resulting from the effects of fluid compression, line and actuator expansion, and external seal deflection. This computation assumed that subjecting all of the fluid in actuators, components, tubing and fittings to 8000 PSI (instead of a mean pressure of 6000 PSI on the pressure side and a mean pressure of 2000 PSI on the return side which would be more realistic) and ignoring the structural expansion of tubing, fittings, components, actuators and seals still gave a reasonable but conservative value for this requirement. Therefore, the volumes subject to high pressure were:

System	#1	System	#2	System	#3
1010.14	in ³	1023.35	IN ³	307.24	in ³

Using 2.06 x 10.5 PSI as the bulk modulus of MIL-H-83282 at 250°F (see Figure 8 of AIR 1362), the change in volume which must be accomodated in the reservoir became:

$$\Delta V_{01} = \frac{Pressurized Vol. (IN^{3}) \times Pressure (PSI)}{Bulk Modulus (PSI)}$$
  
=  $\frac{1010.14 \text{ IN}^{3} \times 8000 \text{ PSI}}{2.06 \times 10^{5} \text{ PSI}}$  For System #1  
=  $\frac{39.23 \text{ IN}^{3}}{2.06 \times 10^{5} \text{ PSI}}$  For System #2  
=  $\frac{39.74 \text{ IN}^{3}}{2.06 \times 10^{5} \text{ PSI}}$  For System #3  
=  $\frac{11.93 \text{ IN}^{3}}{2.06 \times 10^{5} \text{ PSI}}$  For System #3

(h) A fluid volume equivalent to system fluid thermal expansion resulting from 70°F to the maximum operating temperature. (250°F bulk fluid temperature) the fluid volume affected for each system was as follows: (see Item f above)

System #1	System #2	System #3
1175.78 IN ³	1244.22 IN ³	349.43 IN ³

The fluid thermal expansion then became:

$$\Delta Vol (IN^3) = coeff. of thermal expansion (IN^3/IN^3/°F) x heated vol. (IN^3) x Temp (°F)$$

= (.00046) (1175.78) (180) for system #1 =  $97.35 \text{ IN}^3$ = (.00045) (1244.22) (180) for system #2 =  $103.02 \text{ IN}^3$ = (.00046) (349.43) (180) for system #3 =  $28.93 \text{ IN}^3$ 

Based on the above figures the fluid volume capability of each reservoir, without venting fluid overboard then became:

		SYSTEM #1	SYSTEM #2	SYSTEM #3
SUMP VOLUME	(a)	23.00 IN ³	20.00 IN ³	$10.00 \text{ IN}^3$
ACTUATOR VOL. CHANGES	(b)	32.72 IN ³	46.54 IN ³	
ACCUMULATOR VOLUME	(c)		38.00 IN ³	
FUSE VOLUME	(d)		• •	
THERMAL CONTRACTION VOL.	(e)	53.93 IN ³	57.07 IN ³	15.55 IN ³
LEAKAGE ALLOWANCE	(f)	55.99 IN ³	59.25 IN ³	16.64 IN ³
COMPRESSION VOLUME	(g)	<u>39.23 IN³</u>	39.74 IN ³	11.93 IN ³
VOLUME IN RESERVOIR FOR FLUID WEIGHT COMPUTATION PURPOSES		204.87 IN ³	260.61 IN ³	54.12 IN ³
THERMAL EXPANSION VOL.		97.35 IN ³	103.02 IN ³	28.93 IN ³
FLUID VOLUME CAPACITY OF RES. RECERVOIR WEIGHT (SEE FIGURE 53)		302.22 IN ³ 18.25 LB	363.63 IN ³ 18.25 LB	83.05 IN ³ 6.15 LB

4.2.1.13 <u>Hydraulic System Weight Summary</u> - A summary of the elements making up the hydraulic system's weight for Aircraft II is shown in Table ³⁸. This summarization included all power generation, distribution, and utilization elements between the power take off at the AMAD and the various power cutput interfaces points which were common to both Aircraft I and Aircraft II. The total hydraulic system weight is shown in Table 38 as 1197.55 lbs. This compared to the 1362 lbs originally predicted based on a parametric weight analysis (see paragraph 4.2 and Table 28). The 164.45 lb weight reduction of this defined system, versus the parametric evaluation, appeared reasonable. The parametric analysis assumed the use of hydraulic linear actuators and the various bell cranks and levers typically associated with such a system. The defined system used power hinges as the final output device and the weight they represented (approx. 428 lb) was not included as part of the hydraulic system weight. Since hydraulic motors of similar power capabilities, are TABLE 38. HYDRAULIC POWER GENERATION, DISTRIBUTION, AND UTILIZATION SYSTEM (HECARIS) WEIGHT

.

		WEIGHT	((18)*
ITEM	BASIS	ITEM S	UB TOT.
COMPONENTS COMPONENT SUPPORTS	SUM OF SYSTEM WEIGHTS - TABLE 10% OF COMPONENT WEIGHT	428.25 42.83	
TUBING	DERIVED FROM TABLE	66.90 42.69	80.174
SUPPORTS			16.80
CANARD	21.25 × 2 FIGURE	42.50	
EXTERNAL FLAP	10.39 x 4 TABLE	41.56	
THRUST VECTOR VANE	125.97 x 2 " "	251.94	
NOSE GEAR	35,58 × 1 "	35.58	
MAIN GEAR	19.67 × 2 * *	39.34	
R. RAM AIR SCOOP	2.65 × 1 " "	2.65	
L. RAM AIR SCOOP	0.77 × 1 " "	0.77	
EXHAUST DOOR	1.59 × 1 × 53	1.59	
BRACKETS & SUPPORTS	10% OF ACTUATOR WEIGHT	41.59	
		4	57.52
ACCUMULATOR	PARAGRAPH 4.2.1.8 DISCHARGED & PRECHARG	ED   16.21	
RESERVOIR	PARAGRAPH 4.2.1.11	43.25	
SUPPORTS	5% OF RESERVOIR AND ACCUMULATOR WT.	2.97	
SYSTEM FLUID		21.68	52.15
# DRY WEIGHT EXCEPT FOR FLU	ID ITEM TOTAL SYSTEM	NEIGHT 11	97.55

.....

.-



. .....

..........

.....

Figure 55. Reservoir Weight

significantly lighter than hydraulic linear actuators, the fact that the weight shown in Table 38 was less than the parametrically projected weight seemed very much in line.

4.2.2 <u>Aircraft II Electrical System General Description</u> - The primary electrical power system for Aircraft II, as shown schematically in figure 54, consisted of two primary AC generators, two transformer-rectifiers, an emergency AC/DC generator and power distribution (bus) system. External electrical power could be applied to the bus system on the ground and a battery provided electrical power to part of the bus system during an engine start without external power.

Two 120/208-volt, 400 Hz generators were the primary source of electrical power. Each generator was powered by separate, engine-driven, remotely-mounted gearboxes. The two generators were connected for split bus, non-synchronized operation. This meant that with both generators operating, each generator supplied power independently to certain aircraft busses. If one generator failed, it dropped off the line; and, at the same time, power from the remaining generator was provided to the busses of the failed (or turned off) generator. Current protection was provided to prevent a fault in one generator system from shutting down both generators; and either generator was capable of supplying power to the entire system. Each generator was activated auto matically when its control switch was in the ON position, and the generator was connected to its busses when voltage and frequency word within prescribed limits (approximately 50% engine rpm). A protection system within the generator control unit protected against damage due to undervoltage, overvoltage, over- and underfrequency, feeder faults, and generator locked rotor. If a fault or malfunction occurred, the generator control unit removed the affected generator from its busses. Except for an underfrequency condition, the control switch of the affected generator must be cycled to bring the generator back on the line after the fault or out-of-tolerance condition cleared. If the generator dropped off the line due to underfrequency and the prescribed frequency was restored, the generator would come back on the line automatically. A generator might be removed from its busses at any time by placing the generator control switch to UFF.

The electrical power generation and distribution system (EPGDS) was designed to provide electrical power to using subsystems of the aircraft during conditions of normal and emergency operation. Subsystems included avionics/instruments, environmental control systems, fuel, hydraulics, landing gear, lighting, propulsion and weapons delivery. The system was specifically designed to the following requirements:

(1) No single failure of the electrical bus will cause loss of the aircraft.



Figure 54. Aircraft II Electrical Power Generation System Schematic

- (2) Electrical power characteristics for electrical using equipment are equal to, or better than, MIL-STD-704.
- (3) Design and installation of the electrical system conforms to the requirements of specification MIL-E-25499 as specified.
- (4) Triplex redundancy provides assurance of an uninterrupted power supply for the three fly-by-wire (FBW) computer channels of the flight control system.

4.2.2.1 <u>DC Electrical Power</u> - Two 25-ampere, upregulated, static transformerrectifier (TR) units were provided to supply the DC power requirement of approximately 19 amoeres. Normally, each TR unit would deliver 50% of the total DC load, at a nominal operating voltage of 27.5 volts to its bus. In the event one transformer-rectifier failed, the other transformer-rectifier would power the entire DC system. The outputs of the TR's were connected in parallel; however, protection was provided through the use of circuit protectors and rectifier elements in the feeders. The circuit protectors eliminated the possibility of a battery bus fault resulting in the loss of both TR units. The rectifier elements protected against a failure in one TR unit affecting the other. No cockpit warning of single transformer-rectifier failure was provided.

With the AC power input between 195 and 210 volts, line-to-line, the two TR units conformed to the requirements of MIL-P-26517. At this input, the unregulated output was within the limits of 25.1 to 29.0 volts. During normal operation, with each unit sharing the total load, the output voltage might vary between 25.6 and 27.5 volts. Internal radio noise filtering of the TR unit met all provisions of MIL-I-26600 and MIL-I-6181D. A variable-speed, constant-volume blower permited normal in-flight cooling by forced air.

4.2.2.2 <u>Emergency Generator</u> - Emergency electrical power was provided by an all altitude APU/EPU driven AC/DC generator that was sized to provide sufficient power to ensure return of the aircraft to its base in the event of loss of primary AC power. The emergency electrical system was separate from the primary electrical system. If either or both main generators were inoperative or both transformer-rectifiers failed, or some combination of faults occurred. the emergency generator was activated and attached to the essential AC/DC busses.

The emergency generator was an air-cooled, brushless, single-bearing machine with a nominal speed range of 12,000 rpm. It was blast-cooled throughout the entire flight profile and over a pressure altitude range from sea level to 60,000 feet. Cooling air was provided by the environmental control system either normally or through the Ram Air Scoops provided (see Figure 10). The system was designed to operate for a minimum of five minutes without cooling air to ensure availability of electrical power in the event of a temporary loss of the cooling system.

The generator control unit provided the necessary functions for voltage regulation and control of the emergency power system. A static-type voltage regulator was incorporated in the control unit to provide steady-state and transient control of the output voltage at the essential AC and DC besses within the limits of specification MIL-STD-704. Power for buildup and operation of the system was provided by a PMG integral with the generator.

To connect the generator to the essential busses, the control unit provided two 1-ampere, 28-volt DC outputs for closing power transfer relays. Since the emergency power system was a "last-ditch" source of electrical power, the system protection was kept to a minimum. The control unit had an undervoltage sensing function that disconnected the emergency generator from the load bus, after any phase voltage fell below 70 volts, to protect utilizing equipment from damaging exposure to the decaying voltage. Emergency power reset was accomplished by deexciting and subsequently reexciting the generator.

In flight, activation of the APU/EPU was automatic when loss of primary power was sensed and the APU/EPU control switch was in the AUTO position. An ON switch position was provided to activate the APU/EPU, excite the generator, and attach it to the essential buses even when primary power was available. When the emergency generator contactor is picked up, the main DC bus was isolated from the essential DC bus by deenergizing the bus-tie contactor. The APU/EPU system would provide rated speed and power within approximately three seconds from the time when it has been activated.

Annunciators and a control switch were provided for the emergency generator system. The switch was a standard three-position switch, guarded in the AJTO (normal) position, and had the following control functions:

"OVERRIDE" - Overrides undervoltage trip protection function. Pilot option not recommended due to potential for damage to power utilizing equipment.

"AUTO" (Normal) - Normal switch position for all ground and flight modes. When APU/EPU is driving generator at normal speed, the generator shall be automatically excited and connected to essential busses. APU/EPU ELEC advisory legend (green) illuminates when emergency generator connected to essential bus. EMERG CEN FAIL warning legend (red) is displayed when the APU/EPU is on and the emergency generator has tripped off line. 'OFF' (Reset) - Provided for emergency generator to be deenergized and idsconnected from the essential busses via the essential bus contactors. Permitted generator reset by momentarily positioning switch to OFF and subsequently repositioning to AUTO. EMERG GEN OFF caution legend (yellow) illuminated.

4.2.2.2 External Power System - The aircraft-mounted external power system consisted of a standard external power receptacle, external power contactor, and power monitor unit to control application of external power to the aircraft. The power monitor unit prevented actuation of the external power contactor if phase rotation, voltage, or frequency of the external power system were not within specified limits. Only three-phase, 115/200-volt, 400 Hz, AC power was required from the ground power source to energize the aircraft bus system. All DC power was supplied by conversion units mounted in the aircraft.

Control of the external electrical power was by means of an external power switch located in the cockpit. The switch had ON and OFF positions. In the ON position and with no generator power, the external power supplied the total aircraft load. With either engine operating, the external power was automatically disconnected and the total load was supplied by the operating generator(s). With the switch in the OFF position, external power could not be supplied to the aircraft busses. The OFF position also provided reset capabilities in the event external power could not be applied to the air vehicle due to improper voltage or frequency tolerance.

4.2.2.3 <u>Battery System</u> - A battery system was provided in the aircraft to supply power to functions required in support of ground-starting the APU/EPU without the need for external electrical power. A secondary purpose was to provide limited emergency capability in the event of loss of all electrical power.

Normally, the DC start bus was supplied 24-volt DC power from the DC essential bus via the transformer-rectifier units when ground or vehicle powerwas available. A nickel-cadmium, 24-volt battery supplied power to the start bus when essential bus powerwas not available. Use of 24-volt DC powerwas dedicated to safety and special start functions, including fire detection and extinguishing, because the use of AC power offers weight advantages. The battery was maintained in a charged state by its own dedicated battery charger and only specific battery-utilizing systems would be exposed to battery-charging voltages. Load requirements of the DC start bus that determined the size of the battery are summarized as follows:

System	D. Start Bus Loads (Watts)
Intercom	30
Communications	86

No. 1 Digital Control Unit Logic	190
Multimode Display Unit No. 1	80
Display Electronic Unit No. 1	240
APU/EPU Control Unit	70
Aural Warning System	10
Cockpit Utility Lights	8

Total = 714 Watts

The battery selected was a 24-volt, 3 ampere-hour battery that would provide 60 amperes for two minutes at 0°F and for 40 seconds at -20°F. A switch located on the cockpit electrical power panel provided ON-OFF control of the battery power. With the aircraft busses powered by external power on the air vehicle generating system, and if the battery switch was ON, the battery would be charged by the battery charger.

4.2.2.4 Electrical Load Analysis - The housekeeping loads, as shown in Reference 8 provided the basis for determining the AC and DC electrical power required for Aircraft II during various operating modes. These loads provided the design criteria for sizing and selecting the electrical power generator, control and distribution equipment for the aircraft. The maximum demand for primary AC power was on the order of 50 KVA and occurred during the combat portion of the mission. Since the majority of the electrical loads were of the continuously operating type (15 minutes or longer), and the peak 5 sec. load was 74 KVA (70.5 KW x .095 pf). The two primary generator ratings were determined to be 40/50 KVA each.

4.2.2.5 <u>Emergency Generator</u> - The emergency loads of the AC/DC load analysis represented the loads supplied by the APU/EPU-driven generator via the essential busses. The emergency generator was sized at 7 KVA based on the emergency continuous housekeeping load shown on page A-2 of Appendix E. Of this .75 KVA was assumed to require DC power.

4.2.2.6 <u>Fly-By-Wire Power Supply</u> - To supply power for the fly-by-wire (FBW) flight control system, three flight control power supply (FCPS) units were provided. Each FCPS unit was isolated from the others by diodes and was dedicated to one of the three flight control computer channels. During normal operating modes of the electrical power system, each FCPS unit received 28 volts DC from two of the three DC busses (main, essential, and the battery bus), then passed it on to the three flight control channels. To ensure that there were no voltage transients or interruptions as a result of switching operations normal to aircraft power systems, each FCPS unit also contained a 24-volt, 1.4 ampere-hour battery which was connected to its input DC power bus. The FCPS batteries were continually charged by the circuitry in the control unit, but were not sized to provide an emergency source of power for a sustained period of time.

4.2.2.7 <u>Primary AC Generators</u> - As determined by the load analysis of Reference 8 (pages A-1 and A-2 with ECS loads deleted. See Paragraph 4.2.2.4), the generating system of Aircraft II was rated at 40/50 KVA of 115/200 volt, three-phase, 400 Hz power. Based largely upon weight, maintainability, and availability, two variable speed constant frequency (VSCF) generators were selected for tradeoff evaluation: (1) the DC-link generating system; and (2) the cyclo-converter generating system.

The primary advantage of both systems was that they eliminated the constant speed mechanical/hydraulic drive of the conventional integrated drive generator (IDG) system, and coupled the engine gearbox directly to the VSCF generator. With variations in engine speed, the frequency of the generator output was converted to a constant output frequency of 400 Hz by means of an electronic converter. By replacing the mechanical/hydraulic constant speed drive (CSD) with a sold-state power converter, it was felt that the reliability, maintainability, and life cycle cost of the generating system would be significantly improved (see Reference 14 and 23).

The basic difference between the DC-link approach and the cycloconverter was the type of electronic switch used in the converter and the type of input to the converter. In the case of the cycloconverter, the input was a multiphase, varying frequency waveform. The DC-link system, as the name implies, used a DC voltage as the converter input. The electronic switch in the cycloconverter was an SCR, while transistors were used in DC-link systems as the switch elements. Table 39 compares the different types of typical 30/40 KVA generating systems with respect to weight, efficiency, and operating temperatures.

#### Table 39

#### Aircraft Generating System Comparisons

	IDG/CSD	Cycloconverter	DC-Link
Input Oil Temp Limitation	150°F	80°C	120°C
Efficiency (30/40 KVA)	66.4%	71.4%	76.3%
Weight (30/40 KVA)	79 1bs	77 lbs	82 1bs

For both VSCF systems, a high-speed gearbox and a narrow speed range were desirable to minimize overall system weight. As the speed range decreased (i.e. 1.8:1), for a fixed maximum upper speed, the generator weight decreased. Reliable high-speed (27,500 rpm) gearboxes and/or speed increasers were within the state-of-the-art and being flight-tested on the F-18 and F-5G aircraft.

Temperature and type of cooling medium can have a direct impact on the choice of the VSCF system. This criteria directly relates to the temperature capability of the power switching components. The DC-link system power transistors can operate at a higher temperature limitation  $(120^{\circ}C)$  than the thyristors of the cycloconverter system ( $80^{\circ}C$ ). Two predominant cooling methods were employed by the system suppliers: (1) spray-oil cooling by the DC-link system; and (2) conduction-oil cooling by the cycloconverter system. Each of these techniques had its advantages and limitations with regard to system weight, cost, efficiency, and reliability. It also had some bearing on the aircraft oil management system.

Both systems produced a quality of electrical power that met or exceeded the requirements of MIL-STD-704. Technological advances in the area of electrical generating systems have been largely directed toward the development of solid rotor generators, using rare earth samarium cobalt magnets, and developing a microprocessor to perform all the control circuit functions with fewer electronic components. Implementation of most of the new hardware advances improved VSCF size, weight, cost and failure rate very little. Only when most of the control circuits are replaced by a microprocessor could significant improvements be realized.

In spite of the lower weight and better part load efficiency of the cycloconverter approach and the fact that it was chosen for Aircraft I AC load requirements (see Paragraph 4.1.6), the DC Link approach was selected for Aircraft II. This selection was made because of the higher temperature tolerance of the DC Link approach considering the fact that the elaborate evaporative cooling techniques necessary for Aircraft I would not be used in Aircraft II (no inverters) and that fact that the cycloconverter's portion of the total system output was only 25 KVA versus Aircraft II's output of 50 KVA.

4.2.2.8 <u>Fly-By-Wire Control System Arrangement</u> - The fly-by-wire system in Aircraft II was essentially identical to the signal system used in Aircraft I. The Aircraft I arrangement is shown in Figure 37. Like the Aircraft I arrangement, the Aircraft II system employed five microprocessors in the wing and tail, three microprocessors in the nose, and two redundant flight data computers remotely located from each other in the fuselage. Also, like the Aircraft I system, Aircraft II used electro-optical signal interties and the signal transmission lines (see Figure 35) connecting similar components (i.e., motors, actuator clutches, etc.). In the case of Aircraft I, the signal inputs to the actuator were typically fed through an inverter while the comparable item in Aircraft II was a servo valve. In either instance, the signal power and signal characteristics were considered essentially identical. The only significant difference between the signal system used on Aircraft II versus that used on Aircraft I was the fact that Aircraft II had only three power supplies (see Figure 55), whereas Aircraft I had four (see Figure 37).
In spite of this difference the impact on weight and reliability was negligible. Even though Aircraft I had four power supplies, power supply #3 in Aircraft II handled the same power demand as power supplies #2 and #3 in Aircraft I so the power supply weight was considered essentially unchanged. With reference to reliability, Table 40 shows the control and power supply interrelationship on Aircraft II and illustrated that reliability was not impacted. As a typical example, Table 40 illustrates that, if any two power supplies failed to the e wing functions, roll, pitch, and lift control would be maintained. Referring to the table, if power supplies #2 and #3 failed, roll control would be maintained with the right hand outboard flap (OTE-RH) supplemented by the roll function of the left hand and right hand midspan trailing edge flaps (MSTE-IH and MSTE-RH). In a similar manner the pitch and lift functions would be maintained by the right and left hand inboard trailing edge surfaces (ITE-RH and ITE-IH) supplimented by the two midspan trailing edge surfaces. Based on the above analysis, and considering that any differences resulting from the fact that the two systems were supplied different types of power (i.e. 270 VDC for Aircraft I and 115/200 VAC for Aircraft II) would appear as deltas in the distribution system, the fly-by-wire system was eliminated as an item in the trade study.

4.2.2.9 Aircraft II Wiring System - The weight of the Aircraft II wiring, subject to trade, was determined in a manner similar to that used for Aircraft I as discussed in paragraph 4.1.9.2. The wiring weight, using this technique, was found to be 23.0 Lb.

4.2.2.10 <u>APU/EPU Sizing</u> - The Rocketdyne Division of Rockwell had performed extensive development work on all altitude APU/EPU and/or super integrated power units (SIEPU). Therefore, Rocketdyne was asked to evaluate and size an optimum APU/EPU. Four configurations were evaluated. of which. the unit shown in figure 55 was selected. This unit differed from, and/or expanded upon, the unit shown in figure 45 in certain areas. The significant differences between the two were the addition of two heat exchangers and a fuel accumulator to the final selected configuration (figure 55). The weight of the configuration shown in figure 55 was determined as 399.0 Lb and its envelope was 42 in. X 36 in. X 20 in.

4.2.2.11 <u>Aircraft II Electrical System Weight</u> - The weight of the major electrical components making up that portion of the electrical system subject to trade are shown in Table 41. In general these were the power generation and distribution components shown in the schematic of figure 54. Table 41 shows that the total electrical system weight subject to trade was 287 Lb.

4.2.2.12 Aircraft II Total System Weight Subject to Trade - Table 42 shows the total Aircraft II weight subject to trade and lists it as 2,319 Lbs.

AIRCRAFT IN CONTROL AND POWER SUPPLY INTERRELATIONSHIP TABLE 40.

a a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantial and a substantia

				Z	IG A	STUA	1101	2 E	NCTI	NO	SEE	800	E E			
0	TEL	OTE	HSTE	15TE	ASTE	ITE	Ĩ	ITE	ITE	ITE	ITE	<b>JICH</b>	45TE	<b>H</b> 5TE	JU	OTE
1	ΗI	LH	LH	LH	LH	ЕН	H-	LH	RH	RH	RH	R	₹	R	Ð	R
HYDRAULIC		~	-	~	r		0	r	ď	2		đ	3		で	
POWERSYS. NO.		1	•	1	,	-	,	,	5	1	·	,		,		
DRIVEN BY		6	-	0			0			2	-		2	_	2	_
AMAD (ENG) NO.		3	•	3		-	,				•			·		·
DRIVEN BY					"			9	ď		_	(1)	~			
APU/EPU					7			,	5		1	,	1		1	T
MICRO-	4	0	-	0	4	~	0	P	4	a		4	đ		a	-
PROCESSOR	 7	J	•	Ľ	•	-	1	-	•	5	-	•	,	·	5	·
ELECTRICAL (2)	(1	s		6	ন		?	a	'n	0	-	a	0		~	_
POWER SUPPLY	1	J	-	2	2	-	ן נ	2	5	1	-	~	•	•		•

......

- SEE FIGURE 4.21 FOR WING "W" MICROPROCESSOR NUMBERING SYSTEM SEE FIGURE 4.2-13 FOR ELECTRICAL POWER SUPPLY NUMBERS  $\bigcirc$

• •

- OTE MSTE 0
- = OUTBOARD TRAILING EDGE SURFACE = MIDSPAN TRAILING EDGE SURFACE = INBOARD TRAILING EDGE SURFACE (AILERON) = RIGHT HAND = LEFT HAND = LEFT HAND 1 1 1 1

  - RH H



Figure 55. Final APU/EPU Configuration

	QUAN	TITY	UNIT	TOTAL
EQUIPMENT ITEM ()	ĹΡ	ER	WEIGHT	WEIGHT
	$\geq$	<u>A/c</u>	(LB)	<u>(LB)</u>
PRIMARY GENERATOR / CONVERTER (49/50)	KVA)	2	00.88	176.00
TRANSFORMER RECTIFIER (25 AMP)		2	4.00	8.00
EMERG. GENERATOR/CONVERTER (7 KVA-AC, 30	DAMP 0	1	18.00	18.00
BATTERY (JAMP-HR, 24 VOLT)		1	10.50	10.50
BATTERY CHARGER (IDAMP)		1	4.00	4.00
FCPS BATTERIES		3	4.30	12.90
GENERATOR LINE CONTACTOR (3PDT-50K)	/A)	2	3.30	6.60
GENERATOR LINE CONTACTOR (3PDT-7KV	(A)	ł	0.72	0.72
EXTERNAL POWER CONTACTOR (3PDT-50	KVA)	۱.	3.30	6.60
BATTERY CONTACTOR (3PDT-30AMP)	·	1	0.60	0.60
BUS TIE CONTACTOR (SPDT-30 AMP)		١	0.60	0.60
AC CIRCUIT BREAKER (25 KVA)		2	0.18	0.36
DC CIRCUIT BREAKER (30 AMP)		5	0.12	0.60
EXTERNAL POWER CONNECTOR (SOKVI	۹) (۱	1	2.40	2.40
EXTERNAL POWER MONITOR (SO KVA)	•	ł	2.04	2.04
COMPONENT SUPPORTS			-	5.26
WIRING (FEEDER WIRES ONLY)				23.00
CONNECTORS (FEEDER CONNECTORS ON	נא)			12.12
MAJOR EQUIPMENT-TOTAL	L WE	IGH	Т	287.00

1 SEE FIGURE 54

TABLE 42. AIRCRAFT II TOTAL WEIGHT SUBJECT TO TRADE

بالعامية بعدد يعطيهم

	QUANTITY	UNIT	TOTAL
EQUIPMENT ITEM ()	PER	WEIGHT	WEIGHT
	- M/C		(LB)
AMAD-AIRFRAME MTD. ACESSORY DRIVE	2	110.0	220.0
ATS/M - AIR TURBINE START MOTOR	2	32.0	64.0
APU/EPU-AUXILIARY/EMERGENCY POWER	UNIT	3990	399.0
LC - LOAD COMPRESSOR	1	75.0	75.0
PNEUMATIC DUCTING AND FITTINGS			8.4
PNEUMATIC CHECK VALVE (11/2" PORT		0.3	0.3
PNEUMATIC CHECK VALVE (1%" PORT	·) 2	0.2	0.4
PNEUMATIC SOLENOID SHUTOFF VALVE	4	1.2	4.8
PNEUMATIC GROUND CONNECTION		0.8	0.8
APU/EPU START VALVE	1	0.3	0.3
PMG - PERMANENT MAGNET GENERATO	RI	0.7	.0.7
SUPPORTS AND MISC.	-	`	60.7
ELECTRICAL SYSTEM (SEE TABLE 41			287.0
HYDRAULIC SYSTEM (SEE TABLE 38			1197.6
Т	TAL WEI	GHT	2319.0

1 SEE FIGURE 44

.....

. [

1

#### 5.0 TRADE STUDY RESULTS

5.1 Weight Trade - The weight summary for the two primary aircraft configurations studied are shown in table 43. It can be seen in the table that the gross takeoff weight of Aircraft I (the all electric version) was 1245 lbs. heavier than that of Aircraft II (the more conventional hydraulic-electrical configuration). From tables 27 and 42 it can be seen that the difference in the basic system weights subject to trade was 498 lbs. From this it is apparent that the growth factor for this type of airplane was 2.5.

5.2 Reliability and Maintainability Trades - The Reliability and Maintainability (R&M) trade was primarily oriented toward identifying the differences affecting the operating and support costs between the Aircraft I and II configurations. Only those major equipment items impacted by the actuation concept were identified for the R&M trades. The basic approach for the R&M trade was as follows:

- 1. A list of major components affected by the configuration differences was identified including actuators, electric power and generation system, and hydraulic power system.
- For each component identified, R&M parameters based on projecting current operating data to that expected in the 1990+ time frame were estimated. Current operating data for the components used included the following sources:
  - a) F-15 AFM 66-1 Maintenance data for the period Cctober 1978 through September 1979 summarized by the Rockwell International Maintenance Analysis Model (MAM).
  - b) B-1B Aircraft "Reliability and Maintainability Allocations, Assessments, and Analysis," Rockwell International Report, NA-81-745-1, dated 2 April 1982.
  - c) A-7 Aircraft "Design Development and Evaluation of Lightweight Hydraulic System Hardware - Phase I, NADC-77108-30, North American Aircraft Division, Rockwell International Corporation, Contract N 62269-78-C-0363, 30 January 1981.
  - d) Nonelectronic Reliability Notebook, Revision to Section 2, RADS-TR-69-458.
- 3. Estimates were made for:
  - a) Mean-Time-Between-Maintenance (MTBM) including inherent failures, induced failures, and no defects resulting from a suspected failure.
  - b) Mean-Time-Between-Removal (MTBR) to reflect demands on the supply system and the intermediate level maintenance shops.

## TABLE 43. WEIGHT SUMMARY

-----

ad share

		- BASEL	INE
	A/c I	1/c II	
	ALL ELECT	HYD - ELECT	
STRUCTURE GROUPS	(7948)	(7683)	
WING GROUP	1991	1925	
TAIL GROUP - HORIZONTAL	270	261	
· VERTICAL	302	292	
BODY GROUP	3872	3743	
ALIGHTING GEAR GROUP . MAIN	902	949	
			<u></u>
- ARK25TING			<del></del>
AIR INDUCTION SYSTEM	245		
AIR INDUCTION STITEM	- <del>  672</del> -		
			-+
PEOPULSION CROUP	(4610)	(4456)	
ENGINE (AS INSTALLED)	3125	2982	
ACCESSORY GEAR BOXES & DRIVES	180	220	
EXHAUST SYSTEM			
COOLING & DRAIN PROVISIONS	39	30	
ENGINE CONTROLS	43	40	
STARTING SYSTEM	154	166	
FUEL SYSTEM	1075	1018	
FAN (AS INSTALLED)			
HOT GAS DUCT SYSTEM			
	(2040)	(6500)	┈╄╼╼╴╾┥
EQUIPMENT GROUPS	1655	(6505)	
FLIGHT CONTROLS GROUP	1033		
INSTRUMENTS CROUP			
HYDRAULIC & PNEUMATIC GROUP		679	
ELECTRICAL GROUP	1131		
AVIONICS GROUP	2290	2290	
ARMAMENT GROUP	300	300	
FURNISHINGS AND EQUIPMENT GROUP	595	595	
AIR CONDITIONING GROUP	630	630	
ANTI-ICING GROUP			
PHOTOGRAPHIC GROUP			
DRAG CHUTE ASSY			
DRAG CHUIE ASSY		·	
TATAL WEICHT ENBTY	119598	(18648)	
COSW		430	
	176	170	
FUEL - USABLE	11909	11623	
OIL - ENGINE	20	19	
PASSENGERS / CARGO			
ARMAMENT			
A/A MISSILES	515	515	
MISSILE LAUNCHERS	155	155	
PAYLOAD	4330	4330	
PAYLOAU FAIRINGS	5(	51	
PAVE SUIFID EVOCUDARIES		━━ <u>┼</u> ──⋛┤ <u></u> ┼────	
	hazant		
TAKEDEE CROSS WEICHT	37229	36644	
FLIGHT DESIGN GROSS WEIGHT			
LANDING DESIGN GROSS WEIGHT			

FORM 1034-C-9 AEV. 12-74

c) Mean-Time-to-Repair (LTTR) to reflect the average on aircraft time to repair.

-----

ł

d) On aircraft Maintenance-Man-Hours-Per Flight Hour (NMH/FH) to reflect an average maintenance man-hours per flight hour used to maintain the aircraft.

The above parameters were estimated for each listed component and for all the components are shown in tables 44 and 45.

Comparison of the reliability/maintainability results shown by the totals for the two aircraft configurations shows that the hydraulic Aircraft II has 11.68% improvement in the MTBM, and 18.4% improvement in the MMH/FH over the electric Aircraft I. The MTTR values are practically the same for both configurations. Evaluation of the MTBR's indicates that there are approximately five (5) times as many repairs through defective or suspected replacement on the electrical aircraft.

4. <u>Mission Completion Success Probability (MCSP)</u> - Aircraft I and Aircraft II system designs provide practically the same degree of redundancy for the actuation systems. Therefore, relative trends in the MCSP for the affected aircraft configurations can be established on the basis of comparison of total failure or maintenance rates estimated for all the components listed respectively for each type of aircraft.

Based on the data in Tables 44 and 45, the MCSP of the Hydraulic Aircraft II is somewhat higher than the MCSP of the electric Aircraft I.

The unreliability estimate of the electrical Aircraft I based only on the failure/maintenance rate count of the identified components is 11.68% higher than the unreliability of the 8000 psig Hydraulic Aircraft II.

- 5. <u>Design Reliability-Maintainability Comments</u> The following summarizes some of the design features of the two proposed aircraft configurations:
  - a) The 8000 psid hydraulic power generation and distribution system used smaller size components and tubing than the 3000 psig system. This feature considerably improved accessibility, and reduced maintenance times and costs as compared to the 3000 psig system.
  - b) Use of smaller size tubing permitted use of coiled tubing to the exclusion of swivel joints which reduced leakage and, hence maintenance costs, and improved the reliability as compared to the 3000 psi system.
  - c) In the year 1990 + on-aircraft maintenance of the electricalmechanical drive components will be limited primarily to remove/ replace activities. This would reduce flight line and increase intermediate level man-hour requirements.

TABLE	44	RI	ELI	LABILITY-MAI	INTAINA	311	LIIY	DATA
AIF	RCRAF	Т	1	(ELECTRIC)	(SHEET	1	OF	2)

COMPONENT	OUANTITY	MDR PER 10 ⁶ HOURS	MTBM HRS.	MTBR HRS.	MTTR HRS.	MMH PER FL-HR.
INBOARD FLAP ACT.	6	250.0	4000	4400	3.5	0.00153
MIDSPAN FLAP ACT.	6	250.0	4000	4400	3.5	0.00153
AILERON ACT.	4	331.0	3021	3300	3.5	0.00184
UPPER RUDDER ACT.	6	331	3021	3300	3.5	0.00184
LOWER RUDDER ACT.	6	331	3021	3300	3.5	0.00184
LEAD. EDGE FLAP ACT	12	331	3021	3300	3.75	0.00184
CANARD ACT.	4	120	8333	9166	3.5	0.00073
THRUST VECTOR VANE	2	120	8333	9166	3.0	0.00073
ENG. EXT. FLAP ACT.	4	120	8333	<b>916</b> 6	3.2	0.00073
ENG. PLUG THR. ACT.	2	240	4167	4584	3.2	0.00147
ENG. THRUST REV. ACT.	2	180	5556	6110	3.7	0.00110
NOSE L. G. ACT.	1	120	8333	9166	3.2	0,00073
MAIN L. G. ACT	2	120	8333	9165	3.7	0.00073
NCSE GEAR STEER.	1	148.5	6734	7407	3.2	0.00091
M. G. BRAKES	2	148.5.	6734	7407	3.5	0.00091
ECS RH RAMAIR	1	120	8333	9166	3.5	0.00073
ECS LH RAMAIR	1	120	8333	9166	3.5	0.00073
ECS EXCH. DOOR E. B.	1	120	8333	9166	3.2	0.00073

. .

COMPONENT	QUANT.	MDR PER 10 ⁶ HOURS	MTBM HRS.	MTBR HRS.	MTTR HRS.	MMH PER FL-HR
ARMAMENT ACT	1	90.9	11,000	12,100	3.7	0.00056
REFUEL RECEPT. MOTOR	1	90.9	11,000	12,100	3.25	0.00056
GENERATOR-270VDC	4	489.7	2040	5250	3.25	0.00300
DRAIN+FILL CON VERT.	2	40.0	50,000	55,000	3.0	0.00012
TRANSF-RECTIF.	4	11.5	86,956	250,000	3.0	0.00070
CYCLOCONVERTER	4	40.0	25,000	27,500	3.0	0.00024
GENERATOR - CONTROL	4	89,94	11,119	125,000	3.0	0.00055
BATTERY-270VDC	1	348.8	2867	8768	3.0	0.00213
APU MOTOR	1	148.5	6734	7407	3.25	0.00091
APU GENERATOR	2	326.4	3063	9370	3.25	0.00200
APU TRANS-RECT.	2	11.5	86,956	250,000	3.0	0.00006
APU CYCLOCONV.	2	20.0	50,000	55,000	3.0	0.00012
INVERT. (START)	2	20.0	50,000	150,000	2.5	0.00011
REVERSTBLE SCR	2	12.54	79,745	87,720	2.0	0.00007
START RELAYS	10	6.27	159,490	175,439	2.5	0.00003
TOTAL	105	21,155	47.27	63.38	3.25	0.11681

# TABLE 44RELIABILITY-MAINTAINABILITYDATAAIRCRAFT I (ELECTRIC)SHEET 2 OF 2)

		MDR	MTBM	MTBR	MITR	MMH PER
COMPONENT	QUANȚ.	HOURS	HRS.	HRS.	HRS.	FL-HR
INB. FLAP ACT	6	289.6	3452	23014	3.75	0.00153
MIDSPAN FLAP ACT	5	257.4	3884	25893	3.75	0.00136
AILERON ACT	4	231.7	4315	28767	3.75	0.00122
UPPER RUDDER ACT	6	231.7	4315	28767	3.75	0.00122
LOWER RUDDER ACT	6	231.7	4315	28767	3.75	0.00122
L. E. FLAP ACT.	12	231.7	4315	28767	3.75	0.00122
CANARD ACT.	4	148.3	6743	53944	3.5	0.00078
1. V. VANE ACT.	2	148.3	6743	53944	3.5	0.00078
ENC. EXT. FLAP ACT	4	185.3	5394	43152	3.5	0.00098
ENG. PLUG THR. ACT	2	331.1	3020	20,133	3.5	0.00175
ENG. THRUST REV.	2	231.7	4315	28767	3.0	0.00122
NOSE L. G. ACT	1	170.9	5850	46800	3.75	0.00090
MAIN L. G. ACT	2	206.6	4838	38704	3.00	0.00109
NOSE GEAR STEER	1	231.6	4316	28773	3.5	0.00122
MAIN GEAR BRAKES	2	257.4	3884	25893	3.5	0.00136
ECS RH RAMAIR	1	74.1	13486	107,888	3.50	0.00039
ECS LH RAMAIR	1	74.1	13486	107,888	3.50	0.00039
ECS EXCH. DR ACT.	1	148.3	6743	53,944	3.75	0.00078

### TABLE 45 RELIABILITY MAINTAINABILITY DATA AIRCRAFT II (HYDRAULIC) (SHEET 1 OF 2)

		MDR PER 106	MTBM	MTBR	MITR	MMH PER
COMPONENT	QUANT.	HOURS	HRS.	HRS.	HRS.	FL-HR
ARMAMENT-ACT	1	231.7	4315	28767	3.0	0.00122
REFUEL RECEP. MOTOR	1	115.2	8680	37200	3.5	0.00061
PRIM. PIMP	4	356.2	2807	12,047	3.5	0.00188
apu pump	1	296.9	3368	14455	3.5	0.00157
PRIMARY RESERV.	2	30.9	32,320	300,000	3.0	0.00016
EMERG. RESERV.	1	30.9	32,320	300,000	3.0	0.00016
ACCUMULATOR	1	83.3	12,000	120,000	3.5	0.00044
BRAKE RESERV.	2	30.9	32,300	300,000	3.5	0.00016
GENERATOR 40 KVA	2	489.7	2042	6250	3.25	0.00259
GENERATOR 10 KVA	1	326.4	3063	9370	3.25	0.00172
TRANSF/RECTIF.	2	11.5	86956	250,000	3.0	0.00006
1NVERTER	1	20.0	50000	150,00	2.5	0.00011
VOLT. REGUL.	1	89.3	11,200	33600	2.0	0.00047
BATTERY	2	348.8	2867	8760	2.0	0.00184
POWER CONTAC.	12	9.5	105,241	210,000	2.0	0.00005
TOTAL	97	18.942.9	52.7	303.08	3.31	0.09866

# TABLE 45RELIABILITY-MAINTAINABILITYDATAAIRCRAFT II(HYDRAULIC)(SHEET 2 OF 2)

-----

معمدتين والم

ł

- d) Fly-by-wire signal transmission in conjunction with powerby-wire required exceptionally high electrical power reliability for future airborne all-electrical power systems.
- e) To meet the failure and reliability requirement of the 8000 psig hydraulic Aircraft II, the electric Aircraft I had to provide at least 3 completely independent dedicated power systems to match redundancy in Aircraft II.
- f) The actuators of Table 45 (all-electric) included the reliability impacts of their inverters which largely accounted for their deficient reliability with respect to the actuators of Table 5.3 (hydraulic-electric).

5.3 <u>Life Cycle Costs</u> - Life cycle cost estimates were developed for each of the "Airplanes" defined. The baseline aircraft (Aircraft II) was an electric-hydraulic powered aircraft, and the alternate configuration was an "all electric" approach (Aircraft I).

5.3.1 <u>Methodology</u> - The estimates were developed utilizing the Integrated Aircraft Life Cycle Cost Model II (IALCCM II). IALCCM II was a computer program developed by Rockwell in support of previous advanced tactical fighter studies which estimated RDT&E, Production, and Operations and Support Cost. This model provided preliminary cost data during conceptual and preliminary design states.

5.3.2 <u>Ground Rules</u> - The ground rules for developing the cost estimates were as follows:

- 1. Number of flight test aircraft (10)
- 2. Number of production aircraft (500)
- 3. Number of aircraft in the field (432)
- 4. O&S cost for field aircraft
  - a. 10 year operational life
  - b. 5 year buildup

į

- c. 24 aircraft per squadron
- d. 25 flight hours/aircraft/month
- e. Assumed fuel cost \$1.26/U.S. dollars (1982 dollars)
- 5. LCC was submitted in constant 1982 U.S. dollars and 1995 dollars. (The 1995 dollars were developed utilizing inflation factors from the USAF Cost and Planning Factor Manual [AF Regulation 173-13]).

5.3.3 Cost Summaries - Cost data were developed for each of the two configurations. LCC summaries are provided in tables 46 and 47 in 1982 dollars, and table 48 and 49 in 1995 dollars.

TABLE 46 AIRCRAFT I LIFE CYCLE COSTS - 1982 DOLLARS

;

AIRCRAFT:ATSI ACTUATION TRADE STUDY ALL ELECTRIC COSTS IN FY 1932 MILLION DCLLARS

LIFE CVCLE COST	16096°95	TOTAL PRODUCTION AIRCRAFT	503
	~	TCTAL UE	432
		CONMAND SUPPORT	42
R0T&E	1614.16	ATTRITION	26
AI GFRAM "	1012.53		
PR CPUL 3 I DA	544.24	TOTAL PROTETYPE AIRCRAFT	01
AVIONICS	57.40		
OTHER	0•0		
ACGUISTEL JN	6373-10		
PRCOUCTIGN	5985.89		
FLYAU AY	5347.24	UNIT AVERAGE FLYAHAY COST	10.694
INITIAL SPARES	<b>339.</b> HI	AIRFRAME FLYAWAY	7.210
INITIAL CSE	170.52	PROPULSI CN FLYAWAY	2.162
TFAINING EQUIPMENT	69.51	AVIONICS FLYAWAY	1.322
TECHNICAL DATA	58.82	GTHER FLYAWAY	0 • 0
OFFER INVESTHENT	387.21		
TAANS PERTATION	119°72		
INITIAL PERSONNEL ACQUISITION	105.96	-	
INITIAL PERSCANEL TRAIMING	161.53		
Facilities	0•0		
7			
TUTAL OPERATIONS FOR LO YEAPS	8109470	<b>JPERATICNS DATA:</b>	
RECURAL 13 INVESTMENT & MISC. LOGISTICS	5540.43	UE PER SCLADRON	24.0
COMPONIOSE CSF	516.28	UTIL RATE FHRS / UE / MONTH	25.0
AVIATION FUEL & 1.26/6AL	1671.30	CREW RATIC	1.1
PASE LEVEL MAINTENANCE MATERIAL	365.44	PILOYS/CREW	1.0
CEPOT LEVEL MAINTENANCE	2114.13	OTHER OFFICERS/CREW	<b>0 • C</b>
CLASS IN MODIFICATIONS	207.62	MAINTENANCE MAN HOURS/FHR	28.3
STAINING MONITIONS	201.85	PUNIT HAINT MENZUE	6.0
REPLETICHMENT SPARES	455.16	FUEL FLON GPH	1024.0
VEHICULAR EQUIPHENT	8.65	REPL SFARES SFFHR	351.2
PAY AND AL! DHANCES	2014.05	BASE MAINT MTL S/FHR	282.0
HFP- BOJ/RPH SLPPJRT	132.35	DEPOT MAINI \$/FHR	1460.0
MECICAL SUPPORT	90.18	DEPOT PAINT S/UE/YR	51387.4
PEASONVEL SUPPORT (PCS MAVES)	64.54	CCMMCN DSE \$/UE/YP	119508.3
PEFSCANEL ACQUISITION AND TRAINING.	268.15		

TABLE 47 AIRCRAFT II LIFE CYCLE COSTS - 1982 DOLIAPS

ľ

:

The second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second

....

AIRCRAFT: AT 31 ACTUATION TRADE STUDY RASELINE COSTS IN FY 1982 MELLION ONLLARS

LIFE	CYCLE COST	12313.76	TOTAL PRUDUCTION AIRCRAFT	500
			TOTAL UL Command Support	N N 7 4 7
201		1437.60	ATTRITION	26
	A I P F R AN 1	B 4 6 • D 4		
¥\$	PR OPULS TON	534.16	TOTAL PROTOTYPE ATRCRAFT	10
	AVIONICS	57.40		
Ľ	OTHER	0•C		
AC.		5574.29		
	PRODUCTION	5307.36		
	FLYAWAY	4741.10	UNIT AVEPAGE FLYAWAY COST	9.482
	INITIAL SPARES	301.29	AIRFRAME FLYAWAY	6.088
	IWITIAL DSE	151.19	PROPULSION FLYAWAY	2.072
	TRATIZING EQUIPHENT	61.63	AVIDNICS FLYAWAY	1.322
	TECHNICAL DATA	52,15	<b>STHER FLYAWAY</b>	0.0
J	DTHER INVESTMENT	366+93		
	TRANSPORTATION	106.15		
	INITIAL PERSONNEL ACQUISITION	102.80		
2	IMITIAL PERSONNEL TRAINING	157.98		
48	FACIL TTES	୍ଦ ଅ		
101	TAL OPERATIONS FOR 10 YEARS	5701+88	OPERATIONS DATA:	
	RECURRING INVESTMENT & MISC. LOGISTICS	3226.29	UE PER SQUADRON	24.0
	COMMON OSE	199.72	UTIL RATE FHRS/UE/MONTH	25 e U
	AVIATION FUEL 2 1.26/GAL	1623.42	CREW RATIC	1.1
	HASE LEVEL MAINTENANCE MATERIAL	122.97	PILOTS/CREW	1.0
	CEPOT LEVEL MAINTENANCE	459.85	OTHER OFFICERS/CREW	3.0
	CLASS IV 40DIFICATIONS	184.09	MAINTENANCE MAN HOURS/FHR	26.3
	SNOILING SNINLELL	201.35	HUTIT MAINT MENJUE	6.0
	REPLENISHMENT SPARES	425.E9	FUEL FLOW GPH	6.166
	VEHICULAR EQUIPHENT	8.30	REPL SPARES S/FHR	328 . 5
	PAY AND ALLGUANCES	1940.18	HASE MAINT MTL \$/FHR	6**6
. 4	HEP- HJ3/RPM SUMPJRT	127.15	DEPAT MAINT S/FHR	159.9
æ	HEDICAL SUPPORT	86,56	DEPUT MAINT \$/UE/YP	<b>\$6464.3</b>
<b>u</b> t i	ERGONNEL SUPPORT COCS HOVES	62.10	COMMON ASE STUEZER	46231.7
<b>.</b>	DEPRONTEL ACQUISITION AND TPAINING	259.61		

TABLE 48 AIRCRAFT I LIFE CYCLE COSTS - 1995 DOLLARS

MIRGRAFISAISI ACTUMBIUN TRAVÊ SIUNY ALL LLÊLÎRÎC Lusij îr fy 1995 Millien Dûllârs

LIFL CYCLE CUST	51147.62	IUTAL PAUDUCTIUN AIKCKAFT Iŭtal de Cummand Suppürt	500 432 43	
KJ LGL Girtrami	5107.83 1984.02	ATTRETON	20	
PREPULSION Aviualos Uther	1000 57.40 000	TUTAL PAUTUTYPE AIRCRAFT	10	
ACLOINTIAN - FROTOFTIAN	11/64.29 11/14-29			
FLYNNAY	9644.00	UNII AVENAGE FLYAMAY CUST	19.686	
liillal Spakes	dc.250	AIKFKAML FLYAHAY	14.129	
JNET AAL USL	313 <b>.91</b>	PRUPULSIUN FLYAMAY	4.237	
INTELEC CALIFACE	16.7.1	AVIUNIUS FLYAWAY	1.322	
	100.28	UTHER FLYAWAY	0°0	
TAACSUALATICA TAACSUALATICA	ひらょ キャット			
Intitat Peksukket ACQUISITIUN	201103			
INTIAL PENSUMMEL INALMING	50010			
FALLLTLES	0.0			
Lulat yPtrafaustur ju Years		UPERALLUNS DALA:		
AE-UKAING ANVESTMENT & MISE. LUGISTICS	10007-27	UE PLA SQUAUKON	24.0	
CLANDN JSE	20.202	UTIL ANTE FARS/UE/MUNTH	25.0	
AVIAILUN FUEL & COMPANY	3272.03	LALM AAILU	1.1	
BASE LLVEL MAINILIANCE MAIENIAL	710.10	FILUIS/CREM	1.0	
ULP31 LEVEL MAINILNANCE	15.2302	UTHER JFFICERS/CREM	0.0	
Statient tradit verties	2012 + 2010		28.3	
	44.ckc	MUNEL MAINI RENJUE	6.0	
KEPELNISHMENT SPANES	14.100	FULL FLUN GPH	1024.0	
YENTUCAS ELUTPALIT	10.44	KEPL SPAKES \$/FHK	664.7	
TAY AND ALLUNANCES	54°U°43	DASE MAINI MIL J/FHK	947.9	
LAUTIN MAXADO	<b>(), 9</b> 42	DEPUT HAINT */FHK	2804.2	
	110.71	UEFUI MAINI 4/UE/YR	1 (3696.9	
PLALANLE SUPPLAT (PCS MUVES)	120.48	CUMMUN USÉ \$/UE/YK	222006.3	
rendered accults from and inaliance	525.40			

TABLE 49 AIRCRAFT II LIFE CYCLE COSTS - 1995 DOLLAPS

.....

:

<u>AIRCRAFTEATSIT ACTUATION TRADE STUDY BASELINE</u> Costs in Fy 1995 million collars

LITE CYCLE COST	24721.02		
		TITL TRUDUCIION AIRCRAFT	200
		TOTAL SE	61.4
		CGMMAND SUPPORT	
AT REQ AND	2615.73	ATTRITION	
	1511.66		9. N
	10.46.67	10141 DDAYFINDE ITAGE	
	C		10
OTHER			
	) 		
	10400.00		
PRCDUCTION	26466104		
FLYAVAY	9193.05		
INTY N CEREFE	8748.19	UNIT AVERAGE FLYAMAY CACT	
	55.91	A 20 FRANT FI VALAV	974977
	278.97		12.114
THE TRACE TO THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE PART OF THE P	112.71	ARBENTE RETRETENE	4.061
TECHNICAL DAIA	7.0744	AVIONICS FLANAR	1.322
OTHER INVESTMENT	96.23	CTHER FLYAWAY	
	706-88		
	195.86		
INTITAL PERSONNEL ACGUISITION	Solution		
INITIAL PERSCUNEL TRATHING			•
7 FACILITIES	309 <b>.57</b>		
50	G•0		
TATAL OPERATIONS FOR IN VEADS			
RECERPTAG TAUR CTARTS TO A CONTRACT	11605.37	OPERATIONS DATA:	
LOWNER FOR THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STATE OF THE STA	6240.02	UE PER SOUADRAN	÷
	371.74		24.0
CASE FERE STREET STREET	31 79.65	CAFL DATE THROFULLAUNTH	25 • Û
FIGHT FERT ALVIENANCE MATERIAL	239.02		1.1
	891.42		1.0
CLADS IN HODIFICATIONS		UINCH UT FLCERS/CREU	0•0
TRAINING HUNITIONS		MAINTENAACE MAN HOURS /FHR	26.3
SUPLE A DATAEN SUPARES		AUNIT MAINT MENVUE	
VEHICULAR EQUIPHENT	806+73	FUEL FLOW GPH	
PAY AND ALLOUANTS	16.26	REPL SPARES SJEHR	
	<b>+316.17</b>	BASE MATNT MTI * JEUD	
	249.15	NFDAY MATAT ANTI-	184.4
	169.62		384.3
PERSUNNEL SUPPORT (PCS MOVES)		DEVIL PAINT S/UE/YR	91049°7
PERSONVEL ACQUISITION AND TEATNING	464407 500 40 1	COMMEN CSE S/UE/YR	86051-9
	208•12		~ ▶ + 7 ▶ ₽ 0

#### REFERENCES

D. Deamer, S. Brigham, Theoretical Study of Very High Pressure

#### Reference No.

1

- Fluid Power Systems, NA66H-822, North American Aviation, Inc., Columbus Division, Contract NOw65-0567-d, 15 October 1966, Unclassified. AD 803 870 2 J. Stauffer, Dynamic Response of Very High Pressure Fluid Power Systems, NR69H-65, North American Rockwell Corporation, Columbus Division, Contract N00019-68-C-0352, 16 April 1969, Unclassified. AD 854 142 J. N. Demarchi and R. K. Haning, <u>Lightweight Hydraulic System</u> <u>Hardware Endurance test</u>, NR75H-22, Columbus Aircraft Division Rockwell International Corporation, Contract N62269-74-C-0511, 3 March 1975, Unclassified. AD A-013 244 4 J. N. Demarchi and R. K. Haning, Design and Test of an LHS Lateral Control System for a T-2C Airplane, NR76H-14, Columbus Aircraft Division, Rockwell International Corporation, Contract N62269-75-C-0422, May 1976, Unclassified. AD A-032 677 5 J. N. Demarchi and R. K. Haning, Flight Test of an 8000 psi Lightweight Hydraulic System, NR77H-21, Columbus Aircraft Division, Rockwell International Corporation, Contract N62269-76-C-0254, April 1977, Unclassified. AD A-039 717/4GA 6 J. N. Demarchi and R. K. Haning, Lightweight Hydraulic System Extended Endurance Test, Columbus Aircraft Division, Rockwell International Corporation, Contract N62269-78-C-0005, September 1978, Unclassified. 7 J. N. Demarchi and R. K. Haning, Design Development, and Evaluation of Lightweight Hydraulic System Hardware - Phase I, NADC-77108-30 North American Aircraft Division, Rockwell International Corporation, Contract N62269-78-C-0363, 30 January 1981, Unclassified. 8 C. W. Helsley Jr., Airplane Actuation Trade Study First Interim Technical Report NA-79-492, Los Angeles Division, Rockwell International Corporation, 1 April 1980, Unclassified
- 9 J. Riddell III, Feasibility Study of High Voltage DC Secondary Power System Concepts for Future Naval Aircraft NR74H-53 Columbus Aircraft Division, Rockwell International Corporation, April 1974, Unclassified.

#### Reference No.

- 10 D. L. Lafuze et. al., <u>150 KVA Samarium Cobalt VSCF Starter/Generator</u> <u>Electrical System Final Technical Report AFAPL-TR-78-104</u>, General Electric Company Aircraft Equipment Division, Binghamption, N.Y. December 1978, Unclassified.
- 11 <u>Airplane Actuation Trade Study For Rockwell International North</u> <u>American Aircraft Division ATS Aircraft E2182-P2, Sundstrand</u> Aviation Mechanical Unit of Sundstrand Corporation, 29 August 1980, Unclassified.
- 12 D. E. Lautner et. al., <u>Power System Control Study</u>, <u>Phase I –</u> <u>Integrated Control Techniques</u>, <u>Phase II – System Modeling</u>, AFWAL-TR-80-2129 Vought Corporation, Dallas Texas, March 1981, Unclassified.
- 13 J. A. Rhoden, L. E. Scheer, <u>Power Systems Study Advanced Fighter/</u> <u>Attack and V/STOL Airplanes</u> 41-1906 Airesearch Manufacturing Company December 1977, Unclassified.
- 14 <u>Technical Proceedings Aircraft Electrical Power Seminar</u>, General Electric Company, Aircraft Equipment Division, Binghampton, New York, 10-11 May 1977, Unclassified.
- 15 E. Fasol, <u>B-1 Double Voltage (230/400 Volts) Distribution Analysis</u> and <u>Evaluation Study</u>, TFD-71-471, North American Aviation/Los Angeles, North American Rockwell, Los Angeles, Calif. 30 March 1971, Unclassified.
- 16 R. F. Sims, R. L. A. McKenzie, <u>Effect of Altitude on ARC Length in</u> 200 V AC and DC Systems, TR-69259, Royal Aircraft Establishment, Farnborough Hants, England, (1969)
- 17 N.E. Wood, R. A. Lewis Electromechanical Actuation Development, AFFDL-TR-78-150, Airesearch Manufacturing Company of California, December 1978, Unclassified.
- 18 W. W. Billings, <u>Aerospace Technology Development of Solid State</u> Remote Power Controllers For 300 VDC With Current Ratings of One and Two Amperes, One Type Having Current Limiting, NASA CR-135199 Westinghouse Electric Corporation, Lima, Ohio, June 1977 Unclassified.
- 19 J. T. Mitchell, <u>Aerospace Technology Development of Solid State</u> Remote Power Controllers For 120 VDC With Current Ratings of Five and Thirty Amperes, One Type Having Current Limiting, NASA CR-135216 Westinghouse Electric Corporation, Lima, Ohio, June 1977, Unclassified.
- 20 N. E. Wood, et al, <u>Electromechanical Actuation Feasibility Study</u>, AFFDL-TR-76-42, Airesearch Manufacturing Company of California, May 1976, Unclassified.
- 21 R. L. Heimbold, et al, <u>Application of Advanced Electric/Electronic</u> <u>Technology to Conventional Aircraft</u>, NASA Contract Report, NAS9-<u>15863</u>, Lockheed California Company, Burbank, California, July 1980, Unclassified.

22	Ray A. York, Flight Weight Actuators - An Evaluation of the Weight of Aircraft Hydraulic Actuators, Paper presented before the seventy
	third meeting of the Society of utomotive Engineers A-6 Committee, Bertea Corporation, Irvine, California, Oct. 1972, Unclassified.
23	Aircraft Electrical Power Systems, SP-500 A Compilation of Papers Presented At The SAE Aerospace Congress and Exposition, Anaheim, California, Oct. 5-8, 1981 SAE/SP-81/500.
24	W. J. Thayer Supply Pressure Considerations For Servo Actuators, Technical Bulletin 119 Moog Inc. Controls Division, East Aurora, N.Y., Oct. 1967, Unclassified.
25	B. H. Nichols ET. AL, Auxiliary Power System Study For 1975 Fighter Aircraft, AFAPL-TR-67-135 The Garret Corporation, Airesearch Manufacturing Division, Jan. 1968, Unclassified.
26	"Air to Surface Technology Integration Study - Final Summary Report", NA-77-1066 I. I, Rockwell International, Los Angeles, CA
27	"1990 Evaluation of Advanced Fighters - Contract Study Results", Contract F33615-78-C-3026, Rockwell International, Los Angeles, CA
28	"Wing/Fuselage Critical Development Program", Contract F33615-77-C-5229, Rockwell International, Los Angeles, CA
29	"Reliability/Maintainability Allocation Assessments and Reliability Reports - F-15 Proposal Analysis" Report NAAD-69-A050, Rockwell International, Los Angeles, CA
30	MIL-E-7016, "Electric Load and Power Source Capacity, Aircraft, Analysis of
31	"Automatically Processed Electrical Load Analysis", Report TFD-72-194, Rockwell International, Los Angeles, CA
32	"Analysis of the Impact of 270VDC Avionics Power Supplies on the S-3A Aircraft", Report LR-28765, Dated 15 September 1978, Contract N62269-78-C-0007, Lockheed Aircraft Co.
33	"Feasibility Study of High Voltage DC Secondary Power Systems Concepts for Future Naval Aircraft", Report NR74H-53, Dated April 1974.

. . . .

- Contract N62269-74-C-0157, Rockwell International, Los Angeles, CA
- 34 Hydraulic Engineering Design Manual, NA 47-216 Rockwell International, Los Angeles, CA.

والمراجع والمراجع والمروان والمهرين المراجع والمتحاف المحاد والمراجع والمراجع

35 Catalogue - Thermax Insulated High Temperature Wire, Thermax Wire Corporation, Fluishing, N.Y.

### APPENDIX A

## AIRPLANE ACTUATION TRADE STUDY

This appendix includes the initial work done by AiResearch under the direction of S. Rowe, as part of their effort in accordance with service agreement L9FM011231-405 to define the size, weight, and performance characteristics of the inboard flap actuator. With further refinements, this led to the final definition of the "hingeline installation" shown in Figure 25 as well as the applicable entries in Tables 18 through 20.

# ALL ELECTRIC AIRPLANE STUDY

Appendix A



AT NU. TLITE	CALC. NO. 1-221 SHEET NO. 1
INBOARD FLAP PRELIMINARY	DESIGN
SIZE THREE ACTUATORS , I FOLLOWING PERFORMANCE :	EACH HEVING THE
NO-LOAD RATE	100 091
DELIGN- LOAD RATE	50 Ops
RESIGN-LOAD	162 K-1N-13
DESIGN - LOAD & FLATE POWE	R 21.4 HP
STALL - LOAD	216 K-IN-LB
BANDWIDTH (A=12.5°, JL	= 50 N-13-5502)
• Ar	3 HZ - 3 DB
• ø	• 30 • MAX
STIFFNESS	1.82 + 10 1N-18-RAD
ALL THREE ACTUATORS MUST	FIT THE FOLLOWING
	5 DIA
	NOTE : DIMENSIONS I. J IN CHES
256	-14



. . . . . . .

DATE 5-23-20 PART NO. PREPARED BY S. ROWE	CALC. NO. 1-03! SHEET NO. 3 MODEL NO
SELECT A MOTOR ISTICS SHOWN IN THE MOTOR POWER W m = E	WITH THE SPEED-TORQUE CHARACTER- THE FIGURE (10% SPEED DROOP). R RATING IS THEN : PAL * 90% * TH
= (10 : : : : : :	399 K-IN-LE-SEC-1 399 K-IN-LE-SEC-1 0.5 HP
WM = 60 NEXT DETERMIN MANCE WITH FOI	WTOR WITH A POWER RATING OF HP K LE TIME CONSTANT - EVALUATE PERFOR- LOWING MOTOR [1] :
Tm 	100 M.SC 60 HP 24.2 KRPM 21.0 KRPM
J. USE T1-59 002	181.9 IN-LB 8.092 ×10 IN-LB-SECT
ACTUATOR PERFO SHOWN BELOW ( NOMENCLATURE) [2]	H OM-MX FOR TM-MX CONSTANT
	258

*- 34 L

•

a an an a mangangan garapangan ya sa an ana mana da ma mangangan sa

. . .

5. P.:	5v)=	MODEL NO	APPENDIX A
		0000 0000 000 000 000 000 000	
RAATS	IN4 24200. 21000. 181.9 181.9 181.9 181.9 100. 50. 0.85	0UT: 100. 86.7768595 224500.98 59.55428968 1452.	01109 = WM
	PARTER ANA COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSIANA AND COMMENSION AND COMMENSION AND COMMENSION AND COMMENSIA	DANKL DANK GR GR GR	A SR
RAATS	IN: 24200. 21000. 181.9 181.9 11.1 50. 50. 0.0087	DUT: 100, 86, 7768595 224500, 98 42, 11124208 42, 11124208	
- 		C S S S S S S S S S S S S S S S S S S S	EVALUATIO
RARTS	IN1 24200. 21000. 181.9 181.9 181.9 181.9 100. 50. 50. 50.	DUT ? 100. 86, 7768595 224500.98 29, 77714484 29, 77714484	NoToR
	RAATS RAATS RAATS	RhATS         RhATS         RhATS         RhATS         RhATS         IN1         IN1         IN1         RhATS         IN1         IN1	Rhats         Rhats         Rhats         Rhats         IN:         Rhats         IN:         Rhats         IN:         Rhats         IN:         Rhats         IN:         Rhats         IN:         IN:         Rhats         IN:         IN:

259

a105 🔔

TRAATS	CALC. NO SHEET NO
D BY S ROWE	APPENDIX A
ACTUATOR DERFORMA	DE IS ACCEPTABLE SO
SUMMARIZED BELOW	
TIME CONSTANT	100 MSEL
POWER RATING	60 H?
No - LUAD SPEED	24.2 KRPM
MAY TORQUE	101-9 IN-U3
CUERENT LIMIT	208.2 AMP
DIAMETER	3.7 IN
Length	10.5 14
WEIGHT	13.7 B (USE 14 LB)
· · · · · · · · · · · · · · · · · · ·	
· · ·	
	· · · · · · · · · · · · · · · · · · ·
· • · ••	• • • • • • • • • • • • •
• • • • • •	
	,
	•

DATE 6-23-30 CALC. NO. 1-321 SHEET NO. 2 PART NO. ZAATS MODEL NO PREPARED BY S. ROWE APPENDIX A	
HINGELINE DRIVE	
USE DIAMETER OF	
FROM PARAMETRIC DATA [3,4,5] USING THE CURVES MARKED 'ULTIMATE' AND 'LUG'	
$\overline{T} = 3.6 \times 10^{4} \text{ IN} - LB - IN^{-1}$ $\overline{K} = 5.6 \times 10^{5} \text{ IN} - LB - RAD^{-1} - IN^{-1}$	
$W = 1.4 \text{ LB} - 10^{-1}$	
SINCE DATA IS FOR & PLANET DRIVES, INCREASE PARAMETERS FUR CASE OF 12 PLANETS:	
$\overline{T} = 3.6 \times 10^{6} \text{ in - LB - IN - (+ 0%)}$	
$W = 2.1 \text{ LTZ} - 10^{-1} (+50\%)$	
DETERMINE LENGTH OF DRIVE TO ACCOMMODATE ACTUATOR STALL :	
$L = T_{STALL} \div T$	
$= 216 \times 10^{-1} = 3.6 \times 10^{-1}$	
CHECK STIFFNESS !	
K= L × K	
= 6 x 1.1x10" IN-LB-RAD"-IN" = 6.6 x10" IN-LB-RAD"	
	<b></b>

1. .....

TRATS Y S. FRANE	MODEL NO
HE DESIGN IS ACCEPTA HAGELINE. DRIVE - DA:	BLE, SO USE THE PROPOSED TA IS GIVEN BELOW:
REDUCTION (EST)	10-20:1
EFFICIENCY (EST)	90 - 90 %
DIAMETER	N1 2.5
LENGTH	6.0 in
WEIGHT	12.6 LT (USE 13 LB)
Stiffdels	6.6×10 " IN -LB - RAD"
CONFIGURATION	COMPOUND EPICYCLIC, 12 PLANET/SLICE
······································	
	• .
:	

DATE 5-23-50	CALC. NO. 1-001 SHEET NO. 8
REPARED BY S.R. ME	MODEL NO
GEARHEAD	
SCALE GEARHEAD FROM 90° INPUT FOR MOTOR AND DATA FOR THE UN	EXISTING PDU [6]. RETAIN . ADPROXIMATE DIMENSIONS TARE :
REDUCTION 1	- 150 : 1
EFFICIENCY 94	a %
DIAMETER 4.	8 IN
LENGTH 5.	41 С
WEIGHT ~ 11.	6 .8
MAXIMUM INPUT TORQU	E FOR THE UNIT IS
T _{MAX} = 363 IN	-LB
USING CUBE ROOT RUL	E TO SCALE :
$SF'' = \left(\frac{T_{NEW}}{T_{OLD}}\right)$	^y 3
$= \left( \frac{181.9}{363} \right)$	¹ /3
= 0.794	
NEW DIMENSIONS ARE	CALCULATED AS
NEW = OLD + SF	
* SF & SCALE FACTOR	
الم المراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع المراجع الم المراجع المراجع br>المراجع المراجع	
	· · · · · · · · · · · · · · · · · · ·

PARED BY S. RAWE MODEL NO. APPENDIX A	ATE <u>5-23-50</u> TANTE	CALC. NO. 1-021 SHEET NO.
THUS	REPARED BY S. ROWE	MODEL NO. APPENDIX A
THUS		
	THUS	
	= 3.8 1	

L = (5.0) = (0.794)

= 4.0 in Assuming a constant density of  $e = 0.128 \text{ LB-in}^3$ THE WEIGHT DECOMES  $W = V \cdot e$   $= \frac{D^2}{4} \pi L e$   $= \frac{(3.8)^2}{4} \pi (4.0)(0.128)$ = 5.8 LB FROM THE DIMENSIONS, IT APPEARS THAT THE GEARHEAD WOULD BE PRACTICAL FOR HINGELINE APPLICATION NEARLY THE FULL LENGTH OF THE ENVELOPE. THUS USE THE SCALED GEARHEAD:

EFFICIENCY 94% REDUCTION 1-150:1 DIAMETER 3.8 IN LENGTH 4.0 IN

264

.

е <u> </u>	24475	MODEL	NO
ARED BY	5 RAWE		APPENIDIXA
		اليه ومدير الأنابية الأعدار في الألفة فيقط متعادية، مكان <u>مديني التي والعام</u>	
	WEIGHT	5.8 LF (US	E 6 LB)
	CONFIGURATION	SIMPLE PLAN! RIGHT ANGLE	ETARY, WITH E INPUT
		. ·	···· · · · · · · · · · · · · · · · · ·
		•	
	<b></b>		
	-		
	<u> </u>		41°0°
		241	== <u>)</u> ;
		205	

The second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second s

-1.11

CALC. NO SHEET INV	CALC. NO.	1-001	SHEET	NO.	11
--------------------	-----------	-------	-------	-----	----

APPENDIX A

MODEL NO. ____

Numero -----

Ň. . .

JATE	
PATT NO. TZANTS	
PREPARED BY S. ROWE	

INSTALLATION
USING DIMENSIONS AND WEIGHTS FROM PREVIOUS CALCULATIONS, SHOW INSTALLATION DIMENSIONS AND WEIGHTS.
THE INSTALLATION IS SHOWN IN THE FIGURE ON THE FOLLOWING PAGE. THE FOLLOWING NOTES APPLY:
A CLUTCH AS BEEN INCORPORATED FOR MOTOR DISENGAGEMENT. DIMENSIONS AND WEIGHT ARE
D = 3.7  im
L= 1.0 1)
THE DRAWING IS APPROXIMATELY TO SCALE. H. IS NOTED THAT THE UNITS SLIGHTLY EXCEED THE LONGITUDINAL DIMENSION.
(3) ONE INVERTER PER MOTOR WILL BE REQUIRED. DATA FOR A TYPICAL INVERTER ARE [1]:
ilim 200 AMP (+ 10% TRANSIENT)
0 7.2 11
L 18.5 M
W 48 L3
TYPE PASSINE FREDN FUEL COOLED
<b></b> .

266

4401 **-**--



....

ARY NO.	MUDEL NO
REPARED BY	
PERFORMANCE EVALUAT	ION
ESTIMATE ACTUATION	SYSTEM PERFORMANCE
FOR PEST AND WORST	T DRINE EFFICIENCIES,
9-BEST = (0.54	) = ( 0 · 90 )
= 0,944	5
n = (0.94)	$) \times (0.90)$
WORST - COLA	,
	• • • • • •
PRINTOUTS FROM T	1-59 PROGRAM 'TRAATS'
ARE SHOWN ON THE	FOLLOWING PAGE.
and an entry of the second second second second second second second second second second second second second	

. . . . . . . .

. . . .

. . . . . .

۰.

AN AR SHOW AND A CARD AND A CARD AND A CARD AND A CARD AND A CARD AND A CARD AND A CARD AND A CARD AND A CARD A

η=0.152       RAHTS       RAHTS       RAHTS       IN       IN       24200.       RML       24200.       181.9       181.9       181.9       181.9       181.9       181.9       181.9       181.9       181.9       181.9       190.0       11.1       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174       10.174 <t< th=""></t<>
o l
RAATS IN: 24200. DMHL 24200. DMHL 24200. DMHL 21000. DMHL 180. 181.9 JMHL 180. 181.9 JMHL 180. 181.9 JMHL 100. DAHL 0.0174 A 0.0174 A 161. 100. DAHL 162. 1452. GR

.
NAT 9-23-20	CALC. NO. 1-001 SHEET NO. 15
PART NO. 124 ETC	MODEL NOAPPENDIX 4
REFERENCES	
1. A.R. REPORT NO. BU-17:	259,
AIRPLANE ACTUATION TRADE	STUDY : FINAL
REPORT	
2. TI-SU PROGRAM LISTING	
"RAATS : ROCKWELL AI	RPLAJE ACTUATION
TRADE STUDY"	
J. AIR DOCUMENT NO. 5-3	3199,
"ROTORY ACTUATOR PARA	NETRIC DESIGN
WEIGHT AND CONFIGURATION	o"
4. AIR DOCUMENT NO. 5-3	3200,
" ROTORY ACTUATOR PARAN	NETRIC DESIGN
SPRING RATE	<b>_</b>
5. AIR REPORT NO. 78-1531	ο,
PRELIMINARY DESIGN MAN	DAL FOR ROTORY
ACTUATORS	
· · · · · · · · · · · · · · · · · · ·	
6. AIR DRAWING NU. 20226	46
270	

•		_	LOCA DROGRAM 'TAATS' NOUFNELATUEE
41	PPENDIX		
<u> </u>			
	NOMEACL	<u>. 471</u>	DRE
	OMNL	:	MUTUR NO-LOAD SPEED [RPM]
	OMNX	:	MUTUR MAX-LOAD SPEED [RPM]
	TMMX	!	MOTUR MAX-LOAD [IN-LT]
*	JM	:	MOTOR INERTIA [IN-LB-SEC]
	NМ	:	No. of motors [-]
	OANL	•	ACTUATOR NU-WAD SPEED [DPS]
	JL	:	LOAD INERTIA [IN-LB-SEC]
	NGR	;	GEARING EFFICIENCY [-]
	A	:	OUTPUT AMPLITUDE [RAD]
	OANL	:	ACTUATOR NO-WAD SPEED [ DPS]
-	OAMX	:	ACTUATOR MAX-LOAD SPEED [DPS]
	TAMX	:	ACTUATOR MAX - WAD [IN-UB]
	AW	:	ACTUATUR BANDWIDTH [ RPS]
	GR	:	GEAR RATIO [-]

.....

271

-----

# APPENDIX B

# AIRPLANE ACTUATION TRADE STUDY

This appendix includes the initial work done by AiResearch, under the direction of S. Rowe, as part of their effort under service agreement L9FM-11231-405 to define the functional characteristics of the inverters and controllers needed for implementing the "all electric" (Aircraft I) approach to the trade study.



Appendix B

ALL ELECTRIC AIRPLANE STUDY SERVICE AGREEMENT 69FM-11231-405 PROGRESS REPORT FOR THE MONTH OF SEPTEMBER: ADVANCED ELECTROMECHANICAL ACTUATION SYSTEMS

#### 0. PREFACE

This document is submitted in conjunction with Service Agreement LSFM-11231-405, all Electric Airplane Study. The data submitted herein is presented to facilitate completion of the subject study. This document, in addition to previously transmitted data, shall serve as a progress report for the month of September 1980.

# 1. INTRODUCTION

Electromechanical actuation systems (EMAS) are finding increased potential for use in aircraft flight control systems (FCS) [1, 2, 3, 4]. Most advanced EMAS utilize brushless dc permanent magnet (DC-PM) motors. Digital serve control by means of microprocessors is practical for many actuation systems. Systems of this type have a unique configuration and peculiar design requirements.

Near term technology (1990) will provide additional advances in EMAS, although the previous statements will remain unaltered. In anticipation of this, an advanced EMAS is presented in the following sections.

#### 2. DESCRIPTION

A block diagram of a position serve system is shown in Figure B-1. The principal components are the controller, inverter, and actuator. Interfaces between the components are shown, also. The function of each component is briefly described below:

• Controller - The functions of the controller, generally, are (a) close the inner and outer servo response loops, (b) provide a motor current command to the inverter, and (c) provide an interface with the FCS.



Figure B-1. Position Servo Block Diagram and Component Definition

Additional functions (fault detection, built-in-test, adaptive control) may be included in the controller for a particular application.

- Inverter The functions of the inverteriare to provide (a) motor commutation, (b) motor torque/speed control, (c) motor current limiting, and (d) motor drive electronics (power transistors, snubber circuitry, etc.) cooling. Again, accitional functions may be included.
- Actuator The actuator serves a single function, to convert electrical power to a mechanical response as a function of the serve command. The configuration of the actuator will be totally dependent upon the FCS requirements; but generally consists of motors and mechanical drives as a minimum. Each of the above are addressed in more detail in the following sections.

## 3. CONTROLLER

A more detailed block diagram of a general actuation system is shown in Figure B-2. This block diagram is representative of advanced (1990) technology.

The controller consists of blocks I through 6. Block I generates the speed command of the servo motor; block 2 generates the current command of the motor; block 3 allows the servo control laws to be altered during operation; and block 4 provides a monitor function for the FCS. Blocks 5 and 6 provide compensation for the rate and position feedback loops.

Each block may be digital or analog. However, to implement the above control scheme, digital appears to be the most viable. This is due to the availability of high speed microprocessors; and the large number of discrete components required by an equivalent analog system.

Interface with the FCS and EMAS components may be electrical or optical.

#### 4. INVERTER

The inverter consists of blocks 7 through 10 in Figure 8-2. Block 7 generates a pulse train as a function motor current error; block 8 contains commutation and current control switching logic, and power switch driver electronics; block 9 contains the power switches (transistors) required for motor commutation, torque/speed control, and current limiting; and block 10 senses motor current and provides feedback compensation in the current loop. Each of these functions will be explained more fully.

The mechanization of the above loops have been examined previously (5), and will remain analog for the far term. This is due to the frequency response requirements of the current limit loop. A digital system would require a micro-processor with a calculation rate in the gigahertz range. This is considered very unlikely during the near term.

Appendix B

. .



P

.

فالتعاديك ومقدما ومعارية والمتعالي المرار والمكافئة المعاملية والمعارية والمعارية

Figure B-2. Advanced Technology System Block Diagram



Interface may be accomplished by electrical or optical tears; but electrical appears to be nost casinable due to the analog control of the current koop.

#### 4.1 DESCRIPTION

The principal function of the motor inverter assently is to electricative commutate and control motor notation over its entire speec range. A general inverter assently block diagram is shown in Figure B-3.

The transistor inverter converts the 270 vdc source power to variable frequency 3-phase current applied directly to the motor windings. Figure B-4 shows the basic three phase motor drive waveforms. The sequence in which the transistors switch is conveyed by the switch number, SI taru S6.

The inverter assembly may also encompass an input filter for conducted EV: suppression and energy storage; current sensing for feedback to the internacurrent control; drive logic for transistor control; motor position sensing for feedback to the switch logic; and although not shown, logic low level power supplies for control and protection functions.

### 4.2 CPERATION

The conversion of dc power into ac power is accomplished by six switch circuits as indicated in Figure B-4. Each switch (S1 thru S6) consists of a transistor and voltage limiting or snubber circuits (R1, C1, and CR1) as shown in Figure B-5. Internal to the transistor is a parallel free wheel diode that provides a path for the motor lagging reactive current flow. Each transistor switch can conduct up to a 120 electrical degree interval. During this interval the transistors are modulated (cn-to-off) to control the current flow to the motor, thus designated transistors operate in a chopping mode to affectively pulse-width-modulate the ac output voltage amplitude. This technicue inserts notches into the output waveform that cause a reduction in the fundamental current amplitude without causing an objectionable increase in the higher order harmonics.

The inverter is controlled such that the accurrent is synchronized with the rotation or phase via motor rotor position sensors. The control logic uses motor position sensor information to force synchronization between the inverter current and motor phase rotation. In this manner, the correct relationship between motor generated EMF (electromotive force) and applied current from the inverter is continuously maintained.

Figure B-6, Figure B-7, and Figure B-8 illustrate the switching and waveforms involved. Figure B-6 shows the two types of switching, conducting and chopping. Conducting switching is used for basic commutation; chopping is used for current control. Figure B-7 shows the different voltage waveforms and resulting current waveform which exist in the inverter and motor. Figure 8 combines the result of these waveforms over 360 electrical degrees of machine rotation.

Appendix B

1

i



Ì

Figure B-3. Inverter Assembly Block Diagram



Figure B-4. Basic Motor Drive Waveform

. . . . . . . . . .

المستعملية مستندر المراجع

Appendix B



ş

.

Figure 8-5. Inverter Switch Schematic







Figure B-8. Back EMF and Current Relationship

The proposed waveform is the result of a pulse-witth modulation circult (fixed frequency, variable width). This voltage waveform allows the control of current to a pear constant value, as shown in Figure B-7 and Figure B-8.

Each of the switching waveforms (conducting and choosing) may be restricted to particular sets of switches (as implied in Figure B-6); or rotated among all the switches, for improved thermal balance.

4.3 THERMAL MANAGEMENT

Power inverters require cooling due to losses developed during transistor switching and conduction. Figure B-9 illustrates three types of cooling applicable to solid state inverters.



Figure B-9. Inverter Cooling Techniques

Evaporative cooling of electronic devices has grown increasingly popular in recent years as current density ratings of devices have increased [5]. In comparison with conventional air-cooling, the advantages that evaporative cooling offers are significant. They include:

- (a) Fewer electronic components The improved cooling increases the loac-handling capability of a device, hence reducing the number of the devices required.
- (b) Decreased weight and size Evaporative cooling reduces the cooling system size and weight which normally forms the bulk of the electronic package.

------

- (c) Higher reliability Electronic devices are immensed in a dielectric Higher and are not susceptible to dust fouling and resultant proplems. The cooling system is interently more reliable when there is no forcestain involves.
- (d) Less maintenance Periodic maintenance operations, such as dust cleaning, are not required for electronic components and neat sinks.
- (e) Less noise Noise levels will be very low or non-existent.

. . ......

4.4 POWER TRANSISTORS

.

.....

: .

•

ł

.

------

Power transistors have traditionally been the principal limiting factor in the application of brushless DC-PM motors, where significant current levels were required. Transistors are currently available which have very high current ratings; but do not have all the desirable characteristics a designer would like.

Power FET's (field effect transistors) offer potential for the required high in current ratings, in the near future [7]. Existing desirable teatures of power FET's include short switching times, low switching losses, and excellent load sharing.

T-I summarizes probable characteristics of the near term device.

#### T-! Near Term FET Characteristics

FET current rating	50 amp
FET on-state resistance (R _{DC} ) at maximum junction temperature of 150°C	0.10 ohm
FET diode forward voltage drop (V _{CR} ) at 50 Add	1.5 vdc
FET current rise ( $t_r$ ) and fall ( $t_f$ ) time for 50 amp-dc changes	50 nsec
FET thermal resistance from junction to case	0.42°C/watt
FET gate to source capacitance	1000 pF
Lead inductance from the FET assembly to a voltage source	0.3 H
MaxImum allowable FET drain to source voltage caused by switching	320 volts

#### 5. SUMMARY

A general description of a near term EMAS has been presented. Construction, operation, and technology assumptions have been stated.

Appendix B

-----

It is emphasized that there are numerous methods for mechanizing each component of the EMAS (controller, inverter, actuator). Individual applications may dictate specialized approaches to system control. For FDS applications in the near term however, the presented approach appears to be attractive and viable.

------

# REFERENCES

----

- 1. Wood, Ecnolds, Ashmore AFFOL Report No. TR-76-42 <u>Electromechanical</u> <u>Actuation Feasibility Study</u>.
- Grau WACK Report No. 75170-30 Feasibility Investigation for Advances Flight Control Actuation Systems: All Electric Concests (AF CASHE).
- Wood, Lewis AFFDL Report No. TR-73-150 <u>Electromechanical Actuation</u> <u>Development</u>.
- 4. Lewis, Gray, Wood AFWAL Report No. TR-30-3024 <u>Electromechanical</u> <u>Actuation Development</u>.
- 5. Rowe AiResearch Report No. 80-17284 Electromechanical Airplane Actuation Trade Study.
- Rollet Technical Paper "Two Phase Freen Cooling for Electronic Power Equipment".
- Blanchard, Haynie Power Conversion International, V.6, No. 2, Mar/Apr 1980 "Power MCS Transistors: Structure and Performance.

# APPENDIX C

# AIRPLANE ACTUATION TRADE STUDY

This appendix includes the work done by AiResearch, under the direction of S. Rowe, as part of their effort under service agreement L9FM-11231-405 to finalize the definition of the controllers and inverters needed for implementing the "all-electric" (Aircraft I) approach to the trade study. Work included in this appendix also further refines the weight and envelope data for the inboard flap actuator as well as those for the other major flight control actuation functions. These latter data were used directly in the preparation of Figures 25 through 31.

## Al! Electric Airplane Study Service Agreement L9FM-11231-405 Progress Report for the Period October 1980 - January 1981: Advanced Electromechanical Actuation Systems

# PREFACE

This document is submitted in conjunction with Service Agreement L9FM-11231-405, All Electric Airplane Study. The data submitted herein is presented to facilitate completion of the subject study. This document, in addition to previously submitted data, shall serve as a progress report for the period of October 1980 - January 1981.

#### INTRODUCTION

Several objectives of the statement-of-work have been completed during this reporting period. Specific accomplishments are:

- Identification of actuation system performance requirements
- Preliminary design of candidate actuators, and candidate selection
- Inverter analysis and design
- Controller concept selection and sizing

Each of the above are addressed in following sections.

#### REQUIREMENTS

Performance requirements for the actuation systems were defined by the customer during 1980 [3]. Subsequent discussions between AiResearch and the customer modified the flight control system (FCS) actuation system performance requirements to less stringent, but wholly satisfactory criteria.

A summary of FCS actuation system performance may be found in Appendix A.

#### ACTUATORS

.

Actuator performance and other design requirements were examined [1]. Using these data, preliminary designs for FCS candidate actuation systems were generated using the following approach:

- Motors were sized based on duty cycle, and steady-state and dynamic performance requirements
- Mechanical drives (output reduction) were sized based on peak load and life requirements

 Gearheads (intermediate reduction) were sized based on gear-ratio and load requirements

 Actuator dimensions and weights were calculated using the preceding data

Assumptions used to establish actuator configurations were:

- All motors are a brushless direct current-permanent magnet (DC-PM) configuration
- Motor magnets have an energy product (BH) of 22 X 10⁶ Gauss-Oersted
- All reduction is simple planetary, compound planetary, or ballscrew

Candidate designs may be found in Appendix B. A final selection was made after consulting the customer. Selections are indicated in the appendix.

#### INVERTERS

ALL - 2464, 17 - 194

Inverter sizing is usually determined by the type of cooling employed. Natural radiation and convection (heatsink) techniques are a first choice due to their simplicity and low cost. However, size and weight become unacceptable at higher power levels (5-6 kw inverter rating) and alternate cooling schemas must be investigated.

Inverter current requirements for each of the actuation system motors were determined, and may be found in Appendix C. The inverter configuration previously submitted was assumed, and is discussed in Reference [1]. Devices were selected based on current requirements, resulting in two power field effect transistors (FET) with ratings of 25 and 50 amps. FLT characteristics and rationale are presented in Reference [1] (the 50 amp device represents 1990 state-of-the-art).

Cooling requirements were determined next, by calculating inverter losses as a function of duty cycle for each actuation system. Cooling techniques were evaluated, and the following conclusions reached:

- All systems requiring a 25 amp or smaller inverter may be cooled by natural radiation and convection
- All other systems requiring more than 25 amps required alternate techniques, and evaporative cooling was selected

Evaporative cooled inverters suitable for the required application had been previously sized in a similar study [2]. Inverter configuration is depicted in Figure C-1. Existing inverter dimensions were scaled to take into account applicable boundry conditions, and are tabulated in Table C-1. Note that forced convection with surface finning was assumed for inverter cooling.

It may be possible to eliminate any external cooling requirement by increasing inverter fin area and coolant mass. This will require as a minimum, simulation of inverter transient thermal response.



Figure C-1. Evaporative Cooled Inverter (No Fins)

## Table C-1

Evaporative Cooled Inverter Data*

Actuation System	Current Rating	Diameter	Length	Weight
Inboard Flap	150 amp	7.2 in	18.7 in	38.0 Ib
Midspan Flap	50	5.0	10.4	12.5

*Finned surface with forced convection, 130°F, 50 cfm, 2 in-H₂O P

Subsequently, the remaining inverters were sized using individual heatsinks for FET cooling. Figure C-2 shows the heatsink used, and Figure C-3 illustrates inverter configuration. A common design was used for all remaining actuation systems in light of the small size of the 25 amp inverter. Table C-2 summarizes inverter characteristics.

A summary of the thermal analysis for the each inverter configuration may be found in Append'x D to this appendix.







Figure C-3. Natural Radiation/Convection Cooled Inverter (Typical)

## Table C-2

#### Natural Radiation/Convection Cooled Inverter

Actuation System	Current Rating	Length	Width	Depth	Weight
Alleron, Canard,	25 amp	ll in	7 in	3 in	10 I b
Leading Edge Flap, Upper/Lower Rudder					

## CONTROLLER

A controller was configured for multiple actuator control, using the concept of Reference [1]. Up to six inverters and motors, and four actuators may be interfaced with the unit. This approach allowed the use of a common controller for all 270 vdc servos. Controller characterics are summarized in Table C-3.

Figure C-4 is a block diagram of the controller. The unit is full DMX (digital multiplexed). All interfaces are shown as optical, although if transmission distances are short, electrical interfaces may be desirable. A feasible scheme would be optical for the FCS, motor, and actuator interfaces; and an electrical interface for the inverters.

## Table C-3

#### CONTROLLER SUMMARY

Dimensions	4 x 4 x 8 in.
Weight	5 16
Cooling	Air
Interface	Optical/Electrical Bus, Power
Function	Supports up to 6 inverters, 6 motors, and 4 actuators

Two way buses are used for the FCS and inverter interface only, all other buses are one way. This was chosen since only feedback data is necessary from the motors and actuators.



;

291

1

ţ

1

1

1.1

Controller interface is shown in Figure C-5. Note that all motor and actuator data flow to the controller. Motor data is required by the inverter for commutation (rotor position), so any necessary data are passed on to the inverter by an optical-optical or optical-electrical coupling, as required. Figure C-5 indicates that any actuation system data may be made directly available to the inverter, as necessary. Table C-4 summarizes FCS-controller interface.

Current limit control and commutation logic were assumed to be integral with the inverter. A multiplexed interface (optical or electrical) would also be required at the inverter.

Control of the clutches and brakes of the various actuation systems was assumed to be performed at the controller. Discrete wiring was also assumed for clutch/brake power. The devices are controlled by solid state relays located in (or near) the controller. Figures C-4 and C-5 illustrate interface and operation.

#### PROJECTED PROGRESS

CITY WYDAYS Y'S IS ...

During the next reporting period a detailed design of one FCS actuator will be performed. The customer will be consulted prior to the start of the detailed design, for actuation system selection and envelope specification

#### REFERENCES

- 1. Rowe--AiResearch Report No. 80-17351(1), <u>All Electric Airplane Study</u>, Progress Report for the Month of September
- 2. Rowe--AiResearch Report No. 80-17284, <u>Electromechanical Airplane Actuation</u> <u>Trade Study</u>



100 0 84

(n, n)

# Table C-4

# Controller Interface Requirements

Aircraft-Controller

- 1-2 way or 2-1 way data bus
- 270 vdc

Controller-Inverter

- 1-2 way or 2-1 way data bus
- 270 vdc

Inverter-Motor

• 3-phase wiring

Controller-Motor

1-way data bus*

Controller-Actuator

- 1-way data bus
- Discrete wiring (1 per clutch/brake)
- 270 vdc

*May be possible to reduce to 1 bus per actuator

Appendix A

ير د منهودهاد ۲

# Actuation System Requirements

DATE 9-11-90 PART NO. RAATS

Appendix C

CALC. NO. 1-001___ SHEET NO. ____ MODEL NO.

PREPARED BY	S. ROW	Ε			CHE	CKED BY				_
	ACTR'S - OIL - CHANNELS	Pe R Surf≯r£	M	M	ı	C4	ю	~	6	
	TREDUN - DANCY	PER. Surface	F0 ²	Ŀ	FO	FJ	F.0 ²	FO	FΟ	
	τις S	DE SIGN POWER	44 60.41 44 21.11	3.09 HP 7.77 KW	1.11 HP 0.83 KW	0.56 HP 0.41 KW	0.53 HP 0.39 KW	0.72 KW	41 22.0 41 2.0	
	ACTERIS	STALL	151 KIP	30 KIP	26 KIP	SKIP	13 KIP	173 K1P	2.4 KLB	
* *	AD CHAR	DESIGN LOANS # RATE	113 Kip	23 K1P 50 DPS	21 KIP 20 0PS	7 K1P 30 DP3	10 KIP 20 0PS	153 KIP 2.4 DPS	2.0 KLB	
UREMEN.	Γo	No-Land Rate	549 001	100 001	SO OPS	40 DPS	40 005	S DPS	2.3 193	:
भर रिह्ले	TRAVEL		- 50° + 45°	• 30 • 42	+ 25.	± 20°	± 20°	• 02 + 0 -	15.36 14	
ACTUATO	ACTUATOR		BOARD FLAP	SSPAN FLAP	LERON	PER RUDDER	MER RUDIER	.E. FLAP	an ard	
			Ð	WII ()	E V	(f) P	(E) L0	(e) L	Ē	

6-28-80 RUCKWELL UN * BALED ON MINUTES FROM MEETING A.

296

•

# Apeendix C

Appendix B

.

# Actuation Preliminary Designs

DATE <u>9-22-80</u> Appendix C CALC. NO. <u>2-001</u> SHEET NO. <u>1</u> PART NO. <u>RAATS</u> MODEL NO PREPARED BY <u>5. Rowe</u> CHECKED BY	 
INFOARD FLAP INSTALLATION	
USE THE FOLLOWING DATA FOR INSTALLATION PURPOSES :	
1. MOTUR	
<ul> <li>DIAMETER 3.5 IN</li> <li>LENGTH 10.6 IN</li> <li>WEIGHT 12.0 LB</li> </ul>	
2. DRIVE	
DIAMETER 3.5 IN     LENGTH 9.4 IN     WEIGHT 11.1 LB	
3. GEARHEAD	
DIAMETER ~ 5.5 IN ) INVESTIGATE     LENGTH ~ 5.0 IN REDUCTION OF     WEIGHT ~ 15.0 LB DIAMETER	
4. CLUTCH	ł
DIAMETER ~ 3.5 IN   LENGTH ~ 1.0 IN   WEIGHT ~ 2.0 LB	
SEE ATTACHED FUR INSTALLATION .	



DATE         9-22-80         Appendix C         CALC. NO. 2-002         SHE           PART NO.         RAATS         MODEL NO.         SHE           PREPARED BY         S. IROWE         CHECKED BY         CHECKED BY	ET NO
MIDSPAN FLAP INSTALLATION Use the following data for installation	
· DIAMETER 2.3 IN · LENGTH 7.7 IN	
• WEIGHT 2.3 LB 2. DIRIVE • DIAMETER 1.4 IN • LENGTH 9.1 IN	
WEIGHT Z.9 LB 3. GEAIZHEAD DIAMETER 3.0 IN	
• LENGTH 3.5 M • WEIGHT 6.0 LB 4. CLUTCH	
· LENGTH ~ 1.0 WEIGHT ~ 1.0 LB	
······	



DATE <u>9-72-90</u> PART NO. <u>RAATS</u> PREPARED BY <u>S. ROWE</u>	Appendix C	CALC. NO SHEET NO MODEL NO CHECKED BY
ALLERON INSTALLAT	<u></u>	
USE THE FOLLOWS PURPOSES :	ING DATA FOR	2 INSTALLATION
1. MOTOR		
<ul> <li>Diameter</li> <li>Length</li> <li>Weight</li> </ul>	1.9 IN 6.7 IN 1.0 CP	
2. DRIVE		
• DIAMETER • LENGTH • WEIGHT	1.4 m 11.8 m 3.2 LB	
3. GEARHEAD		
<ul> <li>DIAMETER</li> <li>LENGTH</li> <li>WEIGHT</li> </ul>	2.5 IN 3.0 IN 2.5 LB	
9. CLUTCH		
<ul> <li>DIAMETER</li> <li>LENGTH</li> <li>WEIGHT</li> </ul>	~ 1.9 IN ~ 1.0 IN ~ 0.8 СЗ	

-----



. . ....

----

بادغ بمين

-0-0100

:

DATE <u>9-22-80</u> PART NO. <u>RAATS</u> PREPARED BY <u>S. ROWE</u>	Appendix C	CALC. NOSHEET NO.	
UPPER RUDDER	NSTALLATIO	N	
USE THE FOLLO PURPOSES :	WING DATA	FOR INSTALLATION	
1. Motor			
<ul> <li>DIAMETER</li> <li>LENGTH</li> <li>WEIGHT</li> </ul>	0.3 CB		
2. DRIVE			
<ul> <li>Diameter</li> <li>Length</li> <li>Weight</li> </ul>	1.4 m 4.1 m 1.1 LB		
3. GEARHEAD			
<ul> <li>DIAMETER</li> <li>LENGTH</li> <li>WEIGHT</li> </ul>	1.5 14 1.8 14 1.0 LB		
4. CLUTCH			
• Diameter • Length • Weight	~ 1.9 N ~ 1.0 N ~ 0.8 LB		

3C4



<u>9-72-50</u> NO. <u>TZAATS</u> RED BY <u>S. ROWE</u>	Appendix C	CALC. NO. <u>2-00 5</u> MODEL NO CHECKED BY	SHEET NOI
LOWER RUDDER	NSTALLATIO	<u>N</u>	
USE THE FOLLOWING PURPOJES :	A DATA FOR.	INSTALLATION	
1. MOTOR			
<ul> <li>DIAMETER</li> <li>LENGTH</li> <li>WEIGHT</li> </ul>	2.3 IN 6.0 IN 2.6 L3		
2. DRIVE			
<ul> <li>DIAMETER</li> <li>LENGTH</li> <li>WEIGHT</li> </ul>	1.6 IN 4.0 IN 1.3 LB		
. 3. GEARHEAD			
<ul> <li>Diameter</li> <li>Length</li> <li>Weight</li> </ul>	2.0 IN 2.0 IN 2.0 LB		
9. CLUTCH			
<ul> <li>DIAMETER</li> <li>LENGTH</li> <li>WEIGHT</li> </ul>	~ 2.3 IN ~ 1.0 IN ~ 1.2 LB		
		<u>-</u>	


DATE 9-22-80 Appe PART NO. TEAATS PREPAREC BY S. ROWE	endix C	CALC. NO SHEET NO MODEL NO CHECKED BY
L.E. FLAP INSTALLATIO	Ч	
Use the following purposes:	DATA FO	L INSTALLATION
1. Motor		
<ul> <li>DIAMETER</li> <li>LENGTH</li> <li>WEIGHT</li> </ul>	2.3 N 8.1 N 2.6 LB	
2. DRIVE		
<ul> <li>DIAMETER</li> <li>LENGTH</li> <li>WEIGHT</li> </ul>	4.0 IN 7.2 IN 12.8 LB	
3. GEARHEAD		
<ul> <li>DIAMETER</li> <li>LENGTH</li> <li>WEIGHT</li> </ul>	5.5 in 3.0 in 3.0 lb	
9. CLUTCH		
<ul> <li>Diameter</li> <li>Length</li> <li>Weight</li> </ul>	~ 2.3 in ~ 1.0 in ~ 1.2 LB	



í

309

.

9נ

DATE _ 9-22-30 Appendix C CALC. NO 2-00 + SHEET NO 1	
PART NO. RAATS MODEL NO.	
PREPARED BY CHECKED BY	
CANARD INSTALLATION	
Des rels services same son energy aread	
PURPOSES :	
	1
· DIAMETER 2.3 IN	
· LENGTH 5.3 IN	
• WEIGHT U.S CB	
	Į
2. DRIVE	
· RALL ASSY OD LIGIN	
· SCREW OD 2.0 W	
· TOTAL LENGTH 19.9 IN	
• WEIGHT 11.6 LB	
	ļ
S. GEARING	
· DIAMETER ISIN	
• LENGTH 1.8 IN	
• WEIGH ( 1.0 Cg	
4. CLUTCH	
· DIAMETER ~ Z.3 IN	
• LENGTH ~ 1.0 10	
• Weight ~ 1.2 LB	
5. KOD ENDS	
· WEIGHT ZOUB	

;

-----

- I The second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second se



Appendix C

- S - 44

.

فكناف والمتلك فالمتعالية وتفاصينا الانتخاص والأستمنا فيترجز المنتزي كم

inverter Current Requirements

Appendix C

CALC. NO. 3-001 SHEET NO. ____

MODEL NO .-

PART NO._____

K

	-AUPHUXIAATE MO.	rok Curre	NT RATIN	655 & 1AV	erter Se	LECTION S		RED 8Y
	ACTUATUR	Mutor Speers	Motar Poner	Voutare	2	MOTOR	INVERTER RATING	S. Kou
Ð	INGOARD FLAP	ZOKRPM	۳۶۱ ۶۰SE	240 V	6.0	156 A	¥ 051	)E
3	MIDSPAN FLAP	20 KRPM	m× 6.3	240 1	6.0	31 A	\$U ¥	
(1)	ANERON	ZOKRPM	3:0 KW	2401	ດີວ່	4 41	20 A	_
•	Upper Runder	ZOKRPM	0.75 KW	240 0	5. D	3.5.4	א + גי גי	
ତ	Lower Runcer	no krow	my 51	2900	(y. c)	¥ 6.9	** 01	CHECKE
•	L.E. FLAP	Wda yici	3.7 Kw	24.0 v	63.0	174	** 02	D BY
(£)	CANARD	NG KIRPM	WX SLO	2400	6.0	3.5 .	× 4 5	
	* Use inverters de ** Repurcs 25 amp	SIGNED IN	) AIR B	0-17284				

-----

Appendix D-1

Inverter Loss Summary

DATE <u>1 - 8</u> PART NO FREPARED BY	•85 5. Ro,	VE	Apj 	pendix C		CALC. NO MODEL NO CHECKED	<u>3-00</u>	<u>5</u> s	HEET NO
	lora Lusses	W 1061	341	26	0	21	401	<b>.</b>	2406 w
×c.e.	TOTAL	9	و	4	Þ	e	21	<b>र</b>	н К) Ш V)
DUTY C	LO SSES	316.5 2	56.9	و د	ۍ ۲۰	0 M	6.8	5	W W
or. 25%	25%	39 A	8.2	3.5	6.0	E·1	4.3	6.0	-
มั ม ม	INVERTER TYPE 4 KATING	EC **	EC + * 50	20 20	NC NC	U 0	NC 20	N N	80-17284
TER Lo.	MOTOR	156 A	31	<u>م</u>	35	6.9	٤١	3.5	REPORT
INVER	4	AP	9		EK.	2			50% 31 * AIR
	ACTUATO	ARD FL	SPAN FLA	ERDN	ER RUDD	ER RUDDE	FLAP	Arzo	ATA F02 ATA F02
		INBO	WID	AIL	Upp	Low	LE	CAA	ΔΔ *‡

AIRESEARCH MANUFACTURING COMPANY

DATE	1-9-90	Appendix C	CALC. NO3.005_ SHEET NO2
REPARED	BY S. ROWE		CHECKED BY
	INVERTER	NOMENCLATUR	E
	EC	EVAPORATINE	Coounds
	FC	FORCED CONVEC	GOIT
	NC	NATURAL CONV	ECTION
l			
l			
			······································
]			
	· · · · · · · · · · · · · · · · · · ·		
1			

.....

Appendix D-2

معادم ومنابعات والمس

# Natural Radiation and Convection Analysis

DATE	Append	bx C	CALC. NO	3-006 SHEET NO
AATS / ROCK	WELL - IN	IVERTER C	60L/NG	ANALYSIS
TE MPE RA TURE	PLEDICTION	IS FOR	LE FLAP	INVERTERS :
DALL CICLE	TO3 DEVICE	CASE	JUNCTION	Notes
1. SPEED ~ 50 LOAD ~ 25 TIME ~ 00	% 1.8 W	127	/28	θj.c = 0.42. ℃/W
2. SPEED ~ 50% LOAD ~ 50% TIME ~ 120	4.85 W SEC	.137	139 .	0j.c + 0.42 °C/W INITIAL CASE TEMP, Tc; = 127 °C
Tram Lap	sient Temp Res sed time of	per_Inver yeonse off 120 secon	rter Duty Ler Duty	Cycle 1 with
• HEAT SINK / • COOLING APPA	DISSIPATOR (ATT ROACH : NAT.	TACHED) : CONV. +	IERC PN RADIATION	HP1-703- <u>33B</u> To 167°C Ambient
3.0"MIN		2.5	· · · · · · · · · · · · · · · · · · ·	NI plate - U - Y8 - 1/4"
Assumption 3_ 1) No Insulator between JO-3 and Heat Sink 				
	CLAS	S — ATION 318		REPORT



No.waterson .....

Appendix D-3

:

Evaporative Cooling Analysis

#### Appendix D-3

#### Evaporative Cooling Analysis

## Evaporative Cooled Inverter Analysis

As a baseline configuration, a 50-amp inverter with a 50-percent motor speed and current duty cycle was used for analysis [1]. At this current rating, the inverter bridge consists of six electronic circuit packages, each with a FET in a TO-3 can that dissipates approximately 50 watts. The cooling requirement is to maintain the FET junction temperature below 150°C; preferably below 125°C.

R-113 (trichlorotrifluoroethane) was selected as the coolant. It has been used successfully for evaporative cooling of power semiconductors for traction motor choppers. The boiling heat transfer coefficients for R-113 were derived from recent AiResearch heat transfer test results for a thyristor/busbar assembly immersed in R-11 (trichlorofluoromethane) [2].

A cylindrical configuration was used for the inverter. The six electronic circuit packages would be contained in a right circular cylinder which is filled approximately by 90-percent in volume with R-113. The cylindrical container serves as the R-113 condenser. When boiling occurs, a vapor zone above the liquid is generated and condensation takes place on the metal surface of the container and the vapor returns the liquid state. As such, the container/condenser should be designed to meet both inverter packaging and heat transfer requirements. The R-113 temperature selected for the baseline condition was 93.3°C (200°F). The corresponding vapor pressure is 54.7 psia.

Forced air cooling of the inverter assembly was selected due to simplicity. F-1 shows a schematic of the forced convection cooling approach. AiResearch plata-fin matrices are placed over the hollow cylindrical inverter and 130°F air is blown through the fin passages keeping the inside metal surface relatively cool for R-113 condensation. Also shown in F-1 is the effect of air volumetric flow on the overall size of the inverter and on the resultant pressure drop across the plate-fin matrices. With a fixed air flow, the plate fin configuration  $(12R-0.3-0.5(0)-0.006(al))^*$  indicated high heat transfer capability while keeping the pressure drop within acceptable range. To acquire a low fan energy consumption, the pressure drop which is a square function of the air flow should be kept at a minimum. The design air flow of 50 cfm exhibits this characteristic depicting a pressure drop of 2 inches of H₂O. An incremental reduction of the air flow will significantly increase the inverter size as projected in F-1. An increase in air flow results in a substantial increase in the pressure drop.

A diameter of 5 inches (over fins) was chosen for the 50-amp inverter. The data of Figure C-6 reflects this choice. A length of 8 inches for the heat exchanger provided the required surface area. Table C-5 provides a summary of pertinent data.

Dimensions for alternate ratios of length to diameter may be calculated by maintaining a constant heat exchanger surface area. Also, the unit may be scaled for larger current ratings by holding the dissipated watt-inch-¹ constant, and maintaining boundry conditions.

^{*}AiResearch heat exchanger designation.



.....

r

Figure C-6. Effect of Air Flow on Inverter Size

322

1_____

----

# Table C-5

....

50-Amp	inverter	Heat	Exchanger	Summary	
--------	----------	------	-----------	---------	--

•	Diameter	5.0 in
•	Length	8.0 in
•	Configuration	Aluminum, radial fins
•	Coolant Data	
	Туре	R-113
	Temperature	200°F
	Pressure	54.66 psia at 200°F
	Density	97.45 lb-ft- ³ at 80°F
	Volume	90 percent fill
•	Cooling	
	Туре	Air
	Temperature	130°F
	Flow	50 cfm

# References

And the second second second

- 1. Rowe AiResearch Report No. 80-17284, <u>Electromechanical Airplane Actuation</u> <u>Trade Study</u>
- 2. Kim AlResearch Document No. 09303-46685-D11, "Pole Face to R-11 Thermal Resistance - Evaporative Cooling of Power Semiconductors for Rail Vehicle Choppers"

# APPENDIX D

# AIRPLANE ACTUATION TRADE STUDY

This appendix includes the work done by AiResearch, under the direction of S. Rowe as part of their effort, in accordance with service agreement L9FM-11231-405, to provide a detailed design of the inboard flap actuation system. This detailed design was undertaken to give a higher level of confidence in the projected weights, envelopes and life cycle cost figures used for the other 6 flight control actuators. The data generated in this appendix also provided a basis for the motor weights shown in Figure 38.

PEL

# ALL ELECTRIC AIRPLANE STUDY SERVICE AGREEMENT L9FM-11231-405 PROGRESS REPORT FOR THE PERIOD FEBRUARY - MARCH 1981: ADVANCED ELECTROMECHANICAL ACTUATION SYSTEMS

#### PREFACE

This document is submitted in conjunction with Service Agreement L9FM-11231-405, All L actric Airplane Study. The data submitted herein is presented to facilitate completion of the subject study. This document, in addition to previously submitted data, shall serve as a progress report for February - March 1981.

#### INTRODUCTION

Specific tasks completed during this reporting period were:

- Inboard flap actuation system detailed design
- Motor performance trades based on 1990 technology projections
- Study overview and summary

Each of the above is addressed in this progress report.

This submittal satisfies the statement-of-work (SOW) data requirements, and completes the subject study.

#### DETAILED DESIGN

Detailed design of the inboard flap actuation system was performed in accordance with the SOW [1]. The inboard flap was selected by mutual agreement between the customer and AiResearch.

The completed design is described in the following drawings:

Drawing No.	Title
2022824	System Outline, Flight Control
2022794	Gearbox Outline, Flight Control
2022796	Gearbox Outline, Flight Control
2022798	Power Drive Unit Outline

Performance data for the system is tabulated in Table D-1. System weights are shown in Table D-2, including the required motor inverters and a system controller. Additional system and component data may be found on the drawings.

#### Table D-1

# Inboard Flap Performance

+30*

Stroke Rate Load Bandwidth (+1°) Static Stiffness

-45° 100 deg-sec-1 453 X 10³ in-1b > 30 rad-sec⁻¹ (at 0 dB) 58 X 10⁶ in-1b

# Table D-2

Inboard Flap Weight Summary

Controlle	ər :	I	X	5 Ib	=	5	IЬ
Inverter	:	3	х	38	2	114	
Actuator	:	1	х	100	=	100	
Position	Feedback:	1	х	1	=	I	

220 Ib

A brief description of the actuation system and its components is given below.

# Actuation_System Description (2022824)

The inboard flap actuation system was designed to satisfy the operating and performance requirements determined by the customer and AiResearch [2]. Thy actuation system consists of the following components:

I tem	Balloon No.	Qty	Description
1	-	1	Controller
2	-	3	Inverter
3	1	١	Power Drive Unit
4	2	1	Torque Tube
5	3	1	Reduction Gearbox
6	4	1	Torque Tube
7	5	1	Hingeline Gearbox
8	-	1	Position Feedback Assembly

Items 1 and 2 have been addressed in previous reports [1, 2, 3]. Item 8 was assumed to be a module containing three RVDTs and reduction gearing, but was not addressed in detail.

Operation of the actuation system is typical of 270 vdc systems (see Figure D-1 and drawing No. 2022824). The controller commands the inverters as a function of servo position and motor speed feedback. Each inverter in turn controls the current to its respective motor in the power drive unit (PDU) by means of a current feedback loop. Motor response is torque summed in the PDU, and is output through a torque tube at motor speed (1:1 ratio). This torque tube then drives a reduction gearbox (88:1 ratio) and makes a 90° turn, so that the reduction gearbox output lies on the hingeline gearbox centerline. The hingeline gearbox (15:1 ratio) then positions the inboard flap. Feedback loops are provided at the reduction gearbox (actuator position), PDU (motor speed and rotor position), and inverters (motor current).

Actuation system components are described in the following paragraphs.

#### Power Drive Unit (2022798)

Each POU consists of three brushless direct current-permanent magnet (DC-PM) servomotors, each powered from a 270 vdc source via inverters. Motor torque is output through clutches (which can decouple a failed motor from the actuator drive) to a torque summing gear train. PDU output is at 90° to the motor axes due to actuation system installation. All gearing is supported on ball bearing assemblies. Due to motor/gearing speeds (max speed is 22 Krpm), oil sling lubrication is used in the PDU gear housing.

Additional data on motor design may be found in drawing 2022798, and in a later section of this report. Additional data on the PDU gearing is presented in the drawing, also.

# Reduction Gearbox (2022796)

The reduction gearbox provides speed reduction between PDU and the hingeline gearbox. It consists of compound planetary gearing supported on roller bearing assemblies; and a bevel gear at the input supported on ball bearing assemblies. Position feedback for the actuation system is provided by an output shaft which drives a redundant RVDT feedback module.

See drawing 2022796 for additional information.

#### Hingeline Gearbox (2022794)

The hingeline gearbox comprises fourteen identical stages ("slices") of compound planetary gearing operating in mechanical parallel. The load is distributed uniformly (approximately) along the length of the gearbox. An input shaft, which runs the full length of the gearbox, is supported on ball bearing assemblies and engages planets at each slice. The planets then drive each output slice, positioning the inboard flap.

Or awing 2022794 provides additional data on the hingeline gearbox.

-----



A STATE AND A STATE AND A STATE AND A STATE AND A STATE

1.43

South Level & Sugar Street Street

1.16



. . . . . . . . .

A-14023

Appendix D

#### Comments and Recommendations

The design presented in this report for the inboard flap meets or exceeds the operating and design requirements agreed upon by the customer and AiResearch. This has incurred a penalty in terms of system weight, however.

Referring to Table D-1, it is noted that the maximum operating load (stail) of the actuation system is  $453 \times 10^3$  in.-1b, while the maximum load seen by the inboard flap is  $216 \times 10^3$  in.-1b [4]. This results from the design stall torque of the three torque summed motors. The drive of the actuation system, including the hingeline gearbox, was designed to accommodate this stalled condition. If an acceptable assumption is that the maximum drive load is  $216 \times 10^3$  in.-1b under any circumstance, then system weight could be reduced. For example, the hingeline gearbox weight would drop from 50 lb to approximately 25 lb if this assumption were enforced.

It is recommended that the customer evaluate the feasibility of this design assumption.

#### MOTOR PERFORMANCE TRADES

Motor performance trades using assumed 1990 technology were performed early in the study. Data resulting from this effort is presented in Appendix A.

All motors used in actuation systems sized or designed during the study were selected in accordance with the criteria of Appendix A.

#### STUDY OVERVIEW AND SUMMARY

This section presents an overview of the work accomplished during the study, in chronological order; and a summary of study findings.

#### Overview

The study SOW [1] was reviewed during September 1980 and work initiated during that same month. Trades with particular emphasis placed on actuation system configuration and control were performed and used to establish the approach to be utilized [3].

From October 1980 through January 1981 the customer's design requirements were evaluated [4]; and preliminary designs for specific actuation system and related components were completed [2, 5].

February through March 1981 saw the selection of the inboard flap actuation system for in-depth analaysis, and completion of its detailed design (5).

# Summary

BRIDGEN SAME

Conclusions drawn from the study program are presented below:

 Motor designs investigated provided low weight designs with excellent dynamic response capability. For the environment and duty cycle considered, active cooling was found to be unnecessary.

......

- Inverter designs are driven by thermal considerations. For the device characteristics and duty cycle assumed, evaporative cooling appears to be a likely choice for high current units (25 amps or more); while tow current units (25 amps or less) were adequately cooled by means of conventional heat sinking.
- Actuation system control is best accomplished by a microprocessor based controller for surface position, motor speed, and rotor position. Due to required calculation rates, use of digital control for the current loop is impractical and will remain analog.
- 4. Available envelopes for the actuation systems played an important role in establishing actuation system configurations. In most cases, PDUs driving hingeline rotary gearboxes were found to be most suitable.
- Actuation system weight in most cases can be reduced by sizing the mechanical drives for maximum design loads, as opposed to simultaneous motor stall. The acceptability of this approach would require review by the customer.

#### REFERENCES

- Green -- AiResearch Letter No. CBRG:9327:1011, Subject: Rockwell Service Agreement, L9FM-11231-405, <u>All Electric Airplane Study</u>.
- Rowe -- AiResearch Report No. 80-17351 (2), <u>All Electric Airplane Study</u>, <u>Progress Report for the Period October 1980 - January 1981</u>.
- Rowe -- AiResearch Report No. 80-17351 (1), <u>All Electric Airplane Study</u>, Progress Report for the Month of September.
- Heisley -- Rockwell International Report No. NA-79-492, <u>Airplane Actuation</u> Trade Study, First Interim Technical Report.
- Rowe -- AiResearch Report No. 80-17351 (3), <u>All Electric Airplane Study</u>, Progress Report for the Period February - March 1981.

ł

APPENDIX A

- ----

- 10 M

MOTOR SIZING, ANALYSIS AND COMPUTER PROGRAMS

#### APPENDIX A-1

# MOTOR SIZING AND ANALYSIS

Motor selection is probably the most critical phase of EM actuation system design. The motor selection will determine actuator performance, influence actuator configuration, and dictate actuation system control. Design constraints placed on the motor will affect the actuation system design, also.

For primary flight control servo applications, brushless DC-PM (direct current-permanent magnet) motors appear to be the most likely motor candidate [1, 2]. A 270 to 300 vdc power source is probable, due to the availability of 400 Hz 3-phase ac power and past experience with high voltage dc actuation systems [1, 3, 4]. Servo control would be accomplished by means of a transistorized inverter, with control logic for inverter switching (commutation), current limiting, and control law implementation. Figure D-2 illustrates this type of machine.

For other serve applications, induction motors and brush DC-PM motors are likely candidates. Advancements of these types of motors were believed to be of less importance than the brushless DC-PM motor for high performance serve applications and thus not considered for trade studies.

#### A.1 ASSUMPTIONS

Assumptions defining technology advancements and motor configuration were made, and used as a starting point for motor trades. The assumptions developed were believed to be realistic and reasonable, for the time frame involved. Assumptions with rationale are provided below.

#### A.1.1. Magnets

Determination of magnet energy product and characteristics are significant tasks in motor design. Materials with energy products as high as  $30 \times 10^6$  gauss-cersted have been developed, and materials with energy products of 23-26  $\times 10^6$  gauss-cersted are readily available [5, 6, 16]. However, most magnets are presently supplied in the 16-22  $\times 10^6$  gauss-cersted range. While increased energy product is generally desirable for motor design, other factors must be considered in material selection.

Coercive force (H) must be sufficiently large in magnitude to allow full utilization of the material's flux density (B). Insufficient coercive force could allow demagnetization due to motor currents beyond design limits (short circuit, current limit overshoot, etc.). Additionally, elevated operating temperatures decrease magnet energy product, and susceptibility to demagnitization increases. These effects are illustrated by the B-H plots shown in Figure D-3.

increased magnet energy product  $(23-30 \times 10^6$  gauss-cersted) is also associated with increased cost. This is due primarily to the need for significant fractions of the less common rare earths and limited source metals (semarium, praseodymium, and cobalt). Increased demand for these materials would affect cost and availability in the future  $\{17\}$ .



Figure D-3. Comparison of Magnet Materials [16]

The number of manufacturers producing high-energy product magnets is quite limited. Most deal in the  $16-22 \times 10^6$  gauss-persted range. This factor would tend to restrict future supplies, also.

Recent trends in rare earth-permanent magnet development emphasize reduced cost and improved supply rather than increased energy product [17]. In light of the aforementioned factors, the probability of using very high energy product magnets appears to be unlikely. Rather, use of new magnet materials with more desirable characteristics in the  $20-24 \times 10^6$  gauss-persted range appears likely for the near term.

For these reasons an energy product of  $22 \times 10^6$  gauss-persted was used as a reasonable compromise. This energy product represents a balance between B, H, cost, and availability for rare earth-permanent magnets during the near term.

It should be noted for the operating temperatures anticipated (250°F ambient) that any difference in performance between machines using  $22 \times 10^6$  and existing  $30 \times 10^6$  gauss-persted materials would be negligible due to the degraded performance of the highest energy product materials at elevated temperature.

## A.1.2 Motor Speed

A maximum motor no-load speed of 25 Krpm was assumed as a design constraint, for reasons explained later. Running motors at the highest speed possible is desirable since this reduces motor size and weight for a fixed output power.

The following relations define motor power, speed, and torque:

$$\dot{\mathbf{w}}_{\mathbf{m}} = \mathbf{T}_{\mathbf{i}\mathbf{n}} \, \dot{\mathbf{\Theta}}_{\mathbf{m}}$$
 (A-1)

 $T_{m} = K_{\pm} I_{m} \tag{A-2}$ 

$$\dot{\mathbf{e}}_{m} = \mathbf{K}_{m}^{-1} \mathbf{V}_{m}$$
 (A-3)

where

 $\tilde{W}_m$  = motor power

T_m = motor torque

 $\tilde{\Theta}_m$  = motor speed

Im = motor current

V_m = motor voltage drop

K₊ = torque constant

K. = voltage constant

Increasing motor speed will decrease motor copper loss* due to winding changes, allowing reduced machine dimensions. Thermal effects are usually a constraint in motor dimensions.

Motor speed cannot be increased indefinitely, however, due to motor and actuator design constraints. Hoop stress in the rotor magnet sleeve due to centrifugal loads must be considered, as well as rotor surface velocity. In electrical machines, stator hysteresis losses and eddy currents can become excessive as frequency increases. Inverter switching frequency and switching losses may increase. Additional drive reduction is required at high speeds, for a fixed output speed, and gearing is usually velocity limited to 30 Krpm.

Thus, an upper limit of 25 Krpm was used as a reasonable compromise, based on design constraints and past experience.

#### A.1.3 Dynamic Response

Since the motors were meant for use in primary flight control servos, frequency response was a prime consideration as a performance parameter. Most EM actuation systems are acceleration limited, thus determining the maximum possible bandwidth.**

Bandwidth (no-load) for a single motor servoactuator, assuming acceleration saturation, may be calculated from

$$\omega = \sqrt{\frac{\Theta_{\rm m}}{A \times G}}$$
 (A-4)

$$\dot{\Theta}_{\rm m} = \frac{T_{\rm m}}{J_{\rm eff}}$$
 (A-5)

$$T_{m} = K_{\dagger} I_{m}$$
 (A-6)

$$J_{eff} = J_{m} + \frac{1}{G^{2}J_{1}}$$
 (A-7)

where

 $\omega$  = bandwidth

ë_m = motor acceleration

T_m = motor torque

t_m = motor current

* Winding resistance losses; see paragraph A.1.4.

**This is true for operation as a linear system; i.e., up to acceleration saturation. Bandwidth for the system in the nonlinear (saturated) region may be acceptable depending upon the frequency response specification.

AIRESEARCH MANUFACTURING COMPANY 336

- J_m = motor inertia
- J₁ = load inertia
- J_{eff} = effective inertia, reflected to motor
- Kt = torque constant
- A = actuator position command amplitude
- G = actuator gear ratio
- n = gearing efficiency

Equations (A-4) through (A-7) illustrate that there are numerous possible solutions for motor and actuator parameters  $(I_m, K_T, J_m, G, T)$  given  $\omega$ , A, and  $J_1$ . Attempts have been made to develop analytical procedures for "optimum" motor design based on bandwidth requirements; but they failed to address other motor and actuator constraints [1, 7, 8, 9].

In order to develop a family of motors which would provide adequate bandwidth for the various actuation systems under consideration, the following criteria were employed:

$$T_{m} = \frac{e_{rated}}{\Theta_{stall}} = 50 \text{ msec}$$
(A-8)  
$$\frac{1}{d} = \frac{1}{2r} \le 3$$
(A-9)

where

 $\theta_{rated} \approx motor speed at rated power

 <math>\theta_{stall} \approx motor acceleration at stall torque

 <math>\tau_m$  = motor time constant

 d
 = motor rotor diameter

 i
 = motor rotor length

r = motor rotor radius

Figure D-4 illustrates motor dimensional parameters; Figure D-5 illustrates performance parameters.



Figure D-4. Brushless DC-PM Motor Detail



Figure D-5. Motor Performance Points

(A - 11)

 $\tau_m$  is the time required for the motor to accelerate from Point E to Point D of Figure D-5. The motor accelerates under constant torque with no-load, which is accomplished via current limit control in the inverter. This parameter is a measure of the motor's dynamic response; it should not be confused with the time constant which reflects the time required for a motor to reach 63.2 percent of commanded speed for a step commands i.e.,

$$\dot{\Theta}_{m}(t) = \begin{bmatrix} V_{m} \\ K_{\Theta} \end{bmatrix} (1 - \exp(-t/\tau))$$
(A-10)

where R is the motor winding resistance, t is time, and  $V_m$  is a step voltage applied across the motor. The above time constant results from a linear analysis assuming voltage control of the motor, and does not assume constant torque (constant current) operation [10].*

Equation (A-9) is a dimensional constraint owed to manufacturing difficulties and dynamic stability. Generally, if the rotor 1/d ratio exceeds 3:1, winding the motor stator becomes difficult due to motor configuration (see Figure D-6); dynamic stability of the rotor becomes a concern due to bending modes. Also a large 1/d ratio favors lower values of  $\tau_m$ , since this produces a lower rotor inertia than a small 1/d.

Utilizing the two constraints,  $\tau_m$  and 1/d, allowed a family of motors to be designed without regard to detail actuator performance characteristics. Since the motors were expected to satisfy or exceed dynamic response requirements for most actuator applications, motor characteristics could be parametrically taculated for trend analysis. Also, motors could be selected by use of parameters other than dynamic response, greatly simplifying the design process.

# A.1.4 Thermal Considerations

 $\tau = \frac{J_m R}{K_+ K_p}$ 

The performance of any motor is limited by duty cycle. Losses developed by the motor as a function of load and speed must be considered during motor design (and during motor selection), or overheating of the windings may occur.

The two most significant losses which must be accounted for are resistance (copper) losses, and hysteresis and eddy current (iron) losses. Relations for these losses are  $\{1\}$ :

*The analysis used (equations A-10 and A-11) develops the motor mechanical time constant for the case of negligible inductance and viscous losses.



Figure D-6. Stator Cross Section

 $\dot{W}_{cu} = I_{mR}^{2}$   $\dot{W}_{ir} \propto \omega_{m} \kappa_{h} + \omega_{m}^{2} \kappa_{ec}$ (A-12)
(A-13)

where

W_{cu} = copper loss

Ŵ_{ir} = iron loss I_m = motor current w_m = motor (commutation) frequency R = winding resistance K_{ec} = eddy current loss coefficient K_n = hysteresis loss coefficient

Iron losses are usually controllable by judicious machine construction and material selection; copper losses may be difficult to control depending upon stator resistance and duty cycle. In order to maintain manageable steady-state and transient temperatures, a maximum current density of 18,000 amp per square inch of conductor was chosen. The selection of this value was based upon analysis of the specification duty cycle (see paragraph A.2).

## A.1.5 Other Assumptions

. . . . . . . .

-----

Other assumptions for motor design included a six-pole rotor configuration, with radially oriented magnets. Additional constraints and assumptions were made, consistant with experience.

Table D-3 summarizes the principal design assumptions and constraints.

# TABLE D-3

MOTOR DESIGN ASSUMPTIONS AND CONSTRAINTS

Magnets:	BH = 22 X 10 ⁶ gauss-cersted
No-Load Speed:	<mark>∲_{max} ≤</mark> 25 Krpm
Time Constant:	τ _m ≤ 50 msec
Rotor Dimensions:	I/d ≤ 3
Stator Current Density:	18 Kamp-in ⁻²
Rotor Configuration:	6 pole, radial magnet orientation

#### A.2 DESIGN INVESTIGATION

A family of motors using the criteria of Table D-3 was designed by use of a computer program developed at AiResearch [12]. A brief description of the program is given in Appendix A-2.

Motors were designed as a function of peak (rated) power, over a range of horsepower. Key motor data are tabulated in Table D-4. Some of the data were used to calcutate parameters of interest and plotted, also. These data are shown in Figures D-7 and D-8.

Since motor dynamic response was a key parameter in selecting motor design constraints, a more detailed investigation of  $\tau_m$  and its relation to other motor parameters was conducted. Two general cases were examined: (1) motor characteristics as a function of 1/d for fixed rated speed and  $\tau_m$ ; and (2) motor characteristics as a function of  $\tau_m$  for fixed  $\dot{W}_m$  and 1/d.

Four values of 1/d were investigated for the first case. A rated load speed of 20 Krpm was used. Data for this case are tabulated and plotted in Table D-5 and Figure D-9, respectively.

In the second case, three values of  $\tau_m$  were run, assuming a motor power of 60 hp. This power rating was used since  $\tau_m$  can cause extreme variations in motor size and weight for larger machines. Data for this case are presented in Table D-6 and Figure D-10.

TABLE D-4

MOTOR DATA SUMMARY

F

			Re	ted Power	( hp )		
Parameter	-	ŝ	01	15	20	25	30
Housing Diameter (in.)	2.0	2.5	2.7	2.9	2.9	3,1	3.2
Housing Length (in.)	4.7	5.9	7.0	8.0	8.9	9.4	9.6
Motor Weight (1b)	1.0	2.0	3.5	5.0	6.5	9.5	10.5
Current Limit (amp)	3.9	17.9	35.4	53.1	70.8	0.06	106.3
Thermal Flux Density (watt-in. ² )* 25 percent 100 percent	0.83	1.05 16.8	1.14 18.3	1.21 19.3	1.24 20.0	1.24 20.0	1.32
Speed (krpm)** ● A ● B, D	25.6 20.0	23.9 20.0	23.5 20.0	21.0 18.0	21.0 18.0	16.5 16.0	15.1
Torque (in15)** ● D, E	3.1	15.7	31.7	52.8	70.5	98.8	126.6
Time Constant (msec)	46.3	48.0	49.7	49.7	49.8	50.0	50.0

* Copper loss only.
** See Figure A-4.

....

Appendix D



į

Figure D-7. Motor Data Summary



Figure D-8. Motor Data Summary
TABLE D-5

MOTOR DATA AS A FUNCTION OF 1/d

		Katio*,	(-) P/I	
Parameter	1.5 (1.523)	2.0 (2.033)	2.5 (2.528)	3.0 (3.147)
Rotor Diameter (in.)	1.485	1.530	1.557	1.610
Housing Diameter (in.)	2.7	2.8	2.8	2.9
Housing Length (in.)	6.4	7.3	8.1	9.3
Motor Weight (1b)	3.0	3.7	4.8	6.5
Current Limit (amp)	26.6	41.4	56.3	80.6
Thermal Flux Density (watt-in. ² )** • 25 percent • 100 percent	1.11	1.15 18.4	1.20 19.2	1.24 19.8
Speed (krpm)*** • A • B, D	23.3 20.0	23.3 20.0	23.3 20.0	23.2 20.0
Torque (in1b)*** • D, E	23.8	1.75	50.5	72.3
Time Constant (msec)*	50 (52.0)	50 (51.3)	50 (50.8)	50 (51.9)
Rated Power (hp)	7.55	11.76	16.01	22.96

Appendix D

*Specified (actual) **Copper loss only ***See Figure A-4.



CONSTANT VARIABLES: ė_M = 20k rpm (POINT D) τ_M = 50 msec (POINT E TO D)

Figure D-9. Motor Data, Constant  $\dot{\theta}_m$  and  $\tau_m$ 

346

TABLE D-6

A CONTRACTOR OF A CONTRACT

MOTOR DATA AS A FUNCTION OF  $\tau_m$ 

	Ē.	e Constant (msec	*(;
Parameter	50 (50.4)	75 (74.9)	100 (97.8)
Ratio*, 1/d (-)	3.0 (3.089)	3.0 (2.969)	3.0 (2.695)
Rotor Dlameter (in.)	2.949	2.385	2.239
Housing Diameter (in.)	4.8	4.0	3.7
Housing Length (in.)	14.0	11.7	10.5
Motor Weight (1b)	34.8	í <b>8.</b> 7	13.7
Current Limit (amp)	210.5	210.5	208.2
Thermal Flux Density (watt-in. ² )** • 25 percent • 100 percent	1.35 21.6	1.25 20.0	1.41 22.5
Speed (krpm) ^{***} • A • B, D	8.1 7.0	17.4	24.2 21.0
Torque (in1;)*** • D, F	524.4	252.8	181.9
Rated Power*	60. (58.2)	<b>60</b> (60.2)	60 (60.6)

Appendix D

2

*Specified (actual) **Copper loss only ***See Figure A-4



Figure D-10. Motor Data, Constant 1/d, W

### A.3 THERMAL INVESTIGATION

A preliminary thermal analysis of one motor was performed as part of Task 3. The object of the analysis was to determine motor cooling techniques to assure that critical motor elements (i.e., permanent magnet, copper windings, etc.), remained within material temperature limits. A summary of the analytical results completed is presented below.

The motor selected for analysis was the 20 hp motor presented in Paragraph A-2. This motor was selected because its thermal flux density was representative of most of the motors generated. The motor was operated under the following duty cycle:

- (a) A steady-state condition at 10% speed and 25% load.
- (b) A two-minute transient condition immediately after case 1 at 100% speed and 50% load.
- (c) A short time (less than 1 second) transient-state condition immediately after case 2 at 100% speed and 100% load.

The provisions above were used as a baseline condition for the thermal analysis, and were determined to be representative of the aircraft requirements.

There were two motor cooling techniques investigated in the analysis. The first implemented fins on the housing to increase the natural convection and radiation heat transfer to the surroundings (no forced air in the motor). The fins would be cast on the 1/8 inch thick aluminum stator housing, spaced 0.25inch apart with a 0.04-inch fin thickness and a 0.5-inch fin height. Figure D-11 shows this finned housing configuration. A schematic diagram of the second cooling technique appears in Figure D-12. Cooling air is forced through the rotor to keep the Sa-Co (samarium-cobalt) permanent magnets at an acceptable temperature level. Figure D-13 shows a drawing of the rotor cooling flow passages: (1) rotor-stator gap-flow, (2) 6-32/32-inch hole-flow, and (3) rotor web shaft flow. This cooling scheme did not use fins on the housing. The motor thermal nodal network is also shown in Figure D-13. The first of each set of numbers represents the air inlet half of the motor, while the numbers in parenthesis represent the air outlet half. The Sa-Co permanent magnets (nodes 1 & 6) are contained by a sleeve (nodes 43 and 44). The stator stack windings are represented by nodes 11-12 and 26-27 while the end-turn and end-section windings are nodes 13-14, 28-29 and 15, 30, respectively.

The motor nodal model was prepared by means of the AiResearch Stator Armature, and LIM Thermal Model Generation Computer Program H0061 (Appendix A-2). The program connects each of the motor nodes by a conduction and convection resistance array such that the motor thermal model may be combined with the rest of the system. The complete model is then analyzed by the AiResearch Thermal Analyzer Computer Program H0910 (Appendix A-2). The latter program is capable of simulating conduction, convection and radiation calculations and other heat transfer mechanisms.



Figure D-11. External Cooling Scheme



....

A-2234

Figure D-12. Internal Cooling Scheme

.....





Figure D-13. Thermal Nodal Model

351

The losses of the 20 hp motor for the duty cycle are shown in Table D-7, and were obtained from the motor design program [12]. All entries in the table are in watts and copper losses are based on 300°F temperature. The actual copper losses used in the analysis, however, varied as function temperature depending upon the resistance characteristics of copper.

1.67 24

# TABLE D-7 MOTOR LOSS SUMMARY

Condition	Motor Speed (rpm)	Copper	Tooth	Back Iron	Pole Head	Windage	Total
10% speed 25% load	<b>180</b> 0	63.72	8.92	8.16	3.51	0.02	84.33
100% speed 50% load	18000	252.40	135.50	120.0	44.66	5.26	557.82
100% speed 100% load	18000	1021.30	127.80	113.20	115.56	5.26	1383.12

Notes: All losses are in watts Copper losses are given at 300°F

The results of the thermal analysis for critical motor elements are tabulated in Table D-8. Entries under cooling scheme A and B utilized the finned housing configuration where motor cooling was achieved by natural convection and radiation to a 130°F and 250°F ambient, respectively. Cooling schemes C and D, on the other hand used forced air through the rotor to cool the motor with an air inlet of 130°F and 250°F, respectively, both with 250°F ambient. As noted in the table, temperatures under A1, B1, C1, and D1 depict steady state temperature predictions with 10 percent speed and 25 percent duty load under the respective cooling scheme. Entries under A2, B2, C2, and D2 are temperatures at the end of the two-minute transient state condition with 100 percent speed and 50 percent load, immediately after the steady state run. Likewise, temperatures under A3, B3, C3 and D3 are the transient response at indicated elapsed time with 100 percent speed and 100 percent load, immediately after the two-minute transient run.

Although the ambient temperature was specified to be close to  $250^{\circ}$ F, cooling scheme A with 130°F ambient was also analyzed to show the cooling effectiveness of the finned housing configuration. The time-temperature response of a complete cycle under cooling schemes B and C is shown in Figure D-14. The 250°F ambient and 130°F inlet temperature for forced cooling were used as a baseline condition that was specified. The volumetric flow rate in cooling schemes C and D was ~10 cfm with a 130°F inlet temperature. This corresponds to a maximum pressure differential of 0.70-in H₂0 from the inlet to the outlet of the air stream shown in Figure D-12. TABLE D-8

MOTOR ELEMENT TEMPERATURE SUMMARY

	Cool	ing S	cheme - A*	Cool	ing S	cheme - B	Cool	s gul	cheme - C	Cool	ing S	icheme - D
E lemen t	A1+	A2	A3(10 sec)	81	B2	B3(2 sec)	CI	C2	C3(10 sec)	10	D2	03(5 sec)
Permanent Magnets (samariam cobalt)	226	257	264	345	378	380	205	249	257	308	353	357
Copper Windings	230	306	385	352	437	458	230	308	382	324	411	459
Top Tooth	223	276	295	343	401	402	226	29B	318	321	396	405
Back Iron	219	262	270	338	385	386	229	298	309	322	394	399
*Cooling Schemes: A - Finned housin B - Finned housin C - Forced coolin O - Forced coolin	g, na gg, na g air air	tural tural thro thro	convection convection ugh the roto ugh the roto	and r and r witr	adiat adiat h a 1 h a 2	ion cooling ion cooling 30°F air in 50°F air in	t t a t t a t t a t t a	130° 250° emper	F - amblent F amblent ature, no fl ature, ro fl	ns, 2 ns, 2	50 °F	- amblent - amblent
+Canditions:	,											
AI, BI, CI, DI, - A2, B2, C2, D2, -	Stea Two-	dy-st minut	ate temperat e transient	ure p †empe	iratur	tions e response	6 10X	spee load spe	d - 1800 rpm ed - 18000 r	b and pm an	1 1 1	
A3, B3, C3, D3,	- Tra	insler	it temperatur	e res	sponse	at indicat	ed el	apsed	1001	load		



4

Sector Same

S. Cake

وموردها والمربعا بالام استراكات

Figure 0-14. Motor Temperature Profiles

354

## A.4 CONCLUSIONS

Manager and Property of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Party of the Par

Conclusions concerning motor design, performance, and selection were drawn during Task 3. These conclusions are presented in the following paragraphs. The reader should note that these conclusions are valid for the design criteria used.

#### A.4.1 Motor Time Constant

A given motor time constant limits allowable motor rated power or speed.

First, Figure D-9 indicates that if 1/d is a design constraint, then there is a maximum power which can be achieved for a given rated speed and time constant. As an example, Figure A-8 indicates that for

then

 $W_{max} = 30 \text{ hp}$ 

Second, Figure D-10 illustrates that as time constant decreases, the motor rated load speed must decrease. For example (Figure D-10), if

then

reason for these offects is that motor inertia and torque determine enation; and motor rated speed effectively determines  $\tau_m$  for some coloration. Although many individuals associate high motor speeds asponse, the above indicate that the apposite may generally hold true. example, given two motors, the motor with the greater acceleration will not necessarily have the fastest response when coupled to a load through a reduction drive which provides a fixed no-load speed.

## A.4.2 Motor Power Rate

Motor power rate is a near linear function of rated power or time constant.

Figure D-9 shows power rate as a function of rated power. For the design criteria used, power rate is an exceptionally linear function of rated power.

متعطي معيومة والرابية المراري المالية

Note that rotor length-to-diameter ratio varies with motor power also; but this variation may be interpreted as the necessary roror geometry to satisfy the fixed constraints. In this case, power rate may be construed as a direct, linear function of rated power.

By referring to Figure D-10, one sees that power rate is a nearly linear function of time constant; actually, for the presented data, a possible empirical form for power rate may be

where

. . . . . .

W = motor power rate T_m = motor time constant
a, n = empirical coefficients

However, approximating power rate as a linear function of time constant over a limited range appears to be reasonable. This conclusion holds for the design criteria of the figure: constant rotor length-to-diameter ratio and power.

Power rate is sometimes used as a parameter to select motors for servo applications [1, 2, 7, 8, 9]. it is a convenient parameter for comparing various motors, and indicates the ability of a motor to accelerate a load as well as itself. Motor acceleration indicates only the ability of a motor to accelerate itself, and is useful primarily for no-load dynamic performance analysis.

A cursory derivation of motor power rate may be useful to the reader *:

 $\hat{\mathbf{w}} = \mathbf{T}_{\mathbf{m}} \, \hat{\mathbf{\Theta}}_{\mathbf{m}}$  (A-15)

where

 $\dot{W}$  = motor rated power  $T_m$  = motor rated torque  $\dot{\theta}_m$  = motor rated speed

now

*Other deviations of power rate are presented in the literature [7, 9]

For the case of constant torque applicable for our definition of  $\tau_{\rm m}$ 

1

 $\dot{T}_m = 0$ 

so

And a second second second second second second second second second second second second second second second

A REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PROPERTY OF THE REAL PR

$$\tilde{W} \triangleq T_m \tilde{\Theta}_m = T_m \frac{T_m}{J_m} = \frac{T_m^2}{J_m}$$
 (A-17)

The above is defined as motor power rate.

#### A.4.3 Motor Weight

The importance of properly selecting a motor for minimum weight is emphasized by the data of Paragraph A.2.

Figure D-8 shows that motor weight increases in direct proportion to motor power. By referring to Table D-4 one sees, however, that nearly all of the motors are in the same rated speed range (the 25 and 30 hp motors are exceptions).

Examining Table D-6 provides a totally different trend: three different weights for the same rated power and different time constants. For this case, motor weight increases as rated speed declines and as motor time constant declines. This is a penalty associated with increasing dynamic response.

This comparison emphasizes the need to utilize a motor with a time constant (power rate) no smaller (larger) than necessary to satisfy actuator dynamic performance requirements.

#### A.4.4 Motor Thermal Response

All cooling schemes investigated were acceptable for the duty cycle and assumptions utilized. Thus, no motor cooling would be required for any of the FCS actuation systems.

All temperature entries tabulated in Table D-8 are acceptable temperature levels for the different critical motor elements. To indicate the magnitude of temperature response during the 100-percent speed and 100-percent load condition, the elapsed times noted in the table are longer than the required operation time of less than 1 second. Temperature reponse at these conditions and at this short time interval run very close to that at the end of the twominute transient condition with 100-percent speed and 50-percent duty load.

It should be noted that if higher conductor current densities and current limits had been used, it is likely that motor cooling (such as cooling scheme C) would have been required for some of the FCS applications. Thus, the assumptions developed in Paragraph A.1.4 have greatly simplified the actuation system design process.

APPENDIX A-2

- 7 AN Y

- ----

سردمية المروح والمنافر

------

# COMPUTER PROGRAMS

## COMPUTER PROGRAM BIGMAG

The Garrett Corporation has developed a computer program to facilitate the design and evaluation of certain types of permanent magnet machines interfacing with converters or with conventional linear systems. This program is called BIGMAG. Generally, only a few basic inputs have to be specified by the user. BIGMAG then synthesizes a baseline design suitable for optimization studies.

## ELECTROMAGNETICS AND LOSSES

Classic salient-pole, two-reaction, analysis of synchronous machines has been adapted to meet the special cases of tangentially magnetized and radially magenetized permanent-magnet (PM) rotors. The magnetic circuit is represented by an equivalent circuit in which the iron and leakage paths and the magnet and stator MMF's are represented by lumped parameters. Nonlinearity of the iron reluctances vs flux density is taken into account by a look-up routine using B-H data tables for a variety of magnetic and PM materials. Special routines account for configuration effects (i.e., tapered teeth, shaped magnets). The flux leakage paths are estimated by special methods evolved from computerized flux plots, by classic field analysis of elemental situations, or by other methods found in the literature.

Losses calculated include: iron losses, pole head and damper head losses, stray load loss, copper and hysteresis losses, and windage losses.

#### STRUCTURAL ANALYSIS

The structural analysis predicts damper hoop stress, average compressive stress of the magnet, pole root stress, and stress in the non-magnetic hub which supports the poles and magnets. All geometric details are taken into account and a table of stress concentration factors and a look-up routine are included.

## THERMAL ANALYSIS

Thermal analysis is limited to determining a credible stator copper current density to attain a specified final copper termperature with a given fluid, fluid pressure drop, and duty cycle. (Conversely, temperature will be determined if current density is specified.) In addition, certain elemental thermal data are given such as watts/lb in tooth and core iron; watts/sq. in. pole head and damper losses; and watts/sq. in. stator iron loss distributed over the outer periphery of the stack.

#### WEIGHT ANALYSIS

The weight of the basic electromagnetic parts is calculated. In calculating rotor weight, vent holes in the poles and under the magnets are taken into account. Also, the weight of the nonmagnetic hub supporting the poles Is included. To account for shaft extensions, bearings, end frames, stator frame, terminals, and fittings, a weight target equal to 20 percent of the electromagnetic weight is used to estimate total weight.

# RECTIFIER/CONVERTER/INVERTER ANALYSIS

į

In addition to linear three phase systems (such as a linear load impedance or power supply connected to the machine), thyristor converters and transistor inverters may be modeled.

## STATOR, ARMATURE, AND LIM THERMAL MODEL GENERATION COMPUTER PROGRAM (HOO61)

A computer program has been developed which prepares a thermal model of an ac motor or generator stator, a dc motor armature, an ac generator field winding, or a linear induction motor. The program constructs a thermal model in the format of the AiResearch H0910 or H0298 thermal analyzer computer programs. This thermal model may then be combined with the thermal model of the rest of the system to form a complete motor, alternator, generator, turboalternator, or flywheel/motor/alternator.

The input and options to the geometry program are available in the output of the AiResearch motor, generator, and linear motor design and performance programs. Where possible the nomenclature and variable names have been preserved to provide easy and accurate model construction. The program has options provided to construct the thermal model of the following:

- One stack and end turn element model
- One half stack and end turn element model
- Two half stack and end turn element models connected in the center

Options are also provided to consider a constant width slot with rectangular windings or a constant width tooth with round wire windings. Cooling holes may be considered in the back iron arranged in either a square or triangular spaced pattern. The end turns may be considered with separate insulation surface elements for forced convection cooling through the end turns or with only the copper windings elements including conduction heat transfer between the windings and from the end turns into the stack.

#### THERMAL MODEL

CARLE OF COMPANY OF CARLES OF CAR

The thermal model of the stack is constructed with winding elements in the upper and lower half of the slots, Each stack winding is connected to corresponding end turn elements and corresponding elements in the tooth by conduction. The stack winding elements are also connected to adjacent stack winding elements, the top stick element, the center stick element, and the back iron element at the base of the slot. The end turn winding elements are connected to corresponding end section elements and to adjacent end turn winding elements. Separate insulation surface elements may be generated for each end turn and end section winding element which are connected to a fluid stream element by forced convection and to the winding element by conduction. The end section winding elements are connected to both the upper and lower half end turn elements by conduction. The tooth elements are connected to adjacent tooth elements with the top of the tooth connected to the top stick, the center of the tooth connected to the center stick and the bottom of the tooth connected to the back iron element at the base of the tooth by conduction. The top of the tooth and the top stick are connected to the air gap by forced convection of a rotating cylinder in a static housing.

The back iron is divided into NBAK pairs of elements plus NROW elements with cooling holes punched in them. Each of the NBAK pairs of elements includes elements under the slot and under the tooth. These elements are connected to each other by conduction and the NROW elements are connected by forced convection to fluid stream elements in the holes. The other elements of the back iron are connected to element NOUT by conduction.

The initial temperature, heat dissipation, density, volume, specific heat, and thermal conductivity are computed and assigned to each element in the model. The total winding copper losses are divided in proportion to length and number of conductors between the stack, end turn, and end section winding elements. The surface losses are applied to the top of the tooth and the tooth losses are distributed equally to each tooth element adjacent to a stack winding element. The back iron losses are proportioned according to volume in the NBAK elements and the remaining back iron losses are distributed equally over the NROW elements.

As the thermal model is generated the program checks for consistency of the input data by calculating the area available in the slot for the copper conductors and comparing it with the input value of conductor cross-sectional area. When these areas match, the thermal model generated can be punched out by setting TAPE = 1.0.

## H0910

## STEADY-STATE AND TRANSIENT THERMAL ANALYZER PROGRAM WITH COMPRESSIBLE AND INCOMPRESSIBLE PRESSURE DROP AND FILM AND TRANSPIRATION COOLING

#### PROGEAM CAPABILITIES

## Transient Heat Transfer_Calculations

Transient heat transfer calcualtions are developed by an explicit finite difference technique using any element shape with three-dimensional conduction, convection, or radiation heat transfer.

## Steady State Heat Transfer Calculations

Steady-state heat transfer calculations are based on a modified Gauss-Seidel solution to the simultaneous equations in the thermal model. This modified technique involves "accelerated" step substitution with monotonic deceleration until successive substitutions are convergent. A method of "lumping" areas of the problem which are slow to converge is also used to accelerate the calculation procedure. This procedure also provides for any element shape with three-dimensional conduction, convection, or radiation heat transfer.

#### Conduction Heat Transfer Calculations

Conduction heat transfer is input to the program by specifying the element numbers connected by conduction, the cross-sectional area for conduction between the elements, and the conduction length from the center of each element to the interface between them. A mechanical joint thermal contact resistance may also be specified between the elements if they are mechanically separated at the interface. The program obtains the thermal conductivity of each element from a table in which it may be specified as a constant value or as a function of temperature.

#### Convection Heat Transfer Calculations

Convection heat transfer is input to the program by specifying a solid element number connected to a fluid element number by convection, the cross sectional area for convection from the solid element, and the conduction length from the center of the solid element to the convection surface. This program performs the important and often overlooked task of combining conduction heat transfer from the center of the solid element to the surface with convection from the solid surface to the fluid.

The convection heat transfer coefficient may be input to the program by nine different methods. In the first four methods, the heat transfer coefficient may be input as a constant, as a function of time in a table, as a

function of the surface to fluid temperature difference in a table, and as a function of the "film" temperature in a table. In method five, the program calculates the natural convection heat transfer coefficient for both open and enclosed static spaces and enclosed rotating spaces. In method six the program calculates convection heat transfer coefficients for high speed laminar or turbulent flow over external surface- including the effects of the "recovery" temperature in the boundary layer. In method seven the program calculates convection heat transfer coefficients on a free or enclosed rotating disc including the calculation of frictional "windage" heat generation. In method eight the program calculates jet impingement heat transfer coefficients for impingement from a row of holes onto a concave surface. In method nine the program calcualtes convection heat transfer coefficients for flow in a duct, Including the heat transfer "fin effectiveness" of extended surfaces within the duct. This method utilizes tables of Colburn J-factors input as a function of Reynolds number to the program. These tables may be generated for fluid flow in round ducts, square ducts, rectangular ducts, triangual ructs, annular spaces, dimpled tubes, and curved ducts. They may also be generated for fluid flow in tube banks, plate-tin surfaces, pln-fin surfaces, screen matrix surfaces. crossed rod matrix surfaces, and corrugated ceramic surfaces. Entrance effects on heat transfer may be applied using the appropriate multiplying factor at each location. Four techniques for evaluation of the influence of temperature-gependent fluid properties are available in the program. The appropriate fluid peoperties may be input in tabular form as a function of temperature.

Analistication and the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of

. . .

## Radiation Heat Transfer Calculations

والألار فيها ويرتب الطاطة فكوم فروا ستخاط فالمحارث أتكرك المتعا

Radiation heat transfer is input to the program by specifying a solid element number connected to a representative surrounding element number by radiation, the cross sectional area for radiation from the solid element, and the conduction length from the center of the solid element to the radiation surface. This section also includes the important combination of conduction to the radiating surface with radiation from the surface. The emissivity view factor for radiation may be estimated by methods given in "Radiation Heat Transfer" by Sparrow and Cess or by a computer program such as CONFAC 11.

## Initial Temperature, Boundary Conditions, Heat Input, Thermal Capacitance, and Thermal Conductivity Specification

The initial temperature, boundary conditions, heat input thermal capacitance, and thermal conductivity may be specified for each individual element or for blocks of elements which are identical. In transient heat transfer calculation, the initial temperature, the heat input, the density, the volume, the specific heat, and the thermal conductivity of each element is specified. For elements with negligible thermal capacitance the density, volume, and specific heat may be left blank to increase the calculation time step. For steady state calculations, the initial temperature, the heat input, and the thermal conductivity of each element is specified. The boundary condition elements are specified by having a negative value for the density times the volume. This element is then maintained at a constant temperature or may be specified as a temperature versus time function from an input table. Any element in the network may be specified as a boundary condition (constant

.....

temperature) element and any number of elements may be connected to it by conduction, convection, or radiation. The heat input for each element may be input as zero, as a constant value, as a function of time in a table, as a function of its own temperature or another specified element temperature, specified from the frictional "windage" heat generation calculations, or calculated from the ball and roller bearing heat generation calculation computer program which can be supplied. The specific heat and thermal conductivity of each element may be specified as a constant or as a function of temperature in tables.

#### Fluid Stream Heat Transfer and Pressure Drop Calculations

Fluid stream elements may be input with heat transfer to them by conduction, convection, or radiation. Fluid stream heat transfer calculations have provisions for preventing the outlet fluid temperature from "overshooting" the surrounding surface temperatures, a thermodynamic impossibility. The steady state fluid stream calculations are based on thermal capacity rate calculations, while transient fluic stream calculations may be based on the thermal capacitance of each element moving in the fluid stream to simulate "lag" conditions. The energy input of rotational flow may also be added to the fluid stream.

Both steady state compressible and incompressible fluid stream pressure drops may be calculated by the program. The pressure drop calculations include the effects of heat addition, area change, fluid friction, rotational flow, and flow addition or removal. Total head losses due to valves, bends, sharp contractions or expansions, and orifices may be included at the inlet and exit to each fluid stream.

A complete fluid stream network may be simulated with streams branching from previous streams and mixing to form new streams or even returning to a previous stream in the network. The fluid flow rate may be input as a constant, as a function of time, from a table, or as a function of specified element temperature.

### Film Cooling and Transpiration Cooling Calculations

Film cooling calculations have been included in a table of the film effectiveness as a function of the film cooling parameter (x/ms). Local film temperatures on a film cooled surface are calculated by the program from specified element temperatures for the "free stream" and for the film coolant discharge point. Local film temperatures are calculated at specified distances downstream from the point of film injection. The table of film effectivenss as a function of the film cooling parameter (x/ms), may be selected from twelve correlations presented in the transpiration and film cooling effectiveness computer program which can be supplied. The appropriate correlation should be selected for the method of film injection used.

Transpiration cooling calculations may be included by using the Stanton number reduction factor for transpiration cooling from the transpiration and film cooling effectiveness computer program or from papers on transpiration cooling. The thermal analyzer program accounts for the effect of cooling air discharge temperature from the transpiration cooled wall on the Stanton number reduction factor.

#### PROGRAM OUTPUT

1. Each element temperature, heat input, and thermal conductivity for steady state calculations is printed out. Each element temperature, heat input, weight, specific heat, and thermal conductivity for each specified printing time period in transient heat transfer calculations is printed out.

2. The fluid stream inlet temperature and the outlet temperature, the fluid stream flow rate, the fluid density, and the internal fluid heat generation for each section of each fluid stream is printed out.

3. The "free stream" temperature, the film discharge temperature, and the effective film temperature at each location specified is printed out.

4. The printing of the thermal resistance values for conduction, the thermal resistance values and neat transfer coefficients for convection, and the thermal resistance values and effective heat transfer coefficients for radiation may be included or deleted as specified.

5. The fluid stream pressure drop calculations and printout may be deleted if specified. When included, the total and static pressures, the Reynolds number, the friction factor, and the Mach number for compressible flow is printed for each element in each fluid stream.

#### TYPICAL APPLICATIONS

1. Both passive and active electronic cooling system analysis and design with or without heaters or cooling flow controllers.

2. Thermal analysis and design calculations for ambient cooled, forced air cooled, gas cooled, or liquid cooled ac or dc motors, generators, and alternators.

3. Thermal analysis and design calculations for pumps, fans, and compressors including the bearing temperatures and the analysis of the motors for turbines driving them.

4. Thermal analysis and design calculations of gas turbine engines including the axial flow and radial flow compressors and turbines, the combustor, the bearings and seals, the anti-icing system, the lubrication cooling system, the fuel supply system, and the accessory area cooling system. Also the thermal analysis of cooled and uncooled turbine blades.

5. Transient and steady state thermal analysis of heat exchangers including air-oil coolers, fuel-oil coolers, recuperators, rotary regenerators, cryogenic heat exchangers, pool boiling heat exchangers, condensers, periodic flow regenerators, and heat exchangers with more than two fluid streams. The calculations may include the effects of axial conduction, fluid bypassing, perfectly mixed or unmixed fluids, variation of fluid properties through the heat exchanger, condensation of moisture from the air or "wet" heat transfer, and the effect of the variation of fluid to wall temperature difference on local heat transfer coefficients for boiling and condensing.



12401

station. 77

14.5

367

. . .



1201.75

hip the art of the

41.93

Section Section Street

ÿ



•

f. ......









ļ